

Exploring the Care Needs of Stroke Dyads in Hospital-to-Home Transition From Perspectives of Stroke Dyads and Health Professionals

by

Langduo Chen

Registered Nurse, Master of Nursing

Thesis submitted to Flinders University for the degree of

Doctor of Philosophy

College of Nursing and Health Sciences

17 November 2023

Contents

Summary	vi
Declaration	ix
Acknowledgements	xi
Publications and Presentations Produced During the Candidature	xiii
List of Tables	xv
List of Figures	xv
List of Abbreviations	xvi
Glossary	xvii
Chapter 1: Background	1
1.1 Introduction	1
1.2 Stroke in a Global and an Australian Context	2
1.2.1 Stroke and Brain Recovery	2
1.2.2 Aetiology and Clinical Presentation	5
1.2.3 Epidemiology	6
1.2.4 Impact	6
1.2.5 Informal Caregivers	8
1.2.6 Dyadic Interdependence	10
1.3 Post-stroke Services	11
1.3.1 Stroke Rehabilitation	11
1.3.2 Post-stroke Rehabilitation and Community Care Services	12
1.3.3 Nurses' Role in Post-stroke Rehabilitation	14
1.4 Aim and Objectives	16
1.5 Research Design	16
1.6 Significance of the Study	17
1.7 Thesis Structure	18
1.8 Summary	21
Chanter 2: Integrative Review	22
2.1 Introduction	22
2.2 An Integrative Review: Challenges and Opportunities for People With Stroke an	d
Caregivers in Hospital to Home Transition Care	<u> </u>
2 2 1 Abstract	23
222 Introduction	24
2.2.2 Rackground	24
224 The Review	26
225 Findings	35
2.2.6 Discussion	42
2.2.7 Conclusion	
2.3 Summary	
Chanter 2: Theoretical Framework	47
Chapter 3: Theoretical Framework	
3.1 Introduction	41 47
3.2 Theorem and Frameworks informing studies in transition bare	/44
3.2.1 Outdepts of Transition III the Reviewed Studies	4/ 50
3.2.2 The Relevance of Melsie's Transitions Theory to the Dresent Study	50 56
3.2.4 The Need to Adant Meleis's Transitions Theory to the Study Context	56
3.3 Summary	50 58

Chapter 4: Systematic Review	59
4.1 Introduction	59
4.2 Enablers and Barriers in Hospital to Home Transitional Care for People With	
Stroke and Caregivers: A Systematic Review	60
4.2.1 Abstract	60
4.2.2 Introduction	62
4.2.3 Aims	64
4.2.4 Methods	65
4.2.5 Findings	71
4.2.6 Discussion	82
4.2.7 Conclusion	86
4.2.8 Relevance to Clinical Practice	86
4.3 Summary	87
Chapter 5: Research Methodology and Methods	88
5.1 Introduction	88
5.2 Aim and Objectives	88
5.3 Philosophical Perspectives of the Study	88
5.3.1 Paradigms in Research Design	89
5.3.2 Using Constructivism to Guide the Research Design	92
5.4 Research Methodology	93
5.4.1 Overview of Qualitative Research Approaches	93
5.4.2 Overview of Phenomenology	95
5.4.3 Gadamer's Phenomenological Hermeneutics as a Choice for the Current	
Research	97
5.5 Ethical Consideration	104
5.6 Methods	105
5.6.1 Setting and Participant Recruitment	105
5.6.2 Data Collection	110
5.6.3 Data Analysis	113
5.6.4 Rigour	115
5.7 Summary	115
Chapter 6: Findings—Part 1: Outline of Findings, and Findings From the First	
Interviews With Stroke Dvads	117
6.1 Introduction	117
6.2 Characteristics of Participants	117
6.3 Outline of Themes and Subthemes	118
6.4 Findings—Before Hospital Discharge	132
6.5 Exploring the Shared Experiences of People With Stroke and Caregivers in	
Preparedness to Manage Post-Discharge Care: A Hermeneutic Study	133
6.5.1 Abstract	133
6.5.2 Introduction	134
6.5.3 Background	135
6.5.4 The Study	137
6.5.5 Findings	149
6.5.6 Discussion	160
6.5.7 Conclusion	163
6.6 Summary	164
Chapter 7: Findings—Part 2: Themes From the Follow-Up Interviews With Stroke	
Dvads	166
7.1 Introduction	
7.2 Theme 4: Coping With Setbacks in Home Care	
7.2.1 Subtheme 1: Processing Loss and Grief in People With Stroke	168
7.2.2 Subtheme 2: Processing Loss and Grief in Caregivers	170

7.2.3 Subtheme 3: Reconciling With Resilience	172
7.3 Theme 5: Role Transformation	175
7.3.1 Subtheme 1: Impact of Stroke-Related Limitations on People With Stroke	175
7.3.2 Subtheme 2: Impact of Stroke-Related Limitations on Caregivers	178
7.3.3 Subtheme 3: Regaining Control of Life	181
7.4 Summary	187
CHAPTER 8: Findings—Part 3: Themes From the Follow-Up Interviews With	
Stroke Dyads (Continued)	188
8.1 Introduction	188
8.2 Theme 6: Transition With Support	188
8.2.1 Subtheme 1: Support From Family and Friends	189
8.2.2 Subtheme 2: Support From Health Professionals	192
8.3 Theme 7: Expectations in Hospital-to-Home Transition Care	198
8.3.1 Subtheme 1: Enhancement of Discharge Planning and Preparation	198
8.3.2 Subtheme 2: Flexibility of Service Delivery	200
8.4 Summary	202
Chapter 9: Findings—Part 4: Themes From Focus Groups and Interviews With	
Health Professionals	204
9.1 Introduction	204
9.2 Theme 1: Transition Shaped by Care Needs Assessment	205
9.2.1 Subtheme 1: 'Psychological/Mental Health is Underplayed'	205
9.2.2 Subtheme 2: 'Our Focus is on the Patient'	207
9.3 Theme 2: Transition Challenged by Constraints on Service Delivery	209
9.3.1 Subtheme 1: 'Bed Pressures Certainly Impacts'	209
9.3.2 Subtheme 2: 'We're Dictated by Funding Models'	212
9.4 Theme 3: Transition Enhanced by Commitment to the Safety of Care	216
9.4.1 Subtheme 1: 'Empowering Them to Making Decisions'	216
9.4.2 Subtheme 2: 'Working Towards a Common Goal'	218
9.5 Summary	220
Chapter 10: Discussion	222
10.1 Introduction	222
10.2 Balancing Functional Rehabilitation and Psychological Support	223
10.3 Understanding Dyadic Interdependence to Promote Dyad-Centredness in	
Healthcare	227
10.4 The Role of Health Professionals in Enhancing the Hospital-to-Home Transition.	231
10.5 Enriching Meleis's Transitions Theory	234
10.5.1 Nature of Transitions	234
10.5.2 Transition Conditions	235
10.5.3 Patterns of Response	237
10.5.4 Nursing Therapeutics	
10.6 Summary	239
	239 240
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care	239 240
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 . .241
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 241 241
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework 11.1 Introduction. 11.2 Overview of the Framework.	239 240 241 241 241
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 241 241 241 242
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 241 241 241 242 242
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 241 241 241 242 242 242 246
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework 11.1 Introduction. 11.2 Overview of the Framework. 11.3 The Stroke-Specific Transition Care Foundation 11.3.1 Health and Social Care Systems 11.3.2 Health and Social Care Organisations (Facilities). 11.3.3 Health and Social Care Professionals	239 240 241 241 241 242 242 246 247
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 241 241 242 242 242 246 247 249
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 241 241 242 242 242 246 247 249 250
Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework	239 240 241 241 242 242 242 246 247 249 250 250

11.5 Summary	.257
Chapter 12: Conclusion	.258
12.1 Introduction	.258
12.2 Limitations	.258
12.3 Summary of the Study	.259
12.4 Implications	.262
12.4.1 Implications for Policy Development	.262
12.4.2 Implications for Health Organisations	.264
12.4.3 Implications for Health Professionals	.266
12.4.4 Implications for Future Research and Education	.268
12.5 Reflection	.269
12.6 Conclusion	.270
References	.272
Appendices	.320
Appendix 1: Published Integrative Review Protocol	320
Appendix 2: Medline Search Strategy	326
Appendix 3: Summary of Selected Studies	.327
Appendix 4: Quality Appraisal of Reviewed Studies	337
Appendix 4a: Evaluation of Qualitative Studies Included for Review	337
Appendix 4b: Evaluation of Randomised Controlled Trials Included for Review	338
Appendix 4c: Evaluation of Systematic Reviews Included for Review	.338
Appendix 4d: Evaluation of Quantitative Studies Included for Review	.339
Appendix 4e: Evaluation of the Mixed Method Study Included for Review	.339
Appendix 5: Exemplar—Data Extraction by Rule of Thumb for Anchoring Text to	
Context in Restatements of Findings (Sandelowski et al., 2013, p. 1432)	.341
Appendix 6: Themes With Direct Participant Quotations	.343
Appendix 7: PRISMA (Preferred Reporting of Items for Systematic Reviews and Meta-	
Analysis) Statement	.348
Appendix 8: ENTREQ (Enhancing Transparency in Reporting the Synthesis of	
Qualitative Research) Statement	.351
Appendix 9: Published Systematic Review Protocol	.353
Appendix 10: Search Strategies	.359
Appendix 11: Summary of Reviewed Studies	.365
Appendix 12: Developing Descriptive Themes Example	.369
Appendix 13: CERQual Evidence Profile	.370
Appendix 14: Authorisation Letter From the Southern Adelaide Local Health Network	
Ethics Committee	.374
Appendix 15: Participant Information and Consent Form	.376
Appendix 16: Recruitment Poster	.408
Appendix 17: Montreal Cognitive Assessment	.410
Appendix 18: Functional Independence Measure	.411
Appendix 19: Simplified Participant Information and Consent Form	.413
Appendix 20: Interview Guide	.426
Appendix 20a: Interview Guide Before Hospital Discharge (the First Interview)	.426
Appendix 20b: Interview Guide After Hospital Discharge (the Second and Third	
Interviews)	.427
Appendix 20c: Interview Guide for General Practitioners	.431
Appendix 20d: Focus Group and Interview Guide	.432
Appendix 21: Structured Diary	.433
Appendix 22: COREQ (COnsolidated Criteria for REporting Qualitative Research)	404
Unecklist	.434
Appendix 25. Themes with Quotations	.430

Summary

Stroke occurs when cerebral blood flow is disrupted by vascular occlusion or rupture, causing brain tissue damage. It is a leading cause of adult disability worldwide. Hospital-to-home transition is a challenging period for stroke dyads (people with stroke and their caregivers) due to shortened hospital stays and complex care needs, as they must now take charge of care. Although stroke dyads demonstrate dyadic interdependence in the transition, few studies have explored their individual and dyadic care needs and expectations during the transition.

Aim

The aim of the study was to understand the care needs and expectations of stroke dyads during the hospital-to-home transition from the perspectives of stroke dyads and health professionals in South Australia.

Methods

Gadamer's philosophical hermeneutics guided the study. Thirty stroke dyads and 31 stroke care clinicians, including two general practitioners, participated in the study. Data were collected through interviews with stroke dyads before hospital discharge and at 3 and 6 months after discharge. Focus groups and interviews were used to collect data with stroke care clinicians.

Results

Stroke dyads' perceived preparedness in managing post-discharge care was influenced by psychological and emotional state, resilience, and level of engagement in discharge planning. Returning home was seen as a significant milestone in the post-stroke trajectory. However, compared with people with stroke, caregivers perceived profound uncertainties about their role and how to take over care after discharge as a result of inconsistent engagement in discharge planning.

vi

After discharge, stroke dyads experienced a sense of setbacks due to physical and psychosocial factors. They perceived a dyadic interdependence, which generated positive impacts on their adaptation to challenging situations. Caregivers played a key role in providing psychological support for people with stroke, but such a role was not fully recognised nor supported in the healthcare system. Moreover, stroke dyads expected enhanced communication, engagement in discharge planning and needs-driven service provision. Additionally, health professionals believed that prioritising the safety and continuity of care enhanced the transition and helped health professionals to cope with work-related challenges.

Discussion

Findings support previous studies that found that equal attention to stroke dyads' psychological and physical recovery facilitates the development of their self-management ability during the transition. The findings also add new understandings that people with stroke mainly rely on caregivers to provide psychological support, while caregivers also show vulnerability in their role and unmet self-care needs. Such situations underscore the significance of dyad-centredness in care needs assessment and discharge planning. Additionally, findings suggest an innovative nurse-led, stroke-specific hospital-to-home transition care framework. This framework emphasises the partnership between the healthcare organisation, health professional and stroke dyad as the foundation for stroke-dyad-centred transition care at three key time points—during the hospital stay, arriving home and during formal rehabilitation programs, and after separation from formal rehabilitation programs and during community dwelling.

Conclusion

Findings support a dyad-centred and interdisciplinary care approach to hospital-tohome transition care that emphasises collaboration, communication and continuity of care across different care settings and service providers. Findings also support the need for

vii

further studies on the feasibility, acceptability and effectiveness of the nurse-led, strokespecific hospital-to-home transition care framework.

Keywords: people with stroke, caregivers, stroke dyads, transition care, discharge planning

Declaration

I certify that this thesis:

- does not incorporate without acknowledgment any material previously submitted for a degree or diploma in any university
- and the research within will not be submitted for any other future degree or diploma without the permission of Flinders University; and
- to the best of my knowledge and belief, does not contain any material previously published or written by another person except where due reference is made in the text.

Signed ___Langduo Chen_____ Date ____<u>07/09/2923</u>____ To my mother, Saiting Zhang, and my father, Rennai Chen, for their unconditional love and unwavering support over the years.

Acknowledgements

I would like to express my gratitude to all the people who have supported me in completing this research project and the thesis. It would have been impossible to complete this study without the support and guidance I received from the following people.

First and foremost, I would like to thank each participant in this study. It is a privilege to share their stories and experiences. This study would not have been possible without their contribution.

My sincere and deepest thanks to my supervisors, Matthew Flinders Professor Lily Xiao and Professor Di Chamberlain, for their invaluable advice, unwavering support, encouragement and patience over the past 7 years. Their passion for nursing research and striving for excellence will continue to influence me in my professional and personal life. I also would like to thank Professor Stacey George for her constructive feedback and recommendations on the final thesis review. Capstone Editing provided copyediting and proofreading services, according to the guidelines laid out in the university-endorsed national 'Guidelines for Editing Research Theses'.

My gratitude to Ms Petra Bierer, co-director of the Division of Rehabilitation, Aged and Palliative Care at the Southern Adelaide Local Health Network, for her treasured support throughout the entire course of my PhD study.

I would like to extend my sincere thanks to my colleagues, Professor Maria Crotty and Associate Professor Zoe Adey-Wakeling, for sharing with me their experiences and insightful comments. My gratitude also goes to nursing directors Ms Philippa Jones, Ms Tracy Lees, Mr Damon Williams, and associate nurse unit managers Mrs Berenice Broadbent, Ms Nelcy John and Ms Neetu Sharma for their continuous support and inspiration.

xi

Thank you to my friends: Ms Ying Yu for her understanding, encouragement and sharing her learning journey with me; and Mrs Clara Murphy, Mrs Marilou Talens and Ms Xin Xia, whose encouragement and kind help made my PhD study possible.

Finally, my heartfelt thanks to my mother, Saiting Zhang, and father, Rennai Chen, whose unconditional love, encouragement and strong support maintained my momentum.

Publications and Presentations Produced During the Candidature

- Chen, L., Xiao, L. D., Chamberlain, D., & Newman, P. (2018). Stroke survivors' and informal caregivers' experience in hospital to home transition: a metasynthesis protocol. *PROSPERO*. Retrieved October 18, 2019, from: <u>http://www.crd.york.ac.uk/PROSPERO/display_record.php?ID=CRD4201809134</u>
 <u>5</u>
- Chen, L., Xiao, L. D., & Chamberlain, D. (2019). Challenges and opportunities for stroke survivors and carers in hospital to home transition care: a systematic integrated review protocol. *PROSPERO*. Retrieved March 12, 2020, from: <u>https://www.crd.york.ac.uk/prospero/display_record.php?ID=CRD42019136068</u>
- Chen, L., Xiao, L. D., & Chamberlain, D. (2019, October 20-23). Challenges and opportunities in hospital to home transition care for stroke survivors and carers [Conference presentation]. The 4th Annual Scientific Meeting of Rehabilitation Medicine Society of Australia and New Zealand, Adelaide, South Australia.
- Chen, L., Xiao, L. D., & Chamberlain, D. (2020). An integrative review: Challenges and opportunities for people with stroke and caregivers in hospital-tohome transition care. *Journal of Advanced Nursing*, *76*(9), 2253–2265. <u>https://doi.org/10.1111/jan.14446</u>
- Chen, L., Xiao, L. D., Chamberlain, D., & Newman, P. (2021). Enablers and barriers in hospital-to-home transitional care for people with stroke and caregivers: A systematic review. *Journal of Clinical Nursing*, *30*(19–20), 2786– 2807. <u>https://doi.org/10.1111/jocn.15807</u>
- Chen, L., Xiao, L. D., & Chamberlain, D. (2022). Exploring the shared experiences of people with stroke and caregivers in preparedness to manage

post-discharge care: A hermeneutic study. *Journal of Advanced Nursing*, 78(9), 2983–2999. <u>https://doi.org/10.1111/jan.15275</u>

 Chen, L., Xiao, L. D., & Chamberlain, D. (2022, October 20-21). Exploring shared experiences of people with stroke and caregivers in preparedness to manage post-discharge care [Conference presentation]. The 31st National Conference of the Australasian Rehabilitation Nurses Association, Adelaide, South Australia.

List of Tables

Table 2.1 Inclusion and Exclusion Criteria	28
Table 2.2 Brief Summary of the Selected Articles	30
Table 4.1 Inclusion and Exclusion Criteria	66
Table 4.2 Critical Appraisal	69
Table 5.1 Inclusion Criteria	105
Table 6.1 Characteristics of Stroke and Caregiver Participants	119
Table 6.2 Characteristics of Health Professional Participants	126
Table 6.3 Major Themes and Subthemes—Perspectives From Stroke Dyads	129
Table 6.4 Major Themes and Subthemes—Perspectives From Health Profession	nals131
Table 6.5 Interview Guide	143
Table 6.6 Interpretation Process Exemplar	145
Table 7.1 Major Themes and Subthemes in Chapters 7 and 8	166
Table 8.1 Major Themes and Subthemes in Chapter 8	188
Table 9.1 Major Themes and Subthemes in Chapter 9	205

List of Figures

Figure 2.1	Review Flow Diagram2	9
Figure 2.2	Convergent Qualitative Synthesis (Adapted From Treacy & Stayt, 2019)	5
Figure 3.1	Transition: A Middle-Range Theory (Meleis et al. 2000, p. 56)5	5
Figure 3.2	Meleis's Transition Theory (Meleis et al., 2000, p. 56), Adapted to This Study Context	8
Figure 4.1	Review Flow Diagram6	8
Figure 5.1	Recruitment Flowchart10	9
Figure 6.1	Research Process (Adapted From Fleming et al., 2003 and Alsaigh & Coyne, 2019)14	0
Figure 11.	1 Proposed Stroke-Specific Hospital-to-Home Transition Care Framework24	.3

List of Abbreviations

ACSQHC	Australian Commission on Safety and Quality in Health Care
ADLs	activities of daily living
AIHW	Australian Institute of Health and Welfare
APA	American Psychological Association
ARNA	Australasian Rehabilitation Nurses' Association
CASP	Critical Appraisal Skills Programme
ССМ	chronic care model
CERQual	Confidence in the Evidence from Reviews of Qualitative Research
COREQ	Consolidated Criteria for Reporting Qualitative Research
FIM	Functional Independence Measure
GP	general practitioner
ICF	International Classification of Functioning, Disability and Health
MoCA	Montreal Cognitive Assessment
NDIS	National Disability Insurance Scheme
PEO	population, exposure, outcome
PICF	patient information and consent form
PRISMA	Preferred Reporting Items for Systematic Reviews and Meta-Analyses
PSD	post-stroke depression
WHO	World Health Organization
WSO	World Stroke Organization

Glossary

Term	Definition
Care coordination	The deliberate organisation of patient care activities and
	sharing of information among all care providers to achieve safe
	and effective care (Agency for Healthcare Research and
	Quality, 2014).
Depression	The feeling of persistent sadness and/or loss of interest in
	activities an individual once enjoyed (World Health
	Organization, 2017a).
Disability	An umbrella term for impairments of body function or structure,
	activity limitations or participation restrictions. It is not only
	related to pathology and impairment but also considers the
	individual's interaction with environmental and social
	circumstances (Australian Institute of Health and Welfare,
	2022; World Health Organization, 2022a).
Functional recovery	The recovery of cognitive and motor impairments affecting
	activities of daily living, such as mobility, showering, feeding
	and toileting (Branco et al., 2019; Grefkes & Fink, 2020).
Haemorrhagic stroke	Classified into intracerebral and subarachnoid haemorrhage,
	refers to blood vessel ruptures causing abnormal accumulation
	of blood within the brain. This type of stroke accounts for
	approximately 10%–15% of the stroke population (Kuriakose &
	Xiao, 2020).
Health	A state of complete physical, mental and social wellbeing and
	not merely the absence of disease or infirmity (World Health
	Organization, 1948).

Term	Definition
Health system	All organisations, people and actions with the primary purpose
	of promoting, restoring or maintaining health, which includes
	endeavours to influence determinants of health and direct
	health-improving activities (World Health Organization, 2007).
Healthy transition	Subjective wellbeing, role mastery and wellbeing of
	relationships during the transition (Meleis, 2010).
Informal caregivers	People caring for those in need within the context of an existing
	relationship, such as a family member, friend or neighbour
	(Australian Institute of Health and Welfare, 2021).
Ischaemic stroke	Ischaemic stroke is caused by a thrombus or embolus that
	obstructs the cerebral artery, resulting in infarction of the parts
	of the brain the artery feeds. About 85% of stroke patients
	suffer from ischaemic stroke (Kuriakose & Xiao, 2020).
Interdependence	Interdependence presents in a relationship when one person's
	emotion, cognition or behaviours affects those of a partner
	(Cook & Kenny, 2005).
Interprofessional	An approach that enables interprofessional healthcare teams to
approach	practice collaboratively to enhance person-centred care and
	improve patient and health systems outcomes (McLaney et al.,
	2022).
Interdisciplinary team	Multidisciplinary team members collaboratively set goals with
approach	patients and families and implement plans. There is a high
	degree of communication and cooperation among the team
	members (Körner, 2010).

Term	Definition
Interprofessional	An education strategy to promote two or more professionals
education	learning about, from and with each other to enable effective
	collaboration and improve health outcomes (World Health
	Organization, 2010).
Micro-credentials	Certified documents that provide recognised proof of the
	achievement of competency-based skills and learning
	outcomes from shorter-duration educational or training
	activities. (McGreal & Olcott, 2022).
Patient safety culture	A pattern of individual and organisational behaviour, based
	upon shared beliefs and values, that continuously seeks to
	minimise patient harm that may result from the process of care
	delivery (Australian Commission on Safety and Quality in
	Health Care, n.da).
People with stroke	Individuals who have survived stroke and are living with the
	sequelae of stroke.
Person-centred care	An approach that responds to the needs, preferences and
	values of individuals, families and communities in humane and
	holistic ways and through education and support to promote
	their participation in their own care (World Health Organization,
	2015). The terms 'person-centred care' and 'individualised care'
	are used interchangeably within this thesis.
Primary care	The essential, first-contact care that is provided in a community
	setting (World Health Organization, 2021).
Resilience	The 'processes and skills that result in good individual and
	community-health outcomes despite negative events, serious
	threats and hazards' (World Health Organization, 2017b, p. 6).

Term	Definition
Self-management	An individual's ability to manage a chronic condition in
	collaboration with the caregiver, family, community and
	appropriate health professionals (Satink et al., 2016).
Social integration	Reflects the extent of one's social connections and access to
	support, often in important roles such as marriage, friendship
	and in volunteer organisations (Uchino et al., 2018, p. 463).
Stroke	'Stroke is an acute episode of focal dysfunction of the brain,
	retina, or spinal cord lasting longer than 24 [hours], or of any
	duration if imaging ([computerised tomography] or [magnetic
	resonance imaging]) or autopsy show focal infarction or
	haemorrhage relevant to the symptoms' (Hankey, 2017, p.
	641).
Stroke dyad	An individual with stroke and their caregiver form a
	stroke/caregiving dyad (Jacops et al., 2022).
Stroke multidisciplinary	A stroke multidisciplinary team includes stroke-specific medical,
team	nursing and allied health clinicians (Stroke Foundation, 2022).
Stroke nurse navigators	Specialised stroke nurses, practising at an advanced level, play
	a crucial role in facilitating and managing care transitions,
	which leads to improved continuity of care, reduced service
	fragmentation, enhanced quality of care for stroke dyads and
	decreased hospital readmissions (Byrne, Hegney et al., 2020;
	Camicia et al., 2021).
Stroke recovery	Improvement across a variety of outcomes, beginning with
	biological and neurological changes that manifest as
	improvement in performance and activity-based behavioural
	measures (Belagaje, 2017, p. 239).

Term	Definition
Stroke rehabilitation	A proactive, person-centred and goal-oriented process
	facilitated by a multidisciplinary team in a culturally appropriate
	environment and a health strategy aimed at optimising function
	and enabling social reintegration for people with stroke (Stroke
	Foundation, 2022).
Team resilience	The collective capacity of all team members to respond well to
	change, pressure and disruption (Hartwig et al., 2020).
Telerehabilitation	A model of rehabilitation services that uses information and
	communication technologies to remotely deliver rehabilitation
	consultations, assessments and therapies. It forms part of
	various post-stroke services in South Australia (Laver et al.,
	2022; SA Health, 2023).
Transition	'A passage from one life phase, condition, or status to another,
	is a multiple concept embracing the elements of process, time
	span, and perception \dots [it] refers to both the process and the
	outcome of complex person-environment interactions. It may
	involve more than one person and is imbedded in the context
	and the situation Defining characteristics of transition include
	process, disconnectedness, perception, and patterns of
	response' (Chick & Meleis, 1986, pp. 239–240).
Uncertainty in illness	The inability to determine the meaning of illness-related events
	or to accurately predict outcomes (Mishel, 1990).
Wellbeing	A positive state experienced by individuals and societies. It is a
	resource for daily life and is determined by social, economic
	and environmental conditions (World Health Organization,
	2021).

Chapter 1: Background

1.1 Introduction

Stroke (cerebrovascular accident) is a common and serious health condition and a prominent cause of disability worldwide (Feigin et al., 2022; Stinear et al., 2020). Despite a reduction in the age-standardised stroke mortality rate in the past two decades, the overall burden of stroke is substantial and continues to increase with an ageing population (Feigin et al., 2022; Gorelick, 2019). The sudden onset and catastrophic nature of stroke brings enormous physical and psychosocial challenges to people with stroke and their families. Over 60% of people with stroke live with disabilities that require them to be cared for by informal caregivers (hereafter, 'caregivers'; Stroke Foundation, 2020b).

Rehabilitation involves stroke dyads (people with stroke and their caregivers) regaining independence and being prepared to return to the community (Stroke Foundation, 2022). For most stroke dyads, the transition from hospital to home is an exciting but also difficult time. Homecoming marks a great improvement in post-stroke recovery. However, this is also a time to face the real world out of hospital and to adapt to the various changes in their lives (L. Chen et al., 2020; Connolly & Mahoney, 2018). The care needs and expectations of health and social care services from stroke dyads change from time to time during the transition. However, what these needs and expectations during transition are and how they are met is under-researched. Furthermore, most studies on transition care have described issues from the perspectives of people with stroke and/or caregivers. Health professionals' views regarding the approaches to identifying and meeting the care needs of stroke dyads are also largely under-researched.

Nurses have been key members of multidisciplinary rehabilitation teams and act as case managers, care coordinators and educators for stroke dyads to enable continuum of care and optimise rehabilitation and recovery (Camicia et al., 2022; Purvis et al., 2021). It is evident that nurses equipped with comprehensive assessment knowledge and skills are able

to reduce post-stroke readmission rates and adverse events through prevention strategies and timely intervention (Deen et al., 2016; Jun-O'Connell et al., 2023). Stroke-care-nurse-led studies on the care needs of stroke dyads during the hospital-to-home transition are much needed to inform person-centred and consumer-directed quality improvement strategies and nursing interventions in stroke care.

The present chapter highlights the background of the study. It begins with a discussion of stroke in the global and local context, outlining its prevalence, incidence, recovery and the impact that stroke has on people's lives as well as its economic implications for health and social care systems. Next, an overview of stroke care organisations and information on stroke rehabilitation, clinical guidelines for stroke management and available post-stroke community programs for stroke dyads is presented. The social and political context of transition care for people with stroke and support for their caregivers are also discussed. This chapter also outlines the present study's significance, research design, aims and objectives. Finally, the structure of this thesis is introduced.

1.2 Stroke in a Global and an Australian Context

1.2.1 Stroke and Brain Recovery

The brain is one of the most complex organs of the human body. It is the command centre of the nervous system and interprets senses, controls body movement and behaviour, and is 'the source of all the qualities that define our humanity' (National Institute of Neurological Disorders and Stroke, n.d.). Normal function of the brain relies on adequate cerebral blood circulation for the supply of oxygen and nutrients as well as the removal of waste products. Alterations to the cerebral vascular function cause brain injury, resulting in changes to normal human body function (Chrissobolis & Sobey, 2015).

Based on clinical presentation, a stroke, also known as a cerebrovascular accident or brain attack, is defined by the World Health Organization (WHO) as 'rapidly developed clinical signs of focal (or global) disturbance of cerebral function, lasting more than 24 hours or leading to death, with no apparent cause other than of vascular origin' (Aho et al., 1980, p.

114). This traditional definition has been challenged by experts due to advances in stroke management, especially imaging technology, in the 21st century (Coupland et al., 2017). With expert consensus, the definition of stroke has been updated as:

an acute episode of focal dysfunction of the brain, retina, or spinal cord lasting longer than 24 [hours], or of any duration if imaging ([computerised tomography] or [magnetic resonance imaging]) or autopsy show focal infarction or haemorrhage relevant to the symptoms. (Hankey, 2017, p. 641)

This definition was adopted for this study because it is based on both imaging and clinical presentation and thus includes patients who may be excluded by the traditional definition.

There are two major types of stroke: ischaemic and haemorrhagic (Kuriakose & Xiao, 2020). Ischaemic stroke is caused by a thrombus or embolus that obstructs the cerebral artery, resulting in infarction of the parts of the brain the artery feeds. About 85% of stroke patients suffer from ischaemic stroke (Kuriakose & Xiao, 2020). Haemorrhagic stroke, classified into intracerebral and subarachnoid haemorrhage, refers to blood vessel ruptures causing abnormal accumulation of blood within the brain. This type of stroke accounts for approximately 10%–15% of the stroke population (Kuriakose & Xiao, 2020).

Since the groundbreaking discovery in the past century of the brain's ability to remodel, studies on neural plasticity have increased dramatically (B. P. Johnson & Cohen, 2022; Su & Xu, 2020). Neural plasticity refers to changes occurring in the central nervous system in response to intrinsic and extrinsic stimuli. These changes occur at functional and structural levels. They can be beneficial or can become plasticity diseases based on individual or genetic differences (B. P. Johnson & Cohen, 2022; Su & Xu, 2020). For example, ischaemic stroke causes loss of neurons and damages the normal function of synapses. This may result in cognition and memory impairment in people post-stroke (Nie & Yang, 2017). The brain's remodelling ability is reflected in the modulation of synaptic plasticity stimulated by physical exercises. Through energy metabolism and synaptic plasticity, physical exercise training has been found to be beneficial to brain health and

cognition (Nie & Yang, 2017). This also explains why exercise training has been the primary post-stroke rehabilitative therapy (Nie & Yang, 2017).

Among all types of stroke, the highest percentage of stroke patients experience ischaemic insult, including delayed ischaemic events due to arterial vasospasm for some people with subarachnoid haemorrhage (Kuriakose & Xiao, 2020; Tobin et al., 2014). Over the past two decades, there have been advances in acute ischaemic stroke treatment, including the use of thrombolytic agents and endovascular therapy (Hollist et al., 2021). Through the use of multimodal computerised tomography and magnetic resonance imaging, the approach to acute stroke management is now shifting from a standard time-window approach to one of brain tissue preservation and individualised treatment (Hollist et al., 2021). Research on other possible novel therapies is underway as a result of increasing evidence of the complex bidirectional interplay between the central nervous system and the immune system (Magnus & Liesz, 2023). Although further insight is required into the interaction between inflammation and the central nervous system, and the role that the immune response plays after ischaemic stroke, evidence shows that ischaemic inflammation contributes significantly to the irreversible damage of neurons at the ischaemic site. While reperfusion of the site by dissolving the clot re-establishes perfusion of the infarcted area, it can also worsen injury by increasing local inflammation, causing further neuronal damage (Magnus & Liesz, 2023). In other words, this autoimmune response could potentially affect neural plasticity and functional recovery, which may contribute to the incidence of vascular dementia after stroke (Doyle & Buckwalter, 2020).

In addition to post-stroke pathogenesis exerted by activation of the immune system, brain injury after stroke also has a large effect on the system that triggers immunosuppression. It is reflected in the clinical manifestation of systemic infections, such as pneumonia and urinary tract infection, in people with stroke (Faura et al., 2021). Though further understanding is required, the above findings provide exciting avenues for further studies aiming to reduce the burden of disability and death from stroke.

1.2.2 Aetiology and Clinical Presentation

Causes of stroke include genetic factors and modifiable risk factors. Studies on genetic factors have revealed several genetic loci that are related to ischaemic stroke, suggesting that genetic variants may contribute to intracranial haemorrhage (Kuriakose & Xiao, 2020). Further studies are required under this causal category to inform the development of novel therapeutic approaches to stroke prevention.

Researchers have also identified potentially modifiable risk factors associated with stroke. These factors include hypertension, dyslipidaemia, atrial fibrillation, diabetes mellitus, current cigarette smoking, alcohol abuse, obesity, poor diet, low physical activity and psychosocial factors (Feigin et al., 2021; Kuriakose & Xiao, 2020). Furthermore, an acute stroke is usually associated with multiple risk factors. Globally, approximately 87% of stroke is attributed to these risk factors (Feigin et al., 2021). Among the risk factors, hypertension has been found to be significantly associated with all types of stroke, especially intracerebral haemorrhage. Smoking, diabetes mellitus and dyslipidaemia are important risk factors for ischaemic stroke (Feigin et al., 2021; Kuriakose & Xiao, 2020). Therefore, the prevention of stroke should consider patient education with regard to both lifestyle changes and medication management. These intervention strategies should be an integral part of discharge planning.

Despite the aetiology of stroke and the variation of associated symptoms, the most common symptoms suddenly presented by people with stroke include weakness and numbness to one side of body, involving the face, arm and leg; unilateral or bilateral visual loss or disturbance; change in cognitive status, speech or understanding; altered mobility, balance or coordination; and severe headache with an unknown cause (Freytes et al., 2021). Public education on the recognition of these symptoms helps improve clinical outcomes through the early detection and treatment of stroke (X. Chen et al., 2022). Despite mixed results, public health campaigns, such as Act FAST (face, arm, speech, time), have contributed to an increased public awareness of stroke signs and symptoms and timely stroke diagnosis (X. Chen et al., 2022).

1.2.3 Epidemiology

Stroke is recognised as a common healthcare problem and the leading cause of disability in the world (Feigin et al., 2022; Stinear et al., 2020). Globally, stroke is ranked the second most prevalent cause of death (Feigin et al., 2022) and the third main cause of death and disability combined (Feigin et al., 2022). Every year, approximately 15 million people worldwide are diagnosed with stroke, 30% of whom die, and another 30% of whom are permanently disabled because of stroke (WHO, 2023a). In the last two decades, there has been a significant increase in stroke prevalence for people younger than 70 years. It has been suggested that, although the age-standardised stroke mortality rate has decreased, the general global burden of stroke is huge and continues to increase (Feigin et al., 2021).

In Australia, stroke is considered the leading cause of death and disability (Australian Institute of Health and Welfare [AIHW], 2023). In 2021, deaths with stroke recorded as the underlying cause accounted for 6% of all deaths in Australia (AIHW, 2023). In 2018, approximately 387,000 Australians were diagnosed with stroke or recurrent stroke (AIHW, 2023). Approximately 70% of people with stroke were over 65 years of age. This number is predicted to increase with an aging population (AIHW, 2023; Stroke Foundation, 2020a). In 2020, South Australia was one of the Australian states with the highest number of people on a per capita basis experiencing first-time stroke (Stroke Foundation, 2020a).

1.2.4 Impact

Stroke has a massive economic impact. The estimated global cost of stroke is over US\$891 billion (Feigin et al., 2022). In Australia, the total economic costs of stroke were projected to be AU\$6.2 billion in 2020 (Deloitte Access Economics, 2020). The cost for people younger than 65 years with stroke is higher than for their older counterparts due to stroke-induced productivity loss, unemployment and psychosocial complications (Tan et al., 2022).

The effect of stroke is heterogeneous. Depending on the location and size of the stroke lesion, it is common for people with stroke to live with a wide range of disabilities, such as dependence on assistance to walk and to conduct activities of daily living (ADLs);

fatigue; communication or swallowing impairments; visual deficits; incontinence; cognitive and personality changes; and depression and anxiety (Bernhardt et al., 2020). About 80% of people with stroke suffer from impairments that affect control of movement or coordination of one side of the body (Ingram et al., 2021). Among non-motor impairments, the incidence of cognitive impairments is over 70% after stroke, and dementia occurs in about 15% of the stroke population. These cognitive impairments are associated with disability, dependency and morbidity, presenting a major burden to patients, caregivers and healthcare systems (Craig et al., 2022; Rost et al., 2022). In Australia, among all people with stroke with disabilities, about 88% live in households, and the rest reside in cared accommodation (AIHW, 2013). In addition, approximately 65% of people with stroke live with disabilities that require them to depend on others to conduct ADLs (Stroke Foundation, 2018). The majority of people with stroke who return home are cared for by their caregivers (Lobo et al., 2021). Stroke caregiver burden is discussed in the next section.

Apart from post-stroke disabilities, the risk for people with stroke to develop another stroke is high. A number of longitudinal studies have followed up on people with stroke for up to 5–10 years. Despite differences in the location and time of the studies, researchers have reported a similar substantial risk of recurrent stroke (Kolmos et al., 2021). Compared to the onset of the first stroke in the general population, the risk of another stroke after the initial stroke is six times higher (Hardie et al., 2004). However, the stroke recurrence rate remains controversial. The cumulative risk of recurrent stroke is 7% at 1 year, 16%–20% at 5 years and 25% at 10 years (Dhamoon et al., 2006; Hardie et al., 2004; Mohan et al., 2009). After 10 years, people with stroke have an approximately 4% accumulated risk per year of developing a subsequent stroke (Hardie et al., 2004). The good news is that, with a combination of lifestyle modifications and preventative medical treatment, the risk of recurrent stroke could be reduced by about 80% (Linsay et al., 2017). Hence, it is essential for health professionals in hospitals and the community to work collaboratively to support people with stroke in secondary stroke prevention during post-stroke rehabilitation (Stroke Foundation, 2022).

Stroke also has a significant impact on people's mental health. Among various stroke-associated neuropsychiatric disorders, post-stroke depression (PSD) is the most common. Depression refers to the feeling of persistent sadness and/or the loss of interest in activities an individual once enjoyed (WHO, 2017a). It affects approximately 30% of the stroke population, with the cumulative percentage ranging from 39% to 52% within the first 5 years after stroke and up to 70% among all people with stroke over the longer term (da Rocha e Silva et al., 2013; Robinson & Jorge, 2016). Although the relationship between lesion location and the development of PSD remains controversial, it has a significant impact on stroke recovery and rehabilitation.

Studies have reported that depression and cognitive decline are closely related. Neurological and neuroendocrine changes caused by depression may accelerate the rate of cognitive decline (Olazarán et al., 2013; van den Kommer et al., 2013). PSD has a close correlation with the severity of deficits in ADLs and cognitive function, especially executive function and social functioning. Additionally, people with stroke who develop PSD have poor functional outcomes during rehabilitation (Ahn et al., 2015; Robinson & Jorge, 2016). In their retrospective study, Ahn et al. (2015) also found that people with stroke with PSD who were cared for by informal caregivers had better outcomes compared to those cared for by paid professional caregivers. Considering the impact of PSD on the health and wellbeing of people with stroke, early identification and treatment are crucial in post-stroke rehabilitation. Education and information on how to report early signs of PSD and to seek help for timely treatment need to be included in discharge planning for these patients and their caregivers.

1.2.5 Informal Caregivers

Caregivers are people caring for those in need within the context of an existing relationship, such as a family member, a friend or a neighbour (AIHW, 2021). They play an important role in post-stroke rehabilitation and recovery. In Australia, approximately 88% of people with stroke live in households, and 65% of them live with a disability that requires assistance from caregivers to conduct ADLs (Bakas et al., 2022; Stroke Foundation, 2018). Caregivers spend, on average, over 40 hours a week caring for people with stroke (AIHW,

2013). They are unsung heroes in supporting people with stroke in rehabilitation and recovery and are often overlooked in the healthcare system regarding their need for education and psychosocial support (Camicia et al., 2023).

Studies have shown that caregivers experience stress and burnout in caring for people with stroke, and the caregiving experience affects their long-term wellbeing (Akinwuntan et al., 2021; Camicia et al., 2023). During the hospital-to-home transition, caregivers have to commence and adapt to the caregiving role and start to assist people with stroke with ADLs, health condition management and navigate the healthcare system simultaneously. Most have never been a caregiver and, pre-morbidly, had shared household tasks, including financial management, with people now with stroke. Young adult caregivers also have competing duties to their own family as well as their job (Camicia et al., 2023; L. Chen et al., 2020). L. Chen et al. (2016) found that caregivers felt vulnerable and helpless soon after discharge when they were left alone to navigate the healthcare system. In their study, some caregivers felt unprepared, with a lack of caregiver training prior to discharge, and rehabilitation and care services were not designed to accommodate their daily routines. They had to change their daily routines to meet service provision times.

Interventional studies to improve the stroke caregiver experience and reduce caregiver stress have demonstrated some effect. Lin et al. (2022) discovered positive outcomes of a nurse-led intervention for stroke dyads in reducing caregiver burden and unplanned hospital readmission. They recommended a health coaching approach incorporating stroke-related knowledge and self-efficacy building. A meta-analysis of dyadic psychosocial interventions confirmed the above positive outcomes and suggested the significance of early implementation of face-to-face interventions and continuing out-of-hospital support (X.-Y. Zhang et al., 2023). Despite the positive effects, the result of these studies may not be generalisable due to limitations in methodology and the heterogeneity of interventions. Developing individualised interventions to alleviate stress during the hospital-to-home transition is, therefore, a gap in research. Besides this, few studies have specifically focused on the hospital-to-home transition. Consequently, how to realise the early detection

of caregivers' needs and problem-solving during the transition to sustain their caregiving role has yet to be explored.

1.2.6 Dyadic Interdependence

Interdependence presents in a relationship when one person's emotion, cognition, or behaviours affects that of their partner (Cook & Kenny, 2005). As discussed above, stroke affects both the person with stroke and their caregiver and changes family relationships. A person with stroke and their caregiver form a caregiving dyad (Jacops et al., 2022). An important attribute of the caregiving dyad is interconnected mutuality: dyadic members experience and follow a trajectory of the disease together. Their relationship reflects dyadic interdependence, leading to collective emotions and coping (Boyer et al., 2017; Jacops et al., 2022).

Stroke intervention studies have revealed positive outcomes of dyadic interventions. Psychosocial interventions have been found to improve depressive symptoms in stroke dyads (Minshall et al., 2020). The systematic review and meta-analysis by Pucciarelli et al. (2021) indicated that educational interventions improve physical functioning, quality of life and memory in people with stroke and relieve depressive symptoms in caregivers. The reviewers suggested that these interventions should be dyad-focused instead of just dyadbased (Pucciarelli et al., 2021). Another review examining the effectiveness of dyadic psychoeducational interventions showed improvement in functional independence and longterm quality of life for people with stroke and in improving their family caregivers' burden (Mou et al., 2021). However, both reviews suggested further well-designed large-scale trials, prior to implementation in clinical practice, to determine the effectiveness of dyadic interventions in diverse functional and psychosocial health outcomes for stroke dyads (Mou et al., 2021; Pucciarelli et al., 2021). Despite the positive findings regarding these dyadic interventions, their effectiveness is inconclusive. Understanding the shared experiences of stroke dyads as well as their unique individual needs would guide the design of future dyadic interventional studies.

1.3 Post-stroke Services

1.3.1 Stroke Rehabilitation

In 2001, the introduction of the International Classification of Functioning, Disabilities and Health (ICF) provided a new conceptual framework for assessing post-stroke disabilities and facilitating the communication of multidisciplinary teams within stroke care (Quintas et al., 2012; Tempest et al., 2013). The definition of stroke-associated disabilities is not only related to pathology and impairment but also considers the individual's interaction with environmental and social circumstances (WHO, 2022a). Thus, the new definition has significant influence on post-stroke rehabilitation: physical recovery is no longer the only focus. The psychological, social and financial challenges faced by people with stroke and their families also need to be addressed during rehabilitation (Stroke Foundation, 2022).

Post-stroke recovery is a complex, heterogeneous process and is associated with the size and site of stroke lesions. Recovery occurs through processes involving 'restitution (restoring the functionality of damaged neural tissue), substitution (reorganisation of partly spared neural pathways to relearn lost functions), and compensation (improvement of the disparity between the impaired skills of a patient and the demands of their environment)' (Langhorne et al., 2011, p. 1693). Although people with stroke experience spontaneous recovery for weeks and months after initial stroke, rehabilitation therapies that are designed to target specific areas of the central nervous and musculoskeletal system help support or induce these recovery processes (Carey et al., 2019; Stinear et al., 2020).

Rehabilitation is an integral part of stroke management. Stroke rehabilitation refers to a proactive, person-centred and goal-oriented process facilitated by a multidisciplinary team in a culturally appropriate environment and a health strategy aimed at optimising function and enabling social reintegration for people with stroke (Stroke Foundation, 2022). The multidisciplinary team provides rehabilitation interventions with a holistic approach through a cyclical process including 'assessment, goal setting, intervention and reassessment' (Brewer et al., 2013, p. 14).

The rehabilitation process is strongly influenced by support from the health and social care systems. In the current atmosphere of shortened hospital stays and early discharge in healthcare reform, stroke dyads are expected to encounter additional challenges in recovery and rehabilitation after discharge from hospital (Camicia, Lutz, Summers et al., 2021; L. Chen et al., 2020). Carefully structured pathways to enable them to cope with challenges and to seek help from health professionals when they need it are crucial for successful recovery and rehabilitation. Moreover, given the heterogeneity and complexity of post-stroke impairments, rehabilitation interventions are highly individualised and strongly influenced by the care needs and expectations of people with stroke. Engaging stroke dyads in the process is the way to identify and achieve their rehabilitation (Stroke Foundation, 2022; WHO, 2023b).

1.3.2 Post-stroke Rehabilitation and Community Care Services

In Australia, the National Rehabilitation Stroke Services Framework (Stroke Foundation, 2022) outlines essential principles of post-stroke rehabilitation services, stating that all eligible people with stroke are entitled to participate in individualised rehabilitation services. Post-stroke rehabilitation services include inpatient stroke rehabilitation services; ambulatory rehabilitation services encompassing day hospitals, outpatient clinics and homebased services, including early supported discharge services; and telemedicine rehabilitation support (inpatient or community settings).

Telerehabilitation is a model of rehabilitation services that uses information and communication technologies to remotely deliver rehabilitation consultations, assessments and therapies. It forms part of various post-stroke services in South Australia (Laver et al., 2022; SA Health, 2023).

People with stroke can also access transition services. The Transition Care Program, also known as after-hospital care, provides a range of services, including therapy, nursing and personal care assistance and can be delivered in-home or in a facility (SA Health, 2022). The aim is to minimise unnecessarily extended hospital stays, optimise a person's

function and determine their long-term care arrangements. The Transition Care Program is a non-stroke-specific program that is goal-oriented and time-limited (up to 12 weeks) to support eligible people aged 65 years and above (45 years and above for Aboriginal and Torres Strait Islander people; SA Health, 2022).

Furthermore, eligible people meeting the same age criteria can access governmentfunded home care services through My Aged Care (2020). Home care services provide complex support, including personal care, support services, and nursing, allied health and clinical services through the coordinated and consumer-directed home care packages program. There are four levels of assistance, from basic care needs to high care needs (My Aged Care, 2020).

Despite their availability, these packages are difficult to access and navigate (Royal Commission into Aged Care Quality and Safety, 2021). The waiting list is too long to get access to care at home. In 2018–19 financial year, it took at least 7 months to access a Level 1 package and took as long as 34 months for a Level 4 package between the time of assessment and when a home care package was assigned (Royal Commission into Aged Care Quality and Safety, 2021).

Additionally, older people often lack the knowledge to navigate the complex agedcare services to make informed decisions about care services meeting their specific needs (Royal Commission into Aged Care Quality and Safety, 2021). The situation is further exacerbated for those with disability accessing services matching their needs due to the availability of these services only for people younger than 65 years of age (Royal Commission into Aged Care Quality and Safety, 2021).

Moreover, people can also access the Commonwealth Home Support Programme, which provides entry-level service to help with housework, shopping, personal care, meals, transport and short-term respite care to relieve people and their caregivers from usual care arrangements. Short-term restorative care services are also available to support people returning to independence in the home (My Aged Care, 2020).

In Australia, people with stroke who are younger than 65 years of age can access the government-funded disability service, the National Disability Insurance Scheme (NDIS). The NDIS provides funding to eligible people with disability to enable independent living, new skills, jobs or volunteering in their community to lead an improved quality of life (National Disability Insurance Agency, 2022). The NDIS funds a wide range of supports, from household tasks, equipment assessment and provision, to therapeutic support and accommodation (National Disability Insurance Agency, 2022). However, people with disability wait months or years in hospitals to be discharged with NDIS services due to a long application review and approval process (Winkler, 2022). This prolonged waiting in hospitals could have negative psychological impacts on stroke dyads during the transition.

Despite the availability of diverse stroke rehabilitation and community care services for stroke dyads, the National Stroke Audit 2021 on acute services identified that about 24% of people with stroke leave hospital without a discharge care plan (Stroke Foundation, 2021). Furthermore, only approximately 60% of people with stroke were discharged to the community with tailored information regarding stroke rehabilitation and recovery. In addition, as much as 80% of people with stroke did not receive information on post-stroke intimacy (Stroke Foundation, 2020a). These audit results highlight an inadequate preparation of stroke dyads prior to hospital-to-home transition. These results also suggest that an individualised discharge plan and post-discharge follow-up co-developed by stroke dyads and health professionals are imperative for improving the hospital-to-home transition for people with stroke.

1.3.3 Nurses' Role in Post-stroke Rehabilitation

Nurses play a crucial role in post-stroke rehabilitation. Working with stroke patients 24 hours a day, every day, rehabilitation nurses act as teachers, coaches and caregivers to help patients and their caregivers translate the knowledge they learn from therapists into a daily living environment. They also proactively prevent and manage clinical deterioration (Australasian Rehabilitation Nurses' Association [ARNA], 2018; Camicia, Lutz, Summers et al., 2021). As a key member of the multidisciplinary team, they usually act as care

coordinators and case managers in the team to ensure a safe transition from hospital to the community (ARNA, 2018; Camicia, Lutz, Summers et al., 2021).

Despite the vital role of nurses in post-stroke rehabilitation, L. Chen et al. (2016) found that nurses were rarely mentioned by stroke dyads in the transition period. This was echoed by Loft et al. (2019), who found that nurses' role in post-stroke care was yet to be fully acknowledged. They also highlighted the ideal position that nurses are in regarding the provision of holistic care in stroke rehabilitation and advocated for strengthening nursing education on nurses' contribution to rehabilitation to enhance psychosocial support to stroke dyads. Camicia, Lutz, Summers et al. (2021) reaffirmed nurses' essential role in facilitating the transition of care and suggested the role of stroke nurse liaison/navigator to enhance the transition care.

The role of stroke nurse coordinator has been recommended in stroke management and the continuum of care (Camicia, Lutz, Summers et al., 2021; Stroke Foundation, 2023). George et al. (2017) found that nurses activated stroke alerts significantly earlier than physicians in the hospital setting, which greatly improves thrombolytic treatment after stroke onset. In Australia, an observational study revealed a positive association between the presence of stroke coordinators (mainly nurses), the hospital length of stay and the delivery of evidence-based care in a stroke unit (Purvis et al., 2018). In addition, a recent review on stroke nurses' contributions to mobile stroke units provided promising evidence to further expand stroke nurse coordinators' role to the pre-hospital context (Coote et al., 2022).

Furthermore, the role of nurse navigator has emerged in the management of cancer and chronic conditions, including stroke. Cancer nurse navigators enhanced support for patients with cancer early in their course, improved patient experience and reduced problems in care (Wagner et al., 2014). The role of nurse navigator also generated cost savings and improved the quality of perioperative care for patients undergoing elective total knee or hip arthroplasty (Phillips et al., 2019). Moreover, stroke nurse navigators' interventions showed positive outcomes on compliance with medication and follow-up appointments, reduced rehospitalisation and improved quality of life 12 months after hospital
discharge (Deen et al., 2016). The growing role of stroke nurse navigators provides an evolving model of care that may enhance the quality of care and transition experiences for stroke dyads.

It is worth noting that the above-reviewed studies mainly focused on acute care in hospital settings or in the community, lacking follow-up on the continuum of transition. The follow-up period for most studies was limited, except for the 12-month intervention in Deen et al. (2016). However, the main focus of this study was stroke patients' self-reported postdischarge compliance and quality of life measures (Deen et al., 2016). Transition is a nonlinear process, and care needs change over time (Meleis et al., 2000). Hence, it is necessary to obtain further insights into the role of stroke nurses, targeting the individual and dyadic care needs of people with stroke and their caregivers throughout the post-stroke trajectory, with a specific focus on the hospital-to-home transition.

1.4 Aim and Objectives

The aim of the present study was to explore the care needs and expectations of stroke dyads regarding the hospital-to-home transition in an Australian social context. The objectives were to:

- understand the experience and perceptions of stroke dyads during the hospitalto-home transition in a global context
- understand stakeholders' perspectives on approaches to identifying and meeting the transition care and rehabilitation needs of people with stroke in an Australian social context.

1.5 Research Design

This was a two-phased study. In Phase 1, a systematic review using meta-synthesis of qualitative studies was undertaken to address Objective 1. In Phase 2, a qualitative study informed by Gadamer's phenomenological hermeneutics was conducted to address Objective 2. In-depth interviews were conducted with stroke dyads and focus groups and

interviews with health professionals involved in transition care and rehabilitation services to collect data.

1.6 Significance of the Study

Hospital-to-home transition is a challenging time for stroke dyads. The shortened hospital stays and early discharge due to healthcare reform add more challenges to transition care and may result in unintended consequences: for example, post-strokeassociated complications, such as falls, fall-related injuries, infections and functional declines; poor quality of life for stroke dyads; and increased readmission, health service utilisation and premature admission to residential care facilities during the transition. Despite abundant studies on issues and challenges faced by stroke dyads during the transition, studies determining post-stroke hospital-to-home transition care needs and the selection of services to meet these needs largely lack depth. In addition, studies on the experiences of stroke dyads have been descriptive of issues with little focus on shared experiences for stroke dyads and on the most appropriate way to meet their individual and dyadic needs. Furthermore, studies from the perspective of health professionals regarding evidence-based frameworks to determine and meet post-stroke rehabilitation and care needs during the transition are scarce. Finally, in spite of a sophisticated stroke management system in Australia, there is little research exploring stroke dyads' perspectives of the system during the transition.

The present study is significant as it would offer valuable insights into transition care, a crucial aspect for stroke dyads during their journey from hospital to home, from health to illness, and from their roles as spouses/parents to those of caregivers and care recipients. The study would advance the body of knowledge in the field of rehabilitation and care needs during poststroke hospital-to-home transition. In particular, the longitudinal approach of the study would enhance the depth of understandings on the changes of care needs during the transition. Furthermore, perspectives from both consumers and health professionals would contribute to the delivery of consumer-guided and person-centred poststroke rehabilitation

services. This study would also help establish and enhance advanced stroke nurses' role in the field of ever-evolving contemporary nursing profession.

The present study is expected to contribute to knowledge by developing deeper insights of care needs and the approaches to identify and meet the needs of stroke dyads in poststroke hospital-to-home transition care. Employing a longitudinal qualitative approach is expected to address gaps in the research domain by observing and analysing participants' experiences over an extended period, which would empower the researcher to capture changes and develop a comprehensive and nuanced understanding of participants' lived experiences. The temporal dynamics inherent in longitudinal studies would also allow the researcher to track the evolution of transition experiences to help develop insights into how and why changes occur (Holland et al., 2006). Moreover, including health professionals working in different specialised disciplines in the study would enable insights into health professionals' perceptions of transition care for stroke dyads as they transition through different stages of recovery. Collecting data from multiple sources across time, space and person is considered data triangulation, which enables the researcher to validate data and gain multiple perspectives of the studied phenomenon (Denzin & Lincoln, 2018). This methodological strategy creates an innovative way to understand the richness and complexity of stroke transition experiences and addressed gaps in stroke transition care research.

1.7 Thesis Structure

This thesis is presented in 12 chapters. The current chapter, Chapter 1, outlines stroke as a health condition in the global and Australian contexts, detailing the pathophysiology, risk factors, management and prevention of stroke. Post-stroke rehabilitation and community care services for stroke dyads in Australia are elaborated, together with challenges in current healthcare systems relevant to the hospital-to-home transition. This chapter also introduces the significance, aim and objectives of the study.

Chapter 2 includes a published systematic integrative review that explored the challenges and opportunities for stroke dyads in hospital-to-home transition care. The review revealed that the transition care experiences of stroke dyads are influenced by health and social care system service coordination and health organisation resources in supporting health professionals' partnership and engagement with stroke dyads, and by stroke dyads' level of readiness to cope with challenges. Tailored supports for developing resilience in stroke dyads was emphasised.

Chapter 3 reviews the concepts and theories in transition care studies. Meleis's middle-range and situation-specific transitions theory is adapted for this study. In a systematic approach, Meleis's transitions theory guides the development of an interview guide and data analysis to help generate comprehensive insights into the transition experiences of stroke dyads and into health professionals' perceptions.

Chapter 4 incorporates a published systematic review and meta-synthesis that identified the enablers and barriers in hospital-to-home transition care for stroke dyads. The review underscored that partnerships with stroke dyads facilitate discharge preparation, fostering self-management capabilities. Furthermore, a lack of post-discharge support, coupled with inappropriate discharge planning, contributes to unmet needs for stroke dyads after returning home. Hence, stroke dyads expect integrated transitional care that empowers them to be involved in decision-making and enables long-term self-management in the community.

Chapter 5 discusses the methodology and methods used in this study. It describes Gadamer's philosophical hermeneutics and the methodological rationale that informed the research design and methods. It also explains participant selection, data collection and analysis methods, the sample size and ethical considerations.

Chapter 6 introduces the organisation of the findings chapters (Chapters 6–9), including participants' demographics and a summary of major themes and subthemes. This chapter also includes a published article that presents Part 1 of the findings, from interviews before hospital discharge, which encompasses three major themes and subthemes: (1)

psychological stress, (2) resilience and (3) certainty and uncertainty about post-discharge care. The first theme emphasises psychological stress with different causes among stroke dyads. The second focuses on stroke dyads' resilience, fostered by their own positive thinking and support from others. The third theme highlights stroke dyads' inconsistent level of certainty about post-discharge care; they prefer to co-develop and co-implement discharge plans with health professionals.

Chapters 7 and 8 present the study's findings under four major themes, identified from the interviews at 3 and 6 months after hospital discharge. Chapter 7 interprets the first two themes and subthemes: (4) coping with setbacks and (5) role transformation. Theme 4 reflects participants' perceptions of the impact of stroke on their psychological wellbeing and the resilience they demonstrated to reconcile and move forwards with the changes in their lives. Theme 5 focuses on stroke participants' self-management development and caregivers' evolving ability to build caregiving capabilities. The dyadic effects in dealing with challenges enabled them to transform and adapt to their post-stroke lives.

Chapter 8 interprets the remaining two major themes and their subthemes: (6) transitioning with support and (7) expectations in hospital-to-home transition care. Theme 6 presents the support that stroke dyads received from family, friends and health professionals. Positive effects arising from the support between dyadic members are identified. Theme 7 presents stroke dyads' expectations to be equal partners in discharge planning and their desires for flexible service provision to meet their individual needs.

Chapter 9 details the three major themes and subthemes that emerged from interviews and focus groups with health professionals. The themes are (1) transition shaped by care needs assessment, (2) transition challenged by constraints on service delivery and (3) transition enhanced by a commitment to the safety of care. Theme 1 highlights the significance of stroke care needs assessment and identifies a lack of caregiver involvement and an inadequate focus on psychological needs. Theme 2 interprets the challenges that health professionals encountered in supporting stroke dyads, including shortened length of stays and gaps in service accessibility, that may contribute to a mismatch between care

needs and service provision. Theme 3 demonstrates a safety culture fostered through leadership and team resilience that may enhance and facilitate healthy transitions for stroke dyads.

Chapter 10 discusses the major findings of this study in an Australian and international context. These findings are compared with recent studies, looking for similarities and differences. The discussions in this chapter focus on the significance of balancing functional rehabilitation and psychological support for stroke dyads to facilitate the transition and on the benefit for health professionals in developing insights into dyadic interdependence to improve dyad-centredness and the engagement of stroke dyads in their own care planning. Moreover, the discussion highlights opportunities to enhance stroke nurses' contributions to facilitate the transition and relates Meleis's transitions theory to the study's findings that enrich the theory with stroke-specific perspectives.

Chapter 11 proposes a novel stroke-nurse-navigator-enabled hospital-to-home transition care framework. The framework emphasises gaps in the transition process, opportunities to bridge the gaps and the interprofessional collaboration and partnership with stroke dyads during the transition that is enabled by stroke nurse navigators.

Finally, Chapter 12 summarises the key research findings and highlights the main discussion points. It critically evaluates the limitations of the study and outlines implications and recommendations for practice, policy and future research and provides a reflective account of the researcher's experience.

1.8 Summary

This chapter has discussed the background of the study and provided an overview of stroke and stroke-transition-care-related challenges and opportunities in the global and Australian contexts. The aim and objectives of this study were introduced, with discussions on the significance of this study. The next chapter presents an integrated literature review to establish current knowledge and identify gaps in the research area.

Chapter 2: Integrative Review

2.1 Introduction

As introduced in the previous chapter, hospital-to-home transition is the most stressful period for stroke dyads, especially in the context of shortened hospital stays and early discharge. They face enormous challenges in adapting to their home care environments for continuing recovery after discharge. They also encounter great difficulties in exploring suitable rehabilitation and care services. This chapter presents a publication: an integrative review of the research to establish current knowledge and identify gaps in the research area.

The contribution of the first author to the paper is 70% to research design, writing and editing and 80% to data collection and analysis. The second and third authors together contributed 30% to research design, writing and editing and 20% to data collection and analysis.

This section presents a publication arising from this thesis that was published in the *Journal of Advanced Nursing*.

Citation: Chen, L., Xiao, L. D., & Chamberlain, D. (2020). An integrative review: Challenges and opportunities for people with stroke and caregivers in hospital to home transition care. *Journal of Advanced Nursing*, *76*(9), 2253–2265.

https://doi.org/10.1111/jan.14446

This is the peer reviewed version of the following article: An integrative review: Challenges and opportunities for people with stroke and caregivers in hospital to home transition care, which has been published in final form at https://doi.org/10.1111/jan.14446. This article may be used for non-commercial purposes in accordance with Wiley Terms and Conditions for Use of Self-Archived Versions. This article may not be enhanced, enriched or otherwise transformed into a derivative work, without express permission from Wiley or by statutory rights under applicable legislation. Copyright notices must not be removed, obscured or modified. The article must be linked to Wiley's version of record on Wiley Online Library and any embedding, framing or otherwise making available the article or pages thereof by third parties from platforms, services and websites other than Wiley Online Library must be prohibited.

2.2 An Integrative Review: Challenges and Opportunities for People With Stroke and Caregivers in Hospital to Home Transition Care

2.2.1 Abstract

Aim: To identify challenges and opportunities for people with stroke and caregivers in hospital to home transition care.

Background: Due to shortened hospital stays, people with stroke and caregivers must take responsibility for complex care on discharge from hospital to home. Gaps exist in the literature that synthesises studies on hospital to home transition care.

Design: A systematic integrated review.

Data sources: Six databases were searched systematically between 18 June 2018 to 31 October 2018 including Medline, CINAHL, Web of Science, ProQuest, Scopus and Science Direct. The search did not have a date limit.

Review methods: Studies that met the selection criteria were critically reviewed. Data were extracted from the studies for analyses. A convergent qualitative synthesis approach using inductive thematic synthesis was applied to the review.

Results: The analysis of 23 studies identified three major findings. First, health and social care systems influence transition care by either enabling people with stroke and caregivers to manage transition care via well-coordinated services or preventing them from accessing services. Second, health professionals' partnership with people with stroke and caregivers largely decides tailored support for them. Successful partnerships and engagements with people with stroke and caregivers depend on organisational resources.

Third, survivors and caregivers are at different levels of readiness to cope with challenges. Individualised support for them to develop resilience is highly regarded.

Conclusion: People with stroke and caregivers encounter enormous challenges in self-management of hospital to home transition care. Further research is required to address their expectations of support during transition care.

Impact: There is a lack of synthesis of studies on factors affecting hospital to home transition care for people with stroke. Health and social care system designs, health professionals' commitment to individualised care and the self-management capability of people with stroke and their caregivers have a profound influence on the transition care experiences.

Keywords: Stroke, transition care, survivors, caregivers, nurses, health professionals, integrated review

2.2.2 Introduction

Worldwide, 15 million people suffer from stroke (cerebrovascular disease) per annum and 5.8 million die from it (World Stroke Organization [WSO], n.d.). Despite a reduction in the age-standardised stroke mortality rate over the past two decades, the overall burden of stroke is substantial and continues to increase with an ageing population (Feigin et al., 2017; Kunst et al., 2011). Most people with stroke are discharged to home (Dutta et al., 2018; Ouellette et al., 2015). In the global context of shortened hospital stays and early discharge due to cost-saving pressure, people with stroke and caregivers are expected to manage more complex healthcare issues at home (Deloitte, 2018; Langhorne et al., 2017). However, they are largely unprepared with few self-management capabilities to take charge of transition care (Govender et al., 2019; Piccenna et al., 2016). The synthesis of studies exploring this increasingly important research field is much needed to gain a more comprehensive understanding of issues.

2.2.3 Background

Poststroke rehabilitation and recovery is a chronic process that requires ongoing adjustment and interaction with healthcare systems and service providers. Studies indicate

unmet needs of people with chronic diseases due to healthcare system failures (Kadu & Stolee, 2015). The Chronic Care Model (CCM) was developed to transform the healthcare system from reactive to proactive and activate patients' self-management capabilities (WHO, 2016). The CCM consists of six interrelated domains: community, health system, self-management support, delivery system design, decision support and clinical information systems (Accelerating Care Transformation Centre, 2023). The goal of the model is to provide high-quality care and improved outcomes through productive interactions between informed, activated patients and prepared, proactive practice teams supported by community resources and health care organisations (Accelerating Care Transformation Centre, 2023). This review adapted the CCM as a theoretical framework to analyse challenges and opportunities at the healthcare system, organisation, practice, and patient levels for people with stroke and family caregivers after hospital discharge.

Discharge planning is a crucial process to identify education and skill needs for people with stroke and caregivers to take responsibility for rehabilitation and recovery after discharge (Andrew et al., 2018; Waring et al., 2016). Yet, studies reveal people with stroke and caregivers are not actively involved in the discharge planning process, nor receive adequate education and caregiving skills training (L. Chen et al., 2016; Kable et al., 2019). Inadequate patient and caregiver education in discharge planning contributes to increased risk of deterioration and unnecessary hospital readmissions (Reeves et al., 2017; Ulin et al., 2016).

Due to functional impairments, people with stroke must learn to adapt to the home environment after hospital discharge (Connolly & Mahoney, 2018; Hodson et al., 2016). Resilience is described as the "processes and skills that result in good individual and community-health outcomes despite negative events, serious threats and hazards" (WHO, 2017b, p. 6). Survivors and caregivers with resilience are capable of adapting to and coping with poststroke care challenges at home (Z. Liu et al., 2019; Martz & Livneh, 2016). A sense of hope, past life experiences, learning over time and availability of supportive resources are some of the determinants of resilience (Aburn et al., 2016; American Psychological

Association, 2019). People with stroke experience self-stigma due to invisible impairments such as fatigue and depressive symptoms (Balasooriya-Smeekens et al., 2016; Wainwright et al., 2017). This self-stigma may prevent them from taking proactive actions to develop their resilience. Caregivers play a key role in providing everyday care and coordinating care services. For them, resilience acts as a moderator between stress and burnout (Epstein, 2015; Taku, 2014). Programs that support the caregiving role in the context of early discharge are scarce (Atteih et al., 2015; Pindus et al., 2018). This situation may affect their ability to develop resilience to sustain transition care at home.

Studies reveal a mismatch of rehabilitation goals between people with stroke and health professionals (Connolly & Mahoney, 2018; Koh et al., 2014). This care situation is attributed to the lack of partnership with survivors and caregivers in care plan development (L. Chen et al., 2016; Plant et al., 2016). Health professionals' desire for partnership with survivors and caregivers to implement the recommendations of stroke guidelines is limited by a lack of institutional support (Baatiema et al., 2017; Munce et al., 2017). Organisational structures and resource constraints prevent health professionals from delivering flexible and tailored rehabilitation, education and support to survivors and caregivers (Baatiema et al., 2017; Harrison et al., 2017).

Studies with various research designs have explored issues of concern in poststroke hospital to home transition care. However, synthesised research evidence about how best to support people with stroke and caregivers in hospital to home transition care in the first six months post-discharge are scarce. This systematic integrated review addresses this gap in the literature. Findings will inform the design of future research and service development.

2.2.4 The Review

2.2.4.1 Aim

The aim of this systematic integrated review is to synthesise research evidence on challenges and opportunities for people with stroke and caregivers in hospital to home transition care.

The review questions were:

- What are the challenges faced by people with stroke and caregivers in hospital to home transition care?
- 2. What are the opportunities to improve hospital to home transition care?

2.2.4.2 Design

This mixed-studies systematic review applied an integrated design with a convergent qualitative synthesis approach (Pluye & Hong, 2014). This approach enables the synthesis of evidence generated from diverse research methodologies on a topic "to maximise findings -and the ability of those findings to inform policy and practice" (Pearson et al., 2015, p. 122). Quantitative data including those from randomised controlled trials or quantitative components of mixed-methods studies are usually transformed into qualitative themes for synthesis (Pearson et al., 2015; Pluye & Hong, 2014). The review protocol was registered on PROSPERO with registration number CRD42019136068 (Appendix 1).

2.2.4.3 Search Methods

A systematic electronic databases search was conducted between 18 June 2018 and 31 October 2018 for all English articles in six databases including Medline, CINAHL, Web of Science, ProQuest, Scopus and Science Direct. There were no limitations on the year of publication except for ProQuest and Web of Science. The limiter of publication dates between 2008 and 2018 was applied to these two databases due to the large number of articles generated. Search terms used were stroke, cerebrovascular accident, acquired brain injury, survivor, patient, caregiver, health professional, service provider, transitional care, discharge, continuity of care and community services (Appendix 2). We also manually searched reference lists for studies meeting the criteria.

2.2.4.4 Search Outcome

At least two independent reviewers conducted the screening and review of studies for eligibility (LD and either LX or DC). The initial search generated 5855 articles. After removing duplicate citations, 5773 article titles and abstracts were screened. Of those, 258 full-text studies were assessed for eligibility based on the inclusion/exclusion criteria (Table 2.1). Throughout the selection process, consensus was achieved on discrepancies through

discussion. In total, 23 studies (Table 2.2, Appendix 3) were selected for appraisal including 20 studies identified from the electronic database search and three studies identified through the manual search of reference lists (Figure 2.1).

Table 2.1

Inclusion and Exclusion Criteria

Inclusion criteria	Exclusion criteria
Original qualitative, quantitative,	Pilot studies, scoping
mixed method studies	reviews, narrative reviews,
	systematic reviews, reports,
	conference proceedings,
	discussions or commentaries.
Adult people with stroke, caregivers	People with stroke
and health professionals aged 18	younger than 18 years old.
years and above.	Multiple patient population
	including brain tumour,
	traumatic brain injury,
	other neurological
	condition and/or other
	diagnosis where stroke
	specific data cannot be
	separated.
The experience, care needs,	Poststroke long term
expectations of people with stroke,	experiences and needs
their caregivers and health	(over 6 months after
professionals during hospital to	hospital discharge).
home transition or a transition	
	Inclusion criteria Original qualitative, quantitative, mixed method studies Adult people with stroke, caregivers and health professionals aged 18 years and above. The experience, care needs, expectations of people with stroke, their caregivers and health professionals during hospital to home transition or a transition

Items	Inclusion criteria	Exclusion criteria
	facilitating intervention up to 6	Discharges to facilities
	months after discharge.	other than home.
		Studies conducted across
		multiple period of stroke
		trajectory where the
		transition period cannot be
		separated.

Figure 2.1

Review Flow Diagram



Table 2.2

Brief Summary of the Selected Articles

First author,	Brief summary of the selected articles (full description in Appendix
year, country	3)
of origin	
1. Almborg	Cross sectional study examining stroke patients' perceived participation
(2008), Sweden	in the discharge planning process. Descriptive findings showed poor
	engagement in decision making and goal setting.
2. Almborg	Cross sectional study examining stroke patients' relatives' perceived
(2009), Sweden	participation in the discharge planning process. Descriptive findings
	showed 80% perceived no participation.
3. Bakas	Qualitative study exploring self-reported needs and concerns of stroke
(2002), USA	family caregiver after discharged home. Findings indicated needs
	and concerns on various aspects of care.
4. Cameron	Qualitative study exploring perceptions on weekend pass – a discharge
(2014), Canada	planning tool for stroke patients. Weekend pass provided in home
	practice and informs future inpatient rehabilitation.
5. Chouliara	Qualitative study investigating perceived facilitator and inhibitors for a
(2014), UK	stroke supported discharge service. Findings highlighted specificity
	of the service and fragmented stroke care pathway.
6. Cobley	Qualitative study evaluating experiences of people with stroke and
(2013), UK	caregivers on an Early Supported Discharge (ESD) service. Satisfied
	with home-based rehabilitation but post-ESD transition disjointed.
7. Ellis-Hill	Qualitative study to understand poststroke hospital to home transition
(2010),	experience. Recovery needs changed overtime. Stakeholders'
UK	shared models of recovery were needed.

First author,	Brief summary of the selected articles (full description in Appendix
year, country	3)
of origin	
8. Greenwood	Qualitative ethnography exploring stroke caregivers' experience in the
(2009), UK	first 3 months after discharge. Findings revealed uncertainty most
	striking for new caregivers and situation improved over time.
9. Gustafsson	Qualitative study to understand the first month post-discharge
(2013),	experience. Strategies were identified to better prepare for
Australia	discharge. Caregiver support needs were highlighted.
10. Grant	Quantitative survey to identify issues and associated feelings for stroke
(2006), USA	caregiver 2-3 months post-discharge. Caregivers' needs changed
	during transition and coped with caregiving better overtime.
11. Hall (2012),	Mixed method study exploring transition experiences of nontraumatic
Australia	brain injury patients. At 6 months, physical and emotional function
	remained the same for 50% patients with lack of community support
	and difficulty navigating healthcare system.
12. Lou (2017),	Qualitative study investigating experience of ESD for mild stroke
Denmark	patients and partners. ESD supported a safe transition home but
	lacked focus on goal of returning to work.
13. Nordin	Qualitative study describing stroke patients' expectations of coming
(2015), Sweden	home with home rehabilitation service. Mixed feelings with unrealistic
	high expectations of functional status.
14. O'Brien	Qualitative study examining healthcare providers' (HCP) perceptions of
(2014),	stroke caregiver needs. Support for caregivers were crucial at
Australia	discharge point. HCPs' support to caregivers may be challenged by
	workload.

First author,Brief summary of the selected articles (full description in Appendixyear, country3)of origin

15. Perry	Quantitative survey to describe stroke caregiving 1- and 3-months post-
(2011),	hospital discharge. Services were least satisfactory where there
Australia	were most needs. Pre-discharge involvement was lacking.
16. Pringle	Qualitative study to understand the first month discharged home
(2013),	experiences. One-month post-discharge a dynamic time. Family and
UK	social support facilitated reconnection to community.
17. Rittman	Qualitative study describing psychosocial experiences the first month
(2007), USA	after discharge. Survivors with high functional assessment scores
	could benefit from interventions preparing them for home.
18. Shyu	Randomised controlled trial examining the effects of a discharge
(2008), Taiwan	planning program. Experimental group had significantly better
	preparation and greater satisfaction with meeting discharge needs.
19. Simeone	Qualitative study describing experience of people with stroke 3 months
(2015), Italy	after discharge home. People with stroke experienced deeply
	changed life and viewed self as a burden to the family.
20. Wong &	Randomised controlled trial testing the effects of a stroke transitional
Yeung (2015),	care intervention. Significantly better outcomes for intervention group
Hong Kong	in physical and mental wellbeing.
21. Wood	Qualitative study exploring perspectives on the process of community
(2010), Canada	reintegration. Community reintegration is a process of transitioning
	through series of goals. More participation with time.
22. Wottrich	Qualitative study describing expectations of people with stroke 3
(2012), Sweden	months after discharge home. Expectations changed over time. In-
	home practice beneficial prior to discharge.

First author,	Brief summary of the selected articles (full description in Appendix
year, country	3)
of origin	
23. Young	Qualitative study exploring the needs of spousal caregivers during
(2014), USA	hospital to home transition. Unprepared caregivers would benefit
	from an early caregiver assessment with transitional planning.

2.2.4.5 Quality Appraisal

Critical Appraisal Skills Programme (CASP) tools were used to evaluate qualitative studies, qualitative component of the mixed-method study and RCTs (CASP, 2018a, 2018b). The quality of cross-sectional descriptive quantitative studies and quantitative component of the mixed-method study was assessed with a set of questions developed by Büttner and Muller (2015) for appraising epidemiological studies. No scores were placed during the appraisal. Researchers evaluated the quality of each study independently until a consensus was reached. All studies that met the selection criteria were included as every study provides a valuable perspective to achieve the aim of this review (Appendix 4).

2.2.4.6 Data Abstraction

Extracted data included participant quotations from qualitative studies and qualitative component of the mixed-method study, and numerical, narrative findings and quantitative component of the mixed-method study. Extracted data were organised into method-specific tables (see Appendices 5 and 6).

2.2.4.7 Data Synthesis

Informed by the convergent qualitative synthesis approach (Pluye & Hong, 2014), data synthesis was achieved using the inductive thematic synthesis method originally developed by Nowell and colleagues (2017) and adapted by Treacy and Stayt in their review (2019). The inductive thematic synthesis in this review includes 6 steps. First, we extracted data from selected articles that were relevant to the review question and presented data in method-specific data extraction tables for comparison and analysis in parallel (see

Appendices 5 and 6). Second, two researchers independently analysed and coded data from each article. Third, codes were collated into group codes and initial themes were identified and discussed in team meetings based on the review question and the theoretical framework. Fourth, codes, group codes and initial themes were cross-checked by researchers. Changes were made to reflect the raw data. Fifth, initial themes were refined by considering how they best fit into the whole data set, the review question and the theoretical framework. Additionally, final themes and sub-themes were presented with short quotes and/or narrative findings from the original studies to enhance the understanding of findings. Differences were resolved through elaboration and consensus in team meetings. The detailed data analysis and synthesis methods are illuminated in Figure 2.2.

Figure 2.2

Convergent Qualitative Synthesis (Adapted From Treacy & Stayt, 2019)



2.2.5 Findings

This review included 23 studies. Of those, 9 studies took place in stroke units and inpatient rehabilitation facilities respectively. The rest were conducted in inpatient wards (2), outpatient clinics (1), and discharge support services (2). The review report followed the Enhancing Transparency in Reporting the Synthesis of Qualitative Research (ENTREQ) guide (Flemming et al., 2018; Tong et al., 2012) and the PRISMA guidelines (Moher et al.,

2009) with greater emphasis on ENTREQ combined with some elements from PRISMA (Appendix 7). Three themes were identified regarding challenges and opportunities in transition care: (1) health and social care systems factors; (2) health professional and people with stroke/caregiver partnership factors; and (3) people with stroke/caregivers' self-management capability factors.

2.2.5.1 Health and Social Care Systems Factors

People with stroke require considerable social and health care services after discharge from hospital due to complex health conditions underlying stroke and functional impairments. Hospital to home transition care is enhanced if care services are integrated and well-coordinated. Conversely, transition care is compromised by fragmented care services and a lack of services for those with special care needs.

Enhanced transition care via well-coordinated services

Eleven studies discussed the benefits of stroke-specific transition care interventions. People with stroke and caregivers, who are on an ongoing journey of rehabilitation after discharge, perceive they have opportunities to enhance recovery in the transition to home care. They especially value care coordination provided by a rehabilitation team (Lou et al., 2017) and view a network of health professionals that they can call upon for advice as an opportunity to gain support after discharge from hospital (Bakas et al., 2002).

Three studies revealed that early supported discharge from stroke units enhances seamless hospital to home transition (Cobley et al., 2013; Lou et al., 2017; Nordin et al., 2015). Chouliara and associates (2014) described the specialist service as an opportunity in "bridging a big part of the gap in community-based rehabilitation" for people with stroke (Chouliara et al., 2014, p. 374). Newly discharged people with stroke perceive that support from an early supported discharge team provides them with a sense of control and security (Lou et al., 2017; Nordin et al., 2015).

The opportunity for practicing care at home through weekend stay at home programs helps people with stroke and their family attain skills to manage everyday care in a home environment. As a health professional stated: "When people get home, they realise what

they can and...can't do, their limitations, and they realise... this is what I need to work on" (Cameron et al., 2014, p. 861). Weekend leave practice is considered an essential part of preparation for discharge. As a caregiver described:

I think if she'd been in the hospital for the five months without coming home and then suddenly she was home...by herself, ... [it] would be a real drama no matter who it was, So that slower introduction to home works. (Gustafsson & Bootle, 2013, p. 1384)

Compromised transition care due to gaps in services

Sixteen studies reported issues related to the continuity of rehabilitation and care services after discharge from hospital due to gaps in services. Many people with stroke and caregivers find themselves left alone without any help after discharge (Gustafsson & Bootle, 2013; Young et al., 2014). As a caregiver stated: "I felt like I was a little old Eskimo woman that they put on this ice block, chopped it off and sailed it out into the middle of the ocean" (Young et al., 2014, p. 1895). The feeling of being abandoned after discharge is further exacerbated if the decision to discharge is made under pressure (Ellis-Hill et al., 2009; Young et al., 2014). As one person reflected: "...with hindsight, it would obviously have been better for me to stay in C [a ward] because I would have had access to getting physio every day, but I realise that they need the beds..." (Ellis-Hill et al., 2009, p. 67).

The lack of coordination among care services and planning for ongoing rehabilitation not only inhibits poststroke recovery and aggravates uncertainty for caregivers but also creates challenges for service providers (Cobley et al., 2013; O'Brien et al., 2014). One community rehabilitation team reported the dilemma they experienced at the end of their service for people who still had potential for improvement: "There is dependence and reliance which is difficult [for them] to navigate, mainly because people need long-term support and we are the end of the line...There is still much more improvement to make...but there is no one else..." (O'Brien et al., 2014, p. 425).

The care needs of younger and working-aged people with stroke are often neglected as poststroke services are designed to meet the needs of older people who are the majority

of people with stroke (Chouliara et al., 2014; Lou et al., 2017). Working-aged people with stroke felt that rehabilitation and community support services were more suited for retired or older people. The unmet needs of stroke caregivers after discharge and dissatisfaction with services further underscores the deficiency in person-centredness in hospital to home transition care (Perry & Middleton, 2011).

2.2.5.2 Health Professionals and People With Stroke/Caregivers Partnership Factors

Health professionals' commitment to partner with people with stroke and caregivers contributes to tailored services to meet care needs and achieve rehabilitation goals for clients. Otherwise, the lack of partnership leads to poor support for people with stroke and caregivers in transition care at home.

Tailored transition built on partnership with clients

Fifteen studies reported clients' satisfaction with tailored care services. Partnership between a multidisciplinary team of health professionals and people with stroke and their family provides opportunities to co-develop an individualised discharge plan. As a health professional stated: "It's never just one of us that is addressing the problem, if the whole team works together...with the family member and the patient – we can come up with...the best solutions possible" (Cameron et al., 2014, p. 860).

People with stroke and caregivers feel supported during the discharge planning process when they are treated as a valuable partner in discharge preparation to develop a sense of confidence in everyday activities. As a people with stroke described: "The occupational therapist took me into the OT kitchen ... said 'Now I want you to make me a cup of coffee'...I did that without a single spill... so, yes, I knew I was ready to come home" (Ellis-Hill et al., 2009, p. 66).

A sense of confidence in self-management at home is also reflected in evaluation of caregiver preparations before discharge. Caregivers in an intervention group scored significantly higher than the control group on their preparation to care for older stroke patients at home (Shyu et al., 2008).

The lack of engagement with clients

The workload of clinicians hinders health professionals' engagement with families during discharge preparations (Cameron et al., 2014; Cobley et al., 2013; O'Brien et al., 2014). As a health professional said:

I would like to give more time to families but feel constrained by my workload for the shift and sometimes you see...caregivers... are finding things difficult and this is very hard and I feel it is part of my work... (O'Brien et al., 2014, p. 425).

Relatives of people with stroke are poorly involved in information-sharing and expect more information than provided (Almborg et al., 2009). People with stroke with a higher level of dependency perceive less opportunity in decision-making about medical treatment compared to survivors with a lower level of dependency. This might imply use of a standardised rather than individualised approach to information-sharing (Almborg et al., 2008).

Caregivers play a crucial role in people with stroke' continued rehabilitation and recovery after discharge. The lack of engagement with caregivers regarding how to manage emotional and behavioural changes of people with stroke is evident and has a detrimental impact on caregivers' well-being during the transition (Bakas et al., 2002; Cobley et al., 2013; Pringle et al., 2013). As a caregiver highlighted: "I just don't know what to do [regarding his depression]. I cannot cope because I don't know what to do to stop it. When he is continuously crying, it can be really wearing because you are helpless" (Cobley et al., 2013, p. 754).

For caregivers who are to provide physical care to people with stroke, providing skills training is an essential component of discharge preparation. Yet, this vital need is often not met prior to discharge. As a caregiver stated, "I wasn't physically shown the best way to support him...it was all trial and error" (Cobley et al., 2013, p. 754). Even when caregiver training is provided, caregivers may not be able to translate knowledge and skills they have learned into the home care setting without ongoing support from health professionals (Young et al., 2014).

2.2.5.3 People With Stroke/Caregivers' Self-Management Capability Factors

People with stroke and caregivers are at different levels of readiness to take charge of transition care at home. In a supportive environment, they develop and strengthen resilience over the course of the transition. People with stroke can experience a sense of loss and feel burdensome on others which hinders their efforts to develop resilience.

Resilience as a key indicator of a smooth transition

The processes and outcomes in transition care reflect the concept of resilience in health care as defined by the WHO (2017). Factors contributing to the development of resilience for people with stroke and caregivers include spirituality-enhanced hope (Bakas et al., 2002; Rittman et al., 2007), reflection on past experiences in caregiving (Lou et al., 2017; Young et al., 2014) and self-learning of stroke-related knowledge and caregiving skills (Greenwood et al., 2009; Wottrich et al., 2012).

Moving focus away from one's own disabilities and recognising positive progress helps people with stroke to develop resilience and diminishes uncertainty in recovery: "I was in hospital with people ... that couldn't walk properly... couldn't feed themselves... I could have been so much worse ... each day when I see a light improvement ... I am so thankful" (Pringle et al., 2013, p. 1995).

Having faith in God can be a coping strategy for caregivers to develop resilience and sustain caregiving at home (Bakas et al., 2002; Greenwood et al., 2009; Rittman et al., 2007). As a caregiver said: "…whatever happens, I always say that it is best. Better. God never wants to harm you… there is something always positive…" (Greenwood et al., 2009, p. 1129).

Time to learn and develop knowledge, skills and experience to manage transition care at home is closely related to the development of resilience (Greenwood et al., 2009; Lou et al., 2017; Young et al., 2014). Most people with stroke and caregivers show some levels of resilience and gradually adapt to life in the community between three and six months after discharge from hospital (Grant et al., 2006; Greenwood et al., 2009; Wood et al., 2010; Wottrich et al., 2012). As a person with stroke stated: "It was a bit difficult at first

with everything new, but it gets better and better, one enjoys being at home more now than one did when one left hospital" (Wottrich et al., 2012, p. 1222).

Established stroke knowledge and skills from past caregiver experience also contributes to resilience for caregivers: "I didn't have to figure out anything because we had taken care of her mother for 17 years with a stroke...I more or less knew what we had to do" (Young et al., 2014, p. 1897).

Over the course of the transition, some practical solutions emerge through trial and error:

We have learnt from last time...He doesn't have a lot of shirts with buttons because he had difficulty fastening buttons. To give him his independence, we have altered...so that he can still dress himself and doesn't rely on me (Greenwood et al., 2009, p. 1128).

Learning to accept support from others facilitates social reintegration and indicates people with stroke' resilience to adapt to changes: "When it comes to social events...I have to thank my friend because she drives and the same thing with my son...it's encouraging that these people are around me" (Wood et al., 2010, p. 1051).

Sense of loss and burden on others

People with stroke view discharge home as entering an unknown home situation with a new identity. Even for those who do not experience dramatic poststroke changes, life still needs to be adjusted (Chouliara et al., 2014; Lou et al., 2017; Nordin et al., 2015). They face enormous challenges to cope with emotional, cognitive and personality changes. As a caregiver described: "I feel I've got a different man home..., because before he was a very well educated man... and now he's not...even interested in anything" (Pringle et al., 2013, p. 1995).

However, the need for emotional support for people with stroke might not be determined by service providers due to a lack of access to specialist services. As a health professional stated: "I think the support around the emotional journey...is a big gap because

if we had access to psychology or counselling, we could start...offering what they need as things change such as adjusting to their emotions" (O'Brien et al., 2014, p. 245).

The majority of studies in this review recognised the crucial and multiple roles that caregivers play in transition care. Firstly, existing and potential conflicts between people with stroke and caregivers has significant implications for caregiving at home (O'Brien et al., 2014; Young et al., 2014); Secondly, caregivers' readiness to take on the caregiving role impacts their capacity (Cameron et al., 2014; Perry & Middleton, 2011). Thirdly, caregivers' own chronic health issues might jeopardise their ability to sustain the role (Young et al., 2014). Therefore, it is essential to assess caregivers' readiness for their role and the need for support to develop resilience to sustain the role.

People with stroke can feel too guilty to bother family caregivers (Rittman et al., 2007; Simeone et al., 2015). As a person with stroke illustrated: "Do you see the condition in which I put my son? I forced my son to bounce among work, home, his family, and me. My son seems like a yo-yo attached to the rope running back and forth…" (Simeone et al., 2015, p. 166). This perception of burden on caregivers might inhibit people with stroke from seeking help to achieve their recovery goals, which in turn prevents caregivers from taking actions to support people with stroke to reintegrate into the community.

2.2.6 Discussion

Hospital to home transition is the most crucial time for people with stroke and caregivers to build capabilities to optimise poststroke recovery and manage health conditions at home. The synthesis of studies using different research designs enables a more comprehensive understanding of opportunities and challenges arising from health and social care systems, the interactions between health professionals and clients, and the capabilities of people with stroke and caregivers in transition care. This review demonstrates that despite challenges brought by stroke, effective hospital to home transition care can be achieved through the provision of stroke-specific integrated and well-coordinated care services; by enabling health and social care professionals to work in partnership with clients

to determine and meet their individualised care needs; and by supporting people with stroke and caregivers to develop resilience in self-management of transition care at home.

The CCM facilitated a comprehensive understanding of the entwined associations between healthcare systems (system factors), service provider factors (proactive practice team) and individual factors (informed and activated patients) (Accelerating Care Transformation Centre, 2023). The findings of this review confirm the need for the healthcare system to shift from a reactive care to proactive care approach as advocated by the CCM (WHO, 2016). This integrated care approach requires effective partnership between healthcare teams and people with stroke and caregivers (WHO, 2016; Accelerating Care Transformation Centre, 2023). In addition, ongoing support for people with stroke and caregivers during the transition period is imperative to stabilise and instil self-care at home.

People with stroke are discharged from hospital with varying levels of impairments including inability to walk, visual impairment and executive dysfunction (Jokinen et al., 2015; Sörös et al., 2015). Some survivors achieve clinical recovery without physical disability but live with psychological morbidity (Jokinen et al., 2015; Wainwright et al., 2017). About 30% of people with stroke experience poststroke depression and fatigue concurrently (MacIntosh et al., 2017; N. Wei et al., 2015). These limitations are often hidden and cause misconceptions amongst the general public (Schwarz et al., 2018; Wainwright et al., 2017). The stigmatisation of these invisible limitations significantly impacts on ongoing poststroke recovery. Working-aged people with stroke find that the major barrier to returning to work is their self-stigma and colleagues' lack of awareness of these invisible conditions (Balasooriya-Smeekens et al., 2016; Schwarz et al., 2018). Caregivers of older people with stroke, who lack knowledge of depression and fatigue, may relate these symptoms to old age and therefore miss rehabilitation opportunities (Wainwright et al., 2017).

Stroke triggers a local inflammatory immune response in the brain and also causes systemic immunosuppression resulting in susceptibility to infection amongst people with stroke (Anrather & ladecola, 2016; Pagram et al., 2017). Infection occurs in about 30% of the stroke population (Pagram et al., 2017) and is one of the major causes of poststroke

readmission (Lord et al., 2016; Zhong et al., 2016). Identifying high-risk survivors and preparing them and caregivers with knowledge of potential complications and selfmanagement skills empowers them to proactively prevent infection. Discharge preparation, structured post-discharge follow-up and early evaluation by health professionals, such as stroke nurses, could reduce avoidable hospital admissions due to complications (Condon et al., 2016; White et al., 2015).

A mismatch of goals between survivors and health professionals is a major barrier in the implementation of person-centred goal setting (Plant et al., 2016; Rosewilliam et al., 2016). Survivors' goals tend to encompass their aspirations to return to independence. Whereas health professionals' goals tend to focus on impairments and are mainly driven by organisational performance indicators (Plant et al., 2016). It is vital for people with stroke and caregivers to set goals for re-entering the community based on their individual situation. Rosewilliam and associates (2016) suggest changing the culture to facilitate person-centred goal setting by empowering health professionals with knowledge and skills to engage people with stroke and caregivers in the process.

People with stroke and caregivers must adjust to the physical and emotional sequelae of stroke and negotiate complex health and social care systems (Andrew et al., 2018; Nelson et al., 2016). After hospital discharge, people with stroke and caregivers experience a drastic contrast between the intensive support received in hospital and limited access to community rehabilitation and social care services. The perception of abandonment is intensified when expected support is absent (Lamontagne et al., 2019; Piccenna et al., 2016). Maintaining and developing resilience is a protective factor to manage stress and prevent negative outcomes (Epstein, 2015; Taku, 2014). To help sustain resilience, reduce avoidable hospitalisation and prevent institutionalisation, there is an urgent need to develop stroke-specific transition care pathways and care models.

2.2.6.1 Limitations

The review has enabled the synthesis of studies with diverse research approaches providing a range of perspectives. We included all studies regardless of their quality as each

contributes to the narrative of the investigation. Therefore, this may affect the credibility of findings, and findings should be interpreted in this context. We limited the timeframe of transition care to six months not 12 months, as six months is the study timeframe for most poststroke transition care studies. This may have affected narratives, as transition experience is individualised, and transition care for some survivors and caregivers may extend beyond six months after discharge. Their perspectives are missed with the setting of this timeframe. Further reviews need to consider this limitation.

2.2.7 Conclusion

The first six months of the poststroke hospital to home transition is a challenging period for people with stroke and caregivers. They need to adjust to a new life as well as navigate health and social care systems solo. Disjointed service provision further exacerbates the situation, which causes unmet needs and increases potential risk of hospitalisation and institutionalisation. This review highlights complex issues in poststroke hospital to home transition care and identifies gaps for further research.

2.2.7.1 Implications for Future Research

Further research should focus on people with stroke and caregiver expectations of health and social care systems after discharge home, as this was not discussed in depth in the reviewed studies and post-discharge follow-ups were mostly short. Furthermore, health professionals' perspectives of approaches to identifying and meeting transition care needs could be explored further. Moreover, there is scarce research evidence-informed clinical pathways and models particular to poststroke hospital to home transition care. Strokespecific transition clinical pathways and care models will ensure consistent delivery of services and enable stakeholders to work in collaboration to meet care needs and expectations of people with stroke and caregivers.

Anonymised Conflict of Interest Statement

No conflict of interest has been declared by the authors.

This is the end of the publication.

2.3 Summary

This integrative literature review explored the factors that influence hospital-to-home transition care for stroke dyads. A synthesis of the study's findings revealed that hospital-to-home transitions were shaped by the health and social care systems, and health professionals and stroke dyads also had a role in influencing the transition. Multidisciplinary-team-coordinated stroke-specific transitional care programs, and social care services tailored to the needs of people with stroke, were found to facilitate the hospital-to-home transition. Likewise, during the course of the transition, resilience was gradually built on the development of confidence and competence with the mastery of stroke care knowledge and skills. The challenges that inhibited a successful transition included inappropriate discharge preparation, discontinuity of rehabilitation and social care support, and struggles with managing post-stroke disabilities.

The findings urge collaboration between all stakeholders of the stroke care chain to ensure consumer-directed hospital-to-home transition care for stroke dyads. However, the experiences and perceptions of the stroke dyads on their care needs and challenges during transition remain largely unknown. Their expectations of health and social care systems and service providers are key indicators for the development of a stroke hospital-to-home transition care framework.

The next chapter introduces the theoretical framework adapted for this study.

Chapter 3: Theoretical Framework

3.1 Introduction

The previous chapter discussed factors that influence hospital-to-home transition for stroke dyads at the health and social care systems level, health professionals' level and personal level. The self-management capacities of stroke dyads were found to be activated through collaboration with health professionals, continuity of health and social care support, and resilience developed over the course of transition with the mastery of self-management knowledge and skills. This chapter reviews concepts of transition in reviewed studies, analyses Meleis's transition theory and discusses how the framework is adapted for this study context.

3.2 Theoretical Frameworks Informing Studies in Transition Care

Theoretical frameworks enable researchers to integrate findings into a coherent structure and serve to guide researchers' understanding of the studied phenomena in a systematic way. The application of theories in research also provides opportunities for theories to be evaluated and critiqued (Polit & Beck, 2017).

3.2.1 Concepts of Transition in the Reviewed Studies

Nine studies in the integrative review (Chapter 2) linked their research to the concept of transition and transition care, but few had in-depth discussions of the theoretical underpinning. Hospital-to-home transition-related concepts were discussed in a number of studies. Almborg et al. (2008, 2009) connected discharge planning with the concept of 'participation' and referred to discharge planning as receiving information about stroke and related aspects of care and participating in decision-making. These studies mainly focused on one element of multiple factors affecting discharge preparation in hospital-to-home transition.

Ellis-Hill et al. (2009) related 'going home' to the concept of a 'model of recovery'. This model of recovery emphasised the concept of 'moving on through their life, with their

own goals and plans' (Ellis-Hill et al., 2009, p. 69). This model seemed to be associated with the individual's expectations of recovery, their response to the situation and the interaction with health professionals. However, a shared model of recovery between people with stroke and health professionals and that is viewed as a better approach to recovery is overlooked in this model.

Three studies on stroke caregivers discussed concepts related to them during the hospital-to-home transition. The study by O'Brien et al. (2014) on an optimal health program discussed the concepts of self-efficacy and care coordination in caregivers. The program aimed to promote self-management for caregivers and their collaboration with health professionals. Similarly, the study by Shyu et al. (2008) discussed the concept of a family-caregiver-oriented discharge planning program and demonstrated the effectiveness of the program on enhanced caregiver preparation and improved satisfaction in caregivers during the transition. The concept of role preparation for caregivers was further explored by Huang et al. (2003), where the preparation for the new role for caregivers was essential before assuming the role. However, these studies have not elaborated on the transition experience nor examined the perspectives of stakeholders involved in the transition.

Young et al. (2014) advocated for the use of a crisis intervention approach (Palmer et al., 2004) to comprehensively assess caregivers' needs because a stroke event is an unexpected life crisis for caregivers. A crisis intervention approach emphasises situation-specific problem-solving methods for victims of catastrophic events, such as natural disasters, chronic diseases or death (Palmer et al., 2004). Young et al. (2014) believed that this approach might help uncover the unacknowledged grief and anxiety of caregivers and promote a better understanding of their needs. However, this approach only focused on a single element in the hospital-to-home transition.

Stroke was regarded as a biographical disruption in Lou et al.'s study (2017). This concept highlights that stroke brings people's attention to their subconscious awareness of their body state. Stroke stimulates a profound review of one's personal biography and self-concept. People respond to stroke by reorganising resources in the changed situation (Bury,

1982; S. Williams, 2000). The findings in Lou et al.'s (2017) study revealed that stroke was part of much turbulence in the lives that people with stroke encountered. Hence, specific timings, contexts, norms, expectations and the willingness of people with stroke to participate in the process of recovery need to be considered in order to understand their experiences (S. Williams, 2000).

Discharge home after stroke was conceptualised as a psychosocial transition in the study by Pringle et al. (2013). Pringle et al. (2013) viewed returning to home for people with stroke as a process involving changes to a person's 'assumptive world' (p. 1997). Many people with stroke previously assumed that ways of life may be changed after stroke. This theory highlighted that health professionals need to identify good support systems to help people cope with transition and to reconnect with the previous assumptive world.

A study on the psychosocial experiences during the first month home post-discharge by Rittman et al. (2007) was based on the concept that transition was a process, occurring over time and involving changes in identities, roles and behaviours (Schumacher & Meleis, 1994). This concept emphasises the importance for the person to be aware of the transition and to be willing to engage with it (Schumacher & Meleis, 1994). Rittman et al. (2007) found that, despite being scored high on functional assessment, people with stroke still experienced multiple psychosocial changes over time, and they could benefit from interventions to prepare them for these changes.

As discussed above, various conceptual frameworks were used in studies on hospital-to-home transition for stroke dyads. Although these frameworks enhanced understandings of biographical disruption, a model of recovery, crisis intervention, selfefficacy, participation and role development for caregivers, they may not have covered the complexity of transition care derived from health and social care systems, health professionals and stroke dyads as identified in the integrative literature review. Therefore, a comprehensive theory that informs a systematic approach to analysing hospital-to-home transition is much needed.

3.2.2 Meleis's Transitions Theory

The middle-range transitions theory developed by Afaf Ibrahim Meleis and colleagues (Chick & Meleis, 1986; Meleis, 2010; Meleis, 1975; Meleis et al., 2000; Schumacher & Meleis, 1994) was chosen as the theoretical framework to guide this study. Meleis's transition theory has been widely used and tested in various care settings with diverse populations. The theory focuses on the dynamic interactions among the person, community and society, as well as the role of nursing therapeutics in facilitating healthy transitions.

Transition is defined as:

A passage from one life phase, condition, or status to another, is a multiple concept embracing the elements of process, time span, and perception ... [it] refers to both the process and the outcome of complex person-environment interactions. It may involve more than one person and is imbedded in the context and the situation ... Defining characteristics of transition include process, disconnectedness, perception, and patterns of response. (Chick & Meleis, 1986, pp. 239–240)

3.2.2.1 Nature of Transitions

Meleis et al. (2000) suggested four types of transitions with differing patterns of multiplicity and complexity:

- Developmental transitions occur in people's life cycle, such as becoming parents or going through adolescence.
- Situational transitions are changes such as those in family situations, family caregiving and in educational and professional roles.
- Health–illness transitions refer to the effects of illness-related transitions on individuals and their families.
- Organisational transitions focus on the changes in organisations or communities—for example, structural changes in an organisation and their effect on the employees.

Stroke dyads may experience both health–illness and situational transitions after discharge home as they need to adapt to a post-stroke physical and mental state, altered roles, as well as the new home and social environment that was once familiar before the stroke. They may experience these interrelated transitions simultaneously during the post-stroke trajectory.

Meleis's theory has also identified several essential properties that are interconnected in the complex transition process (Meleis et al., 2000):

- Awareness—this is associated with perception, knowledge and recognition of a transition experience.
- Engagement—this refers to the level of an individual's involvement in the process.
- Change and difference—these are essential properties of transitions but are not interchangeable.
 - Transitions both result in and are the result of change.
 - Individuals in transition confront differences at different levels. Changes include unmet expectations, feeling different, being perceived as different or viewing others or the world in different ways.
- Time span—this refers to flow and movement over time. The possibility of changes and fluctuation over time needs to be considered when evaluating transition experiences.
- Critical points and events—these are related to the increasing awareness of change and difference or to increased engagement in managing the transition experience. Critical points may be accompanied by a period of uncertainty and heightened vulnerability. Final critical points may be characterised by a sense of stabilisation in new routines, roles, skills and self-care activities.
3.2.2.2 Transition Conditions

The transition is influenced by a complex interplay between personal and environmental factors (Schumacher & Meleis, 1994). Personal, community or societal factors can facilitate or impede a healthy transition (Meleis et al., 2000).

'Personal conditions' are described as follows (Meleis, 2010, pp. 59–60; Schumacher & Meleis, 1994, pp. 121–122):

- Meanings—these are related to the subjective appraisal of transition in one's life.
 Meanings associated with transitions may be neutral, negative or positive.
- Expectations—these are influenced by previous experiences. The frame of reference may be uncertain or unrealistic. Surprise occurs when expectations mismatch reality.
- Cultural beliefs and attitudes—the transition experience may be influenced by the individual's cultural background and the related stigma attached to the transition.
- Socioeconomic status—low socioeconomic status may inhibit healthy transition.
- Preparation and knowledge—the level of preparation, planning and knowledge about what to expect before and during the transition and problem-solving strategies influences the success of the transition.
- Emotional and physical wellbeing—a wide range of emotions, especially stress and emotional distress, are common during the transition. The level of comfort with changes in the body influences wellbeing during the transition.

'Community conditions' refers to the availability of external facilitating resources, including formal social care support and informal support from family, friends and community (Meleis, 2010, p. 60). Advice from respected sources, consistent and sufficient information obtained from trusted healthcare providers, and timely answers to questions are facilitators during transitions.

'Societal conditions', such as stigmas and stereotypes attached to transitional events, interfere with a healthy transition (Meleis, 2010, p. 61). Other societal factors include policy,

organisational and healthcare system support. These conditions shape an individual's transition experience.

The transition from hospital to home symbolises an achievement in the post-stroke trajectory for stroke dyads. A healthy post-stroke hospital-to-home transition depends on the harmonious interactions of transition conditions. Physical and emotional progress through post-stroke rehabilitation enhances self-appraised coping abilities. Adequate preparation and skills-building before hospital discharge, coupled with post-discharge community and societal support, further augment self-management capabilities that help transformation for both people with stroke and their caregivers and stabilise long-term home care.

3.2.2.3 Patterns of Response

A healthy transition is characterised with process and outcome indicators (Meleis, 2010). Transition is a process that unfolds over time. Identification of indicators of a healthy or at-risk transition prompts early intervention to enable positive outcomes (Meleis, 2010).

Meleis et al. (2000) discussed four dimensions of process indicators:

- 'Feeling connected' refers to contacts and connections with pre-transition
 personal social networks and health professionals who assist with the transition.
 A positive transition experience is indicated when individuals feel connected with
 healthcare providers.
- 'Interacting' denotes intradyadic interaction. The interactions between caregiver and care recipient create a context wherein self-management and caregiving can be balanced and sustained constructively.
- 'Location and being situated' indicates location change during transition. Making comparisons before and after transition provides ways to situate individuals in terms of time, space and relationships.
- 'Developing confidence and coping' refers to an increase of confidence over time in the transition process. Time provides individuals with progressive and

cumulative knowledge about the situation and coping strategies to manage the situation.

The mastery of new skills needed to manage the transition and the development of a fluid integrative identity are two outcomes experienced by individuals in a healthy transition (Meleis, 2010, pp. 62–63). The subjective elements of individuals who experience transitions influence the sense of achieving a healthy transition (Meleis, 2010). When individuals demonstrate proficiency in managing situations in transition, for example, caregivers feel able to take charge of care and navigate the healthcare system, and this indicates a successful transition.

An individual's identity is redefined in the transition process. The redefined identity is dynamic and may change over time in transition. Situational and health–illness transition may be viewed from a bifocal perspective (Meleis, 2010). Settling in the new identity indicates that a healthy transition outcome has been reached.

3.2.2.4 Nursing Therapeutics

Schumacher and Meleis (1994) identified three nursing interventions that are widely used during transitions. First, a transition readiness assessment is developed based on a comprehensive understanding of the individual and through multidisciplinary collaborations. Second, preparation for transition using educational interventions creates optimal conditions in transition. Sufficient time and a gradual approach to knowledge and skills-building and practice are required for adequate preparation. The third nursing therapeutic is role supplementation. As discussed above, individuals experience role and identity change during transition. For example, people with stroke may have difficulty resuming the premorbid role in the household, which results in role insufficiency. Role supplementation nursing interventions include role learning and clarification through a mastery of the knowledge specific to fulfilling the role, peer support, communication and social interaction (Schumacher & Meleis, 1994). During a hospital-to-home transition, the above nursing therapeutics can empower stroke dyads to prepare for the transition and gradually adapt to the new post-stroke identity.

An outline of the elements and the relationships between these elements in Meleis's transitions theory is shown in Figure 3.1.

Figure 3.1

Transition: A Middle-Range Theory (Meleis et al. 2010, p. 425)



Reprinted from *Transition: A Middle-Range Theory: Middle-Range and Situation-Specific Theories in Nursing Research and Practice* (p. 425) by Afaf Ibrahim Meleis, 2010, with permission from the Springer Publishing Company.

This theoretical framework identifies generic conditions for people undergoing transition and the nature of outcomes of the transition experience. It emphasises the interactions among the person, community and society and the conditions that enable healthy transitions. The identification of healthy transition outcomes serves as the impetus for discovering mechanisms used by people during transition that lead to positive transition experiences. The framework also provides opportunities to advance knowledge about nursing therapeutics that may facilitate successful transitions and enhance healthy coping and recovery (Meleis et al., 2000).

3.2.3 The Relevance of Meleis's Transitions Theory to the Present Study

Meleis's transitions theory has been widely used and tested in various care settings with diverse populations, including in studies on health–illness transitions, migration, motherhood and adolescence to adulthood (Meleis, 2010). In the study of health–illness-related transitions, the theory has guided a comprehensive understanding of the transition experience of relatives of older people who were admitted to a nursing home (S. Davies, 2005), of the psychosocial transition of people with stroke after hospital discharge (Rittman et al., 2007), of the experience of living with long-term conditions (Long et al., 2016) and of the caregivers of high-risk infants (Mi, 2016). In Meleis's transitions theory, the domain of 'transition conditions: facilitators and inhibitors' enhances the understanding of enablers and barriers during hospital-to-home transition and uncovers the community or societal conditions for transition (Meleis et al., 2000, p. 56). The element of 'patterns of response' also provides a theoretical foundation for a deeper insight into the outcomes of a successful transition.

3.2.3.1 Limitations of the Theory

Meleis's transitions theory is generic rather than specific to stroke transition care. The theory may not cover all stroke-specific aspects, especially the impact of post-stroke impairments. However, it is this weakness of the framework that provides the present researcher with opportunities to develop a stroke-specific transition care framework. In addition, the theory does not seem to acknowledge the interactive relationship of all stakeholders during the transition, especially health professionals. Health professionals also experience role transition, and they play a crucial role in facilitating an effective transition. Furthermore, in the global context of multidisciplinary collaboration in health care, nursing therapeutics alone may not reflect the need for interprofessional strategies to address issues during transition.

3.2.4 The Need to Adapt Meleis's Transitions Theory to the Study Context

Based on the critique of Meleis's transitions theory, the present researcher adapted this theory to address the aims of the study and to suit the study context. The elements from

Meleis's transitions theory are shown in Figure 3.2. An explanation of these elements and relationships among these elements in the present study are discussed in the following section. It was anticipated that the findings from the study would also add new evidence to inform the further development of a framework specific to hospital-to-home transition for stroke dyads.

In the hospital-to-home transition process, stroke dyads experience multiple and complex transitions that occur simultaneously. In this post-stroke trajectory, they experience health–illness and situational transitions. With the sudden onset of stroke, the daily living functions of people with stroke and the dyads' roles in life significantly change, as do their personal identities, relationships and social networks. These related elements overlap and influence each other during the transition.

The lived experience of stroke dyads uncovers personal, community and societal conditions based on their contexts of life. The interactions of these conditions may facilitate or hinder the achievement of a healthy transition and have an impact on the outcomes of the transition. Early nursing therapeutic interventions developed through interdisciplinary collaboration help equip stroke dyads with the necessary knowledge, skills and problem-solving strategies and prepare them for the imminent hospital-to-home transition. A healthy transition is evidenced by stroke dyads' perceived capability to take charge of the post-discharge life situation, confidence in mastery of their health conditions, navigation of the health system and the ability to maintain caregiving.

Figure 3.2

Meleis's Transition Theory (Meleis, 2010, p. 425), Adapted to This Study Context



3.3 Summary

This chapter has discussed the theoretical and conceptual frameworks used in transition care studies. Meleis's transition theory presents key properties and indicators of the transition experience, which formed the foundation for developing a theoretical framework for the present study. The domains of Meleis's transitions theory served as a guide for the conduct of this study in the development of the interview guide and data analysis and, systematically, for generating comprehensive understandings of the transition experience for stroke dyads and of health professionals' perceptions.

Chapter 4: Systematic Review

4.1 Introduction

As discussed in the previous chapters, challenges and opportunities coexist during the hospital-to-home transition for stroke dyads. The integrative literature review identified a scarcity of global evidence on the care needs and expectations of health and social care in the hospital-to-home transition from the stroke dyad's point of view. Additionally, little evidence exists regarding stroke-specific transition care clinical pathways and models to enable stakeholders to work collaboratively to meet the care needs and expectations of stroke dyads. Hence, it is timely to conduct a systematic review that synthesises the qualitative studies on care needs and expectations of health and social care in the hospitalto-home transition from the perspectives of stroke dyads in a global context.

This chapter presents a published systematic review and meta-synthesis. The contribution of the first author to the paper is 70% to research design, writing and editing and 80% to data collection and analysis. The second, third and fourth authors collectively contributed 30% to research design, writing and editing and 20% to data collection and analysis.

This section presents a publication arising from this thesis that was published in the *Journal of Clinical Nursing*.

Citation: Chen, L., Xiao, L. D., Chamberlain, D., & Newman, P. (2021). Enablers and barriers in hospital to home transitional care for people with stroke and caregivers: A systematic review. *Journal of Clinical Nursing, 30*(19-20), 2786–2807.

https://doi.org/10.1111/jocn.15807

This is the peer reviewed version of the following article: Enablers and barriers in hospital to home transitional care for people with stroke and caregivers: A systematic review,

which has been published in final form at <u>https://doi.org/10.1111/jocn.15807</u>. This article may be used for non-commercial purposes in accordance with Wiley Terms and Conditions for Use of Self-Archived Versions. This article may not be enhanced, enriched or otherwise transformed into a derivative work, without express permission from Wiley or by statutory rights under applicable legislation. Copyright notices must not be removed, obscured or modified. The article must be linked to Wiley's version of record on Wiley Online Library and any embedding, framing or otherwise making available the article or pages thereof by third parties from platforms, services and websites other than Wiley Online Library must be prohibited.

4.2 Enablers and Barriers in Hospital to Home Transitional Care for People With Stroke and Caregivers: A Systematic Review

4.2.1 Abstract

Aims and objectives: To synthesise qualitative research evidence on the experience of stroke survivors and informal caregivers in hospital to home transitional care.

Background: Due to a shortened hospital stay, stroke survivors/caregivers must take over complex care on discharge from hospital to home. Gaps in the literature warrant a meta-synthesis of qualitative studies on perceived enablers and barriers during this crucial period.

Design: systematic review and meta-synthesis.

Methods: A review was guided by Enhancing Transparency in Reporting the Synthesis of Qualitative Research (ENTREQ) checklist where six databases were searched from April to June 2020 including CINAHL Plus, MEDLINE, PsycINFO, Scopus, Web of Science and ProQuest and ProQuest Dissertations and Theses. There was no date limit to the search. Selected studies were critically appraised. A thematic synthesis approach was applied.

Results: The synthesis of 29 studies identified three major findings. First, partnerships with stroke survivors/caregivers empower discharge preparation, foster competence to navigate health and social care systems and activate self-management capabilities. Second, gaps in discharge planning and the lack of timely post-discharge support contribute to unmet care needs for stroke survivors/caregivers and affects their ability to cope with poststroke changes. Third, stroke survivors/caregivers expect integrated transitional care that promotes shared decision-making and enables long-term self-management at home.

Conclusions: Hospital to home transition is a challenging period in the trajectory of poststroke rehabilitation and recovery. Further research is required to deepen understandings of all stakeholders' views and address unmet needs during transitional care.

Relevance to clinical practice: Protocols and clinical guidelines relating to discharge planning and transitional care need to be reviewed to ensure partnership approach with survivors/caregivers in the design and delivery of individualised transitional care. Stroke nurses are in a unique position to lead timely support for survivors/caregivers and to bridge service gaps in hospital to home transition care.

Keywords: Stroke survivors, transitional care, discharge, caregivers, nurses' role, care needs, meta-synthesis, qualitative research

What does this paper contribute to the wider global clinical community?

- Hospital to home transition is a critical period for stroke survivors and caregivers to develop self-management capacities and lay a solid foundation for long-term poststroke recovery and rehabilitation.
- A person-centred and partnership approach by service providers empowers stroke survivors and caregivers preparing for transitioning home from hospital. Integrated community health and social care services augment an individual's self-management capacity.
- Stroke nurses are in a unique position to lead and provide individualised strokespecific transitional care.

4.2.2 Introduction

Stroke is a leading cause of death and adult disability worldwide (WSO, n.d.). Globally, 15 million people are diagnosed with stroke (cerebrovascular disease) every year. The overall burden of stroke is considerable and continues to increase in the context of an ageing population (C. O. Johnson et al., 2019). Most stroke survivors are discharged to home, and the hospital to home transition poses many challenges for them and family caregivers to manage care at home while engaging in rehabilitation and recovery (Lindblom et al., 2020; Reeves et al., 2019). These challenges have become more prominent in the context of shortened hospital stays and early discharge by which stroke patients and their caregivers take charge of care activities and coordinate care services instead of health professionals in hospital (Camicia et al., 2016; L. Chen et al., 2020). Although hospital to home transitional care and post-discharge care services are available in many developed nations, they are often not specific to different categories of stroke and lack standardised care components (Olson & Juengst, 2019; Wang et al., 2017) or they are fragmented, which prevents stroke survivors and caregivers from accessing (L. Chen et al., 2016; Miller et al., 2019). These situations have a negative impact on stroke survivors and caregivers achieving rehabilitation and recovery goals during their transition.

To achieve effective transition to community, the impact of stroke on the physical, psychosocial and spiritual well-being of stroke survivors and caregivers needs to be addressed (Mountain et al., 2020; Sennfält & Ullberg, 2020). The International Classification of Functioning, Disability and Health (ICF) framework developed by the WHO considers that an individual's health condition is influenced by the interplay between functioning, personal and environmental factors (WHO, 2013). Functioning is a hypernym for functions, activities and participation. Functions denotes body function and structure. Activities refer to execution of tasks, and participation means involvement in life situations (WHO, 2013). Environmental factors are external to stroke survivors and should be considered at both micro and macro levels encompassing access to the pre-morbid home, community and workplace, and access to healthcare services and social supports (WHO, 2013). Personal factors are unique

aspects of stroke survivors and caregivers including their pre-morbid relationship, past experiences with the healthcare system and coping strategies (WHO, 2013). The ICF framework was adapted in this review to explore the impact of functioning, personal and environmental factors on stroke survivors and caregivers during hospital to home transition.

Most stroke survivors live with declined functioning such as poor balance, incontinence, swallowing difficulties, visual disturbances, and psychological and cognitive disorders (Jokinen et al., 2015; Wainwright et al., 2017). Hospital readmissions are common after discharge and are associated with increased morbidity, mortality, and risk of early institutionalisation (Abreu et al., 2020; Reeves et al., 2019). Contributing factors to readmissions include falls, urinary tract infection and recurrence of stroke (Abreu et al., 2020; Kilkenny et al., 2020). These factors are preventable via person-centred discharge planning and follow-up support after discharge (Abreu et al., 2020; Lee et al., 2019).

The development of self-management capabilities is hindered by inadequate engagement in discharge planning and caregiving preparation (L. Chen et al., 2016; Miller et al., 2019). Self-management capability refers to an individual's strategies, skills, and confidence to manage their chronic conditions and take charge of their life (Barker et al., 2018). Factors contributing to inadequate engagement in discharge planning are multifactorial and include lack of collaboration amongst multidisciplinary teams and service providers, time constraints, a lack of beds and readiness of survivors/caregivers to participate (Krishnan et al., 2019; Olson & Juengst, 2019; Waring et al., 2019). Poor engagement in the discharge planning process prevents stroke survivors and caregivers from building skills and confidence in the management of their condition and undertaking daily activities at home (Miller et al., 2019).

The WHO defines people-centred and integrated health care as an approach that considers the perspectives of individuals, families and communities which is organised around health needs and expectations rather than diseases (WHO, 2015). Based on individual needs, care services are delivered to ensure a continuum of care at different levels and sites within the health system (WHO, 2015). Whilst person-centredness is well promoted

and advocated in health and social care, the design of care services is largely driven by organisational imperatives rather than helping stroke survivors and caregivers achieve individual care goals. This organisational context of care constrains health professionals' commitment to optimise care services (L. Chen et al., 2020; O'Brien et al., 2014). Discrepancies in goals are reported to inhibit partnerships and collaborations between clinicians, stroke survivors and caregivers (L. Chen et al., 2020; Krishnan et al., 2019). Stroke survivors and caregivers feel alone and abandoned after discharge due to fragmented service provision (Lindblom et al., 2020; Miller et al., 2019). When stroke survivors and caregivers are underprepared before discharge, the lack of support from health professionals further affects self-management of poststroke rehabilitation and recovery.

Studies on issues in hospital to home transitional care for stroke survivors are abundant. However, meta-synthesised research evidence about the perspectives and expectations of stroke survivors and caregivers in hospital to home transition care in the first six months post-discharge is scarce. This meta-synthesis fills this gap in the literature. Findings will inform the design of future trials of poststroke transitional care interventions, research on stroke transition specific pathways and service development.

4.2.3 Aims

The aim of this systematic review is to synthesise qualitative research evidence on the experiences and expectations of stroke survivors and informal caregivers in hospital to home transitional care.

The review questions were:

- 1. What are the contributing factors that enable hospital to home transitional care?
- 2. What are the barriers that impede hospital to home transitional care?
- 3. What are the expectations of stroke survivors and informal caregivers of the healthcare system and health professionals during hospital to home transition?

4.2.4 Methods

This systematic review applied a meta-synthesis approach described by Edwards and Kaimal (2016) that integrates qualitative evidence systematically and discovers novel findings that are "far removed from these findings" from the reviewed primary studies (Edwards & Kaimal, 2016; Sandelowski & Barroso, 2006, p. 18). A six step meta-synthesis framework developed by Lachal and colleagues (Lachal et al., 2017) was used. First, we defined research questions and selection criteria. Second, we selected studies that met the selection criteria. Third, we undertook quality assessment of selected studies. Fourth, we extracted data from selected studies. Fifth, we analysed data, and finally, we presented findings from data synthesis. The review reporting followed the Enhancing Transparency in Reporting the Synthesis of Qualitative Research (ENTREQ) guide (Tong et al., 2012), see Appendix 8. The review protocol was registered on PROSPERO with registration number CRD42018091345 (Appendix 9).

4.2.4.1 Search Strategy

A systematic electronic search was conducted between 9th April and 15th July 2020 for studies published in English, or with an abstract published in English, in six databases including CINAHL Plus, MEDLINE, PsycINFO, Scopus, Web of Science and ProQuest and ProQuest Dissertations and Theses. To provide a comprehensive outline of research in the studied field, there were no limitations on publication year.

The Population, Exposure and Outcome (PEO) framework guided the development of search terms (Khan, Kunz, Kleijnen, & Antes, 2003). Search terms included (P) *stroke, cerebrovascular accident, survivors, caregivers and family carers*; (E) *patient discharge, continuity of patient care, transitional care, aftercare, stroke rehabilitation, hospital to home and care coordination*; and (O) *experiences, perception, perceptive and perspective* (see Appendix 10).

Each term was entered into the databases with truncation where appropriate. All individual searches were combined into a single group using the 'OR' Boolean. Each group was then combined by 'AND' with qualitative filters indicating qualitative data collection. The

final citations were saved in EndNote and screened for duplications. We also manually searched reference lists of selected studies.

4.2.4.2 Selection of Studies

An experienced librarian (PN) helped the team search databases. The screening of studies was carried out by three reviewers (LC, LX and DC) in two stages. At stage 1, the EndNote citations were shared amongst all reviewers. The primary reviewer (LC) screened each article based on title and abstract against the inclusion and exclusion criteria (Table 4.1). The other reviewers screened the titles and abstracts of half of the citations each based on the selection criteria. The review team discussed the screening results. Citations not selected by at least two reviewers were excluded from full text review but maintained for future reference. At stage 2, each reviewer independently reviewed the full text of all included articles for eligibilities. Any discrepancies were discussed until consensus was achieved.

The initial search generated 4660 articles. After removing duplicate citations, 2301 article titles and abstracts were screened. Of those, 131 full-text studies were assessed for eligibility based on the inclusion/exclusion criteria (Table 4.1). In total, 29 studies (Appendix 11) were selected for appraisal including 27 studies identified from the electronic database search and two studies identified through the manual search of reference lists (Figure 4.1).

Table 4.1

Inclusion Criteria	Exclusion Criteria					
People with stroke' and family caregivers'	Studies examining experience poststroke					
experiences, perspectives or needs in	long term (over 6 months after hospital					
hospital to home transition as a primary	discharge) needs.					
aim.	Studies concentrate on discharges to					
Transition Period:	facilities other than home.					

Inclusion and Exclusion Criteria

Inclusion Criteria	Exclusion Criteria						
Preparation for discharge from hospital to	Studies conducted across multiple periods						
home up to 6 months after hospital	of stroke trajectory where the transition						
discharge:	period cannot be separated.						
The Preparation and Implementation							
stage of poststroke recovery trajectory							
defined by Time It Right framework							
(Cameron & Gignac 2008).							
Majority of after-hospital care programs							
last within 6 months (Allen et al. 2014)							
People with stroke and caregivers are adult	Studies that focus on people with stroke						
aged 18 years and above.	below 18 years old.						
Caregivers are defined as unpaid	Studies on multiple patient populations						
caregivers including spouse or partner,	such as traumatic brain injuries, brain						
family members, friends, or significant	tumour, dementia, hip fracture where						
others who provide physical and	data relevant to stroke cannot be						
emotional support to people with stroke.	separated.						
	Studies of multiple population e.g. health						
	care professionals, service provider,						
	where perspectives of people with stroke						
	and caregivers cannot be separated.						
Original studies using qualitative methods of	Studies using quantitative methods or						
data collection & analysis	mixed methods where the qualitative						
	data cannot be separated.						
	Qualitative studies involving statistical data						
	as reporting results.						

Inclusion Criteria	Exclusion Criteria					
	Conference proceedings, discussions,					
	reviews or commentaries.					
Published in English language	Studies published in languages other than					
	English.					

Figure 4.1

Review Flow Diagram



4.2.4.3 Quality Assessment

Quality assessment was undertaken by using the Critical Appraisal Skills Programme (CASP) tool (CASP, 2018a) (Table 4.2). The reviewers evaluated the quality of each study

independently until a consensus was reached. No scores were placed during the appraisal. All studies that met the selection criteria were included as methodologically weak studies may provide novel understandings that may not be produced in methodologically strong studies (Carroll et al., 2012). Every study provided a valuable perspective to achieve the aim and objectives of this review.

Table 4.2

Critical Appraisal

Studies	Questions*									
	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Q9	Q10
Bakas et al. 2002	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Cameron et al. 2013	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Cobley et al. 2013	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Connolly & Mahoney 2018	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Eilersten et al. 2010	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Erikson et al. 2010	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Finch et al. 2021	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Ghazzawi et al. 2016	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Gholamzadeh et al. 2015	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Grant 1996	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Greenwood et al. 2009	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Gustafsson & Bootle 2013	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Gustafsson et al. 2014	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Lou et al. 2017	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Lutz et al. 2017	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Olofssen et al. 2005	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Pereira et al. 2021	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Pringle et al. 2013	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y

Studies	Questions*									
	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Q9	Q10
Rittman et al. 2004	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Rittman et al. 2007	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Sadler et al. 2014	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
Timothy et al. 2016	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
C. L. Williams et al. 2005	Y	Y	Y	Y	Y	?	Y	Y	Y	Y
White et al. 2007	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
White et al. 2015	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Wottrich et al. 2012	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
S. M. Yeung et al. 2011	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
E. H. Yeung et al. 2015	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y
Young et al. 2014	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y

Note. Y = Yes; N = No; ? = Can't tell. Adapted from Critical Appraisal Skills Programme (CASP) (2018a), <u>https://casp-uk.net/wp-content/uploads/2018/01/CASP-Qualitative-Checklist.pdf</u>

* Q1: Was there a clear statement of the aims of the research?

* Q2: Is a qualitative methodology appropriate?

* Q3: Was the research design appropriate to address the aims of the research?

* Q4: Was the recruitment strategy appropriate to the aims of the research?

* Q5: Was the data collected in a way that addressed the research issue?

* Q6: Has the relationship between researcher and participants been adequately considered?

* Q7: Have ethical issues been taken into consideration?

* Q8: Was the data analysis sufficiently rigorous?

* Q9: Is there a clear statement of findings?

* Q10: How valuable is the research?

4.2.4.4 Data Extraction

This review examined the preparation and implementation stage of the poststroke

recovery trajectory as defined by the Time It Right framework (Cameron & Gignac, 2008).

The preparation phase refers to the stage "before patient goes home" from an acute or

rehabilitation hospital. The implementation phase is the "first few months after patient returns

home" (Cameron & Gignac, 2008, p. 308). Data up to six months after hospital discharge

were extracted as the majority of studies on hospital to home transition reported a three to

six months follow-up. Findings from the primary studies, including participants' quotations

and researchers' interpretations, were extracted and organised into a data extraction form guided by Bettany-Saltikov (2012).

4.2.4.5 Data Synthesis

Thomas and Harden's approach to meta-synthesis guided the inductive thematic synthesis of data (Thomas & Harden, 2008). The synthesis involved three stages. First, after reading and re-reading full texts, we extracted data from each study relevant to the objectives of the review. We then conducted line by line coding of the extracted data and compared codes across all reviewed studies. This analytical process enabled the translation of concepts from one study to another (Thomas & Harden, 2008). Second, we assessed similarities and differences between the codes and grouped to create new codes. This step resulted in the development of descriptive themes. Third, based on the review questions, we discussed the descriptive themes (see Appendix 12 for an example) and identified analytical themes and subthemes.

We applied the GRADE-CERQual (Confidence in the Evidence from Reviews of Qualitative Research) approach (Lewin et al., 2018) to assess our confidence in the findings (Appendix 13). The CERQual provides a systematic and transparent framework to support reviewers' reflection on their confidence in each individual review finding from four perspectives: methodological limitations, coherence, adequacy of data and relevance (Lewin et al., 2018).

4.2.5 Findings

Twenty-nine studies were included in the synthesis. Sample sizes across studies ranged from 6 to 125. The total number of participants in the reviewed articles is 517 stroke survivors and 252 caregivers. Most studies were undertaken in the USA (8), with the other studies conducted in Canada (5), UK (4), Sweden (3), Australia (3), Norway, Iran, Denmark, New Zealand, Hong Kong and Portugal (1 each) (Table 2). The most common methodologies were qualitative descriptive, interpretive and grounded theory with semi-structured interviews. The review identified three higher order themes with six subthemes: 1) factors enabling transitional care (subthemes: *partnership approach to individualised*

transitional care; being prepared to navigate health and social care services and developing self-management capabilities); 2) barriers to transitional care (subthemes: gaps in discharge planning; factors affecting self-care at home, and inability to cope with challenges); 3) expectations of transitional care.

A GRADE-CERQual approach to the assessment of confidence in the findings reached 'high confidence' in all but one subtheme (being prepared to navigate health and social care services). We agreed on 'moderate confidence' for this subtheme due to the low number of studies and relatively thin data (Appendix 13).

4.2.5.1 Factors Enabling Transitional Care

This review identified three key factors that enabled hospital to home transitional care for stroke survivors and caregivers: partnership approach to individualised transitional care, being prepared to navigate health and social care services and developing self-management capabilities.

Partnership approach to individualised transitional care

Stroke survivors and caregivers considered engagement in care and decision-making an essential part of discharge planning. They felt well-informed of their progress, which provided them with a sense of trust and confidence (Lou et al., 2017; Sadler et al., 2014). As a survivor stated:

During my process of hospital stay she [partner] was there all through the time with me, so the doctor speak with her and she knows all the changes of what I'm going through and what I'm being there coping for, so she understands everything from the beginning (Sadler et al., 2014, p. 1915).

A family-centred approach to discharge planning not only empowered survivors and their families and fostered independence, but also supported family caregiving and decisionmaking (Cameron et al., 2013; Gustafsson & Bootle, 2013; Pereira et al., 2021; White et al., 2007). As a caregiver highlighted:

Within the first week we had the choice of a family conference, which we elected to have. And we got all the family, and they all sat down with all the staff that ... were

involved with Catherine's rehabilitation. And they explained what they did and what their plans were... that type of planning on behalf of the hospital is very good to give you an idea of what's going to happen (Gustafsson & Bootle, 2013, p. 1384).

Stroke survivors valued the opportunity to practice personal care in a home environment prior to discharge as this helped them attain skills and increased confidence. This also enabled caregivers to assess caregiving capability and identify issues before their loved ones returned home (Cobley et al., 2013; Gustafsson et al., 2014). As a stroke survivor described:

I do say that an invaluable part was coming home for the day and the weekends and the therapy in home was very much worthwhile because it gives you the confidence to do things in your own home that you don't in the hospital (Gustafsson et al., 2014, p. 2247).

As a crucial part of poststroke rehabilitation and recovery, adequately preparing caregivers is a key element of effective transition (Cameron et al., 2013; Gustafsson & Bootle, 2013; White et al., 2007). Caregivers benefited from individualised information provision which combined verbal explanation and written material. As a caregiver praised:

Instead of just handing me information, she sat down and went through each point even though I was so tired and I'm going to remember it so much more. When I need to, I'll know in the sheets that she gave me where to go to look for the information (Cameron et al., 2013, p. 320).

Caregivers also valued opportunities to learn and practice caregiving skills while survivors were still in hospital. Caregivers' training helped them build on personal care techniques and increased confidence in assisting with post-discharge rehabilitation (Cameron et al., 2013; Lutz et al., 2017; White et al., 2007). As a caregiver described:

I worked with the nurses. When he took a shower and different things, helping to get dressed, his exercises...I practised all that before he came home...And then the two weekends that he came home...I thought that was pretty good (White et al., 2007, p. 9).

Health professionals who treated people with dignity and respect empowered survivors and caregivers and enhanced positive experiences during transition. As a survivor shared: "…was very reassuring he didn't tell me not to come [to the hospital] … and um I felt better after I spoke with him and I thought well I'll just, just wait and see if things get better" (Connolly & Mahoney, 2018, p. 3984). Health professionals' compassion and sensitivity towards survivors and caregivers further augmented post-discharge support and served as an impetus for continued participation and reintegration.

Being prepared to navigate health and social care services

Survivors and caregivers perceived that their competence in navigating complex health systems was built by health professionals. As a survivor stressed: "Transition [care team] have been terrific. Very helpful [with the] things that I can get and they've given me names of all the people I can ring who can who help me" (Gustafsson & Bootle, 2013, p. 1383). Their sense of safety was reinforced with the continuity of post-discharge follow-up services. As described by a stroke survivor: "In this process I've felt completely safe. Like, they didn't just send me home and hope for the best; I knew someone would check up on me, that I was not alone" (Lou et al., 2017, p. 305).

Early supported discharge services and community rehabilitation and social care services tailored to individual needs maintained stroke survivors' momentum for ongoing rehabilitation and helped develop self-management capabilities (Cobley et al., 2013; Gustafsson & Bootle, 2013). One stroke survivor was motivated by a therapy approach that addressed his personal goals and helped his reintegration into the community. As he stated: "The team were encouraging and motivating and would take me on a walk to make sure I could get on a bus and that I was able to cross the road, things like that" (Cobley et al., 2013, p. 753).

Developing self-management capabilities

Factors that enabled the development of self-management capabilities included positive thoughts (Connolly & Mahoney, 2018; Gustafsson & Bootle, 2013; Lou et al., 2017; Timothy et al., 2016; S. M. Yeung et al., 2011), positive self-appraisal of functional

improvement and self-efficacy (Olofsson et al., 2005; Pringle et al., 2013; Sadler et al., 2014; White et al., 2007; Wottrich et al., 2012), and family and social support (Greenwood et al., 2009; Lutz et al., 2017; Rittman et al., 2007; Young et al., 2014).

Stroke survivors found it easier to accept poststroke limitations if they realised that their condition was more manageable compared to worse cases. This gave them reassurance and confidence in their capabilities to manage their own conditions: "My friends took me to a testimonial meeting with a man without limbs. Wow! I felt at peace...and positive about my life! I found courage... because I still have arms and legs" (S. M. Yeung et al., 2011, p. 2398).

Over the course of the transition, survivors gradually accepted their situation and understood that recovery is a process that takes time. Accepting that reality enables them to adapt to their new and changed role after stroke. As a survivor reflected: "Mentally, I'm alright, and going to, you know, I've accepted the fact that ... this is what's happened" (Timothy et al., 2016, p. 1570). Caregiver's with a positive view of their role smoothen transition: "It brought us closer together...it's not as sexy as it used to be, but we are still close... we help one another. I think it brings you closer, being with the person all the time" (White et al., 2007, pp. 9-10).

Survivors and caregivers developed more confidence in self-management through positive self-appraisal: "Yes, I mean I think that every time we get a little bit further away, it makes us more secure. It is like dangerous waters and we are gradually sailing out of them" (Greenwood et al., 2009, p. 1126). Caregivers' self-efficacy, built on mastery over caregiving activities, also augments self-management. As a husband described:

I have done so many things that I never thought that I would be able to do around the house – the banking, the shopping, the cooking, giving my wife a shower, making the bed... just something I'd never done...I've learned a lot of things (White et al., 2007, p. 9).

Family and social support played a critical role in assisting survivors' early reintegration during the transition and therefore reinforced their self-management

capabilities: "I'm teaching my son how to cook...him or my wife, cause neither one can cook. She can't cook but she'll try, so...I go in there and I guide her" (Rittman et al., 2007, p. 27). Family and social support also eases the burden on caregivers by serving as a respite to prevent caregivers' burning out and promotes sustainability of long-term caregiving: "My neighbours said they are all going to get their schedule and make a listing and come over even if it's for only an hour, so I can go to the grocery store" (Lutz et al., 2017, p. 884). Family support encouraged caregivers' adjustment, sustained caregiving and expedited the return to normalcy. A caregiver stated:

It was hard to start with but now it is sort of getting in a routine so unless suddenly something goes wrong. . .everyone has some sort of job so we all do that—the brother and sister, the other sister and my brothers (Greenwood et al., 2009, p. 1127).

4.2.5.2 Barriers to Effective Transitional Care

Three major intertwined barriers to effective transition are gaps in discharge planning, factors affecting self-care at home and inability to cope with challenges.

Gaps in discharge planning

Discrepancies between survivors and health professionals regarding discharge priorities and goals were described by stroke survivors and caregivers (Pereira et al., 2021; Sadler et al., 2014; S. M. Yeung et al., 2011). For example, a caregiver described rehabilitation activities suggested by a therapist were not accepted by the survivor due to personal preference:

The therapist wants him to go there, to the shop. He used to enjoy being in the shop serving the public, but now she [the therapist] wants him to pack things up and transport bottles as a way of moving and he doesn't want to (Pereira et al., 2021, p. 2904).

Some participants were exasperated when they expected a compassionate approach, but this was missing in interactions with clinicians (S. M. Yeung et al., 2011,

Cobley et al., 2013, Bakas et al., 2002). They felt that clinicians often cared more about their physical than emotional needs. As a survivor complained:

They took special care with regard to my diet and giving me my medication on time. However, they were rarely concerned with my feelings. Actually, the most important thing for a patient is words of concern (S. M. Yeung, et al., 2011 p. 2399).

A lack of engagement and being discharged with unanswered questions caused survivors and their family unnecessary anxiety and difficulties with daily care (Gholamzadeh et al., 2015; Olofsson et al., 2005; White et al., 2007). A participant shared their frustration: "When you ask a question, they do not answer. There are so many things and tips a nurse knows, which would take months for us to learn, but they did not tell us if we asked them" (Gholamzadeh, et al., 2015, p. 211).

Most caregivers are new to this role. They therefore need to have hands-on training in daily personal care activities. Inadequate preparation before discharge adds to caregivers' burden and anxiety. A caregiver shared the experience:

Last month, the patient was discharged; it was a difficult time for us. We had lots of problems. We did not have any knowledge in this field... We did not know how to move her, how to clean her or how to put diaper on her. They didn't instruct us at the hospital... We need to know how we can help our patients with peace and comfort for both of us (Gholamzadeh et al., 2015, p. 208).

Some caregivers were not prepared for post-discharge care requirements when survivors were discharged home. They struggled to manage care activities and were left without any help. As a caregiver shared:

Only realized how much care my father would need once he was home and no one really gave us any warning at the hospital. Even though I thought I would have help, when I really needed it, it wasn't there (White et al., 2015, p. 1095).

Factors affecting self-care at home

Survivors and caregivers are in a well-controlled environment while in hospital. Once home, stroke survivors rely on support from caregivers to maintain ADLs. Coupled with

inadequate pre-discharge preparation and lack of caregivers' training, inadequate support in the community exacerbates an already challenging transition (Cameron et al., 2013; Eilertsen et al., 2010; Lutz et al., 2017; White et al., 2015; Young et al., 2014). Stroke survivors and caregivers felt that they were forgotten by the healthcare system after discharge. As one stated: "They cast us adrift, you're into the community and you're on your own" (Cameron et al., 2013, p. 320). They also experienced anxiety and uncertainty: "We felt on our own, like we had been abandoned and were so nervous and scared about everything" (White et al., 2015, p. 1096).

Survivors perceived the lack of follow up jeopardised their continued rehabilitation and reintegration. As a survivor questioned:

They don't really help you get back into life, do they? They just sort of, you have a stroke, you have physio and that's it... it [stroke] affected my hand so obviously that was my job, so in effect: How will she move on from there? What can we do to help her get back into work? (Sadler et al., 2014, p. 1915).

There can be a significant gap between inpatient and community rehabilitation services. The considerable waiting time for community rehabilitation programs experienced by some survivors impacted on the achievement of their goals. A survivor described: "All of a sudden, it's like, "Oh, we've referred you to the hospital again to get the physio," which has took...three months. So I've had intense physio for six weeks and then, for three months, I've had nothing" (Cobley et al., 2013, p. 754).

Caregivers are imperative to support stroke survivors' continuous rehabilitation and reintegration into the community. However, some caregivers perceived being taken for granted as the home care service was never offered before discharge. A caregiver expressed her dissatisfaction: "I'm very disappointed that they didn't offer to help me, because obviously he would have had to go into a home or somewhere if I wasn't doing it" (Cobley et al., 2013, p. 754).

Financial constraints have been identified as another significant barrier during transition. Survivors and caregivers reported that poststroke disability brought an

unexpected increase in daily expenses and decreased income due to unemployment (Gholamzadeh et al., 2015, Lutz et al., 2017, Young et al., 2014, White et al., 2007). A survivor stated: "Well I'm anticipating bankruptcy because I don't see any way out of it...bills have gone unpaid simply because I don't have the money that I had when I was working part-time" (Lutz et al., 2017, p. 885).

A lack of financial resources also compromises ongoing rehabilitation with survivors missing out on essential equipment and therapies (Bakas et al., 2002; Ghazzawi et al., 2016; Grant, 1996; Lutz et al., 2017; Sadler et al., 2014; E. H. Yeung et al., 2015). One survivor talked about giving up on buying a walking aid that could potentially improve mobility: "[The doctor] said for me to have a walking stick but my insurance ran out...I was intending to go back and get it, but I don't have the money, \$45, so I let it go" (Grant 1996, p. 897).

Financial situation is further exacerbated when survivors spend more on purchasing healthy food to make lifestyle changes to prevent secondary stroke. As a caregiver shared: "We've had to change the way that everybody eats because of him, that gets expensive because you've got to buy all this fruit now and eating healthy is expensive" (Bakas et al., 2002, p. 248).

Inability to cope with challenges

Stroke survivors' and caregivers' views on and the level of acceptance of poststroke physical, cognitive and emotional impairments have a significant impact on their ability to cope with poststroke disabilities and achieve community reintegration during the transition phase (Erikson et al., 2010; Finch et al., 2021; Gholamzadeh et al., 2015; Grant, 1996; Gustafsson et al., 2014; Olofsson et al., 2005; Pringle et al., 2013; Rittman et al., 2007; Timothy et al., 2016; C. L. Williams et al., 2005). Stroke survivors perceived themself as a stranger. They considered themselves as "half of a person" (C. L. Williams et al., 2005, p. 286) and that 'it feels as if you've gone and somebody else is in your place' (Timothy et al., 2016, p. 1569).

After returning to the community, survivors found it difficult to adjust to their poststroke disabilities, which caused considerable emotional distress and negative thoughts. A survivor resented:

You're alive, but you are worthless. Alive for what? They have to feed you, bathe you... What can you say that you are doing? Living a monotonous life, like a vegetable. Sometimes it is better that they give you something that leaves you dead and it is over (Rittman et al., 2007, p. 29).

Survivors dependent on family caregivers and others to perform basic daily activities perceive themselves as a burden and "troublesome" (S. M. Yeung et al., 2011, p. 2399). One survivor viewed herself as a nuisance to her husband: "There is a lot that he (her healthy husband) is supposed to do, there's actually too much that he has to do, it does not feel quite fair, but when you can't, you can't" (Wottrich et al., 2012, p. 1221).

The difficulty of developing strategies to cope with their changed role resulted in survivors' social disengagement. One survivor described himself as being "apart" from others due to his inability to be independent, which made him think of "taking my life" (C. L. Williams et al., 2005, p. 285).

Caregivers struggled to adapt to the role. They are challenged by their own health limitations, the unexpected increase in responsibilities and personal views on poststroke life (Young et al., 2014, Gholamzadeh et al., 2015, White et al., 2007, Lutz et al., 2017, Bakas et al., 2002). Whilst having to manage their own health conditions, caring added another layer of complexity and anxiety to their lives. A survivor's wife talked about her worries: "I thought I was going to collapse physically, my knees were going on me...my back was hurting. I just thought if I collapse, what's going happen to the two of us. It's very scary" (White et al., 2007, p. 8).

When survivors return home, caregivers are faced with a sudden increase in household responsibilities that they may not have realised or been prepared for. This was described by a wife who cared for her husband on her own: "When he came home—I was

reeling for weeks. My mind was just going a hundred different directions. I had no idea it was going to be as hard as it was" (Lutz et al., 2017, p. 882).

With increased responsibilities and more daily activities to perform, caregivers have less time for social interactions with others. This left them "feeling like a prisoner in your own home" (White et al., 2007, p. 8). The lack of emotional support for caregivers to develop coping strategies caused despair. As a caregiver resented: "There are things worse than death, and this is it. This is the living dead" (Young et al., 2014, p. 1896).

4.2.5.3 Expectations of Integrated Transitional Care

This review has identified stroke survivors' and caregivers' expectations of personcentred integrated transitional care services. Participants expressed their desire for compassionate and holistic care (Connolly & Mahoney, 2018; Gholamzadeh et al., 2015). Participants expected both their physical and psychological needs to be met. As a family suggested:

Doctors and nurses do not care about patient's mental status but only care about patient's physical health, such as doing CT scan and giving them their medicines. However, the main problem of the stroke patients with functional loss is their mental status...the most important measure should be psychotherapy for both patient and family (Gholamzadeh et al., 2015 p. 212).

Survivors and caregivers considered having a central point of contact throughout their trajectory from discharge planning to transitioning home a necessity (Cobley et al., 2013, Lutz et al., 2017, White et al., 2007, White et al., 2015). During the discharge planning phase, they would have preferred a care coordinator who would talk to them regularly about their progress and keep them well-informed of what to expect next (White et al., 2007). A home visit before discharge is desirable to prepare for the reality they will face once the survivor returns home (White et al., 2015).

Most stroke survivors are unfamiliar with how the health system works. Prior to discharge, they prefer to have someone to provide information about services and benefits that they may be entitled to receive. As a survivor stated: "I thought it'd be good to talk about

if you were entitled to any benefits because I've never been on the sick ever. I didn't understand any of it, we had to figure it out for ourselves" (Cobley et al., 2013, p. 755).

Participants also applauded timely follow up and support after returning home. They considered follow up and support as essential to achieving their rehabilitation goals and sustain caregiving (Cameron et al., 2013, Connolly & Mahoney, 2018, Grant, 1996, Lutz et al., 2017). A survivor longed for a follow up phone call from hospital after discharge. As she suggested: "A hospital calling to check in on you and to see how you're doing ... I would have loved something like that" (Cameron et al., 2013, p. 320). Community support for survivors reassures caregivers and helps sustain caregiving. A caregiver expressed her wish: "I would have been more at ease at work if I knew a lady was here every day... just to assist her as far as going to the restroom and dressing her and making sure she is fed right" (Grant 1996, p. 897).

4.2.6 Discussion

The first six months of hospital to home transition is the most critical time for stroke survivors and caregivers to build their capability to maximise recovery, reorient to a changed life and learn to manage health conditions at home. This meta-synthesis of qualitative evidence facilitates an in-depth understanding of stroke survivors' and caregivers' perspectives on the enablers and barriers to an effective transition, and the complex interplay of poststroke functioning, environmental factors and personal factors. Findings from this review demonstrate that opportunities and challenges coexist during the transition. Effective hospital to home transition care can be achieved when health professionals utilise a partnership approach to individualised transitional care; develop stroke survivors' and caregivers' and caregivers' capabilities to navigate and utilise health and social care services to meet their care needs; and active self-management capabilities.

The ICF framework, underpinned by the biopsychosocial model, enabled a comprehensive analysis of the interaction between poststroke functioning and personal and environmental factors (WHO, 2013). The findings of this meta-synthesis confirm that the interplay of stroke survivors' poststroke disabilities, post-discharge support and their self-

management capabilities could impede or facilitate hospital to home transition care. Multidimensional environmental factors and complex personal aspects in poststroke hospital to home transition care call for an integrated approach to care and continuity of rehabilitation processes, which are vital to maintain and foster long term self-care at home (von Koch, 2019; WHO, 2015).

Stroke triggers an inflammatory immune response and systemic immunosuppression which makes survivors susceptible to infection (Boehme et al., 2018; Pagram et al., 2017). About 30% of the stroke population develop an infection and this is one of the major causes of 30-day readmissions after discharge (Boehme et al., 2018, Pagram et al., 2017). Recurrence of stroke is another culprit behind readmissions after discharge. Approximately 65% of stroke survivors are hospitalised with secondary stroke within 12 months of the first stroke onset (Abreu et al., 2020). Providing stroke survivors and caregivers with knowledge of stroke related complications, medication management and lifestyle changes empowers them with self-management skills and proactive actions to manage risk factors and prevent adverse events. Person-centred discharge planning and transitional care coordinated by stroke nurses could reduce preventable hospitalisations resulting from complications (Condon et al., 2016; White et al., 2015).

This review demonstrated that partnerships with stroke survivors and caregivers and consistent engagement in decision-making and discharge planning processes are key factors to the success of hospital to home transitions. Due to shortened hospital stays, comprehensive discharge planning must begin early in the inpatient setting to enable preparation for transitioning to the community (Mountain et al., 2020; Olson & Juengst, 2019). Barriers to a partnership approach during discharge planning include the readiness of survivors and caregivers to participate, the lack of organisational support for health professionals and inadequate person-centredness in goal setting (Busetto et al., 2020; Connolly & Mahoney, 2018; Kable et al., 2019; Krishnan et al., 2019; Pereira et al., 2021). With mismatched care goals and priorities between health professionals and survivors/caregivers, an informed and well-planned transition is unlikely to be achieved,

resulting in unmet care needs and negative impacts on the development of self-management capacities for long-term stroke management at home (Andrew et al., 2018; Hughes et al., 2020). How to collaborate with stroke survivors/caregivers and bridge these gaps in setting goals and priorities to achieve shared discharge planning are important service development areas for organisations and health professionals to explore.

Self-management refers to an individual's ability to manage a chronic condition in collaboration with the caregiver, family, community and the appropriate health professionals (Satink et al., 2016). Support from health systems and community resources is crucial to empower and enable the development and activation of self-management capabilities (Barker et al., 2018; Fryer et al., 2016). Furthermore, it is equally important for stroke survivors and caregivers to actively take part and generate ownership over the management of their condition throughout the rehabilitation and recovery process (Sadler et al., 2017; A. Taylor et al., 2019). The findings of this review confirm that self-management capacities are co-developed by survivors/caregivers and health professionals. Furthermore, the selfefficacy beliefs (or confidence in self-management) of survivors/caregivers are enhanced by their motivation to learn and commitment to take charge of care at home combined with health professionals' encouragement and feedback on performance (Dineen-Griffin et al., 2019; Pereira et al., 2021). Trustful relationships and open dialogues between stroke survivors/caregivers and health professionals motivate stroke survivors/caregivers to learn and self-manage rehabilitation and recovery at home (Osborne & Neville, 2019; Theadom et al., 2018). The activation of self-management capacities begins before discharge and is further augmented with post-discharge health and social care support in the community (Reeves et al., 2019, Mountain et al., 2020).

This review revealed that stroke survivors' and caregivers' personal attributes and contextual factors played an important role in transitional care. Regardless of stroke severity, survivors' and caregivers' introspection triggered both negative and positive thoughts about stroke, their role, relationships with family and post-discharge life (Finch et al., 2021; Hodson et al., 2019). Survivors' and caregivers' pre-stroke family dynamics, past health conditions,

willingness and readiness for caregiving shaped their ability to engage and participate in rehabilitation and discharge planning, and the development of self-management capacity (L. Chen et al., 2020, Hodson et al., 2019, Osborne & Neville, 2019). Moreover, it is estimated that 40% of stroke survivors experienced cognitive impairment that had a significant impact on their self-management capability, and this is associated with a higher level of caregiver burden (Sexton et al., 2019; Wu et al., 2019). Therefore, caregivers need to be prepared with knowledge and skills to cope with cognitive impairment associated behavioural symptoms and functional decline (Rohde et al., 2019). These personal factors interplay with environmental factors, such as availability of financial support, accessibility of health and social supports, and returning to driving and work, and may contribute to readmission and early institutionalisation during the transition (Kilkenny et al., 2020; Pindus et al., 2018). Due to their close therapeutic relationships with stroke survivors/caregivers, nurse-led transitional care interventions targeting the hospital to home transition period could address some of these challenges and help sustain care at home (Condon et al., 2016; Olson & Juengst, 2019).

This review has limitations. We only included literature published in the English language. Twenty-six of the included studies were undertaken in developed countries (United Nations Department of Economic and Social Affairs, 2020). The perspectives of stroke survivors/caregivers in developing countries may resemble or differ compared to their counterparts in developed countries. The review of transitional care in developing countries/regions may further enrich understandings of the issues of concern. Additionally, the investigation timeframe was set at within six months of discharge as most transitional studies' follow up period is three to six months. This may have impacted on the narratives as transition is a personalised experience that may extend beyond six months. Moreover, we only reached 'moderate confidence' in the subtheme 'being prepared to navigate health and social care services' due to concerns on its adequacy based on GRADE-CERQual assessment. This suggests that more primary research is needed to confirm this finding. This review will need to be updated when more original studies are available. Finally, this

review only focused on the views of stroke survivors and caregivers. To generate an overall insight into challenges during the transition, future reviews need to consider perspectives of all stakeholders including health professionals and service providers. These limitations need to be considered when interpreting the findings of this review.

4.2.7 Conclusion

Through the lens of stroke survivors and caregivers, this meta-synthesis underlines challenges and opportunities in the first six months of the poststroke hospital to home transition. Findings from this review support that health professionals' partnering and engaging with stroke survivors and caregivers in individualised discharge planning facilitates self-management during their transition. Integration within multidisciplinary teams and between care service providers enables stroke survivors/caregivers to achieve care goals, develop self-management capacities and paves the way for long-term self-management. This review highlights complex and multidimensional issues in poststroke hospital to home transitional care and identifies gaps for further research.

Future research should focus on the perspectives of all stakeholders in hospital to home transitional care, especially that of health professionals, primary care providers and care service providers/organisations. Their views will deepen the understanding of issues at an organisational and systemic level. Furthermore, the unique care needs and challenges of survivors/caregivers in rural and remote regions need to be explored. Further development of stroke-transition-specific clinical pathways and care models will enhance the consistency and quality of service delivery, facilitate the integration of care services across hospital and community settings, and empower stroke survivors and caregivers to achieve and sustain long-term poststroke management at home.

4.2.8 Relevance to Clinical Practice

Protocols and clinical guidelines relating to discharge planning and transitional care support need to be reviewed to ensure partnership approach with survivors/caregivers in the design and delivery of individualised transitional care. Nurses specialised in stroke care are in a unique position to lead timely support for survivors/caregivers and to bridge service gaps

in hospital to home transition care. A nurse coordinator's role needs to be considered to ensure integrated services across care settings and service providers in transitional care. Tele-health technology needs to be established to ensure those who reside in rural and remote regions are able to receive equitable transitional care services.

This is the end of the publication.

4.3 Summary

This chapter has presented the findings of a systematic review and meta-synthesis of qualitative studies. The systematic review explored, in the global context, stroke dyads' perceived enablers, barriers and expectations during the hospital-to-home transition. The findings highlight the desire of stroke dyads to be part of the planning for the hospital-to-home transition care. They considered that collaboration with health professionals builds their competence in navigating the healthcare system and helps foster self-management capabilities, which are essential for people with stroke to reintegrate into the community and for their caregivers to sustain long-term caregiving after they leave hospital.

The next chapter reports the methodology and methods for the qualitative study, which aimed to investigate, in an Australian context, the perspectives of people with stroke, caregivers and health professionals regarding care needs and expectations during hospitalto-home transitions.
Chapter 5: Research Methodology and Methods

5.1 Introduction

The previous chapter presented the findings from a systematic review of care needs and expectations in the hospital-to-home transition from the perspective of stroke dyads in a global context. The results highlighted the need to foster a collaborative approach to hospital-to-home transition care among all stakeholders of the stroke care chain to ensure that the care needs and expectations of stroke dyads are met. The scarcity of evidence on such a collaborative approach to transition care informed the present qualitative study. This chapter describes the study's philosophical underpinning, research methodology around Gadamer's phenomenological hermeneutics, and methods.

5.2 Aim and Objectives

The aim of the study was to explore the care needs and expectations in hospital-tohome transition care for stroke dyads in an Australian social context. The objectives were to:

- understand the experience and perceptions of stroke dyads during the hospitalto-home transition in a global context;
- understand stakeholders' perspectives on approaches to identifying and meeting the transition care and rehabilitation needs of people with stroke in an Australian social context.

5.3 Philosophical Perspectives of the Study

The word 'philosophy' originates from the Greek word '*phílosophía*', meaning the 'love of wisdom' and can be referred to as 'the study of the fundamental nature of knowledge, reality and existence' (Oxford Learners' Dictionaries, n.d.). Philosophy provides the principles of theoretical thinking and the philosophical underpinnings for research. It guides researchers in the design, conduct, analysis and interpretation of research and its outcomes (Moon & Blackman, 2014).

5.3.1 Paradigms in Research Design

A paradigm is a world view and a basic set of beliefs that guide the inquirer's action (Denzin & Lincoln, 2018). Major paradigms that influence research design include positivism, post-positivism, constructivism, critical theory and participatory action frameworks (Denzin & Lincoln, 2018). Each paradigm has its unique ontological, epistemological, axiological and methodological basis (Lincoln et al., 2018). They address different concerns of a study. Three paradigms are widely used in nursing research and are discussed in the following sections.

5.3.1.1 Positivism

Positivism is also called 'logical empiricism' and emphasises the rational and the scientific (Liamputtong, 2017, p. 14). Positivists view the world as a 'real reality' that can be studied and known. They assume that nature is stable and organised, that an objective reality exists and that it is separate from human observation (Lincoln et al., 2018, p. 110). Hence, positivists apply empirical tests with scientific methods to study hypotheses under carefully controlled conditions. They believe there is a real world that is driven by causes and effects (Lincoln et al., 2018).

The fundamental difference between positivism and other paradigms is its ontological perspective of 'realism' (Lincoln et al., 2018, p. 110). Positivists value objectivity as their epistemological stance. They strive to put aside personal beliefs and thoughts and avoid contamination of the studied phenomena. In the methodological dimension, positivists believe in scientific, quantitative methods with a 'gold standard' and tight control of the research situation (Lincoln et al., 2018, p. 117).

Under the same umbrella, the post-positivist paradigm evolved over time as strict positivist thinking was challenged. However, the revised post-positivists still believe that a true reality is out there. However, they are more conscious of the impossibility of total objectivity, though they continue to strive for objectivity and minimum interaction with research subjects (Lincoln et al., 2018).

Successful research guided by positivism results in replicable and objective law, which leads to generalisation. The strengths of positivism are its clarity, precision, rigour, standardisation and generalisability (Bryman, 2016; Lincoln et al., 2018). However, to remain objective, this approach does not consider the voice of participants. Only the researchers' voices are heard in the research, which denies human uniqueness and individuality (Bryman, 2016; Lincoln et al., 2018). Hence, research results may not be applicable to human beliefs, values and people's lives in different contexts and cultures.

5.3.1.2 Constructivism

Constructivism is also known as 'interpretivism' (Liamputtong, 2017, p. 13). Contrary to positivism, constructivists view reality as multiple and subjective. It is understood as a construction of individuals and is based on their beliefs and the sociocultural environment they interact with (Houser, 2015). Researchers under this paradigm believe that knowledge is co-developed and obtained through interaction between the researcher and the participants (Lincoln et al., 2018).

In the ontological dimension, interpretivists hold the world view of 'relativism'. They believe that multiple realities exist, and they depend on the individual's social and experiential situation to know the realities (Lincoln et al., 2018). Knowledge is constructed through lived experiences and interaction with others in society. To the interpretivists, produced knowledge is reflective of the reality of the participants. This ontological stance is opposite to that of the positivist (Lincoln et al., 2018).

In the epistemological dimension, interpretivists believe that new knowledge is coconstructed by the researcher and participants through interaction. The relationship between the researcher and the participants is central to new knowledge development. The main goal of inquiry based on constructivism is to inform improved practice through understanding and interpreting the meanings of phenomena (Lincoln et al., 2018). Thus, the researcher needs to actively participate in the entire research process.

The methodologies commonly used by constructivists include qualitative approaches such as phenomenology, grounded theory and case studies. Dialogue between the

researcher and the participants is valued as a way to collaboratively construct a meaningful reality (Lincoln et al., 2018). Therefore, methods used to collect information include interviews, focus groups, observations and the analysis of existing texts. Evidence is generated through an inductive process. However, there are a number of limitations to using constructivism in research. Under this paradigm, information is collected through human beings as a study instrument. Human beings are intelligent and sensitive but prone to errors. Additionally, the subjective nature of constructivism may yield different conclusions by two constructivists in similar study settings and populations.

5.3.1.3 Critical Theory

Critical theory belongs to approaches within the 'transformative paradigm' (Polit & Beck, 2017, p. 480). In the ontological dimension, critical theorists view the world as 'historical realism' (Lincoln et al., 2018, p. 111). They believe that the operation of human nature in the world is based on a struggle for power. Virtual reality is shaped by social, political, cultural, economic, ethnic and gender values. Studies under this paradigm are driven by the aim to create change and benefit those oppressed by power (Lincoln et al., 2018). In the domain of epistemology, knowledge is viewed as subjective, historical consciousness and insights and developed through collaborative critical reflection with participants. It involves the self-reflective aspect of the researcher. The researcher must account for their own transformative effects to reveal changing directions and catalyse changes (Lincoln et al., 2018).

Critical researchers triangulate multiple methodologies to inform and catalyse changes. During the interactions with participants, researchers particularly focus on the expertise of participants. While carefully presenting knowledge based on their own paradigm, critical theorists remain sensitive to the views of participants (Lincoln et al., 2018).

When used in the nursing discipline, critical theory allows for the consideration of the sociocultural and political contexts of nursing care. It provides a framework for nursing research on the investigation of social inequalities related to health and the healthcare system and for transforming these situations (Mosqueda-Díaz et al., 2014). Critical

ethnography is one example of critical theory's application in nursing study; it is considered suitable for health promotion research due to its focus on enabling people to have control of their own situation (Polit & Beck, 2017).

In summary, paradigms guide the design and the conduct of research. All paradigms have their strengths and weaknesses. Positivism emphasises scientific and statistical methods and the objectivity of evidence but is criticised for its ignorance of the values and beliefs of human beings. Other paradigms focus on subjectivity but are critiqued for their lack of replicability and generalisability of findings. Nevertheless, the knowledge of paradigms enables the appropriate use of a suitable methodology to achieve research goals.

5.3.2 Using Constructivism to Guide the Research Design

The design of this study is guided by constructivism. Constructivism views the existence of reality as being multiple and subjective. Reality is self-created by individuals based on their own way of thinking about life and their role in that life. The aim of an inquiry underpinned by constructivism is to understand and interpret the lived experiences of human beings (Lincoln et al., 2018).

The purpose of the present study was to explore stroke dyads' experiences, perspectives and expectations in transition care and rehabilitation. The constructivist approach helps researchers to develop in-depth insights into the experiences and perspectives of people with stroke, their caregivers and health professionals. In addition, constructivism emphasises that knowledge is co-constructed through the interaction between the researcher and participants through hermeneutical discussions (Lincoln et al., 2018). In this study, interviews with stroke dyads and focus groups with health professionals served as means to ensure sufficient dialogue and interaction between the researchers and the participants. New knowledge of transition care is gained through the interpretation of the participants' perceptions embedded in their cultural and social context (Lincoln et al., 2018).

Furthermore, the constructivist approach examines goals and knowledge to inform praxis (improved practice; Lincoln et al., 2018). One of the objectives of the present study was to develop an evidence-informed hospital-to-home transition care model and pathways

to enhance collaborative and consumer-directed transition care and rehabilitation services. As a stroke rehabilitation clinician, the researcher's original intention for conducting this study was to gain a deeper understanding of the perceived issues related to stroke transition care to improve clinical practice. This feature of constructivism well serves this intention.

As discussed above, the study aimed to understand the care needs and expectations of hospital-to-home transition care for stroke dyads in the Australian social context. Qualitative research design underpinned by constructivism tends to be holistic and enables the researcher to strive for an understanding of whole phenomena. It requires the researcher to be deeply involved in the research process and to be part of the research instrument. The collection and analysis of data are concurrent, which requires the researcher to be flexible and able to adjust to new information (Polit & Beck, 2017).

5.4 Research Methodology

5.4.1 Overview of Qualitative Research Approaches

The history of qualitative research can be traced to the last quarter of the 19th century when the term 'ethnography' was used by anthropologists to describe the way of life of particular local groups of people living in colonial situations (Erickson, 2018, p. 38). There are five main approaches to qualitative inquiry: ethnography, phenomenology, grounded theory, narratives and case studies (Creswell, 2013). The following section focuses on the first three methodologies that have been widely used in the nursing and social sciences.

Rooted in anthropology, ethnography describes and interprets the cultural behaviours of a particular group. Ethnographers participate extensively in fieldwork and engage actively in the cultural activities of the studied group (Creswell, 2013). Ethnographers are particularly interested in information such as cultural behaviours, cultural artefacts and cultural speech. Used in health practice, ethnography helps researchers to develop an understanding of the health beliefs and behaviours of a cultural or subcultural group (Polit & Beck, 2017).

Phenomenology has its roots in both philosophy and psychology. Its focus is on an understanding of lived human experience. Phenomenological researchers describe the

experiences of participants and how they experience them (Creswell, 2013). Through indepth conversations (e.g., interviews), the researcher and the participants enter into a 'coparticipant' relationship, which enables the researcher to enter into the world of the participant so as to obtain a more enriched understanding of their perspectives and expectations. Besides in-depth conversation, observations and introspective reflections are also data sources to gain deeper insight into the studied phenomenon (Creswell, 2013; Polit & Beck, 2017).

Grounded theory was developed in 1967 by two sociologists, Barney Glaser and Anselm Strauss (Creswell, 2013). Although grounded theorists also study people's experiences, they attempt to understand a process or action as grounded in reality (data). The theory generated from this kind of methodology is embedded in participants' actions and their meanings and, in return, is used to explain and anticipate human behaviours (Creswell, 2013; Polit & Beck, 2017). Data in grounded theory are sourced mainly through interviews and observation (Creswell, 2013; Polit & Beck, 2017).

All of the above approaches involve active interaction with participants to achieve research goals. The research methods shared among these research approaches include interviews and observations. In addition to these methods, ethnographers also collect data through diverse sources, including symbols and artefacts. Grounded theorists constantly compare data during the data collection process. To evolve theory, researchers using grounded theory usually go back and forth between participants and conduct new interviews (Creswell, 2013). The aim of study for these approaches is different. Grounded theory focuses on developing a new theory grounded in the collected data. Involving extensive fieldwork, the goal of ethnographers is to develop understandings of the culture of a group. Phenomenology, by contrast, strives to obtain in-depth knowledge of participants' lived experiences and interpret what they have experienced and how they have experienced it.

Any of these three approaches could be used in this study. However, phenomenology was chosen as the most appropriate research approach to address the aim and objectives of the study. The aim of this study was not to describe cultural behaviours or

generate a theory but to explore the hospital-to-home transition care experiences, needs and expectations of stroke dyads from the perspectives of stakeholders. It coincides with the inquiry aim of phenomenologists, who focus on people's lived experiences based on their individual historical, cultural and social context. Additionally, the goal of phenomenological inquiry does not stop at identification and explanation but also critiques the meanings generated from data (Crotty, 1996). Insights into the shared experiences of a particular group of people can be of great value to professionals, such as health professionals, policymakers and therapists (Creswell, 2013). One of the objectives of this study was to understand stroke dyads' transition care experiences and expectations to inform clinical practice; a phenomenological approach was well suited to achieving this objective.

5.4.2 Overview of Phenomenology

Phenomenology, developed in the early 20th century, is both a philosophy and a research methodology (Dowling, 2007). The definition of phenomenology is diverse, depending on its philosophical underpinnings. The simplest definition is that it is 'the study of things (phenomena)' (B. Taylor & Francis, 2013, p. 77). There are many types of phenomenology. Phenomenology can be classified into six types: descriptive (to accurately describe), essential (to explain essences and their relationships), phenomenology of appearance (how phenomena appear), constitutive (processes of phenomena) and hermeneutic (aims to reveal hidden meanings in phenomena; B. Taylor & Francis, 2013, p. 78). These types were established based on the inspiration of other philosophical thoughts on phenomenology. So, they are not exclusive of each other; rather, they are related (B. Taylor & Francis, 2013, p. 78). The next section discusses the key concepts of three main originators of phenomenology.

Developed by Edmund Husserl, descriptive (transcendental) phenomenology asked the question, 'What do we know as persons?', and emphasised the description of people's lives as they experience it through objectivity. The main feature of descriptive phenomenology is so-called bracketing, a process of identifying and holding aside the

researcher's pre-existing beliefs and understandings of the studied phenomenon and avoiding the influence of those pre-existing beliefs on the findings of the study. Therefore, to facilitate the emergence of potentially different possibilities of understanding, Husserl advocated for an objective approach (Carpenter, 2017, p. 159; B. Taylor & Francis, 2013, p. 79).

Building on Husserl's philosophical tradition, Martin Heidegger, a student of Husserl, developed an existential-ontological hermeneutics that rejected Husserl's objectivity through bracketing. Heidegger was concerned with human existence, or '*Dasein*' ('being-in-the-world') and believed that understanding could not be separated from people's pre-cognitive awareness(B. Taylor & Francis, 2013, p. 82). In another word, understanding is connected to one's history and moves back and forth between parts and whole of experiences. Heidegger developed the concept of the hermeneutic circle to envisage a whole regarding a reality that was situated in the detailed experience of everyday existence by an individual (the parts) (B. Taylor & Francis, 2013). Researchers who follow Heidegger's approach consider their own historically situated self and bring this into their inquiry. Thus, understanding is context-based.

Hans-Georg Gadamer adopted Heidegger's view of the hermeneutic circle and further developed philosophical hermeneutics. For Gadamer, understanding is hermeneutical and historical. He emphasised that all understandings are linguistic. People engage in a conversation with their own knowledge of that encounter. This pre-knowledge needs to be acknowledged when questioning understanding (Gadamer, 1979).

With the similar aim of studying lived human experiences, the main difference between Husserl, Heidegger and Gadamer's philosophical traditions is that phenomenologists following Husserl's approach strive for objectivity in understanding. They believe in putting presuppositions in abeyance to generate new understandings, and the understandings remain at the level of describing a specific experience of a phenomenon. On the contrary, hermeneutic phenomenology values the pre-existing knowledge of the

researcher, and the researcher's pre-understanding guides the development of meaningful inquiry (Carpenter, 2017).

5.4.3 Gadamer's Phenomenological Hermeneutics as a Choice for the Current Research

The term 'hermeneutics' is derived from the Greek word '*hermeneuein*', which means 'to interpret' or 'to understand' (Crotty, 1998, p. 88). In ancient Greece, this word implied 'translating' something strange that was separated in time or space to make it present and easily understandable (Crotty, 1998). It was originally used to guide scholars in interpreting biblical texts in the 17th century. Apart from its use in interpreting scripture, it was also used to study human actions to bring about understanding (Crotty, 1998).

5.4.3.1 Gadamer's Hermeneutics Circle as Applied to the Study

Gadamer claimed that understanding (interpretation) is inseparable from language and is achieved through the experience of conversation (Gadamer, 1979). He described the interpretive process as a 'hermeneutic circle', whereby the inquirer brings historical awareness (called 'prejudices' by Gadamer) and enters a questioning process with an open mind (Gadamer, 1979, pp. 235, 238). During this process, understanding emerges or 'happens' to the inquirer (Gadamer, 1979, p. 345). 'A fusion of horizons' (explained below) forms as a new understanding is established. In other words, a fusion of the past (the inquirer's presupposition) and the present (the inquirer's interpretation by reflection) develops (Gadamer, 1979, p. 273). Understanding deepens as the fusion of horizons continues.

Gadamer argued that it is impossible for interpreters to bracket their preconceived ideas or leave them in abeyance when they are engaged in the process of interpreting human actions. Gadamer (1979) emphasised prejudices ('fore-understanding') as a condition of understanding:

The important thing is to be aware of one's own bias, so that the text may present itself in all its newness and thus be able to assert its own truth against one's own fore-meanings. (p. 238)

He believed that this was the only way by which interpreters could understand the meanings underneath the appearance of human actions.

To raise a consciousness and develop a global view of issues around post-stroke hospital-to-home transition care, the present researcher conducted an integrative literature review and a systematic review. As a stroke rehabilitation clinician, the researcher's work experience and understanding of challenges faced by stroke dyads after hospital discharge formed 'prejudices'. Maintaining consciousness of her history (pre-understanding) and being open to the meaning of the findings, results and interview excerpts ('text') of the reviewed studies, the researcher's 'own thoughts have also gone into the re-awakening of the meaning of the text' (Gadamer, 1979, p. 350), whereby they formed a 'fusion of horizon' (the pre-understanding fused with new understanding through reflection). As Gadamer (1979) emphasised:

in fact, the horizon of the present is being continually formed, in that we have continually to test all our prejudices ... Hence the horizon of the present cannot be formed without the past. (p. 273)

As the fusion of horizon continued, the hermeneutic circle widened, and the understanding of issues under study deepened.

In Phase 2 of this study, the researcher applied the hermeneutic circle to achieve an in-depth understanding of the care needs and expectations of people with stroke and their family caregivers. Bringing the understanding established from the two reviews, the researcher was engaged in long-term follow-ups with participants through two to three interviews at different time points. Interview transcripts, observation notes and semi-structured diaries completed by participants were used in the interpretation process to generate deeper insights into participants' transition care experiences. Likewise, applied to the focus groups with health professionals, the hermeneutic circle facilitates an enriched understanding of the perceptions of health professionals on care needs during the hospital-to-home transition.

5.4.3.2 The Application of Gadamer's Hermeneutic to the Study

5.4.3.2.1 The Researcher's Prejudices and Enhancing Understanding

In Gadamer's philosophical hermeneutics, prejudices are 'a judgment that is rendered before all the elements that determine a situation have been finally examined' (Gadamer, 1979, p. 240). Prejudices are shaped by traditions and play a critical role in analysing human actions because they are 'regularly greeted with current thought' and influence individual judgements (Crotty, 1998, p. 103).

As a stroke rehabilitation clinician, the researcher oversees the discharge and admission of stroke patients. It has been observed that, despite the existence of various community rehabilitation and social care services, consumer feedback indicates a breakdown of the continuity of care after people with stroke are discharged from hospital. Stroke dyads often report uncertainty about what happens and who to contact after discharge in spite of the provision of information and stroke nurse education prior to discharge. An eagerness to establish a better understanding of the issues and a passion for service quality improvement motivated the present researcher to embark on this study. This motivation served as the researcher's prejudice to initiate the inquiry. Her clinical experience and existing knowledge helped her to identify the research problem and address problems through the development of in-depth knowledge of the experience and care needs of stroke dyads.

5.4.3.2.2 Conversation/Dialogue and Relationship-Building With Participants

One of the pivotal aspects of Gadamer's hermeneutic approach is that 'language is the middle ground in which understanding and agreement concerning the object takes place between two people' (Gadamer, 1979, pp. 346–347). In Gadamer's philosophy, the inquirer and the participants do not talk at 'cross purposes' when conducting a conversation (Gadamer, 1979, p. 330). Talking at cross purposes causes misunderstanding between the inquirer and the participants. Therefore, the use of questions and answers is essential in a conversation/dialogue: 'Questions always bring out the undetermined possibilities of a thing' (Gadamer, 1979, p. 338). Therefore, when a conversation begins, neither party knows the

conversation's trajectory or where it will finally land; understanding or a failure to understand is unplanned (Gadamer, 1979).

To achieve shared understanding, Gadamer also suggested that, to conduct a conversation, 'it requires that one does not try to out-argue the other person, but that one really considers the weight of the other's opinion ... [It is] the art of thinking that is able to strengthen what is said by referring to the object' (Gadamer, 1979, pp. 330–331).

Gadamer considered two issues that need to be addressed during interviews and focus groups – 1) people involved in conversations are not talking at cross purposes and 2) they remain open-minded to each other's opinion. To ensure that the researcher and participants talk at mutual purposes, an interview guide, informed by findings of the literature and systematic reviews and the existing clinical knowledge of the researcher, was developed. Semi-structured questions in the interview guide facilitated meaningful conversation between the researcher and participants, enabled participants to elaborate on their experiences and provided the researcher with the opportunity to ask probing questions to explore participants' experiences more deeply. In this way, the researcher's additional questions engaged with the participants' descriptions, resulting in open-ended meanings rather than an expected gain for either the participant or the researcher.

Furthermore, during the process of interviews and focus groups, remaining openminded, the researcher developed mutual trust and respect with participants. This allowed participants to freely express their thoughts and elaborate on their experiences. The establishment of a reciprocal relationship with participants facilitated the generation of a shared understanding. As Gadamer (1979) stressed, 'dialectic ... is the art of thinking that is able to strengthen what is said by referring to the object' (p. 331).

5.4.3.2.3 Conditions to Reach Fusion of Horizons in the Present Study

Gadamer claimed that understanding is always the fusion of horizon between the historical horizon and the horizon of the present (Gadamer, 1979). Gadamer encouraged the inquirer to see beyond the horizon but to be limited by it. Thus, the inquirer must stand in the 'tradition (historical horizon)' in which human actions are rooted and 'bring together the

horizon of present (the human actions) and the horizon of the past (the tradition)' (Gadamer, 1979). The fusion of horizons is achieved through the use of language in dialogue (Gadamer, 1979).

In Gadamer's hermeneutic circle, the inquirer's prejudices (the past) constantly fuse with new knowledge (the present), obtained through an interpretation of reviewed studies interviews with participants and the inquirer's reflection. When the pre-knowledge of the inquirer agrees with the interpretation of participants, the limited, historical prejudices die away, and a genuine understanding of the studied phenomenon emerges (Gadamer, 1979). The fusion of horizons bridges the gap between the familiar and the unfamiliar: it enables the inquirer to reach a deeper understanding of either the perspectives or participants in the conversation, or both.

In the current study, the researcher stood in the 'tradition' of the socio-politicaleconomic context of transition care in Australia. As a clinician, and through work experience in the stroke rehabilitation field and reflection on clinical practice, she developed knowledge about the Australian healthcare system, health professionals and people with stroke. With this clinical knowledge and the findings from the integrated literature and systematic reviews, the researcher developed further insights into three-tiered factors that affected the hospitalto-home transition experience of stroke dyads. These system factors, health professional factors and individual factors of stroke dyads helped the researcher reach a historical consciousness. The established historical consciousness served as prejudices for Phase 2 of the study.

5.4.3.2.4 A Holistic Understanding of Care Needs and Expectations

Gadamer believed that understanding is constantly moving from the whole to the part and back to the whole (Gadamer, 1979). The inquirer attempts to understand 'the whole through grasping its parts and comprehending the meaning of the parts divining the whole' through the hermeneutic circle (Crotty, 1998, p. 92). Gadamer suggested that 'the harmony of all the details with the whole is the criterion of correct understanding' (Gadamer, 1979, p. 259). Understanding is formed in 'ever-widening circles', created by 'constant movement and

extension', which allow the generation of new knowledge (Crotty, 1998, pp. 102–104). Hence, new knowledge forms the prejudices for another circle of interpretation.

When applied to this study, Gadamer's concepts highlight a holistic understanding of care needs and expectations. Stroke brings enormous challenges to people with stroke and their families, including physical deficits, mental and cognitive impairment, and social and financial difficulties. When planning for hospital discharge, their physical, psychosocial and spiritual care needs and relevant services need to be considered. Consideration should be based on an assessment of their individual circumstances.

5.4.3.3 Reasons for Choosing This Methodology

The aim of this study was to explore the care needs and expectations in hospital-tohome transition care for stroke dyads in an Australian social context. Therefore, an interpretive approach, underpinned by Gadamer's hermeneutics, was considered the most appropriate methodology to address this study's aim and achieve its objectives.

In Gadamer's hermeneutic circle, understanding is achieved through dialogue between the inquirer and the participants, using a question-and-answer technique. In this process, new understanding is generated through constant movement from the whole (historical context) to the part (human actions) and back to the whole, and by the fusion of the inquirer's prejudices (the past) and newly obtained knowledge (the present) through reflection. As the circle widens, the understanding deepens (Carpenter, 2017). Different from natural science, which focuses on parts that comprise the natural world, this attention to the whole is unique to human science: 'To understand a text bearing upon human affairs or a culture that guides human lives, one needs to be able to move dialectically between part and whole, in the mode of the hermeneutic circle' (Crotty, 1998, p. 92).

In this study, the human interest focused on how the needs and expectations of stroke dyads in Australia regarding transition care and rehabilitation services may be better met, within the constraints of the current clinical pathway and model of care, after discharge from hospital to home and the community. New knowledge was generated by understanding the underlying meaning of stroke dyads' perceived enablers and inhibitors during hospital-to-

home transition care and of health professionals' perspectives of approaches to identify and meet the transition care and rehabilitation needs of the consumers. This knowledge enables the development of an evidence-informed hospital-to-home transition care model and pathways to enhance consumer-directed transition care and rehabilitation services.

5.4.3.4 Limitations of Gadamer's Hermeneutics

Gadamer was criticised for his embrace of tradition (Crotty, 1998). Gadamer's hermeneutics requires the inquirer to bring in their historical awareness (tradition) to the process of interpretation. Gadamer claimed that, in the process of fusion of horizons, 'those prejudices that are of a particular and limited nature die away' (Gadamer, 1979, p. 266). In other words, when the researcher brings their own presupposition of stroke transition care, some prejudices may be 'false ones by which we misunderstand' (Gadamer, 1979, p. 266). These potentially false prejudices may affect the understanding.

Gadamer claimed that 'hermeneutics surpasses methods, because it reminds us of our practical life experience and only through this awareness might the innate power of dogmatism be overcome' (Gadamer, 1976, p. 12–13, as cited in Cuff, 2019). Gadamer concentrated on the phenomenon of understanding as an event, not a method for 'objective' interpretation (Gadamer, 1979). Therefore, Gadamer did not develop concrete methods to apply his philosophy to research (Cuff, 2019). However, in critiquing Gadamer's hermeneutics, Cuff (2019) revealed that Gadamer believed methods that are in line with the principles of his philosophic hermeneutics would help interpreters to achieve an understanding of a phenomenon of interest. The researcher of the present study planned to overcome the limitations through careful inspection and reflection on her own knowledge of transition care, especially when her horizon was significantly different from that of the participants. In addition, the interview guide served as a means to enable the development of meanings and understanding from the participants' own words.

5.5 Ethical Consideration

This study was approved by the Southern Adelaide Clinical Human Research Ethics Committee (Appendix 14) prior to commencement. Based on the requirements from the committee, permissions were obtained from the head/director(s) of division/services where recruitment took place. The participant information and consent form (PICF; Appendix 15) and participant response sheet for people with stroke, their caregivers and health professionals were developed for distribution.

The PICF included a brief introduction of this study and its objectives; the participant selection criteria and justification; the contact information of the researcher, including the researcher's non-research professional role; details on the participants' rights, including voluntary participation, the right to refuse to participate or answer particular questions, freedom to withdraw from the study without prejudice, and freedom to choose the number of interviews/focus groups to attend; what the participation entailed; and how their confidentiality and anonymity would be maintained. Written informed consent was obtained from each participant prior to the interview or focus group. The consent form clarified that interviews and focus groups would be audio-recorded and, with their permission and on the condition that their identities would remain confidential, that records may be made available to other researchers.

Throughout the research, the ethical guidelines of the National Health and Medical Research Council (2018) were strictly complied with. Participants were informed that, if emotional distress occurred due to the questions asked during the interviews, they could be referred to a psychologist if deemed necessary with their consent.

Audio records and interview transcripts were maintained securely on the cloud service OneDrive at the Flinders University College of Nursing and Health Sciences. All collected data were de-identified. The transcripts of interviews and focus groups were coded so that participants' responses were de-identified. Data were analysed collectively. To

further protect their identity, participants were assured that pseudonyms would be used in this thesis and all relevant publications.

5.6 Methods

This section details the study's settings, participant recruitment, data collection and analysis. The rigour of the study is discussed at the end.

5.6.1 Setting and Participant Recruitment

The study was conducted at a stroke rehabilitation ward and at the ambulatory rehabilitation service in a tertiary metropolitan public hospital in Adelaide. Stroke dyads were recruited through the stroke rehabilitation ward. Health professional participants were recruited from the stroke rehabilitation ward and the ambulatory rehabilitation service of the hospital.

5.6.1.1 Criteria for Inclusion

Setting selection criteria facilitates the identification of participants who have experienced the studied phenomenon and are able to share rich information in the area of study (Polit & Beck, 2017). Based on the aim of this study, and informed by the literature review and systematic review, inclusion criteria were developed and are listed in Table 5.1.

Table 5.1

Inclusion criteria	Rationale
Criteria for p	eople with stroke
Clinical diagnosis of stroke.	Stroke is the focus of this study.
Discharged from hospital to home, not to	Hospital-to-home transition care is the scope
a facility or supported accommodation.	of this study.
Discharged with further rehabilitation and	This group of people with stroke is the focus of
community care services.	the study.

Inclusion Criteria

Inclusion criteria	Rationale						
MoCA: scored 20 or above.	Including people with stroke with mild						
FIM: scored 4 or above in each item	cognitive or speech impairment helps the						
except Expression and	researchers to understand the care needs						
Comprehension, which need to be	of this group.						
scored 5 or above.							
Have sufficient language and cognition to	Informed consent is important for participation.						
consent.							
Able to communicate verbally in English	English is the language used in this study.						
before stroke.							
Criteria for informal caregivers							
Principle informal caregivers who have	Only informal caregivers who have daily						
daily contact with people with stroke.	contact with people with stroke are able to						
	provide rich information.						
Aged 18 or above.	A person aged 18 or above is considered an						
	adult (ABS, 2013). Adults are the focus of						
	the study.						
Having no cognitive impairment or	These impairments make an interview						
aphasia.	impossible.						
Able to communicate in English.	English is the main language used in this						
	study.						
Criteria for he	ealth professionals						
Staff members of stroke or rehabilitation	Only those involved in stroke care are able to						
care team, not students.	share thoughts on stroke transition care.						
GPs	Inputs from GPs will enrich the understanding						
	of stroke transition care.						

Inclusion criteria	Rationale						
Aged 18 and above.	A person aged 18 or above is considered an						
	adult (ABS, 2013).						
Able to communicate in English.	English is the main language used in this						
	study.						

Note. MoCa = Montreal Cognitive Assessment; FIM = Functional Independence Measure; ABS = Australian Bureau of Statistics; GP = general practitioner.

5.6.1.2 Identification and Recruitment of Potential Participants

For stroke dyad participants, potential participants were identified by ward clinicians based on the inclusion criteria. The researcher, who was the nurse unit manager of the ward but did not have a direct care relationship with the patients, approached potential participants with the recruitment poster (Appendix 16) to ascertain their intention to participate. After obtaining their intention, a research assistant (hired by the researcher and external to the ward) made an appointment and met with potential participants. This was to ensure a separation from clinical practice so that potential participants did not feel coerced to participate in the study. For people with stroke who had a caregiver, the appointment was made to meet with both. During the meeting, the research assistant provided an information pack including the PICF (Appendix 15), explained the details of the research project (including their rights to withdraw) and answered questions.

For stroke health professionals, the flyer and information pack (containing a PICF and participant response sheet) was made available in staff tea rooms and offices. Potential participants expressed their intention via email, text or phone call to the researcher. For general practitioners (GPs), information packs were posted to all GP clinics in the Southern Adelaide Region. Potential participants expressed their intention via email, text or phone call to the researcher. Upon receiving their expression of interest, the researcher screened for eligibility of participation. The researcher remained available if potential participants required further information. In this case, the researcher made appointments with the potential participants at a venue of the potential participant's choice to explain relevant details.

5.6.1.3 Consent

For people with stroke, upon receiving instruction from the researcher, the research assistant obtained consent after a detailed explanation of the research project and an assessment of eligibility for inclusion based on cognitive assessment.

The cognitive assessment was conducted using the Montreal Cognitive Assessment (MoCA) tool (Nasreddine et al., 2005). MoCA is a widely used cognitive assessment tool to screen for mild cognitive impairment (Appendix 17). MoCA scoring is as follows: (1) 26 and above = normal cognition; (2) 25–20 = mild cognitive impairment; (3) below 20 = moderate to severe cognitive impairment (Webb et al. 2014). Participants meeting the inclusion criteria then consented to participate in the study.

In this study, speech difficulty was determined by the Functional Independence Measure (FIM; Uniform Data System for Medical Rehabilitation, 2012). The FIM is an 18item measurement tool that explores an individual's physical, psychological and social function (Appendix 18).

Based on the ward clinicians' recommendation, a simplified PICF (Appendix 19) was developed and used for potential participants who required simplified text and pictures to aid understanding. The non-simplified PICF was used for all other potential participants.

For people with stroke who had caregivers, consent was obtained from both. This was to ensure that the person with stroke was not recruited against their caregiver's wishes. For caregivers, the research assistant obtained consent upon receiving confirmation of their intent to participate. For health professionals, the researcher obtained consent upon receiving confirmation of their intention to participate.

5.6.1.4 Sample Size

According to Creswell (2013), the sample size for phenomenological studies is 5–25. The sample size usually depends on the scope of the study questions and the sensitivity of the studied phenomenon, and the initial sample size is only an estimate rather than a fixed number (Carpenter, 2017). For qualitative studies, the richness of information supersedes the number of participants (Carpenter, 2017). It was planned for this study to recruit 15

stroke dyads (dyads comprising a person with stroke and their caregiver) for interviews. The recruitment process is illustrated in Figure 5.1. In total, 30 participants were recruited to this study: 12 dyads and six caregivers.

Figure 5.1

Recruitment Flowchart



The sample size for focus groups is dependent on the research topic and the social context of participants. Focus groups normally consist of 5–15 participants. Ideally, focus groups should include members that share similar characteristics—for example, members from the same professional group (Davidson et al., 2017). In this study, focus groups comprised health professionals from different disciplines involved in stroke care. This was because of the interprofessional approach to acute stroke and post-stroke rehabilitation. Focus groups that include multidisciplinary members help to develop a coherent as well as diverse insight into the care needs of stroke dyads during the hospital-to-home transition.

A few studies selected for the literature review (Chapter 2) included allied health team members and community service providers. These health professionals provided valuable information and their perspectives on stroke hospital-to-home transition care. However, only one study included a medical officer. In this study, health professionals included stroke care team members such as medical officers, nurses, physiotherapists, occupational therapists, speech pathologists, dietitians and social workers. Efforts were made to include medical officers due to the scarcity of data from them in the reviewed studies. Additionally, community service providers, especially GPs as the primary health practitioner in the community, have more encounters with stroke dyads after discharge. Their input may enhance the understanding of challenges regarding the continuity of transition care.

Purposive sampling was the main strategy for recruiting participants. This sampling strategy facilitates the selection of participants that benefit the study the most (Polit & Beck, 2017). Under the condition that inclusion criteria were met, snowball sampling was also used to complement the purposive sampling strategy, especially with the recruitment of health professionals. With the combination of these strategies, the planned sample size was achieved within the time constraints of the PhD program.

5.6.2 Data Collection

According to Gadamer's (1979) philosophical hermeneutics, language and dialogue are essential components for developing understanding. Thus, in-depth interviews, using a semi-structured interview guide, were used to collect data from stroke dyads.

In-depth interviewing is a well-developed method of data collection that enables researchers and participants to work together to develop an understanding of the studied phenomenon (Serry & Liamputtong, 2017). Semi-structured interviews were used as the interview framework. The interview guide (Appendix 20) provided directions for the interview but allowed some freedom for participants to express their thoughts. Compared to an informal conversational interview, where spontaneity is encouraged, and a standardised open-ended interview, which pursues asking similar questions, the semi-structured interview provided a balance that allowed some space for spontaneous thoughts and expressions in a guided condition (Serry & Liamputtong, 2017).

Face-to-face interviews were the main interviewing approach. For participants who were unable to attend face-to-face interviews, especially due to COVID-19 restrictions, telephone interviews were conducted. The limitation of telephone interviews is the lack of visual contact between interviewer and interviewee. However, collected data can still add value to a face-to-face interview data pool. Bakas et al. (2002) used telephone interviews to establish insights into the needs and concerns of stroke caregivers in the first 6 months following discharge.

People with stroke and their caregivers were interviewed separately. Family dynamics and the dyadic relationship may have become barriers for the researcher to obtain truthful information. However, if they requested to be interviewed together, their request was respected. The interview location was the participant's choice. The length of each interview was between 30 and 45 minutes. The researcher's observations and field notes were maintained to supplement interviews. The Research Assistant was only involved in the consenting with stroke dyads during the recruitment process.

A structured diary with instructions (Appendix 21) was developed for participants to take home to record activities, experiences and thoughts during the follow-up period. The diary contains a list of questions about participants' experiences/thoughts on significant events relating to their coping with post-stroke limitations, with caregiving activities and daily living activities, and how they manage other health conditions. Participants were also encouraged to record the challenges they experience when using care services. Diary entries were voluntary. The content was reviewed at the follow-up interviews. This information complemented the interview data and generated in-depth knowledge.

Transition is a continuum process, and the care needs of stroke dyads change over time. Informed by the findings of the systematic review, participants in this study were followed up from 1 month up to 6 months following discharge. Interviews were conducted at three time points—within 5 days prior to discharge and at 2–3 months and 5–6 months following discharge. The intervals allowed them to settle and progress. Participants attended 2–3 interviews at their choice.

The collection of data from health professionals was through focus groups as a first preference and one-on-one interviews if they were unable to attend focus groups. Focus groups, as a data collection method, have been widely used in the healthcare field. It is used to explore experiences, knowledge and beliefs. Focus groups enable synergy within the group and provide a collective perspective on the studied topic (Davidson et al., 2017). It can be challenging to control the dynamics of the group and monitor verbal and non-verbal responses due to the number of participants. However, careful planning and appraisal of risks in advance can minimise these challenges (Davidson et al., 2017).

In this study, it was planned to conduct 3–5 focus groups with 15–25 health professionals. The number of focus groups depends on data saturation (i.e., when no new data is being produced from subsequent groups: Davidson et al., 2017). The inclusion of multidisciplinary team members in the same focus groups may not cause heterogeneity in power relationships and social classes because all team members are working with the same goal: to ensure quality care to people with stroke. Each discipline works independently as well as collaboratively with others. So, there may not be heterogeneity in power relationships or social class within the group. Multidisciplinary participation enhances the richness of perceptions.

There have been debates on phenomenological focus groups. C. Webb and Kevern (2001) criticised the application of focus groups in phenomenological studies. They suggested that focus groups and phenomenology are methodologically incompatible in that phenomenological research pursues the essence of a phenomenon, which is achieved through an individual's description of their lived experiences in an uncontaminated way, but focus groups, involving interactions between group members, disintegrate the totality of an individual's lived experience that forms the foundation of phenomenological studies (B. Webb, 2002; C. Webb & Kevern, 2001).

Despite this criticism, focus groups have been used extensively as a method to collect and analyse data in phenomenological studies (Holm & Dreyer, 2018; Love et al., 2020; Robertson et al., 2019). In their research methodology discussion paper, Bradbury-

Jones et al. (2009) critiqued Webb's views from their own perspectives as researchers and that of other scholars (B. Webb, 2002). They argued that focus groups support the concept of collaboration and dialogue in phenomenological emphasis. The individual's perspectives can be preserved by allowing the individual to share their narratives with limited interruptions first, followed by other members of the group adding their insights and views to stimulate reflection and enrich the understanding of a phenomenon (Bradbury-Jones et al., 2009). They concluded that, instead of causing methodological tension, small and well-mediated focus groups complement phenomenological endeavours and that the collaboration within groups enhances the development of new insights (Bradbury-Jones et al., 2009; Nielsen et al., 2020).

A total of four focus groups for 11 stroke clinicians were conducted, with three participants in each group on average. Focus groups were facilitated by the researcher in the meeting room at the participants' workplace and lasted for 60–90 minutes. Each participant was given the opportunity to share their individual views, and this was followed by reflection and group discussions.

Other stroke clinicians (n = 18) had in-depth interviews due to their inability to attend focus groups. Each interview lasted 30–45 minutes. GP participants (n = 2) were interviewed individually in their consulting rooms.

5.6.3 Data Analysis

Data analysis is aimed at organising data and producing meaning from data (Polit & Beck, 2017). Data analysis in this study was concurrent with data collection. It was a cyclical process that allowed analysed data to inform and guide upcoming data collection so as to address issues that arose during the initial data analysis (Liamputtong & Serry, 2017).

5.6.3.1 Data Analysis Model

A three-step interactive model of data analysis was used in this study. This model included data reduction, data display and conclusion (Liamputtong & Serry, 2017).

As a crucial first step of data analysis, data reduction paves the way for deeper analysis. Audio-recorded interviews and focus groups were listened to repeatedly. Raw data

from interviews and focus groups were transcribed verbatim. The preliminary analytical phase commenced at this stage, with the transcribed data transforming into initial codes, summaries and introductory themes. These initial analytic activities can be modified later as analysis further grows (Liamputtong & Serry, 2017). The transcripts were coded line by line to prepare for analytical coding—namely, focused coding (synthesis of codes), interpretive coding (categorising codes) and selective coding (identifying core themes)—as analysis deepened. This coding process allows data to be reduced and facilitates the formulation of propositions. (Liamputtong & Serry, 2017).

As the analysis continued, data display occurred because data were transformed into codes, mind maps or graphs. Conclusions were drawn from the transformed data as patterns or themes emerged. At this stage, the researcher remained open-minded and continued to draw on the analysis of upcoming collected data until enough evidence arose to verify the conclusions (Liamputtong & Serry, 2017). In the process of data analysis, a fusion of horizon occurred between the prejudice of the researcher and participants, which is in line with Gadamer's hermeneutical circle.

5.6.3.2 Data Analysis Based on Gadamerian Principles

Gadamerian philosophical hermeneutics guided a systematic approach to interpretation but did not offer specific data analysis methods (Gadamer, 1979). Inspired by a group of researchers who explored data analysis methods based on Gadamerian principles (Alsaigh & Coyne, 2019; Fleming et al., 2003), this study adapted their methods, involving five interconnected and cyclic steps to enhance interpretation and understanding. The five steps were 1) immersion in raw data; 2) understanding through line-by-line coding; 3) abstraction to identify subthemes by grouping open codes into clusters; 4) synthesis and theme development, involving a continuous moving back and forth between earlier analyses, text and literature, and across datasets to identify similarities and differences. In the process, themes were informed by, and in turn, enriched the researcher's pre-knowledge; and 5) the illumination and illustration of phenomena resulting in a reconstruction of participants' stories in their own words through linking themes and subthemes with the literature (Alsaigh &

Coyne, 2019; Fleming et al., 2003). The data analysis was closely linked to the aim and objectives of the study during the whole process. The qualitative data analysis software NVivo 12 was used to assist in coding, organising data and developing mind maps.

5.6.4 Rigour

The rigour of a qualitative study concerns the trustworthiness of the quality of its findings. The criteria established by Lincoln and Guba (1985) were used to ensure the rigour of this study. Credibility (confidence in the truth of the data) was enhanced by audio-recording and verbatim transcription of interviews and focus groups; cross-checking participants' statements and opinions during and after the study; debriefing and discussion with other research team members; and reflexivity, involving self-reflection and introspection about personal bias and every decision made (Morse, 2018; Polit & Beck, 2017).

Strategies—including maintaining a decision tree for future reference, using participants' own words to support themes and extensive discussion among the research team—were used to ensure the dependability (stability of data over time) and confirmability (originality of findings derived from participants) of data (Morse, 2018).

Qualitative studies are context-based, so findings may not be generalisable. However, they may be transferred to a similar social context. Transferability of the findings can be enhanced by citation of existing confirming research evidence from other studies to verify the interpretations and conclusions. When producing a report, the original excerpts of participants were used to enhance the understanding of readers (Morse, 2018). A reflexive journal was maintained throughout the research process to ensure the overall trustworthiness of this study. The researcher self-consciously critiqued, appraised, and evaluated how her subjectivity and context influence the research process.

5.7 Summary

The chapter has discussed Gadamer's philosophical hermeneutics as underpinning this study. Ethical considerations, methodology and methods were also described. The next

chapter presents Part 1 of findings of the qualitative study on the perspectives of stroke dyads.

Chapter 6: Findings—Part 1: Outline of Findings, and Findings From the First Interviews With Stroke Dyads

6.1 Introduction

The previous chapter presented the constructivist paradigm used in the project; Gadamer's philosophical hermeneutics, which informed the methodology of this study; ethical considerations; and the methods used in the study.

This chapter, along with Chapters 7–9, presents the findings of the qualitative study of stroke dyads (Chapters 6–8) and health professionals (Chapter 9). This chapter outlines the characteristics of the participants and the themes and subthemes generated from interviews and focus groups with stroke dyads and health professionals. It presents the findings from interviews with stroke dyads within 5 days before discharge or within 2 weeks after discharge. Chapters 7 and 8 report the findings from follow-up interviews and structured diaries at 3 and 6 months after discharge for stroke dyads. Chapter 9 presents the findings from focus groups and interviews with stroke care health professionals.

6.2 Characteristics of Participants

In total, 35 stroke dyads consented to participate in the study. Interviews were conducted with 12 people with stroke and 18 caregivers; five people with stroke were unable to participate in the interviews due to cognitive and/or speech impairments. Equal numbers of male (n = 8) and female (n = 9) stroke participants were included in this study. Their ages ranged from 55 to 96 years (M = 74.9). Of the 17 people with stroke, 14 suffered from ischaemic stroke, and 16 had comorbidities or chronic conditions. Most of the caregivers were women (n = 13), and their ages ranged from 48 to 89 years (M = 61.5). Caregivers were spouses or ex-spouses (n = 11), siblings (n = 1) or children (n = 7) of the person with stroke. Table 6.1 outlines the characteristics of the stroke and caregiver participants.

In total, 31 health professionals participated in this study: rehabilitation allied health staff (n = 16), registered nurses (n = 6), service managers (n = 4), doctors (n = 3) and GPs (n = 2). Of the 31 health professional participants, 27 were women, and four were men, with years in profession ranging from 5 to 39 years (M = 13.8). The demographic information of these participants is presented in Table 6.2.

6.3 Outline of Themes and Subthemes

Seven major themes and 16 subthemes (Table 6.3) were identified in interviews with stroke dyads before hospital discharge and at 3 and 6 months after discharge from hospital. These themes represented the shared experiences of people with stroke and their caregivers during the first 6 months after discharge.

The three major themes identified from before hospital discharge revealed that similarities and differences existed between people with stroke and their caregivers in terms of their preparedness to manage post-discharge care. Stroke dyads demonstrated resilience, which was built on the same foundations, such as their positive thoughts and having the support of family and health professionals. However, they also showed psychological stress that was triggered by different factors: for example, loss of function in people with stroke and a lack of confidence from caregivers about providing home care. The findings also indicated stroke dyads' inconsistent involvement in discharge planning, which contributed to the differences in perceived certainty about post-discharge care between people with stroke and their caregivers.

Table 6.1

Characteristics of Stroke and Caregiver Participants

	Stroke patient participants									
Code	Gender	Age	Type of	Rehabilitati	FIM	МоСА	Comorbidity	Number of	Number of	
	(M = 8;	(years;	stroke	on length of	score***	score	/chronic	interviews	diary entries	
	F = 9)	<i>M</i> = 74.9)		stay (days:			conditions			
				M = 28)						
SS01*	М	75	Left ischaemic	23	4/4	_	Ischaemic	0	0	
							heart			
							disease,			
							knee surgery			
SS02	F	96	Left ischaemic	16	6/6	20/30	Hypertension	3	7	
							, atrial			
							fibrillation			
SS03	М	64	Bilateral	14	7/7	30/30	Ischaemic	2	0	
			ischaemic				heart			
							disease,			

Stroke patient participants									
Code	Gender	Age	Type of	Rehabilitati	FIM	MoCA	Comorbidity	Number of	Number of
	(M = 8;	(years;	stroke	on length of	score***	score	/chronic	interviews	diary entries
	F = 9)	<i>M</i> = 74.9)		stay (days:			conditions		
				<i>M</i> = 28)					
							diabetes,		
							cancer		
SS04	М	65	Left ischaemic	13	6/6	20/30	Renal failure	3	0
							on dialysis		
SS05*	М	55	Bilateral	55	1/1	—	Atrial	0	0
			ischaemic				fibrillation		
SS06*	F	66	Left	66	3/1	—	Stroke,	0	0
			haemorrhagic				hypertension		
							, Type 2		
							diabetes		
SS07	М	97	Left ischaemic	14	7/6	24/30	Hypertension	3	0
							, stroke, falls,		

				Stroke patie	nt participa	nts			
Code	Gender	Age	Type of	Rehabilitati	FIM	MoCA	Comorbidity	Number of	Number of
	(M = 8;	(years;	stroke	on length of	score***	score	/chronic	interviews	diary entries
	F = 9)	<i>M</i> = 74.9)		stay (days:			conditions		
				M = 28)					
							Ischaemic		
							heart		
							disease		
SS08*	М	68	Left ischaemic	14	1/1	-	Hypertension	0	0
							, asthma		
SS09	F	61	Left ischaemic	29	7/7	25/30	Atrial	3	4
							fibrillation,		
							rheumatoid		
							arthritis		
SS10	F	77	Left ischaemic	48	7/7	24/28	Atrial	3	5
							fibrillation,		

Stroke patient participants										
Code	Gender	Age	Type of	Rehabilitati	FIM	MoCA	Comorbidity	Number of	Number of	
	(M = 8;	(years;	stroke	on length of	score***	score	/chronic	interviews	diary entries	
	F = 9)	<i>M</i> = 74.9)		stay (days:			conditions			
				<i>M</i> = 28)						
							stroke,			
							cancer			
SS11	М	82	Right	11	7/7	21/30	Nil	3	5	
			ischaemic							
SS12	F	72	Left ischaemic	10	7/7	26/30	Cancer	3	0	
SS13	F	83	Left ischaemic	11	7/7	24/30	Hypertension	3	4	
							, depression			
SS14*	F	91	Right	62	4/4	_	Falls,	0	0	
			ischaemic				peripheral			
							vascular			
							disease			

	Stroke patient participants									
Code	Gender	Age	Type of	Rehabilitati	FIM	МоСА	Comorbidity	Number of	Number of	
	(M = 8;	(years;	stroke	on length of	score***	score	/chronic	interviews	diary entries	
	F = 9)	<i>M</i> = 74.9)		stay (days:			conditions			
				<i>M</i> = 28)						
SS15	F	71	Left	30	7/7	30/30	Hypertension	3	0	
			haemorrhagic				, cancer,			
							ischaemic			
							heart			
							disease			
SS16	F	81	Left ischaemic	19	6/6	25/30	Hypertension	3	0	
							, atrial			
							fibrillation			
SS17	М	70	Bilateral	44	7/7	24/30	Hypertension	3	0	
			ischaemic				, cancer			
Stroke caregiver participants										
-------------------------------	-----------------	--------------------------	--------------	-----------	------------	---------------				
Code	Gender	Age	Relationship	Co-living	Number of	Number of				
	(M = 5; F = 13)	(years; <i>M</i> = 61.5)			interviews	diary entries				
SC01	F	72	Spouse	Yes	3	0				
SC02	F	65	Daughter	Yes	3	0				
SC03	F	62	Spouse	Yes	3	8				
SC04	F	61	Spouse	Yes	2	0				
SC05	F	48	Spouse	Yes	3	0				
SC06	М	68	Spouse	Yes	3	0				
SC07	F	61	Daughter	No	3	0				
SC08	F	64	Spouse	Yes	3	1				
SC09	М	63	Brother	Yes	3	4				
SC10	М	89	Spouse	Yes	3	0				
SC11	F	79	Spouse	Yes	3	6				
SC12	М	75	Ex-spouse	Yes	2	0				
SC13	F	57	Daughter	No	3	0				
SC14**	F	62	Daughter	Yes	2	0				

Stroke caregiver participants						
Code	Gender	Age	Relationship	Co-living	Number of	Number of
	(M = 5; F = 13)	(years; <i>M</i> = 61.5)		interviews	diary entries
SC15**	F	65	Daughter	No	2	0
SC16	Μ	68	Spouse	Yes	3	0
SC17	F	48	Daughter	No	3	0
SC18	F	63	Spouse	Yes	3	0

Note. M = male; F = female; SS = stroke patient; SC = stroke caregiver. MoCA = Montreal Cognitive Assessment (Nasreddine et al., 2005); FIM = Functional Independence Measure (Expression/Comprehension; Uniform Data System for Medical Rehabilitation, 2012).

* Stroke patients who were not interviewed. **Caregiver caring for SS14. SS14 had two caregivers with one co-living providing domestic care support and the other coordinating services. *** FIM Score Expression/Comprehension.

Table 6.2

Characteristics of Health Professional Participants

Participant	Code	Qualification	Years in profession	Gender	Age range
group			(<i>M</i> = 13.8)		(years)
Registered	NS03	Bachelor	14	Female	31–40
nurse	NS04	Bachelor	12	Female	31–40
Allied health	AH01	Bachelor	7	Female	21–30
staff*	AH02	Bachelor	6	Female	21–30
	AH03	Bachelor	15	Male	31–40
	AH04	Bachelor	6	Female	21–30
	AH05	Bachelor	5	Female	21–30
	AH06	Master	15	Male	31–40
	AH07	Bachelor	25	Female	41–50
	AH08	Bachelor	20	Female	41–50
	AH09	Bachelor	22	Female	41–50
	DOC01	Bachelor	14	Female	31–40
Medical officer	DOC02	Bachelor	5	Female	31–40

Participant	Code	Qualification	Years in profession	Gender	Age range
group			(<i>M</i> = 13.8)		(years)
	DOC03	Bachelor	9	Female	31–40
Service	MG01	Bachelor	16	Female	31–40
manager	MG02	Bachelor	20	Female	41–50
	MG03	Bachelor	15	Female	31–40
	MG04	Bachelor	28	Female	41–50
General	GP01	Bachelor	39	Male	61–70
practitioners	GP02	Bachelor	27	Male	41–50
FG01**	AH10	Bachelor	12	Female	31–40
	AH11	Bachelor	8	Female	31–40
	AH12	Bachelor	15	Female	31–40
	AH13	Bachelor	6	Female	31–40
FG02**	AH14	Bachelor	5	Female	21–30
	AH15	Bachelor	5	Female	31–40
	AH16	Bachelor	9	Female	31–40
FG03	NS01	Bachelor	20	Female	61–70

Participant	Code	Qualification	Years in profession	Gender	Age range
group			(<i>M</i> = 13.8)		(years)
	NS02	Bachelor	5	Female	21–30
FG04	NS05	Bachelor	8	Female	31–40
	NS06	Bachelor	15	Female	31–40

Note. NS = registered nurse; AH = allied health; DOC = medical officer; MG = manager; GP = general practitioner; FG = focus group.

*Allied Health Staff included in interviews: Occupational therapist n=3; Physiotherapist n=2; Speech Pathologist n=2; Social Worker n=2;

**Allied Health Staff included in focus groups: Physiotherapist n=3; Speech Pathologist n=2; Pharmacist n=1; Dietitian n=1.

Major themes identified at the 3- and 6-month follow-ups after discharge indicated that people with stroke experienced setbacks in functional recovery that triggered their psychological stress, which also caused psychological stress in their caregivers. Stroke dyads continued to demonstrate resilience and coping strategies that enabled them to reconcile with their pre-stroke lives. Additionally, stroke dyads gradually developed selfmanagement strategies through collaboration between dyadic members. The dyadic effects in dealing with challenges enabled them to transform and adapt to their post-stroke lives. Moreover, stroke dyads acknowledged the crucial role of family and friends during the hospital-to-home transition, particularly the substantial support that people with stroke received from their caregivers. Mutual support between dyadic members enhanced their transition experience.

Stroke dyads also recognised professional services as essential in supporting their transitioning home. They considered that professional support focusing on both dyadic members enabled functional improvement and psychosocial recovery during the transition. Furthermore, stroke dyads reflected on the implications of engaging both dyad members, especially caregivers, in discharge planning before returning home. They also perceived that continuity of care could be enhanced through flexible service delivery in the community to better support their transition.

Table 6.3

Theme	Subtheme	Chapter presenting
		the findings
1. Psychological stress	1. 'I am looking forward to going	Chapter 6
	home but feeling anxious'	
	2. 'Not knowing what the future	
	holds'	
2. Resilience	1. 'I've come from zero to hero'	Chapter 6

Major Themes and Subthemes—Perspectives From Stroke Dyads

Theme	Subtheme	Chapter presenting
		the findings
	2. 'I look at this as a change in the	
	road for life'	
3. Certainty and	1. 'They boost you up'	Chapter 6
uncertainty about post-	2. 'We don't know what's going to	
discharge care	happen'	
4. Coping with setbacks	1. Processing loss and grief in	Chapter 7
	people with stroke	
	2. Processing loss and grief in	-
	caregivers	
	3. Reconciling with resilience	
5. Role transformation	1. Impact of stroke-related	Chapter 7
	limitations on people with stroke	
	2. Impact of stroke-related	
	limitations on caregivers	
	3. Regaining control of life	
6. Transitioning with	1. Support from family and friends	Chapter 8
support	2. Support from health	
	professionals	
7. Expectations in	1. Enhanced discharge planning	Chapter 8
hospital-to-home transition	and preparation	
care	2. Flexibility of service provision	
	3. 'Working towards a common	
	goal'	

The three main themes identified from the focus groups and from interviews with health professionals (Table 6.4) revealed inadequate assessment of psychological needs for stroke dyads. The care needs assessment was largely focused on the functional needs of people with stroke, such as personal care skills and cognitive functions; a formal assessment of caregivers' care needs was lacking. Furthermore, care service delivery was constrained by the limited length of hospital stays due to bed pressures and the funding model of services. This may inhibit a dyad-centred approach to service delivery and increase stress for health professionals. However, despite these constraints, health professionals strived for excellence through their commitment to safety and quality care. Their determination and flexibility in service provision seemed to enhance the hospital-to-home transition for stroke dyads.

Table 6.4

Theme	Subtheme	Chapter presenting
		the findings
1. Transition shaped by	1. 'Psychological/mental health is	Chapter 9
care needs assessment	underplayed'	
	2. 'Our focus is on the patient'	
2. Transition challenged	1. 'Bed pressures certainly	Chapter 9
by constraints on service	impacts'	
delivery	2. 'We're dictated by funding	
	models'	
3. Transition enhanced by	1. 'Empowering them to making	Chapter 9
commitment to the safety	decisions'	
of care	2. 'Working towards a common	
	goal'	

Major Themes and Subthemes—Perspectives From Health Professionals

6.4 Findings—Before Hospital Discharge

From the next section onwards, findings (major themes and subthemes) identified in the interviews with stroke dyads before hospital discharge are presented. These findings have been published in a peer-reviewed journal. Therefore, the presentation of findings is in a published manuscript format, as detailed in the following sections.

This section presents a publication arising from this thesis that was published in the *Journal of Advanced Nursing*.

Citation: Chen, L., Xiao, L. D., & Chamberlain, D. (2022). Exploring the shared experiences of people with stroke and caregivers in preparedness to manage post-discharge care: A hermeneutic study. *Journal of Advanced Nursing*, *78*(9), 2983–2999. <u>https://doi.org/10.1111/jan.15275</u>

This is the peer reviewed version of the following article: Exploring the shared experiences of people with stroke and caregivers in preparedness to manage post-discharge care: A hermeneutic study, which has been published in final form at

https://doi.org/10.1111/jan.15275. This article may be used for non-commercial purposes in accordance with Wiley Terms and Conditions for Use of Self-Archived Versions. This article may not be enhanced, enriched or otherwise transformed into a derivative work, without express permission from Wiley or by statutory rights under applicable legislation. Copyright notices must not be removed, obscured or modified. The article must be linked to Wiley's version of record on Wiley Online Library and any embedding, framing or otherwise making available the article or pages thereof by third parties from platforms, services and websites other than Wiley Online Library must be prohibited.

6.5 Exploring the Shared Experiences of People With Stroke and Caregivers in Preparedness to Manage Post-Discharge Care: A Hermeneutic Study

6.5.1 Abstract

Aims and objectives: To explore the shared experiences of people with stroke and caregivers in preparedness to manage post-discharge care.

Background: People with stroke and caregivers show dyadic effects in dealing with post-discharge care challenges. However, few studies have explored their shared experiences and unique challenges for each dyadic member in preparedness to manage post-discharge care. This study addresses this research gap.

Design: An interpretive approach underpinned by Gadamer's philosophical hermeneutics.

Methods: Semi-structured interviews were conducted with people with stroke and caregivers within five days before hospital discharge (n=26) or two weeks after discharge (n=4) between July and December 2019. The study followed a five-step data analysis method aligning with Gadamerian hermeneutics. The study reporting followed the COREQ checklist.

Results: Three themes and six subthemes were identified. First, both people with stroke and caregivers experienced psychological stress, although the sources differed. Second, stroke dyads demonstrated resilience built on positive thoughts, confidence and support from family and health professionals. Third, stroke dyads exhibited different levels of certainty about post-discharge care. They desired to be equal partners in co-developing and co-implementing discharge plans.

Conclusions: Similarities and differences in perceived preparedness to manage post-discharge care existed between stroke dyadic members. A co-design approach to developing and implementing discharge plans would enhance planned post-discharge care.

Impact: Hospital to home transition is a challenging time for people with stroke and their caregivers. Understanding the shared experiences of stroke dyads in preparedness for post-discharge care enables nurses to take proactive actions to enhance managing post-discharge care. Early identification of those at risk of developing psychological stress will enable nurses to co-develop stress-coping strategies. These will have a positive influence on the dyad when facing setbacks due to stroke-related complications.

Keywords: Caregiver, discharge planning, people with stroke, dyads, qualitative research, nurse, stroke, transition care

6.5.2 Introduction

Globally, over 13 million new stroke cases occur per annum (World Stroke Organization, n.d.). With advances in stroke management, morbidity and mortality have steadily reduced, resulting in more people with stroke returning home after hospitalisation (Olson & Juengst, 2019). After discharge, they and their caregivers must take over complex care that in hospital was usually provided or coordinated by nurses. However, most are illprepared with insufficient knowledge and skills to effectively manage stroke-related health issues, underlying chronic diseases and engage in poststroke rehabilitation (Camicia, Lutz, Harvath & Joseph, 2021; L. Chen et al., 2021). This situation is further compounded by shortened hospital stays that challenge nurses, who usually lead the multidisciplinary team implementing discharge plans, to prepare this patient population to take over complex care within a short window of opportunity (Camicia, Lutz, Harvath & Joseph, 2021; Mountain et al., 2020). The readmission rate of people with stroke can be as high as 43% within 12 months after discharge, and most readmissions can be prevented through carefully designed discharge plans (Kilkenny et al., 2020; Sennfält & Ullberg, 2020).

Most people with stroke have functional and self-care impairments due to hemiplegia or other disabilities (C. O. Johnson et al., 2019). Therefore, they heavily rely on family caregivers to care for them post-discharge (Camicia, Lutz, Harvath & Joseph, 2021; L. Chen et al., 2020). A study by Chung et al. (2016) revealed that an interdependent relationship existed in stroke patient-caregiver dyads in which one member's strengths in coping with

poststroke health issues, for example poststroke depression, can influence the other dyad member. The dyadic effect was also reported in a systematic review and meta-analysis which found that psychoeducation interventions targeting both members can significantly improve functional independence of people with stroke and simultaneously reduce caregivers' burden in the immediate term or within one month after discharge (Mou et al., 2021). Research evidence suggests that stroke patient-caregiver dyad-centred discharge plans should be routinely developed in clinical settings to enable nurses to detect dyad strengths and promote a strength-based dyadic approach to managing poststroke care (Mou et al., 2021, Pucciarelli et al., 2021). However, few studies examined shared experiences of the dyads in preparedness to manage post-discharge care. This study addresses the gap by exploring the similarities and differences in perceived preparedness to manage post-discharge care for people with stroke and their caregivers. Findings will enable nurses and other stakeholders to identify unique challenges to each dyadic member and foster discharge plans that support them as a dyad.

6.5.3 Background

Preparedness for discharge refers to the state of being ready or willing to take over care from health providers in people with stroke and their caregivers (Camicia, Lutz, Harvath & Joseph, 2021). Patients' and caregivers' perceived preparedness for discharge is the aim of discharge planning. Discharge planning is a core component of safety and quality standards in healthcare and is best built on partnerships with patients and their caregivers to prepare them to take charge of care after discharge (Australian Commission on Safety and Quality in Health Care [ACSQHC], 2019), (Mountain et al., 2020). However, studies revealed poor discharge planning inhibited people with stroke and caregivers from preparing themselves with adequate self-management capabilities to take over hospital to home transitional care (Camicia et al., 2020; L. Chen et al., 2020). Building the required skills for people with stroke and caregivers to provide ADLs and prevent stroke-related complications (e.g., falls and pneumonia) takes time and requires hands-on practice with nurses prior to discharge (Green et al., 2021). In addition, stroke is associated with chronic health

conditions, and both people with stroke and caregivers need to develop capabilities in chronic disease management, lifestyle modification, problem-solving skills, and utilisation of health care resources to manage other chronic conditions and cope with stress (Frantz et al., 2021). Patient-centred discharge planning is suggested in the hospital care standards (ACSQHC, 2019). This patient-centred approach may overlook interdependent relationships between people with stroke and caregivers (Chung et al., 2016, Mou et al., 2021); and may impede nurses from taking proactive actions to assess the strengths of people with stroke, encouraging active participation in self-care and supporting caregivers.

Studies revealed that people with stroke do not have a full understanding of the impact of stroke. Information about post-discharge services was not communicated to them at hospital discharge (L. Chen et al., 2020). Young people with stroke encountered challenges sourcing post-discharge age-appropriate support (Hersh & Armstrong, 2021; Stokman-Meiland et al., 2022). Shortened hospital stays added more challenges for nurses and multidisciplinary teams to assess and address these issues during discharge planning. Incorporating stroke patient-caregiver dyad in discharge planning that considers the dyadic effects may help nurses identify innovative solutions to address these challenges. Any solutions should be built on nurses' careful analysis of unique challenges each member face within a dyad regarding post-discharge stroke care.

Studies revealed that stroke caregivers faced enormous uncertainties about their roles and responsibilities in poststroke care (Camicia, Lutz, Harvath & Joseph, 2021; L. Chen et al., 2021). Caregivers' needs were often not assessed and met before the discharge of people with stroke from inpatient wards or rehabilitation facilities into their care at home (Camicia, Lutz, Harvath & Joseph, 2021, L. Chen et al., 2021). Caregivers' unmet care needs ranged from insufficient information on how to assume their role to a lack of basic care skills and problem solving and coping strategies in dealing with challenges arising from changed emotions, cognition and/or altered behaviours of people with stroke (Camicia, Lutz, Harvath, & Joseph, 2021). Camicia and colleagues strongly suggested that nurses use the preparedness assessment tool to explore stroke caregivers' needs and address their

concerns before hospital discharge (Camicia, Lutz, Harvath & Joseph, 2021). The assessment outcomes should be interpreted in the context of an interdependent relationship within the stroke patient-caregiver dyad and consider the influence of people with stroke over caregivers.

Stroke, as a sudden adverse event or family crisis, has an enormous impact on the mental health and psychological wellbeing of people with stroke and their caregivers which weakens their active preparedness for discharge from hospital (Minshall et al., 2021). Depression and anxiety affect approximately 30% and 25% of the stroke population respectively and contribute to poor mental health and guality of life, high mortality rates and prolonged disabilities (Medeiros et al., 2020; Minshall et al., 2021). Mental stress along with stroke-related disabilities adds vulnerabilities to people with stroke developing adverse events after discharge and increasing hospital readmissions (Stein et al., 2020). Studies also indicated that caregivers experienced high-levels of emotional stress from this sudden family crisis and 40%-60% showed depression which affected their ability to undertake and sustain the caregiving role (Mou et al., 2021). Hospital readmissions can be prevented through timely detection of mental disorders, psychological and emotional stress, and individualised support for patient and caregiver populations (Denham et al., 2022; Stein et al., 2020). A study by Chung et al. (2016) suggested that strategies building on dyadic members' strengths in preventing stroke-related depression are imperative. Discharge planning provides an opportunity for nurses to initiate interventions addressing stroke-related depression within stroke dyads. However, understanding their shared experiences and unique challenges for each dyad member in preparedness for discharge is a prerequisite.

6.5.4 The Study

6.5.4.1 Aim

The aim of this hermeneutic study was to explore the shared experiences of people with stroke and caregivers in preparedness to manage post-discharge care. This study is part of a longitudinal study that explored hospital to home transition care for people with stroke and informal caregivers.

6.5.4.2 Design

This study applied an interpretive approach underpinned by Gadamer's philosophical hermeneutics. Gadamer described the interpretation process as a "hermeneutic circle", where the inquirer enters an interpretation process with an open mind and historical awareness of a phenomenon (pre-knowledge) (Gadamer, 1979, p 235). Gadamer emphasised that a deeper understanding can only be gained when the inquirer consciously identifies, provokes and revises their own pre-knowledge (Gadamer, 1979). Reaching a deeper understanding in a study using Gadamer's hermeneutic design has the key components shown in Figure 6.1 and detailed in the following (Alsaigh & Coyne, 2019, Fleming et al., 2003). First, the researcher needs to decide on the aim of study or develop research questions based on a research gap. In this study, we conducted two systematic reviews that helped us identify the research gap (L. Chen et al., 2020, 2021) and the interdependent relationship in the stroke patient-caregiver dyad (Chung et al., 2016, Mou et al., 2021). Second, the researcher needs to be equipped with pre-knowledge in the study field. LC currently practises in stroke care, LX and DC are experienced in chronic diseases management and nurse-led interventional study field. Our commitment to the two systematic reviews and reflections on current practice prepared us to achieve this level of preknowledge. Third, the researcher needs to conduct interactive dialogues with participants by which understanding is developed through fusion of the horizons of participants and the researcher. Fourth, the researcher needs to evolve their understanding through dialogues with text (transcripts and field notes) during the data analysis phase. Fifth, the researcher needs to constantly relate participants' point of views to findings reported in previous studies (historical awareness) to ascertain a new understanding generated from the dialogues with participants. In addition, the researcher needs to revise their own understanding and gain a deeper understanding (new knowledge) by reflecting on current practice including current policies and standards guiding the clinical practice individually or collectively within a team. This deeper understanding represents a shared understanding co-developed by the participants and the researcher. These key components set Gadamer's hermeneutics apart

from other phenomenology methodologies (e.g., Husserl's phenomenology) in which the researcher is required to 'bracket' their pre-knowledge in the research field in order to minimise the researcher's influence on the findings (Crotty, 1998).

Figure 6.1

Research Process (Adapted From Fleming et al., 2003 and Alsaigh & Coyne, 2019)



6.5.4.3 Settings and Participants

This study is part of the first author (LC)'s PhD study and was conducted in a stroke rehabilitation unit at a tertiary care public hospital in Adelaide, South Australia. Australia has a unique hybrid health system with the foundational public health system (Medicare) complemented by private health insurance (Department of Health and Aged Care, 2019). Australians can choose to utilise Medicare or private health insurance to fund inpatient care, primary health care and community healthcare services (Department of Health and Aged Care, 2019). In South Australia, based on individual situations and age, discharge destinations include 1) home with community care services, ambulatory rehabilitation services, community transition care services or National Disability Insurance Scheme (NDIS) (for people 64 years and younger); 2) inpatient rehabilitation services, residential transition care services, or residential aged care facilities.

The rehabilitation unit where this study was conducted is led by an interprofessional team. Discharge education and information provision occurs throughout the admission with particular focus on developing a post-discharge plan two to three days before discharge. Discharge planning usually entails activities with people with stroke and their caregivers including goal setting, caregiver training, education, home visits and family meetings. A patient's target length of stay is assessed based on the rehabilitation outcomes measure – Functional Independence Measure (FIM) on admission (Uniform Data System for Medical Rehabilitation, 2012). However, their actual length of stay and discharge destination depends on individual factors, such as severity of stroke deficits and caregiver availability and capacity, and availability of community rehabilitation and care services etc.

Participants were selected based on the inclusion criteria (see Table 5.1). The criterion sampling enabled identification and selection of information-rich participants relevant to this study (Polit & Beck, 2012). Recruitment occurred between July and December 2019. During this period, there were 142 stroke admissions to the unit. Based on the inclusion criteria, unit clinicians recommended potential participants. LC, who is the Nurse Manager of this unit but does not have a direct care relationship with patients, initially

approached potential participants to determine their interest. If interested, a Research Assistant provided and explained the participant information sheet and consent form. Among 22 eligible people with stroke-caregiver dyads, two declined, two were excluded due to unstable living conditions and one deceased. In total, 17 people with stroke-caregiver dyads were invited to the study and one person with stroke had two caregivers, making a total of 35 potential participants. However, five people with stroke were unable to participate in an interview due to moderate to severe cognitive and/or speech impairment. Therefore, only 30 of them participated in interviews. The mean length of stay of people with stroke in the rehabilitation unit was 28 days. All people with stroke were discharged home into the care of their family with community support.

6.5.4.4 Data Collection

We developed separate semi-structured interview questions for people with stroke and caregivers (see Table 6.5). These interview questions were developed based on the research aim, the findings of literature reviews and previous studies on stroke rehabilitation and hospital to home transition care (L. Chen et al., 2016, 2021). These questions served as a prompt for people with stroke and caregivers to share their point of views regarding preparedness to manage post-discharge care. The interviews reported here are the first round of interviews in a longitudinal study with follow up interviews occurring three and six months after returning home.

LC collected data through face-to-face semi-structured interviews with participants. People with stroke and caregivers were interviewed separately except people with mild cognitive impairment (n=3) who were interviewed with their caregiver to cross check the narratives. Of 30 participants, 26 participants including those with cognitive impairment were interviewed 24 to 48 hours before discharge except one dyad who was interviewed 48 hours beforehand. Due to the participant's deteriorating medical condition, discharge was subsequently postponed until three days after the interview. Two dyads were interviewed within two weeks after discharge due to time constraints before discharge. Participants

chose an interview time and venue. The average time of each interview was 30 minutes and was audio-recorded and transcribed verbatim for analysis.

Table 6.5

Interview Guide

Participants	Stroke Patients	Stroke Caregivers
Question 1	How have you been since the	How have you been since the
	stroke?	stroke of your relative?
	How have you been dealing with	How have you been coping with
	the fact that you had a stroke?	the fact that your relative had a
	How do you think your family	stroke?
	cope with what happened?	• How do you think your relative is
		dealing with what happened?
Question 2	Now you are going home soon,	Now your relative is going home
	please tell me about your	soon, please tell me about your
	rehabilitation experience.	thoughts on the caregiving role.
	How prepared do you think you	How prepared do you think you
	are for returning home?	are for supporting your relative at
		home?
Question 3	What are your expectations after	What are your expectations after
	returning home?	your relative returns home?
	Expectations on your	• Expectations on the relative's
	rehabilitation/recovery.	rehabilitation/recovery.
	Expectations on community	• Expectations on the support
	health and social care services.	services.

6.5.4.5 Ethical Considerations

This study was approved by the the Southern Adelaide Clinical Human Research Ethics Committee (Approval No. 39.19). Participants received a detailed explanation of the study and participation was voluntary. Participants were informed of strategies to ensure confidentiality and their freedom to participate or withdraw from the study. They provided informed consent before the interview. Deidentified data were stored on a secure university cloud drive.

6.5.4.6 Data Analysis

Gadamer advocated a systematic approach to achieve a deeper understanding although he did not offer a detailed data analysis method (Gadamer, 1979). Subsequently, a group of researchers explored directions for undertaking data analysis (Alsaigh & Coyne, 2019, Fleming et al., 2003). We have adapted these methods and detailed our dialogues with text (transcripts and field notes) in the five interrelated and cyclical steps outlined below to enrich understanding. Nvivo 12 software was used to assist with data analysis.

Step 1: Immersion. Interviews were transcribed verbatim and read repeatedly while listening to the original record to immerse researchers in the raw data. From this an understanding of the whole text was obtained, which influenced understandings of other parts of the text. At this step, text was encountered with anticipation developed through our pre-knowledge.

Step 2: Understanding. Every single sentence of the text (the part) was examined carefully to expose its meanings and identify first order (participant) constructs. This was achieved with line-by-line coding in NVIVO. Open codes were developed by LC to represent participants' ideas (their horizon).

Step 3: Abstraction. The researchers identified subthemes by grouping open codes into clusters and labelling each cluster based on the meaning. Regular team discussions were conducted to elaborate potential subthemes based on their relevance to the aim of the study and the researchers' pre-knowledge. The subthemes represented the researchers' horizon.

Step 4: Synthesis and theme development. Subthemes were grouped into themes through elaboration and clarification on the relationship between subthemes and themes (see Table 6.6). This involved continuous moving back and forth between the literature, our earlier analysis and the text, and across people with stroke and caregiver datasets to identify similarities and differences. Moving from parts to the whole and back again is known as the rhythm of Gadamer's hermeneutic circle. In this process, themes were informed by and in turn enhanced researchers' pre-knowledge. From this process, interpretations on the perceived preparedness of people with stroke and their caregivers to manage postdischarge care, and the differences between them evolved.

Step 5: Illumination and illustration of phenomena. The researchers linked the literature to the themes and subthemes derived from the entire dataset. Based on the themes, subthemes and their interrelationships, the team reconstructed the perceived preparedness to manage post-discharge care for people with stroke and caregivers using their own words and highlighted key findings from the data. In this process, the research team examined the quality of stories to ensure the constructed stories reflected participants' perceptions and their voices were represented in the findings.

Through dialogues with the text, our pre-knowledge changed through data collection and interpretation. The team met and discussed these changes to facilitate the hermeneutic circle process (Fleming et al., 2003).

Table 6.6

Transcripts	Coding	Grouping and summarising coding	Sub-theme	Theme
Out of here being symbolic. I've done well, and I'm on my way again (SS17).	SS17Co1: Out of hospital a great achievement SS17Co2: Satisfied with recovery SS17Co3: Ready for	Focusing on capabilities SS09Co1: Clear about limitations and solution SS11Co1:	"I've come from zero to hero."	Resilien ce
	a new journey after discharge	Accepted		

Interpretation Process Exemplar

Transcripts	Coding	Grouping and	Sub-theme	Theme
		coding		
I've come from zero to hero. A fortnight ago, I	SS03Co1: Consider recovery a great success	poststroke impairment SS11Co2 ⁻		
couldn't sit up in bed. [Now] I can walk some distance independently (SS03).	SS03Co2: Significant achievement in 2 weeks	Observed stroke patients in worse conditions SS11Co3:		
I just know that I have to be careful and go	SS09Co1: Clear about limitations and solution	Satisfied with remaining physical functions		
those things I'm not terribly	SS09Co2: Plan to continue progress	SS13Co1: Observed other		
good at but I'll try. My brother	with caregiver	disabilities		
will fix the shopping and I will do the cooking. And we'll go back to how we were	SS09Co4: Confidence to resuming pre- stroke life	SS13Co3: Appreciate remaining body functions		
before (SS09).	SS13Co1: Observed	Positive thoughts about recovery		
disabilities that had happened to them, and I	other stroke patients' disabilities SS13Co2: Felt lucky	SS17Co1: Out of hospital a great achievement		
saw this man and he was just so badly stricken with	having mild stroke SS13Co3: Appreciate remaining body	SS17Co2: Satisfied with recovery		
his stroke that it really upset me And it made me	functions	SS03Co1: Consider recovery a great success		
lucky I was that I had just had a minor stroke, was still able to		SS03Co2: Significant achievement in 2 weeks		
think and feel (SS13).		SS13Co2: Felt lucky having		
After really accepting that I had a stroke, I	SS11Co1: Accepted poststroke impairment	mild stroke		
understand I'm not going to get back to where I was and things	SS11Co2: Observed stroke patients in worse conditions			

Transcripts	Coding	Grouping and summarising coding	Sub-theme	Theme
will be all slower And I see around here a lot of people are far worse and I'm certainly happy with my physical side of things in many ways (SS11).	SS11Co3: Satisfied with remaining physical functions	SS09Co2: Plan to continue progress SS09Co3: Clear plan with caregiver SS09Co4: Confidence to resuming pre- stroke life		
I do have the facility and I've already had it. When there's a drama, I gone dead calm and deal with it. [It is] my personality' (SC11). I face things front- on, so this is what it is, look at the positives. I 've got him and he's luckythe family's lucky. We don't look at this as a disability, I look at this as a change in the road for life. All my life and we've had a lot of ups and downsI just accept it and work from there (SC08).	SC11Co1: Natural ability to manage crisis SC11 Co2: Calm personality to manage unexpected situations SC08Co1: Personality to look at positives SC08Co2: Felt lucky husband still alive SC08Co3: Interpret stroke as a change rather than a disability SC08Co4: Past life experience facilitate acceptance of stroke	SS17Co3: Ready for a new journey after discharge Coping through internal strength SC11Co1: Natural ability to manage crisis SC11 Co2: Calm personality to manage unexpected situations SC08Co1: Personality to look at positives SC08Co4: Past life experience facilitate acceptance of stroke SC01Co1: Activated coping strategies	"I look at this as a change in the road for life."	
I'm as prepared as I can be for what's going to come, but I really don't know what's going to come. I think it's just	SC14Co1: Prepared for uncertainties SC14Co3: Pre- empted to face ups and downs	Coping through positive thoughts and preparation		

Transcripts	Coding	Grouping and summarising coding	Sub-theme	Theme
going to be again a rollercoaster for some time to come (SC14).	SC01Co1: Activated coping strategies SC01Co2: Coping strategies to alleviate stress	SC08Co2: Felt lucky husband still alive SC08Co3: Interpret stroke as a change rather than a		
So, Thave done a lot of mindfulness practice and that has been hugely beneficial. And I would really recommend that there is support for people in that area to learn – to be able to manage their anxiety or anything that might come up (SC01).		disability SC14Co1: Prepared for uncertainties		
		SC14Co3: Pre- empted to face ups and downs		
		SC01Co2: Coping strategies to alleviate stress		

Note. SS = people with stroke; SC = stroke caregiver; Co = code.

6.5.4.7 Rigour

We applied the criteria established by Lincoln and Guba (1985) to ensure rigour of this study. Credibility was enhanced by audio-recording interviews and transcribing verbatim, crosschecking statements with participants during interviews and debriefing and discussion among the research team (Morse, 2018). Strategies applied to fulfil dependability and confirmability of data included maintaining a decision trail, using participants' own words to support themes and extensive research team discussions (Morse, 2018). During data analysis, LC maintained a research journal in which the research team's reflections and discussions were recorded. This practice also enhanced dependability and confirmability of data. Transferability of the findings was enhanced by citation of existing research evidence to verify interpretations and conclusions. When reporting, original excerpts from participant interviews were used to enhance readers' understanding (Morse, 2018). A reflexive journal

was maintained throughout the research process to enhance the overall trustworthiness of this study. Study reporting followed the Consolidated Criteria for Reporting Qualitative Research (COREQ) checklist (Tong et al., 2007) (see Appendix 22).

6.5.5 Findings

In total, 17 stroke dyads were invited to the study and interviews were conducted with 12 people with stroke and 18 caregivers only as five people with stroke were unable to participate due to cognitive and/or speech impairments. Table 6.1 outlines participants' characteristics. Three main themes with two subthemes under each theme presented perceived preparedness of people with stroke and caregivers in managing post-discharge care. First, both people with stroke and caregivers showed psychological stress that may affect them to take over complex post-discharge care. The causes and triggers contributing to the stress differed and required nurses and the multidisciplinary team to consider individualised discharge planning. Second, despite the stroke crisis in the family, both people with stroke and caregivers demonstrated similar resilience that might enable dyadic effects to overcome challenges and stress arising from post-discharge care. Third, perceived certainty about post-discharge care and rehabilitation differed between people with stroke and caregivers. Most people with stroke were certain about post-discharge rehabilitation due to considerable support they received from nurses and therapists during the hospitalisation. However, most caregivers exhibited uncertainty due to lack of engagement in discharge planning. Both people with stroke and caregivers expected to co-develop and co-implement discharge plans to meet their concerns regarding post-discharge recovery and care. These themes are detailed in the follows and additional information regarding findings is presented in Appendix 23.

Theme 1: Psychological stress

Before discharge, both people with stroke and their caregivers exhibited psychological stress resulting from the unexpected and overwhelming nature of the stroke. The origins seemed to differ between them. People with stroke were anxious about strokerelated loss of functions whereas caregivers expressed uncertainty about their caregiving

role and the impact of the stroke on their future lives. Understanding the dyads' psychological stress and different causes would facilitate the assessment of risks and development of dyad-focused discharge plans by nurses and the multidisciplinary team.

Subtheme 1.1: "I am looking forward to going home but feeling anxious."

People with stroke expressed psychological and emotional stress arising from the onset of stroke. Stress seemed to affect their mental strength to deal with anxiety about returning home.

The loss of functions seemed to have caused anxiety. A person with stroke described his impairments in calculation and writing: "It is frustrating. I'm annoyed about it, and I wrote this morning, well a page in the notebook...it probably took me half an hour to write, and I made some mistakes" (SS11). The loss of functions triggered various psychological and emotional reactions, for example, frustration, depression, fear and worry in people with stroke. Inability to walk safely was distressing to people with stroke, but it also seemed to form a major part of their rehabilitation goals as a person with stroke stressed:

A big focus for me was the walking part because that was quite devastating to find that you'd stand up and you couldn't put any weight on the left hand side. And when I started using the walker, it was kind of very stilted... it became I suppose a focal point for me to actually keep doing that (SS12).

People with stroke were challenged with the unexpected and overwhelming nature of stroke-associated functional impairments and the detrimental impact on their lives. They would benefit from professional counselling that helped them develop positive thoughts, a critical indicator of psychological preparedness to self-manage post-discharge care.

They were also concerned about the consequences of stroke on their health after discharge: "[I feel] a bit depressed, worried about having another one [stroke] and not going back to my previous life. Worried about my independence" (SS09). In this case, psychological stress was triggered by the lack of knowledge about preventing a second stroke. Discharge planning that incorporates self-care education would relieve people with

stroke from this stress. Another person with stroke resonated her anxiety about imminent discharge:

I am looking forward to going home but, at the same time, I'm also feeling anxious about my level of functioning in the home situation...what impact it's going to have on me and how is it going to perhaps affect me emotionally as well (SS12).

Findings indicated people with stroke were psychologically unprepared to selfmanage care after discharge. Early detection of psychological stress via assessment and individualised interventions could support them to better cope with stress before discharge. Findings also indicated the need to integrate psychological counselling into post-discharge care services.

Subtheme 1.2: "Not knowing what the future holds."

This subtheme revealed various sources that triggered psychological stress for caregivers. Caregivers perceived that they were assigned to the caregiving role and expected to take over complex poststroke care while still affected emotionally by the vivid memory of stroke onset and facing potential financial impacts. They were unconfident about providing home care and unclear about the post-discharge plan.

Stroke suddenly changed the lives of caregivers. Remembering events at the onset of stroke continued to emotionally impact caregivers, which heightened their anxiety as a caregiver recalled: "I worry about him at home. I've been the one that's found him on the floor. And I worry about that [another stroke] ... and falling again" (SC18). Another caregiver described: "The initial shock was terrible because we nearly lost him…it seemed unreal, but gradually, I know from the moment he had the stroke my life will change, so will his" (SC08).

The emotional trauma challenged caregivers to have the mental strength to take over care and assume their role in hospital to home transition. The need for sufficient time to be mentally prepared to take charge of care at home was mentioned by a caregiver interviewed on the day of her husband's discharge: "It's been difficult...Just running everywhere, making sure that everything's okay when he comes home. I'm usually very organised... things have sort of gone a bit haywire. I lost my keys, forget my phone" (SC18). The psychological and

emotional distress of caregivers may be underestimated due to the focus on people with stroke, rather than their caregivers, during hospital stays. An assessment of caregivers' psychological health would help identify their needs and facilitate early interventions. Caregivers may need additional support during the transition in order to be mentally stable to take over care of the stroke patient.

Stroke can have a negative impact on family income, especially when the people with stroke were the main breadwinner. In such a circumstance, financial stress becomes another source of mental stress for caregivers:

Well because XX [the husband] was the main breadwinner we've got a mortgage, young family, kids in school, I needed to stabilise my work... But I've cut back one day so I'm three days a week, but I need to build back up to doing just so that we can pay off our [bills] (SC05).

Caregivers were anxious before discharge because they were unsure what the future would hold and whether they would have sufficient capacity to provide care. A wife described her concerns about providing care at home:

I do [worry] because he's not always easy to tell to slow down, and we'll stress at each other over those things. And just not knowing what the future holds, how I can care for him and help him better... [I feel] really stressed, depressed, just generally a lot of sadness because of not knowing what the future holds for him. Just coping, coping not as good as I used to be (SC03).

In this case, caregivers may benefit from an assessment on their preparedness for the caregiving role and a psychological consultation to alleviate anxiety and other mental health issues.

Findings demonstrated that various sources may contribute to caregivers' mental stress during hospital to home transition. Their perceived preparedness for providing care depended on their mental status and pre-discharge support received to help stabilise this.

Theme 2: Resilience

In this study, both people with stroke and caregivers demonstrated resilience despite the devastating outcomes of stroke. Their resilience was built on the same foundations such as positive thoughts and appraisal of their situation. Resilience of people with stroke was enhanced by their functional achievements and family support, whereas caregivers' resilience was built on their internal strength, past life experience and pre-existing coping strategies. Insights into the mechanisms underlying resilience would guide nurses and rehabilitation clinicians in tailoring interventions that address the needs of both people with stroke and caregivers from individual and dyadic perspectives.

Subtheme 2.1: "I've come from zero to hero."

This subtheme demonstrates resilience and acceptance of stroke-related limitations for people with stroke despite a great sense of loss of functions. The source of their resilience came from their positive thoughts, positive appraisal of attaining rehabilitation goals and family support.

By focusing on their functional capabilities and comparing themselves with worse cases, people with stroke generated positive thoughts about their recovery as they stated: After really accepting that I had a stroke, I understand I'm not going to get back to where I was and things will be all slower... And I see around here a lot of people are

far worse and I'm certainly happy with my physical side of things in many ways

(SS11).

I saw the awful disabilities that had happened to them, and I saw this man and he was just so badly stricken with his stroke that it really upset me ... And it made me realise how lucky I was that I had just had a minor stroke, was still able to think and feel (SS13).

Acceptance of stroke-related functional impairments seemed to enhance optimism for people with stroke about their ability, rather than their disability. Recognising these positive thoughts may help nurses to activate and motivate people with stroke to selfmanage their chronic conditions and rehabilitation.

People with stroke who viewed discharge from hospital as a huge milestone reinforced their resilience: "Out of here being symbolic. I've done well, and I'm on my way again" (SS17). The same attitude was echoed by another participant: "I've come from zero to hero. A fortnight ago, I couldn't sit up in bed. [Now] I can walk some distance independently" (SS03). Their positive appraisal of recovery may activate and motivate them to self-manage care after discharge and engage in further rehabilitation.

People with stroke appeared prepared to overcome challenges ahead: "I'm just so looking forward to going home. I am probably going to make mistakes, but I've got things organised" (SS15). Support from families also enhanced resilience:

I just know that I have to be careful and go slowly. All those things I'm not terribly good at but I'll try. My brother will fix the shopping and I will do the cooking. And we'll go back to how we were before (SS09).

The findings highlighted that resilience is a crucial personal attribute that could enable self-management of care after discharge. Through discharge planning, nurses and rehabilitation health professionals can enhance resilience in people with stroke by engaging them in self-assessing their abilities to manage their conditions and by providing resources to support self-management in the community setting.

Subtheme 2.2: "I look at this as a change in the road for life."

Caregivers demonstrated resilience and revealed mechanisms to manage change including their personality, past life experiences, views about life and death, and strategies previously used to alleviate stress. These helped them face the sudden onset of stroke and perceived challenges after discharge.

Caregivers discussed how the coping strategies that came naturally as part of their personality had helped them respond to the stroke event. A caregiver linked her personality with coping mechanisms: "I do have the facility and I've already had it. When there's a drama, I gone dead calm and deal with it. [It is] my personality" (SC11).

The combination of a positive outlook and past life experiences seemed to have further strengthened caregivers' perceived capacity to manage the uncertainties and challenges ahead of them. A wife's statement represented:

I face things front-on, so this is what it is, look at the positives. I 've got him and he's lucky...the family's lucky. We don't look at this as a disability, we look at this as a change in the road for life. All my life and we've had a lot of ups and downs...I just accept it and work from there (SC08).

Past work experience and prior knowledge about the healthcare system alleviated anxiety and made it easier for caregivers to coordinate care. As a caregiver shared:

My job is working with people that live at home with disabilities. So, it's more setting up rosters and doing care plans and working with the NDIS [National Disability Insurance Scheme]. So, I have set up support for XX [the husband] with the NDIS just for when I'm working (SC05).

Caregivers' openness towards life and death facilitated the development of resilience and enabled them to embrace adversity with courage and an open mind as a caregiver described their advanced care planning:

We've talked about death and dying, we're very straight forward about the future. We've got things in place with the family and with our executor as to what we want and we both realise he's 82, I'm nearly 80 that our life is coming towards the end (SC11).

Caregivers who showed resilience might be better prepared to overcome challenges and stress they may face after discharge. A couple of caregivers shared their thoughts:

I'm as prepared as I can be for what's going to come, but I really don't know what's going to come. I think it's just going to be again a rollercoaster for some time to come (SC14).

So, I have done a lot of mindfulness practice and that has been hugely beneficial. And I would really recommend that there is support for people in that area to learn – to be able to manage their anxiety or anything that might come up (SC01).

Caregivers' resilience activated their coping mechanisms. Resilience may enhance caregivers' capability to support people with stroke and overcome challenges ahead in hospital to home transition.

Theme 3: Certainty and uncertainty about post-discharge care

This theme highlighted the perceived certainty about post-discharge care and rehabilitation differed between people with stroke and caregivers. People with stroke had more interactions with nurses and other health professionals to discuss their post-discharge rehabilitation. Therefore, they showed a sense of certainty about post-discharge rehabilitation plan and implementation. In contrast, caregivers perceived uncertainty about post-discharge care due to lack of clarity about post-discharge support services and the exclusion from discharge planning. The findings are detailed in the following subthemes.

Subtheme 3.1: "They boost you up."

Setting clear and realistic goals with rehabilitation therapists for continued rehabilitation at home gave participants hope of returning to norm after discharge. Several people with stroke shared their views on what they planned to achieve at home: "I think probably a week, 10 days, I'll be managing to walk...There are stairs at home, well I won't be able to do those yet ... but I think eventually I'll be able to do all those things" (SS15); "I'd like to be able to strive to get back to a normal life. I'm hopeful that given a week at home, I can push a vacuum around..." (SS03).

Interventions tailored to individuals and positive feedback from nurses and therapists while in the rehabilitation unit improved adherence to rehabilitation programs during the hospital stay and further enhanced self-determination of people with stroke to succeed at home. Tailored rehabilitation was built on partnerships with nurses and other health professionals. A couple of participants' comments exemplified:

They [nurses and therapists] know where I'm comfortable to work within that risk parameter, but if they weren't as professional, I may be nervous and not as willing to undertake those next exercises. And they encourage you all the time that makes me feel like to do more (SS17).

[They] explain what was happening all the time. They [nurses] have given me a lot of confidence. They said: 'Oh, you're wonderful. You did it all really well'. So apart from the actual showing you how to do things, they boost you up (SS15).

These examples showed a true partnership approach between people with stroke and health professionals in co-developing and co-implementing discharge plans. This cooperative approach may have helped improve health professionals' knowledge about care needs and facilitated shared decision-making.

Awareness of the continuum of rehabilitation and care services available after discharge seemed to reassure people with stroke that support to manage poststroke recovery and care at home was available. A participant shared his experience of how a wellorganised post-discharge rehabilitation plan increased his self-confidence: "I am confident, I'm not intimidated or nervous by it, rehab will go on but in a different way, from the home and visits from here. So, I'll still get that support, I'm just thrilled to be going home" (SS17).

However, the lack of a clear rehabilitation plan after discharge was also reported: "I don't know exactly what the services are at this stage" (SS11). Another person with stroke echoed: "I believe that I'll be getting some in-home rehab and that's one of the things. The other things are sort of not specifically sure in the home situation" (SS12). These contrasting findings highlighted the significance of information sharing to enhance preparedness of people with stroke for discharge. A clear follow-up plan increases confidence as well as supporting self-management capabilities at home.

Findings indicated that people with stroke demonstrated determination for transition through their hope for rehabilitation. Nurses and the interprofessional team who used a codevelop approach enhanced sense of certainty, confidence, motivation and preparedness for discharge in people with stroke.

Subtheme 3.2: "We don't know what's going to happen."

Different from perception of people with stroke, caregivers demonstrated uncertainty about post-discharge care due to lack of opportunities to engage in discharge planning with nurses and other health professionals. Some caregivers perceived that they were not fully

engaged in developing and implementing discharge plans. As the handover of care neared, they expected health professionals to treat them as an equal partner and help them to build capabilities to take over care at home.

Caregivers were unsure what to expect and felt things would be out of their control. A mother of three young children discussed her uncertainty about rehabilitation and associated outcomes at the point of discharge:

No, I just know that really there can't be expectations because we don't know what's going to [happen] – my commitment is only to put the work in, to put the therapies in place, that's what I requested from all of them [service providers] to give me as much as we can with therapy ... but you know we'll just keep strengthening him and see what we can glean out of it (SC05).

In this case, education about post-discharge care for the caregiver was much needed to activate and empower them to work with health professionals as an equal partner in developing and implementing discharge plans.

Caregivers desired to be fully informed of the post-discharge rehabilitation plan. However, approaching discharge that information could be unclear. A wife expressed her concerns on the day of discharge:

I don't know what to expect. I believe someone came and spoke to XX [the husband]. But he's not reliable [due to memory issues]. And here we are going home. I'd like to know who I'm expecting to come into my home, how often. Now because of Christmas it's not going to happen. So, it's just bad luck that you've got the public holidays (SC18).

This case revealed the lack of a consistent partnership approach by health professionals to co-developing post-discharge plans with caregivers. It also indicated the lack of information provision about stroke-related community resources. Together, these situations triggered additional psychological stress for this caregiver that affected her confidence and preparedness to manage post-discharge care.

Caregivers continued to cope with life changing situations caused by the stroke crisis in the family. Hence, they might be unavailable for meetings scheduled by discharge nurses and the interprofessional team. As a daughter complained: "It was a little bit disappointing because we had phone calls about that we all get together and have a chat about Mum. And it just never happened" (SC13). This example reveals the difficulties in engaging all family members in discharge planning during a short hospital stay. More realistic strategies that enable family members to engage in discharge planning, for example publicly available education programs for family members and meeting with the family soon after discharge, may be needed.

A husband who has cared for his wife for the past 30 years echoed: "But also not knowing on the support side what's available, what's going to be available. It really was a matter of trying to get information from XX [an organisation] (SC06)". This case indicates that people with prior long-term experience in the caregiver role still feel a lack of capability to navigate information about stroke-related post-discharge care. The need for additional caregiver education about post-discharge care is evident.

In contrast, some caregivers perceived that they were included in the decisionmaking and planning process, and they benefited from these activities. Caregivers commended their interactions with the interprofessional team that prepared them to care for their mother after discharge: "[We have been] totally involved and where we've thought that we haven't had an answer quickly enough, we've rung; so, there was a time where somebody didn't get back to us and we followed it up straight away (SC 14 & 15)". This case supports the dyad-centred discharge planning that may be enhanced with two-way communication and collaboration between caregivers and discharge nurses or other health professionals.

The availability of timely access to rehabilitation team support after discharge improved caregivers' perceived ability to assume their role. As a wife who would be caring for her husband with cognitive impairments specified: "I also feel confident – I am being told
all the time that I can access support here, ask any questions. That will be important for anything that comes up" (SC01).

Compared to people with stroke, caregivers may have limited time capacity to discuss their concerns with discharge nurses and other health professionals. This may impact their participation in discharge planning. Innovations such as video/tele discharge planning conferences could improve communication and information-sharing. Discharge nurses' collaborative approach could also help maintain an open dialogue with caregivers. Caregivers' desires to co-develop and co-implement discharge plans with discharge nurses and other health professionals were evident.

6.5.6 Discussion

The use of Gadamerian philosophical hermeneutics (Gadamer, 1979) enabled us to holistically interpret the shared experiences of people with stroke and their caregivers regarding perceived preparedness to manage post-discharge care. The interpretation of findings has been further enhanced through the first author's reflections on her own practice as an experienced stroke nurse. Our study revealed that perceived preparedness of people with stroke and caregivers in managing post-discharge care is influenced by their psychological and emotional status, resilience and the level of engagement in discharge planning. The findings indicated that the event of returning home triggered distinct feelings in people with stroke and caregivers. Returning home after stroke was celebrated as a significant milestone in the poststroke trajectory by both. However, compared to people with stroke, caregivers demonstrated feelings of profound uncertainty about their role and how to take over care after hospital discharge.

In this study, people with stroke and caregivers were to experience transition from professional care in hospital to self-management at home. Transition can be defined as "a passage from one life phase, condition, or status to another" (Chick & Meleis, 1986, p. 239). Meleis transition theory described personal, community and societal transition conditions that could facilitate or hinder the transition (Meleis, 2010). In 'personal conditions', Meleis discussed that a wide range of emotions, especially stress and emotional distress, are

common during the transition. The level of comfort with changes in the body influences wellbeing during the transition (Meleis, 2010).

Aligning with Meleis transition theory, our findings demonstrated that people with stroke and caregivers experienced different psychological stressors in the lead up to discharge. Psychological stress for people with stroke seemed to be related to the loss of function, worries about prognosis and future independence. Whereas caregivers' emotional stress appeared to be triggered by lack of knowledge and confidence about their caregiving role and unclear post-discharge plans. Psychological and emotional stress have detrimental impacts on caregivers' preparedness to manage post-discharge care, suggesting the need to provide them with psychological counselling services in the poststroke trajectory. However, discharge plans largely focus on physical and cognitive functions of people with stroke with substantially less attention to psychological and emotional stress (Hughes & Cummings, 2020; Stroke Foundation, 2019b). A recent stroke services audit revealed that only 56% of people with stroke had a mood assessment and only 37% of audited hospitals provided psychology services (Stroke Foundation, 2019a). Nurses leading discharge planning in a multidisciplinary team need to initiate and coordinate psychological counselling services. Considering the dyadic effects in poststroke depression prevention reported by Chung et al (2016), assessments of dyadic members' characteristics relevant to depression prevention, such as self-esteem and optimism, need to be undertaken and development of interpersonal skills that foster positive dyadic effects on preventing and coping with depression need to be incorporated into psychological counselling services to increase perceived preparedness for discharge.

Findings revealed uncertainty as a common psychological stressor for both people with stroke and caregivers. Uncertainty in illness refers to the inability to determine the meaning of illness-related events or accurately predict outcomes (Mishel, 1990). As found in this and previous studies, uncertainty could be heightened by a lack of information and the unpredictability of poststroke outcomes (Ramazanu et al., 2021). In the reconceptualisation of uncertainty in illness theory, Mishel (1990) advocated that uncertainty could become an

opportunity for the individual to reconstruct a new view of life by developing a probabilistic thinking on uncertainty. Probabilistic thinking considers uncertainty as natural and part of reality (Mishel, 1990). By assessing their learning needs and implementing tailored health education for stroke dyads, nurses can help people with stroke and caregivers to maintain probabilistic thinking and manage uncertainty (Ashcraft et al., 2021). Providing dyads with evidence-based information on poststroke functional ability improvement rates and the reduction and/or prevention rates of stroke-related complications, medication side effects, acute episodes of chronic diseases and secondary stroke will not only reduce uncertainty-induced psychological stress, but also motivate dyads to actively participate in and take responsibility for post-discharge care. A nurse-led intervention encompassing informational support, shared decision making, and practical skills showed preliminary benefits for stroke couples to cope with uncertainty (Ramazanu et al., 2021).

Up to 56 % of people with stroke also live with one or more chronic diseases (e.g., hypertension, diabetes) and the stroke recurrence rate one year after the first-ever stroke can be as high as 25% in stroke survivors (Feigin et al., 2022; Lin et al., 2021). While an inpatient, nurses spend far more time with people with stroke than other health professionals. Nurses are therefore well-positioned to identify high-risk patients and enhance the dyad's preparedness to manage stroke-related complications, pre-existing comorbidities and secondary stroke prevention through health education. Health education interventions empower dyads at home to proactively monitor stroke risk factors, take action to stabilise chronic conditions and seek timely medical advice. Adequate discharge preparation, structured post-discharge follow-up and early evaluation by specialist nurses could reduce unplanned hospital admissions due to complications (Lin et al., 2022; Pugh et al., 2022).

Our study confirmed previous studies that people with stroke and caregivers are not consistently engaged in discharge planning although they have strong desires to be equal partners (L. Chen et al., 2021; Krishnan et al., 2019). Co-design in healthcare is defined as a partnership between patients, their family/caregivers and health service providers to develop a person-centred care plan/intervention with the aim of achieving better outcomes or

improved efficiency (Ward et al., 2018). Engaging people with stroke and their caregivers in co-designing poststroke care leads to better care outcomes (Jones et al., 2020; Lobo et al., 2021). Increased understanding of stroke dyads interdependent nature in post-discharge care would encourage stroke nurses to lead the revision of conventional practice in discharge planning and advocate for a co-design approach. Any innovative discharge planning practice needs to engage both people with stroke and caregivers in assessment and care plans that address the dyads mutual concerns as well as acknowledging individual needs (Mou et al., 2021).

6.5.6.1 Limitations

This study has limitations. The findings may not be generalisable due to the nature of the hermeneutic study, which is built on specific social and cultural contexts, but they may be transferrable to similar social contexts. Additionally, some participants with stroke in this study had mild cognitive impairment. Findings do not represent those with moderate to severe cognitive impairment. Moreover, the recall bias may impact on narratives from two dyads interviewed after discharge. Furthermore, dyadic relationship may influence narratives of dyads interviewed together. Finally, this study excluded non-English speakers and findings may not represent the perceptions of those from ethnic minority groups.

6.5.7 Conclusion

This study developed new knowledge on the shared experience, similarities and differences of perceived preparedness to manage post-discharge stroke care between people with stroke and caregivers. The use of Gadamer's philosophical hermeneutics enabled us to reach new insights into the co-existence of psychological stress and resilience, sense of certainty and uncertainty in people with stroke and caregivers and the different factors that contributed to these conditions in the dyadic members. The research methodology also enabled us to explore the expectations of the stroke dyads for co-developing and co-implementing discharge plans to meet their care and information provision needs.

Findings strongly suggest the review and revision of conventional discharge planning guidelines in a healthcare system using a co-design approach by inviting stroke dyads to participate in the process. Any innovative discharge planning practices for these populations should address dyad concerns and consider dyadic effects in health promotion and post-discharge stroke care while acknowledging individual needs of people with stroke and their caregivers. Stroke nurses and nurse practitioners are in an ideal position to lead and coordinate a multidisciplinary team to implement discharge planning. A stroke nurse coordinator's role should be developed to facilitate early identification of high-risk patients with multiple chronic conditions, support stroke patient-caregiver dyads to navigate the complex healthcare system through the continuum of hospital to home transition and achieve stabilisation of home care.

Anonymised Conflict of Interest Statement

No conflict of interest has been declared by the authors.

This is the end of the publication.

6.6 Summary

This chapter has outlined the demographic characteristics of the stroke dyads and health professionals who participated in the study and has outlined the major themes identified from interviews with the former and focus groups with the latter. In addition, this chapter detailed the findings generated from the interviews with stroke dyads before hospital discharge.

The first set of interviews of this longitudinal study explored the shared experiences of stroke dyads in terms of their preparedness for managing post-discharge care. The findings highlighted the unique challenges that people with stroke and their caregivers experienced individually and as a dyad before transitioning home from hospital. Their perceived preparedness in managing post-discharge care was affected by their psychological and emotional status, resilience and level of engagement in discharge planning. People with stroke and their caregivers manifested distinct feelings about returning home from hospital. Although they both viewed homecoming as a significant success in post-stroke recovery, caregivers perceived, in comparison to people with stroke, profound uncertainty as to how to assume the caregiving role.

The next two chapters (Chapters 7 and 8) report findings from the interviews of dyads at 3 and 6 months following hospital discharge.

Chapter 7: Findings—Part 2: Themes From the Follow-Up Interviews With Stroke Dyads

7.1 Introduction

As discussed in the previous chapter, people with stroke and their caregivers experienced significant challenges before discharge both individually and dyadically. Despite their optimism about returning home from hospital, they experienced psychological stress. People with stroke were frustrated about their loss of function and were worried about the prognosis and future independence, whereas caregivers perceived insufficient time to take on the caregiving role. Caregivers also doubted their capabilities to manage post-stroke care at home.

The dyads were followed up for 6 months after discharge. They were encouraged to record their experiences in managing care at home using a structured diary (see Appendix 21) and to participate in interviews at 3 and 6 months after discharge. The findings about their experiences in hospital-to-home transition care are reported in this chapter and the next chapter. The findings include four major themes and 10 subthemes (Table 7.1). This chapter reports the first two major themes: coping with setbacks and role transformation. The other two main themes—transitioning with support and expectations in hospital-to-home transition care are reported in the next chapter.

Table 7.1

Theme	Subtheme	Chapter
4. Coping with setbacks	 Processing loss and grief in people with stroke 	Chapter 7
	2. Processing loss and grief in caregivers	

Major Themes and Subthemes in Chapters 7 and 8

Theme	Subtheme	Chapter
	3. Reconciling with resilience	
5. Role transformation	1. Impact of stroke-related limitations	Chapter 7
	on people with stroke	
	2. Impact of stroke-related limitations	
	on caregivers	
	3. Regaining control of life	-
6. Transitioning with	1. Support from family and friends	Chapter 8
support	2. Support from health professionals	
7. Expectations in	1. Enhanced discharge planning and	Chapter 8
hospital-to-home transition	preparation	
care	2. Flexibility of service provision	

The findings represented participants' experiences individually and dyadically regarding the challenges they encountered, efforts they made and support they received from family and those in the community when managing care at home after hospital discharge. The findings revealed new understandings of both positive and negative dyadic effects between people with stroke and their caregivers in home care settings after discharge from hospital. Dyadic members demonstrated various strategies for coping with setbacks and adapting to changes in their lives.

These findings were supported by data collected from interviews and structured diaries at two time points: 3 and 6 months after hospital discharge. The codes SS and SC are used to indicate quotations from a person with stroke and the stroke caregiver, respectively. The codes T1 and T2 are used to indicate quotations from interviews at 3 and 6 months following discharge, respectively.

7.2 Theme 4: Coping With Setbacks in Home Care

The full impact of stroke-related impairments on everyday life only became clear to stroke participants on their return home. Home was a very different environment from the carefully controlled and well-supported hospital environment. This theme reflects participants' perceptions of the impact of stroke on their psychological wellbeing and the resilience they demonstrated in reconciling and moving forwards with the changes in their lives in the 6 months after hospital discharge. Subtheme 1 revealed that people with stroke experienced setbacks in functional recovery that triggered their psychological stress. Subtheme 2 highlighted that psychological stress in people with stroke triggered psychological stress in their caregivers. Stroke dyads demonstrated resilience and coping strategies that enabled them to reconcile with and adapt to their pre-stroke lives.

7.2.1 Subtheme 1: Processing Loss and Grief in People With Stroke

Unlike the anticipation of getting on with a life at home that stroke participants perceived before hospital discharge (Chapter 6), participants were disappointed by the fact that their expected functional improvement did not occur. They further experienced a profound sense of loss and grief in the home care environment that had a negative impact on their psychological wellbeing. However, they rarely mentioned the impact that caregivers and health professionals had on them when dealing with the setbacks. The findings are detailed in the following sections.

Participants felt a sense of setback when they did not achieve the recovery goals they had set before discharge: 'I thought I'd come home and 3 or 4 weeks I'd be doing everything normally. But it hasn't happened that way now' (SS02, T2). The setbacks triggered their frustration throughout the transitional care period: 'What I find a little bit frustrating is I expected to get to continue on another curve [of improvement]. But it's flattened out and nothing much has changed' (SS07, T2). The setbacks to recovery also contributed to their grief on the loss of independence brought about by stroke: 'It's pretty hard to take that independency away from me that all that's gone now to a degree because I

can't drive anymore' (SS04, T1). The loss of independence also contributed to the perception of the loss of their usual role and responsibilities in a family: 'I've suddenly felt outside my family. I'm not important to anyone anymore' (SS04, T1). Without proper interaction, situations like this would result in a sense of loneliness in their own family for people with stroke, although they might have had family members around them. Loneliness in people with stroke not only affected their mental health but also that of their caregivers due to the dyadic effects. Therefore, the need to prepare caregivers to engage their care recipients in meaningful activities in the family is evident for hospital-to-home transition care.

Stroke participants also shared their emotional reactions towards the setbacks in post-stroke recovery. These included feelings of resentment and fear, being teary and having negative thoughts about their future. For example, one participant felt that 'the living situation is worse because I get angry quickly ... I get emotional, teary ... Just a thought. My grandson's in Grade 3. Am I going to see him when he is in Grade 4?' (SS08, T2). Some participants described their fear of having another stroke: 'I'm afraid to get into my bed again because that's where the stroke happened. I sleep in the lounge chair every night, which is not really the best thing to do' (SS05, T1). This case revealed a possible lack of knowledge in preventing secondary stroke that resulted in behavioural changes in people with stroke. Preparing people with stroke with knowledge about stroke prevention in the transitional care period is a way to reduce their psychological stress and change their behaviour while engaging them in self-care. Such an education strategy would reduce burdens for caregivers, considering the dyadic effects.

It was evident that unrealistic recovery goals contributed to the feeling of setbacks and triggered psychological stress in people with stroke in the first 6 months after returning home. Such situations underscored the need to support people with stroke to develop achievable and realistic recovery goals throughout the transitional care period. Considering the dyadic effects, caregivers also need to be included in the discussions of the recovery goals to foster their positive impact on the care recipients.

7.2.2 Subtheme 2: Processing Loss and Grief in Caregivers

Caregivers also experienced setbacks in taking on the caregiving role. Returning home after stroke seemed to have provided caregivers something to look forward to (Chapter 6). However, when faced with the changes in a real-life situation, caregivers were overwhelmed and did not seem to find hope and positivity in their situation. They experienced loss and grief due to their loved ones' health conditions and the changes in the lives in the family. One caregiver revealed her sense of loss and grief at the 3-month interview:

His speech was shocking, his cognitive skills were very bad. So, I was living with half a shell of a man, which it becomes very, very stressful because you've lost your partner, your confidant, your husband, your friend, it's all gone, and you have to adjust to that. That's very, very difficult. (SC08, T1)

After hospital discharge, caregivers were faced with not only practical care activities but also significant psychological adjustment to develop strategies for living with their loved ones with stroke-related impairments, especially cognitive and speech deficits. It cannot be overemphasised that service providers need to continue to monitor caregivers' care needs and identify psychological needs timely after discharge from hospital so that rehabilitation teams and community service providers can continue addressing these concerns and develop a care plan specific to individual life situations.

The sense of loss was particularly significant in couples who used to work closely in everyday life before the stroke. Grief, to an extent, seemed to have left caregivers in despair, as one young caregiver shared: 'I do grieve the fact that XX [husband] has changed so much. Because we used to have lots of talks and conversations. I want to have hope, but it's quite hard; I do feel quite alone at times' (SC05, T1). In this case, tailored interventions to address the psychological care needs of caregivers would be warranted for community care providers when developing care plans for people with stroke. Counselling services may improve the feeling of helplessness and further support the development of resilience.

The unpreparedness to assume the caregiver role before hospital discharge also contributed to the feeling of setbacks, loss and grief that had a negative impact on the psychological wellbeing of caregivers in the follow-up period. At the 3-month follow-up, their stress seemed to present in the form of anger and resentment when talking about the changes in their life since their loved ones returned home. This was reflected in a wife's explanation of her mental state: 'I just felt he [husband] consumes me. I have no time for myself ... I struggled with resentment of my own emotions. Just the guilt I felt the whole time. I thought one time if he died it would be better' (SC03, T1). The caregivers' psychological and emotional distress seemed to be intensified after taking over care from professionals. Caregivers would benefit from a thorough review of their psychosocial needs by community service providers so that a tailored plan targeting each caregiver's specific circumstances could be put in place to provide emotional support and address some concerns that might not have been identified before hospital discharge.

At the 6-month follow-up point, people with stroke demonstrated functional improvements, including in cognition and speech. While functional improvement signified a milestone for people with stroke, the anxiety and fear that came with it complicated the transition for caregivers. One wife described mixed feelings regarding her husband's improved cognition:

XX [husband]'s brain function has improved. But with that, his cognitive understanding of where he's at has made him quite depressed. He feels anxious that he's never going to get better. He's scared that this family is going to abandon him when I had to work extra hours and gradually back to normal working hours. So, that was very stressful. (SC05, T2)

In this case, the dyadic effect was evident. The mental state of people with stroke triggered psychological stress for caregivers. Understanding such dyadic effects would help nurses and other stroke care professionals to better support caregivers and those they care for.

As demonstrated in these findings, stroke-related disabilities in people with stroke appeared to be a significant impact on their caregivers' psychological stress. However, people with stroke rarely described the impact of their caregivers' psychological stress on them. Interventions in hospital and continued psychological support in the community for caregivers would not only empower them to cope with stress but also help their care recipients to deal with setbacks and the sense of loss and grief.

7.2.3 Subtheme 3: Reconciling With Resilience

Despite setbacks and the sense of loss and grief, stroke dyads seemed to actively seek strategies individually and dyadically to cope with stress, manage their negative feelings and work with stroke-related limitations. It was evident that time and functional improvement for people with stroke had helped with the adaptation process. Furthermore, participants perceived dyadic effects in the psychological adaptation process in the home care setting, as detailed in the following sections.

7.2.3.1 Resilience in people with stroke

Most stroke participants demonstrated resilience, which was evidenced by their acceptance of and positive thoughts about stroke-related changes and their determination to find ways to resume everyday life activities. While adapting to changes, participants persevered with strategies that could help them move forward. One participant expressed his gratitude that he was 'lucky to be still alive and can still participate' (SS04, T1). This was echoed by another participant, despite his disappointment regarding his progress 6 months after discharge: 'It is not changing; I got to live with it. It is not ideal, but I can walk. Well, I see people that are in wheelchairs or people who can't move legs and say to myself, I'm lucky' (SS07, T2).

People with stroke demonstrated an ability to identify positivity when faced with challenges. In the process of working together with them to achieve their rehabilitation goals, recognition and encouragement from health professionals may support the cultivation of positive thoughts.

Participants discussed the positive thoughts that enabled them to accept strokerelated limitations and adjust to these changes. One participant with memory issues stated, 'some things I just keep forgotten. In the long run I thought I'm better off just being cheerful and getting on with it. And I'm not feeling sorry for myself and negotiating everything if it comes up' (SS10, T2). Other participants reconciled themselves with their limitations. The statement from a participant who was living with stroke-related fatigue and sensation changes exemplified this: 'I think probably more coming to terms with the fact that this is what it's like ... the level of getting better is probably very little increase; however, the acceptance of it being as it is, is probably more accepted' (SS08, T2). Reconciliation with stroke-related physical and emotional limitations seemed to help participants leave behind the stroke event and approach the post-stroke adjustment realistically.

With time, participants developed strategies that enabled them to acknowledge limitations while focusing on what they could do. One participant discussed how he had to adapt to balance and dexterity issues: 'I won't say I have adapted. I'm still adapting and adjusting. I am looking at my improvement is that I can work the heights and multitask' (SS02, T2). Similarly, when reflecting on what made him persevere and move forward in spite of stroke-related limitations, one participant identified silver linings in the midst of adversity: 'But my life has changed because I don't want to go back to the old [unhealthy] life. People saying now how much weight I've lost ... Recovery from stroke had also shown me a way that I could be healthier' (SS12, T2).

Over time, people with stroke demonstrated resilience in the presence of challenges brought on by stroke through reappraising their functions, reinforcing positive thoughts, reconciling with limitations and identifying positivity out of negative situations. Resilience enabled people with stroke to focus on the priorities in the transition and attempt to resume pre-stroke life. These findings may inspire health professionals to consider strategies to instil hope and provide support, validation and guidance in their interactions with people with stroke to activate and enhance their self-management capabilities.

7.2.3.2 Resilience in caregivers

Caregivers also demonstrated resilience and talked about strategies to help them reconcile with their changed lives. Their resilience seemed to be enhanced with time, the functional improvement of their loved ones, spiritual beliefs and past life experiences. Three months after hospital discharge, caregivers expressed their gradual acceptance of a changed post-stroke life. They identified even the smallest functional improvement in their loved ones, which played a part in their acceptance. As one wife shared:

It's only just now you accept what's happened ... You do accept because my husband will never speak like he did, but his speech is getting better, and he can have a little conversation ... The grieving and acceptance is better as he's getting better. (SC08, T1)

The findings implied that both people with stroke and their caregivers have positive dyadic effects in their approach to managing stroke-related limitations and enhancing resilience.

The approach to the changed life, coupled with past life experience, helped caregivers to cope with the challenges during the hospital-to-home transition: 'My life has not always been easy. So that's why I am accepting. I look at my life now, it's not bad, but it's different. And I'm just going to live a little bit of a quieter life' (SC08, T2). In this case, life experience and personal value seemed to act as sources of strength to enhance caregivers' resilience, which gave them the courage to continue to support their loved ones. A lack of this may hinder an adjustment and return to pre-stroke life for the dyads.

Some caregivers had not quite accepted the changes yet, even at the 6-month follow-up. However, they exhibited the ability to live with uncertainty and 'just expecting the unexpected' (SC06, T1) and move on with life. As one caregiver's statement represents:

I have not completely accepted [stroke]. I am wanting him back to the way he was. And I've got to keep saying the same when he wants more of my time: 'You've got to do that [exercises]' ... My mind is at the painting while listening to him talking about cars. That's just something that I've got to live with. (SC11, T2)

Viewing uncertainty as part of life may symbolise caregivers' ability to live with it. As presented in this case, caregivers seemed to have developed strategies to balance their own needs and those of their loved ones. These strategies may enable them to continue supporting people with stroke and become new ways to adapt to stroke-related limitations and better sustain the caregiving role.

During the first 6 months following hospital discharge, caregivers were in the process of adapting to their changed lives and reconciling the present with the pre-stroke life. With time and functional improvement in their loved ones, they developed coping strategies and enhanced resilience to face the challenges. Positive dyadic effects were evident in the enhanced resilience in stroke dyads and functional improvements in people with stroke. These findings encourage stroke nurses and rehabilitation teams to formulate interventions targeting individuals' specific life situations, to guide them in balancing expectations and the capacity to optimise their relationship, and to strengthen the adaptation.

7.3 Theme 5: Role Transformation

This theme focuses on stroke participants' self-management development with the support from their caregivers and on caregivers' evolving ability to build capabilities in collaboration with people with stroke despite being confronted with physical, communication and cognitive limitations and psychological distress. The dyadic effects in dealing with challenges enabled them to transform and adapt to their post-stroke lives. Stroke participants demonstrated their ability to accept stroke-related limitations, create new ways to work with these limitations, and regain a sense of control of health, wellbeing and social integration. Stroke caregivers appeared to be in the process of gradually transforming and settling into a new role in the family and community. They also discussed what could have been done to support the development of self-management and caregiving capabilities.

7.3.1 Subtheme 1: Impact of Stroke-Related Limitations on People With Stroke

This subtheme illustrates the challenges related to various stroke-related limitations experienced by people with stroke, especially at the early stage of returning home. Prior to

hospital discharge (Chapter 6), they might have already expected a different life after stroke. However, it was when they encountered these differences in real-life daily activities that they realised the implications of these limitations.

When they first came home, some stroke participants perceived home as a strange place and experienced distinctive differences between their expectations of their functioning and their actual status in the home environment. A participant described her feeling when she first arrived home after a long hospital admission: 'You think, if I get home, everything is going to be normal, but it's a bit of a shock when you get home and discover that you can't move around like before, you can't do the things like before' (SS06, T1). The unexpected nature of stroke, functional impairments and length of hospital stay complicated transitioning home for people with stroke. A health-professional-supported home visit, with trials of daily activities in the home environment before discharge from hospital, may help put in place some strategies and alleviate anxiety and frustration at the initial homecoming stage.

Stroke participants experienced difficulties affecting their participation in daily activities. For example, a woman who was independent with ADLs found her 'energy level gets smashed' (SS08, T1); she had to break down her tasks, and everything became slower. Another participant was 'very depressed and unhappy' (SS05, T1) that she needed antidepressants to continue participation in telerehabilitation at home. Stroke participants would benefit from education about stroke-related complications and the implications for their daily activities. Strategies such as energy conservation and managing negative emotions may foster the development of self-management capabilities.

Stroke participants were not only confronted with the impact of these limitations in self-care activities, but they also expressed concerns about the impact of stroke on their caregivers. Their concerns extended over the 6 months after hospital discharge. One stroke participant emphasised the need for support for caregivers: 'Me coming home was an enormous challenge and burden for her [wife]. While the focus was on me, she probably needed emotional support. Some support on really what is the impact to the partner when the patient comes home' (SS12, T2). This case implied that a stroke participant's perception

of their caregiver's emotional distress and caregiving burden being ignored by health professionals could distract those people with stroke from concentrating on developing selfmanagement capabilities during the transition. This highlights the interdependence within a stroke dyad and the possible dyadic effects that could promote or hinder the progression of regaining control of life.

Resuming driving was an important symbol of regaining control of post-stroke life. Several stroke participants expressed their frustration at the prolonged and seemingly incomprehensible assessment process to regain their driver's license. They understood the need to suspend their license after stroke, but they did not seem to know the assessment process and felt they were expecting the unexpected in the process. A participant's statement represented this: 'The issue is that you are not kept informed what's going on. It's just like the process of getting a license again. The process was unknown to me ... I know the principle, but nobody told us that' (SS07, T2). Stroke participants' experiences indicated a lack of adequate information provision before hospital discharge. Comprehensive processes, like the assessment for a driver's license, should be explained with a step-bystep guide that considers an individual's stroke-related limitations. This may alleviate stress when people with stroke need to face it in the community.

Perceived stroke-related limitations and their impact on the life of people with stroke was complex. Physical and psychological impairments limited their participation in self-care and home-based rehabilitation programs. Before they returned home, people with stroke were probably mentally unprepared as to the extent of the impact of these limitations on their life, especially the impact on their caregivers. The perceived unmet emotional needs of stroke caregivers may hinder the concentration of people with stroke on developing self-management capacities. The in-depth insights of health professionals into the complexity of the perception of people with stroke can help in developing strategies for discharge planning and for the process of delivering home-based or community rehabilitation programs.

7.3.2 Subtheme 2: Impact of Stroke-Related Limitations on Caregivers

People with stroke returned home with varied levels of physical, emotional and cognitive limitations that caregivers must manage together with them. Caregivers revealed inadequate knowledge and skills to deal with these limitations, which placed people with stroke at risk of adverse events, such as rehospitalisation, and placed caregivers at risk of mental stress. For example, a caregiver brother was concerned about stroke-related incontinence that was neither managed before hospital discharge nor reviewed in the community. He felt helpless because he was unable to support his sister: 'Since she's had the stroke, she is suffering from incontinence. No one has done anything about it, which I find it frustrating. Because she's got it and cannot control it, she suffers from depression so badly. Nobody's helping her' (SC10, T2). This case demonstrated that a stroke participant's physical limitations and mental state could negatively affect the caregiver's caregiving experience. Understanding the interdependent interaction within a stroke dyad and the factors contributing to positive or negative dyadic effects would inspire stroke nurses and rehabilitation teams to co-develop care plans that can be implemented in the community and promote self-management for both people with stroke and their caregivers.

This sense of unpreparedness was echoed by another caregiver caring for her mother, who was on a modified diet:

We found it very hard when Mum first came home because she was on a soft moist diet. The speech pathologist went through what Mum could and couldn't have. But Mum didn't know what to have, so it was a lot for us to work out what she can eat. (SC02, T1)

This demonstrated a lack of effective pre-discharge preparation for both people with stroke and their caregivers regarding the knowledge about stroke complications and the management of these complications. They might have received education and training on caregiving, but the evaluation of effectiveness seemed opaque. Gaps in assessment and communication between care settings were evident, which highlighted the significance of a

holistic approach from nurses and rehabilitation clinicians to care needs identification and intervention before hospital discharge.

Apart from dealing with visible stroke-related limitations for people with stroke, such as walking and speech difficulties, incontinence and the requirement for a modified diet, caregivers were also faced with challenges from invisible limitations, such as fatigue, emotional lability, memory issues and behavioural changes for people with stroke. A wife caregiver shared her struggle with her husband's memory impairment:

I've really struggled with the fact that his [husband] memory has interrupted his ability to see how he's improving ... He doesn't see the whole idea of being dependent and me offering my observations and giving him feedback—and so we've had to work through a lot of things but having the counselling once a week has really helped. (SC01, T1)

Since the onset of stroke and before hospital discharge, caregivers might have expected difficulties managing stroke-related limitations. These limitations only became more prominent and posed challenges when they were manifested in real-life situations. As applauded by the caregiver in the above example, counselling services were essential for supporting caregivers in managing their own emotions and in developing strategies to cope with the changed cognitive state of people with stroke. Psychological support in the community may also help prevent caregivers from burnout.

Intimacy and sexuality form an important part of quality of life. Satisfaction in intimate relationships may be impacted by stroke-related sequelae. This seemed to have complicated an already challenging transition time for caregivers, as one wife expressed: 'Everything was breaking down. His confidence took a real blow. Life now is pretty sexless—we had a fairly healthy sex life before. We were very confident with each other. He's just lost that. He's a lot less patient' (SC03, T1). Stroke dyads may experience relationship challenges due to a changed level of intimacy caused by stroke-related limitations. In this case, the interdependence within the dyad may create unhealthy dyadic coping that challenges the dyad's adjustment to their changed roles in their relationship. This suggests that there is a

need to include topics about post-stroke intimacy and dyadic relationships in discharge planning and stroke education so that stroke dyads understand the potential impacts of stroke on their relationships and can co-develop interventions to prevent deterioration and enhance their relationships.

All caregivers in this study talked about how busy they became after stroke throughout the whole hospital-to-home transition period, especially at the early stage of the transition. They had to juggle different roles and multiple tasks to maintain some pre-existing routines and to motivate people with stroke to continue with rehabilitation activities. A couple of caregivers described the situation at different time points:

First 6 weeks I think—there's been shifts and changes. Initially, it was just really hard work. I felt like I was battling against time the whole time because I wanted time for XX [husband] to practice those things that are difficult for him to do, such as processing information. (SC01, T1)

Life has picked up a much faster pace. Because XX [husband] has increased the amount of therapy. He's going to hydro three times a week, gym four times a week. He's got appointments. It's been hard just juggling these. And on top of that, I've got teenagers. (SC05, T2)

Findings demonstrated that some care needs may be addressed before discharge through education and hands-on caregiver training. Other care needs may emerge and change with time during the transition. Closing the gaps in care needs identification and support between the hospital and the community may be achieved through the early engagement of caregivers with discharge planning in the hospital setting and continued involvement with community service providers. It was clear in the findings that, despite the stress, caregivers were able to find ways to adapt to the changed life. Support and guidance from nurses and other health professionals may ease the intensity of care during the transition.

Echoing people with stroke, at the 6-month follow-up point, caregivers expressed different views about post-stroke progress in people with stroke and their adaptation

approaches. Six months after hospital discharge, people with stroke made further progress in functional abilities. The progress seemed to have slowed down and plateaued to a level that both people with stroke and their caregivers had to face and accept. One wife expressed her concerns regarding her husband's mental state: 'XX [husband] hasn't really gone forward, and things have been fairly flat. He's very keen to be independent. But he's not independent' (SC11, T2). With functional improvement in people with stroke, stroke dyads may have established daily routines and resumed some pre-stroke activities, but the feeling of helplessness continued to confront caregivers, as one wife shared: 'So, my life is back. But I haven't got a partner I had before stroke. The responsibility is yours. That's the transition that takes time. It comes with anger, sorrow, grieving because that word "carer", I don't like it' (SC08, T2).

The journey to manage stroke-related limitations for caregivers was filled with strength, resilience, stress and sadness. Caregivers had to manage their own feelings as well as be the resource person for people with stroke to encourage and motivate them through the transition. Understanding the differences in managing stroke-related challenges for stroke dyads would encourage nurses and other health professionals to shift focus from functional recovery in people with stroke to an encompassing view of the transition for the dyad. Health professionals' appreciation of dyadic effects in dyads' coping with the changed life would enhance their engagement with stroke dyads, treating them as a unit of care so that outcomes of both dyad members could be optimised during the transition.

7.3.3 Subtheme 3: Regaining Control of Life

This subtheme presents the strategies that stroke dyads developed to meet the challenges of stroke-related limitations and changed roles over time. Dyads were in the process of balancing expectations and capacities and had achieved different levels of adaptation. Positive dyadic effects between stroke dyad members were reflected in their ability to work together and achieve functional improvement.

With time, stroke participants developed their own strategies to enable pre-stroke activities. As one participant indicated, 'the best physio I've had is actually coming home'

(SS02, T1). He shared his experience of working out specific ways to complete a task: 'Picking up an electric drill, I have to think through new processes because previously I'd just pick it up and drill. Now I have to put my body behind the drill because I don't have the strength and dexterity' (SS02, T1). Finding alternative ways to accommodate stroke-related limitations signified the development of self-management and adaptation.

To achieve their goals of resuming pre-stroke activities, participants not only tried different ways to adjust, but they also explored and sourced services that met their own needs. One participant shared her story of finding a neurophysiotherapist who made a difference to her physical improvement:

I did get to see my first good physiotherapist 3 weeks ago. That has made a great difference. It was very hard finding one ... Now I can climb up and down extremely steep stairs at home, which is a real landmark for me. (SS10, T2)

The initiatives from people with stroke to find services customised to their individual needs reflected their progress in post-stroke adaptation and the beginning of taking responsibility for their own health.

At the 6-month interview, a majority of participants described their physical and cognitive improvement and increased confidence in performing pre-stroke activities: 'My hand was not working well. Now it is fairly normal. My memory is much better—I am not mixed up with appointments anymore. I had trouble banking on that computer initially. I could do that right now' (SS05, T2). This case demonstrated that people with stroke developed confidence from re-evaluating their recovery over time, which enhanced their resilience and acted as an impetus for further rehabilitation. However, their progress and increased confidence may not have been possible without the support from their caregivers. One stroke participant attributed his post-stroke improvement to his own efforts and support from his wife:

My wellbeing is still very high. With the support from XX [wife], I maintain exercise regime in my room. I am walking longer distances now with assistance from family.

There has been some development with my vision. XX [wife] and I are positive about the way things are. (SS12, T2)

This example indicated that functional improvement and resilience in people with stroke would facilitate stroke dyads' reconciliation with stroke-related changes and the resumption of pre-stroke life. Positive dyadic effects between dyadic members could foster optimism and strengthen dyadic coping in the recovery process.

As summarised by a participant who returned to work after discharge, resuming prestroke activities was viewed as a significant achievement during the transition: 'I went from probably being very dependent on the great help I got there [workplace] to having to look around for and try and find other ways of doing things' (SS10, T2). It symbolised a new chapter in their lives as well as facilitating regaining control of their lives.

Findings showed that, with time, people with stroke adapted to the changed life during the transition. Acknowledging their emotional stress and physical and cognitive limitations, they focused on managing and learning how to live with the consequences of stroke, gaining confidence from improvements and maintaining optimism even when challenged with discrepancies between their desired functional gains and reality. Positive dyadic effects demonstrated during the recovery process seemed to have augmented their adaptation.

Caregiver participants reported that they assumed the caregiving role regardless of their different views about the changed role in life and their relationship with the person with stroke. Time, functional improvement in people with stroke, and improved caregiving skills seemed to have helped caregivers transform and increase their strength to look beyond the current situation.

Caregivers' views about their changed role were related to the type of stroke-related impairment in people with stroke. Caregivers for people with stroke who had cognitive or speech impairment and changed behaviours seemed to struggle with the caregiving role even 6 months after discharge. A wife who cared for her husband with significant communication deficits shared her struggles in the adaptation process:

I am a lucky carer. He could live on his own: just need somebody to come in every day to make sure everything was okay. But becoming a carer made me feel not good at all in the beginning because when you become a carer, you're totally responsible. When you have a partner, you share. So, all of a sudden, you can't share. (SC08, T2)

However, for people with stroke who had physical impairments with no or mild speech and cognitive deficits, caregivers seemed to have experienced positive dyadic effects and found it rewarding to care for their loved one, as shared by one husband:

It has been tiring and demanding, but it's also been very rewarding. I can see every day she makes progress. Every day she can do some little thing that she couldn't do the day before. She's been really focused and determined to get well. (SC16, T1)

Echoing the above spousal caregiver's experience, non-spousal caregivers also expressed their positive views about becoming a caregiver, as one daughter caregiver commented: 'Well, it's better in one way because I was lonely and on my own as well, and it's given me a purpose and given me things to do for my dad, and I don't mind' (SC07, T1).

Caregivers' perspectives on their changed role seemed to be associated with the status of the person with stroke. Understanding dyadic interdependence in stroke dyads and incorporating dyadic perspectives in support services design may facilitate the hospital-to-home transition.

Regardless of their views, all caregivers demonstrated the ability to renegotiate the new role with a changed mindset and through working together with the person with stroke. Over time, caregivers developed coping strategies: for example, doing exercises for physical fitness, taking up hobbies and chatting with friends. A couple of caregivers who looked after their spouses with memory and communication deficits shared how they managed to cope with all the changes:

I think generally I'm coping. I'm really grateful that I've used meditation and done a lot of personal growth work, especially in a time of crisis. It's basically breathing and

centring and letting go and practising mindfulness techniques ... and I suppose because of this and also my attitude has supported me. (SC01, T1)

I have got a theory that if I can do something about a problem, I do it. If I can't, I've got a rubbish bin over my shoulder, and I throw it in the rubbish bin and don't worry about it anymore. So, I have been just going along with the changes. (SC11, T2)

Caregivers' experiences in handling crises in the past seemed to have sustained them in the stroke event. They seemed to be able to utilise resources to develop and enhance resilience to ease the adjustment process. Listening to and understanding caregivers' sources of strength and pre-existing coping mechanisms would assist nurses and the stroke team in creating interventions to instil positivity and hope during the transition, which may have a far-reaching impact on caregivers' ability to sustain long-term care.

When people with stroke and their caregivers worked in partnership, the positive outcomes of their partnership not only encouraged both but also promoted their ability to return to pre-stroke life. One husband caregiver shared their achievement:

We've gotten deep into the garden down quite a steep path. I just stand there in case she stumbles so I can catch her. We think it's better for her to do it herself ... Except those stairs, there's almost nothing around the house that she used to do that she can't do now. So, I'm sure that's been a great boost to the morale of us. (SC16, T1)

Positive dyadic effects were evident when people with stroke and their caregivers worked together to achieve mutual goals, which seemed to have strengthened their relationship, deepened interactions and enhanced dyadic coping.

Caregivers reflected on the fact that the first few weeks of hospital discharge were much busier compared to the stages after that. Over time, they started to look into their own health and wellbeing:

It was getting me down about trying to get the housework done. Now I've realised that if it doesn't get done, don't worry about it. I'm not so hard on myself anymore.

You learnt along the way. We'll just have to accept how you are and what you can do. (SC18, T1)

Caregivers were in the process of balancing their capacity and expectations of how their life should be. Shifting the focus to people with stroke alone, to the realisation of caregivers' own wellbeing may pave ways for caregivers' transformation during the transition.

All caregivers who participated in both the 3- and 6-month interviews expressed their hope and determination to move forwards to a better future. Despite the uncertainty, they began to think about plans for the future or about strategies to sustain their life for the long term. A couple of wives shared their visions:

I said to him, getting away, that's something we've got to work together to sort out. We just got to try and find a happy medium so I can do what I want to do, and he can do what he wants to do. (SC03, T2)

I'm just trying to look at strategies going forward and not using the kids as my means because they'll burn out too. I am going to meet with the support coordinator to see if there are some other things that she would suggest making it work a little bit better. (SC05, T1)

Findings revealed that becoming a caregiver was a huge adjustment for relatives of people with stroke. Losing their identity in the family relationship, taking over full household responsibility and assuming the new role of caring for and motivating people with stroke—for caregivers, these all required both psychological and physical adaptation. Depending on individual circumstances, and at different stages of the hospital-to-home transition, caregivers began to adapt and regain control of their lives over time. Understanding the unique challenges of stroke caregivers and dyadic effects derived from dyadic interactions would enable stroke care providers to co-develop interventions with stroke dyads to strengthen caregivers' ability to adjust and ease the adaptation process.

7.4 Summary

Stroke dyads shared their experiences and views about the process of adapting to the changed life after hospital discharge. This chapter presented two themes and six subthemes focusing on the psychosocial and physical challenges encountered after hospital discharge and the efforts made to find ways to adapt to these changes. Over time, stroke dyads reflected on the impact of stroke; they moved on from the negative thoughts and initial shock when they first returned home. Additionally, interdependent interactions within stroke dyads produced dyadic effects that were closely associated with functional and psychological adjustment of the dyads and their capabilities for partnership to overcome everyday challenges.

One-way dyadic influence was observed in this study, in that caregivers' psychological stress seemed to be affected by the physical, mental and relationship status of the person with stroke, whereas the person with stroke rarely mentioned the caregivers' influence on them regarding psychological stress. Furthermore, they were at different levels in terms of regaining a sense of control through facing limitations with resilience, negotiating daily activities with self-developed strategies, and balancing capacities and expectations. Moreover, while both experienced a huge adaptation, people with stroke seemed to focus more on their physical recovery, while caregivers seemed to perceive the complex losses and reconciliation journey for achieving the adaptation.

The next chapter (Chapter 8) offers the perspectives of stroke dyads regarding support during the hospital-to-home transition and their expectations of post-stroke services during the transition.

CHAPTER 8: Findings—Part 3: Themes From the Follow-Up Interviews With Stroke Dyads (Continued)

8.1 Introduction

As discussed in the previous chapter, this study identified four major themes in relation to the experiences of stroke dyads in the 3- and 6-month follow-ups after hospital discharge. These themes are coping with setbacks, role transformation, transitioning with support, and expectations in hospital-to-home transition care. The first two themes (Themes 4 and 5) were presented in the previous chapter (Chapter 7); this chapter presents the last two themes (Themes 6 and 7). Both themes include subthemes, as shown in Table 8.1.

Table 8.1

Major Themes and Subthemes in Chapter 8

Theme	Subtheme
6. Transitioning with support	1. Support from family and friends
	2. Support from health professionals
7. Expectations in hospital-to-	1. Enhancement of discharge planning and
home transition care	preparation
	2. Flexibility of service provision

8.2 Theme 6: Transition With Support

Both people with stroke and their caregivers acknowledged that they were unable to achieve rehabilitation goals and adapt to the changed lives without support from family, friends and health professionals. Subtheme 1 revealed the indispensable role of family and friends throughout the hospital-to-home transition journey for stroke dyads, especially the considerable support people with stroke received from their caregivers. Positive effects arising from the support between dyadic members were identified. Subtheme 2 detailed

participants' perceptions about professional service provision that either enhanced or could work better to facilitate their transition.

8.2.1 Subtheme 1: Support From Family and Friends

All stroke participants perceived the critical role their families and friends, especially caregivers, played in supporting them to adapt to living with stroke-related disabilities at home in the first 6 months after hospital discharge. Positive dyadic effects arising from perceived support between the dyadic members fostered transitional care. However, caregivers were in the midst of developing their caregiving capability, and the support they received from other family members varied in the early stage of homecoming. Stroke dyads also received support from those in their social network, including friends, neighbours and colleagues, which had a positive impact on their recovery experiences and enabled them to cope with stress and self-care despite living with stroke-related disabilities. In addition, adequate community resources enhanced their transition experiences.

The support from caregivers and other family members encouraged people with stroke to improve self-management, as one stroke participant emphasised:

I love my wife and kids. I really understand more now post-stroke. XX [wife] is always there. My son and daughter drop in frequently ... They really inspire me. It's almost the case I owe them that I go in to work through this and come out on the other side. I am well on my way to doing that. (SS12, T2)

It is evident that perceived support from the caregiver and other family members motivated the person with stroke to achieve his rehabilitation goals and to adapt to strokerelated disabilities. The findings also indicated that stroke caregivers and other family members, as a care unit, played a critical role in supporting people with stroke.

Diary and interview data from participants revealed positive dyadic effects arising from caring, understanding and encouragement within the dyad. For example, a stroke participant reflected on the support from his wife in the diary at different time points. At Week 1 post-discharge, he wrote, 'XX [wife] is very understanding of my limitations. But she allows me to try things out' (Diary SS02, T1); At Week 2, he described, 'I am starting to do

small things in the kitchen. XX [wife] does not have to provide personal care now' (Diary SS02, T1). At the last interview, he said, 'I sometimes need motivation to do things. XX [wife] tries different ways to help me with this' (SS02, T2). Likewise, his wife was also inspired by his improvement, as the wife wrote in the diary at Week 4, 'I can see good and positive changes happening. I'm happy because he is not so negative' (Diary SC03 [wife of SS02], T1). These examples indicated that the dyad had a mutual understanding of stroke-related limitations and self-management goals. Moreover, they also demonstrated partnership in post-stroke recovery and care, which benefited their wellbeing.

However, each family is unique. Some participants perceived less support at the early stage of returning home than they had expected, for various reasons (e.g., the caregiver's advanced age). A stroke participant shared her perspectives on family support when she needed it the most upon arriving home from hospital. Her elderly husband caregiver was only able to provide minimal assistance, and there was no active home care service at the time:

My family have been lovely, but I think once you leave hospital, they think you're better [and do not need help] ... XX [husband] can't help much. I don't like to press the family; I know they're busy. My daughter works, and my daughter-in-law has also got her mother living next door, can't do [much for me]. (SS06, T1)

This case highlighted the significance of caregiver capacity assessment before hospital discharge. In some family situations, both the primary caregiver and other family members should be engaged in discharge planning to activate family support for stroke dyads during the transition.

People with stroke highly regarded the support from caregivers during the transition. They suggested that it is particularly important for health professionals to understand caregivers' needs and their preferred approach to address these needs during the transition:

I was prepared for that [discharge] physically ... [but] we are not prepared as patient, partner and even other members of the family ... challenges and responsibilities all of a sudden bestowed upon my wife, particularly she is employed ... There were

probably small incidents where we weren't all on the same page, and that can affect me as patient, and could also affect my principal carer. (SS12, T2)

Findings indicated that support from caregivers is imperative for people with stroke to cope with challenges in their reintegration into the community. However, other family members also play important roles in supporting stroke dyads in the transition. Perceived unpreparedness for stroke dyads and their family seemed to have intensified psychological stress for stroke dyads. Engaging all key family members, including principal caregivers, in care planning and decision-making before hospital discharge appeared to be essential to empower caregivers and other family members to appropriately support people with stroke.

Stroke caregivers in this study highlighted the support they received from people with stroke and other family members that was essential for alleviating stress and strengthening their caregiving capabilities during the transition. One caregiver, who cared for her husband with cognitive impairment, talked about the support from family that enabled her to sustain caregiving: 'I feel like I've been blessed with the people around me. When I am getting up towards a bit of overwhelm maybe, they come along and just take off things that they can, which have been really good' (SC01, T1). Another husband caregiver concurred and expressed his gratitude for the support he received from his wife (the person with stroke) and son:

She [wife] has never complained or been self-pitying. She's just focused on what the next step is ahead. Our son has also been splendid. He came over here from XX [city] and helped me a lot ... I'm optimistic. I hope and expect her [wife] to be with us for a good number of years. (SC16, T1)

As described in this example, optimism demonstrated by people with stroke, coupled with support from family and friends, seemed to have cultivated hope, strengthened caregiving capability and enhanced resilience in caregivers. This case also demonstrated positive dyadic effects in the dyad.

Caregivers also revealed that their ability to cope with daily activities and caregiving during the transition was impossible without support from friends, neighbours and

employers. They identified that 'neighbours are wonderful and have walked the dog for us' (SC16, T1) and that friends had been 'fantastic. From that point of view, we've had really good support' (SC06, T1). For caregivers who were still in the workforce, their employers also played an important role in supporting them in coping with the changes: 'I'm very tired, but coping. I think I have a very strong faith, along with the workplace that's very supportive' (SC05, T2).

The experiences described by the caregivers above indicated that the support from social networks helped them adapt to the new norm. The physical and emotional support not only helped alleviate some of the caregiving burden but also enhanced caregivers' resilience. Support from family and social networks enabled stroke dyads to cope with challenges and setbacks during the transition. It served as a protective barrier against threats to their wellbeing. Health professionals' knowledge of pre-stroke family dynamics, social support networks, the family's caregiving capabilities and potential dyadic effects would prompt a tailored care plan to support stroke dyads through the hospital-to-home transition.

8.2.2 Subtheme 2: Support From Health Professionals

Recovery from neurological injury following stroke is complex and influenced by individual situations. This requires stroke care service providers and health professionals to work collaboratively to apply collective knowledge and specialist skills for the benefit of stroke dyads. Stroke dyads perceived that support that focused on both dyadic members facilitated effective functional improvements for people with stroke and enabled psychosocial recovery for both of the dyad's members in the transition period.

All participants commended the support from health professionals who enabled the person with stroke to achieve goals of resuming pre-stroke daily activities and facilitated caregivers' adaptation. Compared to their perceptions at the initial homecoming stage, stroke participants gradually adapted to the 'new' home environment and daily activities at home. They perceived that they were 'getting more confident and much stronger' (SS09, T1) and continued rehabilitation helped them regain functions and maintain motivation: 'I'm

basically pretty lazy anyway, what I could've done extra on my own, I never did. I just did my exercises with them [therapists on iPad]. It got me to the stage where I could walk without the frame' (SS05, T1).

Rehabilitation was imperative in the first 6 months after hospital discharge. It not only helped improve physical functions, but it also strengthened and maintained participants' motivation to move on with their lives. Suspension of planned rehabilitation due to the global coronavirus pandemic clearly affected participants' desires for continued functional improvement: 'I'd been to the hospital [day rehabilitation centre]. I really came out with strength and determination. Then it was COVID. The rehab was deferred. I was a bit low; it was frustrating that I was marking time' (SS12, T2). This example indicated the need for innovation in community rehabilitation service delivery. Telehealth may have filled in the gaps during the pandemic by improving the accessibility of rehabilitation services and continuity of quality care for stroke dyads.

Stroke caregivers echoed stroke participants' positive views about health professionals who provided post-discharge rehabilitation programs. They considered that the continuity of rehabilitation motivated them: 'It gave her [person with stroke] some confidence to get up and go around and doing things she had to do, and I felt so relieved' (SC17, T2). Post-discharge rehabilitation follow-up also reassured caregivers:

I think when you're in the carer role ... initially it's kind of a scary outlook. These services help you in a lot of ways with that. They help balance your crazy thoughts at that time and offer those things that help settle you in that way, which I think is worth gold. (SC02, T2)

Continued engagement with caregivers after discharge and stroke participants' increased confidence and functional recovery seemed to have strengthened caregiving capabilities and promoted their adaptation during the transition.

In addition to the benefits brought by post-discharge rehabilitation programs, participants also discussed the support they received from community care services. The continuity of these care services enabled stroke dyads to focus on functional improvement,

psychological adjustment and self-management capability development. One stroke participant, who had carer responsibility for her daughter who had disability, commended the support she received from the service provider for her daughter:

When I was in hospital, they [service provider] put in extra support for XX [daughter]. Initially, when I came home from hospital, they actually came on Saturday/Sunday as well just to make sure I was okay to handle things ... now I'm at a stage to cut back. However, I could always call on them to help out again. (SS08, T1)

This case revealed that individualised care services and flexibility with service delivery offered reassurance as well as alleviated anxiety for people with stroke in case their health condition changed.

Continuity of community care services for people with stroke after completion of rehabilitation programs seemed to have also supported caregivers and enhanced their sense of certainty. One caregiver expressed his satisfaction:

[Rehabilitation therapists] were terrific. Then they handed us over to XX [community care service] ... They made regular phone calls to see how things are going ... it makes me feel we're in control of the situation; we're not waiting for something to happen unexpectedly. (SC06, T2)

Participants' experiences indicated that community care services played a crucial role in supporting stroke dyads through this challenging transition period. These services enabled stroke dyads to adapt to the changed life at their own pace and enhanced their self-management capabilities to improve independence and resume pre-stroke life.

When discussing service delivery, stroke dyads acknowledged the essential role health professionals played throughout their post-stroke recovery journey, especially during the hospital-to-home transition. Stroke participants were inspired by health professionals during therapy interactions. They commended the tailored approach of health professionals that carried them through the transition and helped them increase confidence and gradually resume meaningful activities and reintegrate into the community. One stroke participant

shared how the physiotherapist from the home rehabilitation service supported her to achieve her goals:

I said to him [physiotherapist], 'I want to get on the bus and go up to the city'. So, at the last session, we walked up to the bus stop, got on the bus, went to the city, and caught the bus back. We were really pleased that I'd really achieved what I wanted to do. (SS09, T1)

This participant's experience exemplified the positive outcomes generated through collaborative efforts between health professionals and people with stroke. Health professionals' partnership approach to service delivery appeared to have promoted a return to pre-stroke daily routines for people with stroke that helped improve their physical and psychological wellbeing.

Stroke participants' positive perspectives resulting from their interactions with health professionals were also reflected at the completion of a service program. Appropriate preparation focusing on developing self-management helped to maintain motivation for rehabilitation:

I knew it was time to move on. XX [therapist] and I discussed what I supposed to do when that rehab [program] stopped. He pointed out the benefits of continuing exercise. And the responsibility was shifting to me to maintain the programs. It, in a way, was a turning point. He also provided me with paperwork on organisations and some physios contact. (SS12, T1)

This example augmented the significance of self-management development guided by health professionals. Encouraging people with stroke to take an active role in their own rehabilitation seemed to motivate people with stroke and strengthen their self-management.

Caregivers' perceived engagement in the hospital-to-home transition process appeared to have increased their confidence to take over care from health professionals. One caregiver, caring for her husband with speech and cognitive impairment, described, 'they [rehabilitation professionals] were trying to give strategies to me to support XX [husband] at home. I felt respected and felt like I was an important part. There was a sharing
of information and collaboration' (SC01, T1). A focus by health professionals on caregiving capacity development would facilitate problem-solving during the transition and potentially improve caregiving sustainability.

Health professionals' compassionate approach in their interaction with stroke dyads was described by caregivers as 'encouraging that instilled confidence and helped him [husband with stroke] improve so much' (SC18, T1). Moreover, caregivers viewed the provision of tailored information by health professionals as educational in that it 'answered all of my questions about what's available and gave us tremendous information of things that we didn't know' (SC06, T1).

These caregiver perceptions further highlighted the benefits of stroke-dyad-focused care provision, which could potentially motivate stroke dyads, improve their physical and psychological wellbeing and enhance the hospital-to-home transition.

However, contrary to the above, participants also discussed what could have been done to facilitate their transition and areas for improvement to enable the building of selfmanagement capabilities. One stroke participant, who was involved in telerehabilitation after discharge decided to cancel the rehabilitation service, described her interactions with the therapist:

She [therapist] didn't really give me any exercises on iPad after the first session. She rang once again and said 'why can't you come down to the hospital'? I said I live in the hills and can't drive yet. She didn't ring again. I just didn't get any physio at all. And I really still needed it badly. (SS10, T1)

It was evident in this case that a partnership approach by health professionals would promote service utilisation. Assessment and knowledge of the specific situation of people with stroke would facilitate individualised rehabilitation intervention for them.

People with stroke who participated in rehabilitation programs aimed to achieve their personal goals. They perceived that, when rehabilitative interventions were delivered in a one-fits-all approach, the intervention was ineffective and resulted in frustration. A participant with reading and writing difficulties perceived that he was asked to read the same article

many times and write three main points over the 6 weeks of the program without receiving feedback on what he wrote: 'I didn't initially but tried to mention at the end. I think they understood what I was saying ... but one of them even said "No, got to tick this box". So, I thought this is just a ticking box exercise' (SS07, T1).

Participants' experiences emphasised the person-centredness and shared goals of service delivery. The approach and attitude of health professionals seemed to decide the level of engagement of people with stroke. The perceived box-ticking approach did not consider the differences of the individual and their needs. Declining therapy participation during the transition may affect the motivation of people with stroke and jeopardise the building of self-management capabilities.

Differences regarding perceived challenges working with health professionals existed between people with stroke and their caregivers. Difficulties experienced by people with stroke seemed to be associated with post-stroke rehabilitation therapy. For caregivers, when working with health professionals, their perceived challenges appeared to be related to information provision and the flexibility of service provision. Caregivers appreciated postdischarge support but found that 'we just got so much information. We understand that they [therapists] had a lot to do in a short time. But to try and process it all was really tricky for us' (SC02, T1). Information provision based on health professionals' workload rather than stroke dyads' needs and capacity may create stress and confusion for stroke dyads due to information overload. This not only defeats the purpose of building self-management capacity but may also intensify caregiver distress.

During the transition, caregivers experienced different levels of flexibility in service delivery, ranging from positive experiences—'they really give you the opportunity to make a time that you're available' (SC02, T1)—to less positive ones —'I don't think in the whole process that people understand just what's involved from the carers' perspective. For example, they would just expect you to be there the next day for an appointment' (SC05, T2). Caregivers' distinct experiences indicated inconsistent post-discharge service delivery.

A person-centred service delivery model appeared to ease anxiety and stress during the transition.

Findings highlighted that post-discharge rehabilitation and community care services were essential to enable the hospital-to-home transition for stroke dyads. Participants identified health professionals' central and critical role in supporting stroke dyads during the transition. The person-centred approach strengthened the confidence of stroke dyads and guided them to grow and accomplish taking over care from health professionals. Positive interactions with health professionals seemed to encourage and motivate people with stroke and enable caregivers to develop self-management capabilities. This further emphasised the significance of health professionals' involvement in the transformation of stroke dyads during the transition.

8.3 Theme 7: Expectations in Hospital-to-Home Transition Care

Reflecting on their hospital-to-home transition journey, study participants identified what could have been done to make their transition better. Subtheme 1 highlighted the significance of engaging both people with stroke and their caregivers, though especially caregivers, in the discharge planning process before returning home. It also highlighted that continuity of care could be enhanced by stroke-specific services, especially for those residing in regional areas. Subtheme 2 detailed the expectations of stroke dyads on the flexibility of service delivery in the community that could better support stroke dyads, especially caregivers, during the transition.

8.3.1 Subtheme 1: Enhancement of Discharge Planning and Preparation

After stroke dyads arrived home, they started to identify issues that could have been addressed before hospital discharge. They recognised that adequate preparation before discharge, with continued rehabilitation and care services, were essential to support their post-stroke adaptation and transformation. Participants discussed strategies that could better support them in coping with the transition, including information provision and selfcare and caregiving skills-building.

Stroke dyads expressed their preference for engaging in the discharge planning process so that they could feel prepared and informed for homecoming. They recognised that a lack of involvement could have a negative impact on their ability to adapt to stroke-related consequences. One stroke participant, who had continence issues complained, 'I'm just disappointed that I've been left with it [incontinence]. I don't understand why. I'm quite devastated. I was hardly at all [involved in discharge planning]. The [rehabilitation] goals weren't done until the day before I left the hospital' (SS05, T1). From the caregivers' perspective, involvement in discharge planning would help them organise life and prepare for the arrival of their loved ones. However, some caregivers were not adequately prepared with personal care skills. One caregiver, who needed to help her husband with personal care, stated, 'no, I did not have training. You just do it. I just accepted it and did it. We had a seat in the shower. So, it's just all been about him' (SC18, T1).

These examples reinforced that stroke dyads' full engagement in their own care and skills-building for ADLs were essential parts of discharge planning. Training and practising these activities, guided by health professionals, seemed to empower stroke dyads and ease some uncertainties after they returned to the community. This had a significant impact on their perceived capabilities to self-manage daily life situations.

Some stroke caregivers for people who had long hospital stays suggested that a meeting with the rehabilitation team at the early stage of admission and before discharge would have made them feel more engaged because 'after these meetings in the beginning, as time goes by, you sort of left to your own bosses and you just don't have that contact [with the team]' (SC18, T2). A second family meeting 'just gives a sense of what everybody thought was beyond the hospital and what to do to prepare for the coming discharge' (SC14 & 15, T2). Caregivers also indicated that family meetings on the day of discharge would not work because 'at the time you're trying to get them out [of hospital], you don't always think of the questions to ask' (SC02, T2).

The above suggestions illustrated the timing and format of communication and involvement in discharge preparation that could improve stroke dyads' transition experience.

Their suggestions further emphasised that discharge plans should be based on individual situations and tailored to specific needs—for example, people with long hospital stays may require comprehensive preparation before discharge.

Participants also highlighted that the content of information, and the approach to delivery, could be improved before hospital discharge, which may alleviate their sense of uncertainty. They desired a clear rehabilitation plan after discharge as well as a clear explanation of the person-specific challenges that they may face after returning home:

I just think there's not enough time or written material available to just spell out, 'hey, when the patient gets home, you're going to be confronted with this, this and this. You'll be [with] home rehab and then you're back in the day rehab'. So, I think that can be modified and inclusive, certainly more inclusive of the carer, the wife. (SS12, T1)

Conversely, some caregivers were overwhelmed by the quantity of information they received before discharge. As one caregiver, looking after her mother with swallowing difficulty, stated, 'it was good that they all came and spoke to us. But some of that food thing, which I find overwhelming. Maybe that's just a personal thing for me' (SC02, T2).

Findings revealed the unique information needs of stroke dyads that could be influenced by stroke-related impairments, the family situation or the caregiver's capacity to digest information. Insights into their preferred approach to information-sharing may guide health professionals to improve collaboration with stroke dyads in planning and preparing for discharge.

This subtheme reinforced the significance of health professionals' partnership during the post-stroke recovery journey for stroke dyads. Findings stressed stroke-dyad-centred discharge planning, self-management capability-building and information-sharing that could play essential roles in enhancing stroke dyads' hospital-to-home transition.

8.3.2 Subtheme 2: Flexibility of Service Delivery

During the first 6 months after hospital discharge, stroke dyads seemed to mainly focus on maintaining existing functions and making continuous improvements. However,

some participants perceived receiving inadequate rehabilitation and care services due to the unavailability of services at their residential address and the inflexible design of available services. Challenges arising from the inadequacies of services were perceived as barriers to a smooth hospital-to-home transition.

Participants who lived in regional areas considered themselves to be disadvantaged in accessing community services and follow-up rehabilitation programs after they completed formal post-discharge programs due to their location. The lack of stroke-specific services and support groups in regional suburbs seemed to have posed great barriers for them to make further progress in achieving their goals. One stroke participant discussed his desire to continue improving his speech and writing: 'There is nothing [services] around this area. The system seems to be all geared to metropolitan area' (SS07, T2). Another participant echoed the lack of community and rehabilitation services in regional areas: 'There wasn't community help or rehab here. I am still classified as city living. But when I spoke to XX [service provider], they said I am too far out. Not living close to town has made a big difference' (SS10, T2).

The findings unveiled gaps between hospital and home that may hinder the process of adjusting to stroke-related impairments and the desire to make further functional improvements for people with stroke during the transition. It was evident that participants expected service integration between hospital and the community and expected strokespecific services to be available in all residential locations after they completed formal rehabilitation.

Stroke dyads who had access to rehabilitation and care services perceived being overwhelmed by the approach of service delivery. Some felt the 'seamless' home rehabilitation service was 'far too early for anyone. We checked out of the hospital at 10 to 12; we had the first home visit at 12:30. You need to come home to settle first and not to have these visits' (SC08, T1). Additionally, the long waiting list for care services left stroke dyads, especially caregivers, devastated: 'she [wife] is qualified for a Level 4 [care package] and there is probably a 2–3 year waiting list. But we need it now! ... Even for a Level 2 in the

meantime, the waiting list is 6–12 months!' (SC06, T1). Furthermore, due to the timeframe of programs, stroke dyads felt overloaded: 'they [therapists] tried to integrate Mum into the home in those 2 weeks. It was a lot of effort for us. Mum found it hard' (SC02, T1). Moreover, some caregivers commented on the available rehabilitation interventions for people with stroke that 'could improve physical side. But very little on the mental side' (SC06, T2).

Findings call for service design and delivery to be flexible and to target individualised needs. Balancing service providers' imperatives and the capacity of care recipients may enable stroke dyads to maximise the benefits of care services at their own pace.

The majority of participants in this study desired a contact person after discharge. Some preferred an in-person visit; others preferred at least phone calls, emails or a contact number that they could call when needed. They perceived a contact person would be 'just to reassure you that you're doing the right to follow you in the right path' (SS06, T2), and 'if you're caring for somebody, sometimes you have a bit of self-doubt, or you need to discuss a change in something, you need that sounding board and somebody to talk about it' (SC02, T1).

Findings unfolded the complexity of post-stroke rehabilitation and care services in the community pertaining to the availability, accessibility and flexibility of these services and their impact on stroke dyads' hospital-to-home transition experience. It cannot be overemphasised that service providers' service design and delivery must be customised to individuals' life situations and care needs. Stroke dyads' expectations imply the need for a co-design approach in service development.

8.4 Summary

Stroke dyads reflected on the support they received during the transition that either empowered or negatively affected their adaptation to post-stroke life and their expectations of service provision. This chapter has presented two themes and four subthemes detailing the essential support stroke dyads received from family, friends, neighbours and employers.

Dyadic effects were evident between stroke dyad members supporting each other, generating positive perceptions of physical and psychological wellbeing. Furthermore, professional support from service providers facilitated their transition through collaboration and partnership. Moreover, stroke dyads identified issues during the transition that could be improved to strengthen their adaptation. They expected enhanced engagement in discharge planning and preparation and flexible service provision that considered their individual life circumstances and specific care needs.

The next chapter (Chapter 9) presents the perspectives of stroke health professionals regarding the hospital-to-home transition for stroke dyads.

Chapter 9: Findings—Part 4: Themes From Focus Groups and Interviews With Health Professionals

9.1 Introduction

The previous three chapters presented seven major themes pertinent to the hospitalto-home transition experience of stroke dyads. Themes identified before hospital discharge were psychological stress, resilience, and certainty and uncertainty about post-discharge care. Themes identified from the 3- and 6-month follow-ups after discharge were coping with setbacks, role transformation, transition with support, and expectations of hospital-to-home transition care. In the early stage of the transition, stroke dyads experienced psychological stress pertaining to stroke-related physical and cognitive limitations. However, with time and support, they gradually adapted to the changes and began to resume pre-stroke life.

This chapter presents findings from focus groups and interviews with stroke rehabilitation health professionals on factors affecting hospital-to-home transition care for stroke dyads. Participants in this study were recruited from both the inpatient rehabilitation wards and ambulatory services, including home-based and centre-based rehabilitation services. In total, 31 health professionals participated in this study, including rehabilitation allied health staff (n = 16), registered nurses (n = 6), service managers (n = 4), doctors (n = 3) and GPs (n = 2). The demographic information of participants is presented in Table 6.2 (Chapter 6).

Data analysis of interviews and focus groups identified three major themes: (1) transition shaped by care needs assessment, (2) transition challenged by constraints on service delivery and (3) transition enhanced by a commitment to the safety of care. These themes and their subthemes are listed in Table 9.1.

Table 9.1

Major	Themes a	and Subtheme	əs in	Chapter 9)
-------	----------	--------------	-------	-----------	---

Theme	Subtheme		
1. Transition shaped by care	1. 'Psychological/mental health is underplayed'		
needs assessment	2. 'Our focus is on the patient'		
2. Transition challenged by	1. 'Bed pressures certainly impacts'		
constraints on service delivery	2. 'We are dictated by funding models'		
3. Transition enhanced by	1. 'Empowering them to making decisions'		
commitment to the safety of care	2. 'Working towards a common goal'		

9.2 Theme 1: Transition Shaped by Care Needs Assessment

Health professionals perceived stroke care needs assessment as the foundation of a rehabilitation program that helped realistic goal-setting for rehabilitation, determined the length of stay for people with stroke in the program and identified the support that was needed for stroke dyads on the completion of the program. However, rehabilitation care needs assessment mainly focused on the physical and cognitive functions of people with stroke. Psychological needs assessment was largely overlooked (Subtheme 1). Moreover, the assessment only focused on people with stroke. Caregivers' involvement in the assessment was lacking (Subtheme 2). These subthemes are detailed below.

9.2.1 Subtheme 1: 'Psychological/Mental Health is Underplayed'

Participants described care needs assessment based on the physical and cognitive functions of people with stroke. For example, on admission to a rehabilitation program, care needs for people with stroke were identified 'by different members of the team depending on what they've been focusing on' (DOC02). The initial assessment focused on the level of function 'so that we [clinicians] can determine how long they will need the program for. And it's also a way of informing us how much they will be gaining at the end of their rehab program' (DOC01). Rehabilitation goals were set based on these discipline-based

assessments. Goals were discussed in team meetings throughout the life of the rehabilitation program. When preparing for discharge from hospital, care needs were unpacked based on 'how much physical assistance they are going to need ... showering ... home modification ... then as a team, problem-solving, how to best support the client and carer to transition into the community' (MG01).

These examples demonstrated that health professionals prioritised rehabilitation to improve the functional recovery of people with stroke. There seemed to be a limited focus on the assessment of psychological needs for people with stroke: 'the emotional side, perhaps we don't address that as well as we could. Clinical psychology is an area that we're struggling with. We've got quite low FTE [working hours] and patients' feedback that they'd like more emotional support' (MG01). Even when psychological health for people with stroke was assessed and addressed during inpatient stay, this may not have been carried out in primary care, as a medical officer reported:

Psychological/mental health is underplayed in our medical model, where we focus on physical health. We have psychologists coming to review patients, but they don't attend team discussion ... All our summaries and handovers to GP, other services are generally about the physical and cognitive recovery, not so much about coping and psychological ... If it's something mild [psychologically], we won't necessarily pick it up (DOC02).

The unmet psychological care needs were also observed by nurses at the stroke follow-up outpatient clinics: 'When it comes to the questions around mood, I'd say roughly 40% of patients say they have been down. It's not something they come out straight away; you have to really dive into that' (NS06, FG04).

Health professionals' perceptions of factors affecting care needs assessment for people with stroke revealed gaps in meeting the psychological needs of people with stroke. With the trend of shortened hospital stays, psychological needs might have to be managed in the community setting after hospital discharge. Identifying and effectively communicating

these needs to the next service may address the psychological and emotional support needs for stroke dyads.

Additionally, the focus on physical recovery seemed to be driven by the decision of hospital discharge. Participants indicated that 'the challenge is particularly when someone can walk, and physically their care needs aren't great. Often there's a big push to get someone home. [For those with] cognition or even communication [deficits], sometimes it is hard to advocate' (AH01). Health professionals' perceptions suggested that people with stroke who had fewer physical impairments but had cognitive and/or speech impairment might be disadvantaged in hospital discharge decision-making, which was based on physical functioning. People with stroke and with cognitive and/or speech impairment and their caregivers might encounter more challenges in coping with these types of stroke-related disabilities during the transition if they missed rehabilitation as inpatients before hospital discharge.

Although health professionals perceived that cognitive or speech impairment were significant components of post-stroke rehabilitation, they showed limitations in advocating for stroke patients with these impairments due to system pressure on discharging patients. Health professionals also perceived that mental health and psychological wellbeing played crucial roles in post-stroke transition. However, they had limited time and resources to facilitate the early identification and treatment of psychological health issues for both people with stroke and their caregivers.

9.2.2 Subtheme 2: 'Our Focus is on the Patient'

All participants discussed the significance of care needs assessment to enable discharge planning and the transfer of care for stroke dyads. However, throughout the continuum of hospital-to-home transition and at different care settings (inpatient and in the community), health professionals mainly assessed care needs for people with stroke, and caregivers' needs and the dyadic effects were not formally assessed:

Obviously, the patient is the primary focus. (MG02)

It's mainly if the caregiver coping or not coping ... It's more are they able to take care of the patient, and do they need more support? But I think our focus is on the patient, and we often expect the caregiver to just do everything we're recommending. (DOC02)

Additionally, caregiver care needs assessments were 'more just informally discussing what their capacity is and what their concerns are' (AH07). Caregivers' needs were largely around the hands-on care activities for people with stroke, as revealed in a focus group: 'In the interdisciplinary assessment tools that we use, there are questions around their [people with stroke] care needs. So how they manage showering, dressing, shopping, transport ... And that encompasses the carer' (AH14, FG02).

Moreover, caregivers' care needs may be identified shortly before discharge from the services. For example, 'sometimes those [care needs] conversations come up when we start talking about discharge. You suddenly get that response: "If this is coming to an end, where am I going to get my respite? Where am I going to get ..." ' (AH11, FG01). In some cases, 'when we're going to discharge the patient in a few days, then we say, oh, they [patient and carer] haven't had carer training' (DOC03).

Furthermore, caregivers' psychological and emotional stress was largely ignored during hospital stays despite being viewed as crucial care partners in the National Safety and Quality Health Service Standards. One inpatient clinician indicated, 'I think carers aren't coping well at all. It's quite sad that we haven't probably identified upset in a carer' (AH08).

Findings indicated that, as a member of stroke dyad, caregivers' care needs were not formally assessed. As stroke affects both people with stroke and their caregivers, early identification of caregivers' specific needs would facilitate discharge planning and enhance stroke dyads' preparedness for the transition.

Care needs assessment is an essential part of discharge planning. Findings revealed the need to include both people with stroke and their caregivers in the assessment and to meet their needs accordingly as a dyad and as individuals. Moreover, the identification of

psychological and mental health issues of both people with stroke and their caregivers would support intervention plans and enhance their transition experience.

9.3 Theme 2: Transition Challenged by Constraints on Service Delivery

This theme focused on the challenges that health professionals encountered in supporting stroke dyads during the hospital-to-home transition. Health professionals perceived that shortened hospital stays only provided a narrow window of opportunity for them to support stroke dyads in the discharge planning process, which might affect the stroke dyads' preparedness for the hospital-to-home transition. The shortened length of stays could also create increased workload and additional stress for health professionals (Subtheme 1). Furthermore, gaps existed in service accessibility for stroke participants who resided in non-metropolitan suburbs. Community care services with a specific cutoff age (65 years old) may contribute to the mismatch between stroke dyads' care needs and available community care services for them (Subtheme 2). These subthemes are presented in the following sections.

9.3.1 Subtheme 1: 'Bed Pressures Certainly Impacts'

Health professionals identified that hospital length of stays and the support from community rehabilitation services affected them in setting out and implementing the rehabilitation goals of people with stroke. For example, discharging home early with support from community rehabilitation services was viewed as beneficial because 'if someone doesn't have significant deficits from stroke, sometimes it's better for them to go home so that they don't become institutionalised but get the opportunity to be in their own environment that might actually enhance their recovery' (AH01). Moreover, regarding services meeting targeted length of hospital stays in the healthcare system, health professionals perceived that:

We are accountable to improving patient outcomes ... to maximise function and support a safe discharge home. But there's also accountability around funding ... Our

service is funded on a length-of-stay target. We do need to look at meeting those targets. (AH06)

And safety for discharge was a key consideration, as one manager indicated: The longer we keep them [patients] in hospital, they are more likely to develop complications or become less confident about returning home ... we need to make room for the patients coming through the door. But also, if they're not able to safely be home, then they're only going to be readmitted, which doesn't help the system. So those targets just give us a reference point when we're setting goals with patients and their family. (MG01)

Health professionals' perceptions about the length of hospital stay targets indicated that they needed to initiate early discharge for people with stroke if it was safe for them. Meanwhile, health professionals needed to set rehabilitation goals for stroke dyads at the point of discharge, which may need further adjustments after discharge. Such a situation required health professionals to partner with stroke dyads to set up follow-up rehabilitation goals that were based on their needs and preferences during the hospital-to-home transition period.

Participants also recognised that longer inpatient stays may provide more chances for rehabilitation therapists to support people with stroke and to further improve strokerelated limitations before returning home. However, due to hospital bed pressures, they were at times discharged earlier than desired, as one therapist stated:

Bed pressures certainly impact. For those people that can walk and got someone at home with them. But got cognitive changes ... if we had kept them, we could have made them more independent from a cognitive perspective. But the fact that they are supported at home and safe with their mobility, then it's unfortunately less of a priority. (AH02)

Shortened hospital stays could also affect the delivery of community rehabilitation services. People with stroke may be discharged too soon to be ready for community rehabilitation, as discussed in one focus group:

Patients may get discharged potentially early, and they might not be as suitable [for community rehabilitation]. So, the amount of intervention that you can provide is limited because they are still acute, not feeling very well [for rehabilitation]. (AH14, FG02)

Findings demonstrated that shortened hospital stays have a significant impact on the ongoing care people with stroke after discharge. Shortened hospital stays might have contributed to inadequate education and discharge preparation for stroke dyads, which may in turn compromise their continued rehabilitation and recovery at home. The feedback at the stroke follow-up outpatient clinic reflected the potential risks, as one clinician stated: 'I think it's mainly anxiety. So, they go home, they're not sure about what appointment they need to attend, when they leave hospital, they're not clear of a contact person' (NS05, FG04).

Health professionals perceived that they were sometimes unable to provide support for stroke dyads within a limited timeframe 'Quite often we've got an agenda and things that we need to look at. And giving that time for people to have more open discussions, I think, is few and far between sometimes because we've got such a schedule' (AH02). Additionally, the perceived high workload due to quick hospital bed turnover seemed to have prevented therapists from providing rehabilitation interventions needed for people with stroke, as one therapist reported: 'Workloads are really high. I think we are more heading towards acutebased therapy model than rehab model. We probably would be building therapeutic relationships. But for the acute, you're kind of in and out. You don't build those' (AH09). Furthermore, shortened hospital stays created additional stress to everyday practice for clinicians, as one medical officer shared:

You'd like to have a bit more time to work this patient out. So, you got that additional stress ... You constantly have patients in the back of your mind, we'd like to know how they're going now or whether we'd done enough for them. That's probably slightly higher with a higher turnover in shorter stays. (DOC03)

Findings revealed that hospital bed pressures and shortened hospital stays may have contributed to inadequate preparedness for discharge in stroke dyads. However, some

people with stroke may benefit from short hospital stays with enhanced recovery in their home setting, subject to their safety at home, their family support and continuity of community rehabilitation. For health professionals, working within the constraints of shortened hospital stays may challenge them with increased workload and additional stress. These challenges could prompt care services designers to reflect and develop strategies to support health professionals and improve service sustainability.

9.3.2 Subtheme 2: 'We're Dictated by Funding Models'

Community care services facilitate continuity of care after hospital discharge, support continued rehabilitation and help meet the care needs of stroke dyads. However, health professionals identified gaps associated with community care services that may pose barriers to meeting the specific needs of people with stroke at different ages.

While inpatient and ambulatory rehabilitation services (home rehabilitation and centre-based rehabilitation services) are not age-based, there is a cutoff age for other hospital-to-home transition services and community support services: 'People over 65 get aged-care services while those under 65 get disability care services. Unless people have support in the community to help them, it could be quite challenging finding those services' (AH07). Although community-based care services were consumer-directed, some stroke dyads may not be ready to navigate the system:

All of the services now are supposed to be consumer-directed, which is great. But it presents huge challenges for people who can't manage to navigate through the system. We have to try and do a lot of that for them whilst they're on the program during that short period of time. But, sometimes, we just don't know what happens to them after that. (MG02)

Findings indicated the need to prepare stroke dyads to master consumer-directed aged-care or disability care services so that they were empowered to identify care services to meet their specific care needs. Findings also implied the potential risk of unmet care needs within the limited length of stay in a community rehabilitation program.

Another challenge brought by age-oriented service design was that the services may not have met the specific care needs of individuals when their lifestyle did not fit the age group, as discussed in a focus group:

I've got a 66-year-old who really live like 40-year-old. So just because of their age, they are set into this funding, which means they get next to nothing going forward. But if they were 50, they get XXXX [a disability service] and could have a lot more. So, I think that age [cutoff] makes it difficult sometimes. (AH14, FG02)

Findings suggested that care services considering both the age and specific needs of individuals may provide more opportunities for people with stroke to reintegrate into the community and manage care for the long term. This lack of person-centredness may generate additional challenges to the transition for stroke dyads.

To reduce hospital length of stays, people who were deemed unsafe to return home with ambulatory rehabilitation services sometimes had to be moved across different interim inpatient services while waiting for community support at home. One therapist shared her concerns about these interim services:

In my opinion, the kind of waiting between services and programs is not always an effective way of doing things. Some patients are quite frustrated about having to go to a queue and move across [different interim inpatient services] ... By that time, they've spoken to five different therapists and the risk of miscommunication, handover breakdown ... and I'd question mental health wise how those people actually coping, being waiting for so long for services. (AH01)

Long waits for community support services and moving between different settings could jeopardise post-stroke recovery and increase the risk of psychological distress for people with stroke and their families. It may also increase the risk of institutionalisation and reduce people's confidence to return home.

Health professionals perceived that people living in non-metropolitan suburbs might be disadvantaged due to the lack of services, especially stroke-specific services, in these suburbs, as discussed in one focus group:

I think in metro, it's probably easier to be referred to services that specialise in adult rehab. But as you get further out, there's not the luxury of that, and you're referring to a generalist service. Some sites have rehab, but they might not be stroke-specific. (AH16, FG02)

Stroke-specific rehabilitative interventions are essential to maximise functional gains for people with stroke during the first 6 months after stroke. Their recovery may be compromised without these interventions. Lacking services in non-metropolitan areas could create barriers to continuity of care after completion of time-limited post-discharge rehabilitation programs.

Participants expressed their expectations of service integration and flexibility in service provision that could enhance transition and continuity of care. For example, before hospital discharge, 'it'd be helpful for us [home-based rehabilitation therapists] to come up here [ward] first, meet them to get a sense of strategies for discharge' (AH02). Participants perceived that, if therapists from inpatient and ambulatory rehabilitation services had time to meet people with stroke so that 'there is a clear communication of expectations and differences in the programs ... that might help to ease any anxiety about the transition ... some greater transparency or communication at that point would be great' (AH05). Suggestions from health professionals highlighted the potential benefits of integration between services before hospital discharge to alleviate the anxiety heightened by hospital discharge for stroke dyads, and the integration may also improve communication between services and facilitate the development of individualised care plans.

Health professionals from the ambulatory rehabilitation services particularly conveyed their expectations of flexibility in providing home-based and centre-based rehabilitation services. Due to funding models, two services were operated independently, which meant that, when people with stroke received the home-based service, they could not access the centre-based service until they were referred to the service after completing the home-based service. However, in reality, people with stroke may benefit from both services at the same time to better achieve their goals, as a service manager advocated:

Ideally, home rehab and day rehab would almost merge, and patients would get whatever they needed, wherever they needed it ... Transitioning from a team on the ward and then to home rehab and then to day rehab would be very confusing [for patients]. A lot of duplication, handover ... Unfortunately, we're a bit dictated by funding models. (MG04)

Shifting between multiple services or programs may increase the risk of communication breakdown and compromise individualised care delivery. Instead of funding-based services delivery, care-needs-oriented services delivery may better meet individuals' specific care needs and ease the stress of navigating a complex health system for stroke dyads.

Additionally, the flexibility of service delivery seemed to be constrained by service design. For example, the home-based rehabilitation service was designed as 'service where somebody gets a service everyday ... it's a hospital substitution and replicates what happens on an inpatient ward ... Seeing a patient two or three times a week doesn't feel like an intensive program for people to meet their goals' (MG02). Some people with stroke may not be able to tolerate the intensity of the program. However, they would still benefit from the home-based program, as one therapist expected:

More complex patients would probably benefit from more therapy in their own home and community setting. Say getting back to shopping or go on public transport ... they [patients] generally don't have the time to do that. So, having that sort of support for a little bit longer would probably make that transition from hospital a bit smoother and make sure that they're getting back to all those meaningful activities rather than just surviving. (AH07)

Findings suggested that providing a tailored rehabilitation service to people with stroke is a way to help them overcome challenges during the hospital-to-home transition. Supporting people with stroke to resume meaningful activities would promote community reintegration and caregiving sustainability.

This theme revealed that shortened hospital stays may affect preparation for the hospital-to-home transition for stroke dyads. It may also increase the workload and stress for health professionals. Findings highlighted the need for a co-design approach to care service development, which reflects a true partnership with stroke dyads and may strengthen their self-management capabilities.

9.4 Theme 3: Transition Enhanced by Commitment to the Safety of Care

Despite the challenges in supporting people with stroke during the hospital-to-home transition, health professionals demonstrated their commitment to safe discharges from their services and identified strategies to enhance safety after discharge. Safety enhancement strategies were implemented during discharge planning and in the process of supporting stroke dyads navigating the complex health system (Subtheme 1). Collaboration of health professionals within and across services enabled enhanced safety and supported continuity of care (Subtheme 2). Details of both subthemes are presented below.

9.4.1 Subtheme 1: 'Empowering Them to Making Decisions'

In the context of shortened hospital stays for people with stroke, health professionals remained proactive in discharge planning in that they 'should be thinking about discharge planning as soon as we're assessing and admitting patients. We need to identify discharge barriers early in the stay as a team' (AH06). Health professionals tried to 'be flexible and available to answer any questions from family' (DOC03). They also tried to involve stroke dyads in discharge planning at the early stage of admission as they believed this approach would 'empower them [caregivers] to making decisions and feeling comfortable saying they're taking on extra care burden and that they've got some support for that' (DOC03). To improve caregivers' engagement, health professionals would invite them to therapy sessions. For example, speech pathologists would invite caregivers to 'communication partner training in aphasia [communication disorder] because, when they go, it's the family that are communicating with the patient' (AH09). Additionally, caregiver training was arranged in a way to save caregivers' time and accommodate their specific situations where

possible. For example, for a person with cognitive impairment, therapists organised 'a wellrounded [caregiver] education session with physio, speech pathologist and occupational therapist so that they [caregivers] didn't have to come in more than once' (AH02).

Other strategies to involve caregivers and family in discharge planning in the inpatient setting included encouraging 'family to take their loved ones home or out in community, come back and let us know how is going' (AH08), inviting caregivers to 'stay overnight in hospital, so they can get a better idea of how much assistance is required ... or goes on overnight leave for carer to carry out all the duties. Then they come back and give us some feedback' (NS01, FG03).

Despite the challenges with shortened hospital stays, health professionals demonstrated endeavours and flexibility to partner with stroke dyads. These efforts reflected health professionals' dedication and determination to support a safe hospital-to-home transition for stroke dyads.

Regardless of service settings, all participants considered safety as 'first and foremost' (AH07) in discharge planning process, and 'if they're physically at risk of, say, fall because they didn't have the right equipment yet, even we're getting pushed [due to bed pressures], we wouldn't send them home' (MG02). Additionally, ambulatory rehabilitation services had implemented strategies to fill in the gaps for the next services. For example, when the next service provider was unknown, 'to bridge that gap, I just gave my email address to the patient's partner to email me and let me know [the provider] so I can [handover]' (AH11, FG01). Similarly, when the next service could not take over the care, the ambulatory services extended the length of the program to fill in the gaps: 'in some cases, we go over our length of stay to bridge a gap ... we can advocate for patients, so they don't miss out on any therapy ... it's always what's best for the patient' (AH14, FG02). Furthermore, to meet the specific needs of people with stroke, sometimes a check-in service after completion of a program would be in place to support the transition:

When there's ones that we are feeling like there's a particular space that is standing out, what we've sometimes done, like we've finished this 6-week [program], but let's

just checking in a week's time by a telerehab and to see how that's going, especially [for] the return-to-work ones (AH13, FG01).

These examples highlighted that health professionals were enabled to be flexible in service provision to enhance safety of care. Health professionals attributed this enablement to the safety culture at the workplace:

I do feel we are very lucky that executive in this division and leaders on the wards do genuinely care about the patients' wellbeing and outcomes. And I think there is a recognition that, yes, we need to try and make room for the patients coming through the door. But, if we do that at the expense of patients who are not able to safely be home, system-wide [it] doesn't help. (MG01)

Findings revealed health professionals' commitment to partnership with and their advocacy for stroke dyads to enhance safety and quality of care. A focus on safety, driven by strong leadership, seemed to foster a positive safety culture and enable health professionals to empower stroke dyads to take over care and be supported during the transition.

9.4.2 Subtheme 2: 'Working Towards a Common Goal'

All participants identified that effective communication among multidisciplinary team members and between services was key to supporting the hospital-to-home transition for stroke dyads. As nursing leaders described, 'we're all kept on the same page ... we all have our own roles, but we're working towards a common goal, which is the best, focused outcome for patients' (NS01, FG03). The benefit of the interdisciplinary approach was that 'everyone having an input towards what the patient requires, and what their level of functioning is and working towards that effective discharge with the patient or their family' (DOC01). Additionally, efficient communication among team members may have also helped fill in gaps between hospital and home, as one therapist discussed:

Sometimes even when they're coming from an inpatient [rehab] ward, it could only be quite a short stay there. And staffing issues mean that potentially they've ever been seen a couple of times from XX [therapy] perspective. So, it's through [setting] goals

and lots of multi-D discussion in our team that helps with discharge planning. (AH16, FG02)

Effective communication and an interdisciplinary approach to service delivery and discharge planning facilitated shared visions on care planning in a multidisciplinary team. It may also enhance the quality and safety of the transition and strengthen continuity of care for stroke dyads.

Participants also highlighted the support among team members that fostered collaboration: 'Some people in the team feel burdened, especially after a prolonged busy period ... It's very challenging. But they support each other very well. We try to do things to lighten the load and lighten the mood and help each other' (MG02). Furthermore, trust and confidence in the expertise of colleagues seemed to enhance safety and continuity of care:

We have good communication avenues between us [inpatient and ambulatory services] to support discharge, flagging any issues that perhaps we couldn't manage in an inpatient setting ... I think it's because the staff move around that provides them with a better understanding of what each service can provide and how we best tackle that transition. (AH05)

In contrast, when referring to external services, health professionals may not have had the same level of confidence:

We feel the most comfortable transitioning those dysphagic patients if we know where they're going, the home rehabs or day rehabs because there's the feedback, and you know they're getting the service. The ones like XX [community service] where you don't have that feedback. (AH09)

Health professionals' perceptions indicated that communication and trust built on understanding among team members were important elements to promote team collaboration regardless of the care setting. The findings also suggested that collaboration of service providers in the chain of stroke care could potentially enhance the transition experience of stroke dyads.

To enable reflective practice and further improve the quality of care, participants expressed their expectations of feedback from stroke dyads. One therapist suggested:

I think it is so important to get that patient feedback in a meaningful way ... that's the really challenging thing for someone to negotiate a system after they've left here ... If we don't really get that feedback, we just don't know how to change our practice or reflect more on our practice. (AH01)

Reflective practice supports continuous quality improvement. A feedback system would strengthen the partnership between health professionals and stroke dyads to enhance their experience in the hospital-to-home transition.

Findings suggested that effective communication and collaboration between health professionals would enable safe and quality care and improve outcomes during the hospitalto-home transition for stroke dyads. Health professionals and service providers would benefit from a system established to receive feedback and to evaluate the effectiveness of discharge planning and continuity of care.

9.5 Summary

Stroke rehabilitation health professionals reflected on the challenges and opportunities in the first 6 months of hospital-to-home transition for stroke dyads. This chapter has presented three themes and six subthemes. Involving both people with stroke and their caregivers in the assessment was essential to enable an individualised approach to care planning as a dyad and as individual dyadic members. Additionally, focusing mainly on the functional recovery of people with stroke may overlook their psychological needs and caregivers' pivotal role in the rehabilitation of people with stroke during the transition. Moreover, health professionals' commitment to safety and continuity of care was evident despite the challenges they encountered, such as shortened hospital stays and limited resources. Their dedication and collaboration were considered paramount to enhancing the transition experience of stroke dyads.

The next chapter (Chapter 10) explains the significance of the findings and new knowledge generated from this study.

Chapter 10: Discussion

10.1 Introduction

The previous four chapters presented 10 major themes and multiple subthemes regarding hospital-to-home transition care from the perspectives of stroke dyads and health professionals. The experience of stroke dyads was explored across the time period from before hospital discharge to 6 months after returning home. Seven themes identified for stroke dyads were (1) psychological stress, (2) resilience, (3) certainty and uncertainty about post-discharge care, (4) coping with setbacks, (5) role transformation, (6) transition with support and (7) expectations in hospital-to-home transition care. Interviews and focus groups with health professionals from inpatient and ambulatory rehabilitation settings generated three major themes: (1) transition shaped by care needs assessment, (2) transition challenged by constraints on service delivery and (3) transition enhanced by commitment to the safety of care.

Gadamer's (1979) philosophical hermeneutics was used in this qualitative study. This methodology enabled the researcher, who has extensive experience in stroke care, to bring her pre-understandings of factors influencing hospital-to-home transition care for stroke dyads to the interactions with participants during the interviews and focus groups to achieve rich data for analysis. The researcher also experienced numerous hermeneutic circles during the interpretation of data (Gadamer, 1979). The interpretation and reflection processes in the hermeneutic circles helped the researcher to reach a new understanding and generate new knowledge pertinent to hospital-to-home transition. Gadamer's philosophical hermeneutics also enabled the researcher to link study findings to the socio-political-economic context to achieve a historical consciousness about stakeholders' views of factors affecting the hospital-to-home transition in the current health and social care system in South Australia. This chapter presents a discussion of the main findings and relates the findings to similar studies in a global context.

The discussion is organised in four sections: (1) balancing functional rehabilitation and psychological support, (2) understanding dyadic interdependence to promote dyadcentredness in healthcare, (3) the role of health professionals in enhancing hospital-to-home transition and (4) enriching Meleis's transitions theory.

10.2 Balancing Functional Rehabilitation and Psychological Support

The study's findings shed light on the coexistence of psychological stress and coping with resilience in stroke dyads during the hospital-to-home transition. However, the findings suggest that their psychological needs could be addressed more appropriately during this critical period. Participants highlighted several contributing factors that hindered the provision of appropriate psychological support, including a function-oriented rehabilitation service that concentrated on functional recovery over emotional wellbeing, limited access to psychological/counselling support and a lack of awareness among stroke dyads about their psychological wellbeing and its impact on a healthy recovery transition. These findings offer a novel perspective on the care needs of stroke dyads across the transition process, highlighting the need for a more comprehensive and integrated approach to care that prioritises the psychological wellbeing of stroke dyads alongside their physical health.

Psychological stress refers to the negative emotional status of an individual when they perceive a lack of resources to cope with a real or imagined threat (Gyawali et al., 2020; Monroe & Slavich, 2016). The present study concurs with previous findings that both people with stroke and their caregivers experience psychological stress during the hospital-to-home transition (L. Chen et al., 2022; Lin et al., 2022). They presented fear, frustration and uncertainty before hospital discharge, and their stress continued through the hospital-tohome transition (Lin et al., 2022; L. Chen et al., 2022). Psychological stress, if left unaddressed, may cause increased bone marrow activity and arterial inflammation, leading to intensified amygdala activity, which is associated with recurrent stroke (Booth et al., 2015; Tawakol et al., 2017). Studies have also highlighted the psychological sequelae of stroke leading to mental health issues such as depression and anxiety, which accounts for

approximately 30% and 25% of the stroke population, respectively (Dos Santos et al., 2018; Rhudy et al., 2020).

The present study reveals that sources of stress for caregivers differed from those experienced by people with stroke. For people with stroke, their psychological distress and sense of achievement were closely associated with their functional recovery. For caregivers, their psychological wellbeing was not only affected by stroke-related limitations in people with stroke but also by the changed relationship and their adaptation to the caregiving role. An interesting finding in the present study was that people with stroke rarely described the impact of caregivers' psychological stress on them. They attributed their progress to the support from their caregivers. The findings suggest that caregivers might have hidden their emotions and put aside their own needs because the main focus of life during the transition was to support the person with stroke with progressing in functional recovery and resuming their pre-stroke lifestyle (Bucki et al., 2019; Quinn et al., 2014). By hiding their true feelings, caregivers might be misinterpreted as if they were effectively coping with the transition. This could increase the risk of vulnerability in coping during the transition (Quinn et al., 2014). Insights into stroke dyadic differences and their psychological trajectory are of paramount significance for stroke nurses to identify stroke dyads at high risk of developing psychological distress and to create tailored interventions to enhance continuity of care during the transition.

The findings of this study parallel with other studies that have shown that post-stroke rehabilitation mainly focuses on functional recovery with limited emphasis on psychological recovery or support in hospital and community settings (Hughes & Cummings, 2020; K. F. Liu et al., 2022). Such a practice disregards the association between psychological stress, low resilience and poor post-stroke outcomes in people with stroke (Gyawali et al., 2020; Ianni et al., 2022). In a stroke services audit, approximately 44% of people with stroke did not have a mood assessment and as much as 63% of audited hospitals did not provide psychology services (Stroke Foundation, 2019a). Additionally, caregivers' preparedness for the caregiving role is significant for a positive hospital-to-home transition (Camicia, Lutz,

Harvath & Joseph, 2021; Pucciarelli et al., 2021). A formal caregivers' needs assessment, including psychological assessment, is lacking. In their study of challenges to stroke caregivers' resilience, Qureshi et al. (2022) suggested that caregivers need to be engaged in care planning and that their needs assessment would enable person-centred planning.

Functional recovery after stroke refers to the recovery of cognitive and motor impairment affecting ADLs such as mobility, shower, feeding and toileting (Branco et al., 2019; Grefkes & Fink, 2020). Psychological wellbeing in stroke dyads and functional recovery in people with stroke are positively correlated (Ezema et al., 2019; Perna & Harik, 2020). Therefore, psychological support for stroke dyads should be viewed as an equally important care element as functional recovery during the post-stroke trajectory. Psychological symptoms may be hidden and may not be readily observable by therapists during therapy interactions (Perna & Harik, 2020; Vlachos et al., 2021). Stroke nurses providing 24-hour care are well poised to assess and identify stroke dyads at risk of developing psychological symptoms and to work collaboratively with the interdisciplinary team to develop a care plan tailored to an individual's specific situations. Shifting the mindset of health professionals from a focus on the physical health (i.e., functional recovery) of people with stroke to both physical and mental health (i.e., psychological wellbeing) in both dyad members may potentially improve the hospital-to-home transition experience and stroke outcomes (Meyers et al., 2020; Qureshi et al., 2022).

The American Psychology Association (APA) defines resilience as 'the process and outcome of successfully adapting to difficult or challenging life experiences, especially through mental, emotional, and behavioural flexibility and adjustment to external and internal demands' (APA, 2023). When applying this definition to the stroke population, resilience refers to a person's ability to cope with adversity in life, adapt to changed situations and engage in meaningful activities to maintain quality of life (Gibbs et al., 2020; Zapater-Fajarí et al., 2021). Resilience can be influenced by an individual's views about and engagement with the world, the availability and quality of social resources, and coping strategies (APA, 2023). This individual attributor has been found to protect stroke dyads against mental health

issues and predict quality of life after stroke (Kim et al., 2019; Z. Liu et al., 2021). The findings of the present study align with previous studies that show that stroke dyads demonstrate coping with resilience, drawing on their past life experiences, functional improvement in people with stroke, and the support from family, friends and health professionals despite the stress they face (L. Chen et al., 2020; Pereira et al., 2021). They were able to identify positivity, adapt to changes and regain health, wellbeing and social integration in the continuum of the transition (L. Chen et al., 2020; Pereira et al., 2021). Therefore, support from nurses and rehabilitation clinicians to build and enhance resilience should be incorporated into post-stroke rehabilitation. Recent studies have highlighted some preliminary benefits of psychoeducational interventions to enhance caregivers' resilience and the importance of understanding the predictors of differing levels of resilience before designing interventions (Qureshi et al., 2022; W. Zhang et al., 2020).

Functional recovery and psychological wellbeing are two core elements in poststroke rehabilitation for stroke dyads. They are interrelated and contribute to stroke dyads' reconciliation and regaining control of health, wellbeing and social integration during the hospital-to-home transition (Baker et al., 2018; Pereira et al., 2021). They also influence the sustainability of caregiving and the self-management of stroke-related health issues at home (Baker et al., 2018; Pereira et al., 2021). Stroke care stakeholders' perceptions of the equal importance of functional recovery and psychological wellbeing in stroke dyads is crucial for designing and funding stroke care services. Perna and Harik (2020) have suggested that psychology services are embedded in stroke care to enable a focus on psychological wellbeing and to help optimise rehabilitation outcomes in the continuum of hospital-to-home transition care. With the opportunity to work with stroke dyads closely during their poststroke journey, stroke care nurses are ideally positioned to assess the psychological needs and resilience of stroke dyads, develop tailored strategies in collaboration with the interdisciplinary team and reinforce these strategies in their daily encounters with stroke dyads (Diaz et al., 2020; Le Danseur, 2020).

10.3 Understanding Dyadic Interdependence to Promote Dyad-

Centredness in Healthcare

Consistent with previous studies, the findings of this study highlight dyadic effects between stroke dyad members that either facilitate or hinder their hospital-to-home transition (Bakas et al., 2022; Pucciarelli et al., 2021). In this study, positive dyadic effects were evident between dyadic members in supporting each other, demonstrating positive coping resilience and positive views when facing uncertainty. The level of physical, cognitive and functional ability, and resilience in people with stroke either enhanced or challenged their caregiver's confidence in caregiving and the psychological distress those caregivers experienced. Likewise, support from caregivers motivated people with stroke to continue their rehabilitation and reintegration into the community. It was evident that caring, understanding and encouragement within the dyads could produce positive outcomes.

A caregiver and a care recipient form a caregiving dyad (Ferraris et al., 2022). Previous studies on patient–caregiver dyads living with chronic conditions, such as stroke, cancer and cardiac diseases, have suggested dyadic interdependence. In other words, a caregiver and a care recipient share a mutual experience in the adaptation process of a health condition, and their wellbeing is interconnected (Badr et al., 2019; Bakas et al., 2022; Ferraris et al., 2022). The concept of interdependence originated from the theory of interdependence developed by Kelley and Thibaut (1978). They suggested that interpersonal relationships are shaped through interpersonal interdependence, which is the process whereby interacting people affect one another in the context of a specific interdependence situation (Rusbult & Van Lange, 2008). Stroke affects caregiving dyads as a unit, which results in dyadic members affecting each other regarding their health, wellbeing and social integration (Ferraris et al., 2022; Welten et al., 2023).

Stroke dyads experience relationship challenges, such as perceived power imbalance, difficulties adjusting to changed identities as care recipient and caregiver, and incongruent views about the physical and cognitive capabilities of people with stroke

(Hodson et al., 2019; McCarthy et al., 2021). These interpersonal relationship challenges may increase the risk of psychological distress in stroke dyads. Previous studies have revealed that dyadic effects have a vital role to play in the physical, psychological and quality-of-life outcomes for both spousal and non-spousal dyads (Ferraris et al., 2022; Pucciarelli et al., 2021).

Caregiving dyads in this study comprised couples (n = 11), parent and adult child (n = 6) and sibling (n = 1) dyads. Regardless of their relationship, this study found that stroke dyads experienced and navigated the hospital-to-home transition together. Their shared understanding of stroke and its impact, and mutual support, as well as support from health professionals and from their social networks, enabled their community reintegration. The present study accords with previous stroke dyadic studies. Pereira et al. (2021) revealed that mutual goals and priorities between people with stroke and their caregivers facilitated their regaining control of life after hospital discharge. A systematic review and meta-analysis by Pucciarelli et al. (2021) agreed that stroke dyadic educational intervention studies generated improved outcomes on functional independence and psychosocial distress for stroke dyads. These positive findings from previous studies (Pucciarelli et al., 2021; Pereira et al., 2021) are promising and support a dyad-centred approach for enhancing stroke dyads' hospital-to-home transition.

The present study adds new knowledge to the existing literature in that, apart from rehabilitative interventions delivered by health professionals, people with stroke in this study attributed their psychological and functional recovery to the positive influence that caregivers had on them during the transition. In their view, caregivers had not only taken on household responsibilities that were shared by both before the stroke but also provided strong support, encouragement and navigated the transition with them. Contrastingly, caregivers discussed the negative impact of psychological distress, behaviour changes and functional limitations in people with stroke on them and the psychological adjustment they had to make. Ample studies have revealed challenges experienced by stroke caregivers, including unpreparedness for the new role, changed family relationships, physical and mental

caregiving burden, and stress (Achilike et al., 2020; Kazemi et al., 2021). The present study suggests that caregivers may have put on a brave face and tried to be strong for their loved ones because the focus of post-stroke recovery during the transition was on the functional recovery of the person with stroke (Hughes & Cummings, 2020; Bucki et al., 2019). This approach to coping may overwhelm caregivers and add extra burden to the already challenging situation. Previous studies explored the use of caregiver assessment tools to understand caregiver's preparedness for care and their prioritised needs from a caregiver's perspective (Camicia et al., 2023; Toye et al., 2016). Researchers found that preparing caregivers for their role is pivotal to home care sustainability and caregivers' long-term health and well-being (Camicia et al., 2023; Toye et al., 2016). Findings of this study support a comprehensive caregiver preparedness and support needs assessment before hospital discharge and at key timepoints in the continuum of hospital-to-home transition such as before separation from care services and when health condition of care recipient changes.

Camicia et al. (2022) advocated for a family-integrated care model for caregivers of patients in the rehabilitation inpatient setting. The family-integrated care model acknowledges the interdependence of the caregiver and care-recipient dyad and provides an avenue to build caregivers' confidence through comprehensive assessment and targeted strategies. Recognising dyadic interdependence and taking a dyad-centred approach have implications for health professionals' engagement with stroke dyads in the hospital-to-home transition. A dyad-centred approach enables health professionals to consider the stroke dyad as a unit and to integrate the dyad into all aspects of their care. This approach would enhance the development of tailored plans that consider the needs and preferences of caregivers as well as care recipients (see Family Integrated Care,

http://familyintegratedcare.com/).

Person-centred care is widely recognised as a foundation for safe, high-quality health care (ACSQHC, 2021). The WHO describes people-centred health services as an approach that responds to the needs, preferences and values of individuals, families and communities in humane and holistic ways and through education and support to promote their

participation in their own care (WHO, 2015). This care approach, focusing on people rather than disease. is described in the National Standards for Quality and Safety in Healthcare in Australia (ACSQHC, 2021). Findings from the present study support a person-centred approach but also advocates for dyad-centredness in stroke care, emphasising the equal importance of partnering with stroke dyads in all facets of care.

The present study, along with existing research evidence, highlights the crucial and indispensable role of caregivers in hospital-to-home transition for people with stroke (Camicia et al., 2023; Pont et al., 2020). A co-designing approach to developing care plans and incorporating their situation-specific priorities in care design empowers stroke dyads to develop self-management capabilities, which has a far-reaching impact on long-term care in the community (Jones et al., 2020; Lobo et al., 2021). Acknowledging the benefits of person-centred care principles in dyad-centredness, the present study recognises the challenges in the operationalisation of this approach in clinical practice. Byrne et al. (2020) have argued that person-centred care would require organisational culture development to facilitate implementation. This involves policy, funding and education to align the concepts and principles with reality (Byrne, Baldwin & Harvey, 2020; McCormack & McCance, 2017).

Valuing interdependence within stroke dyadic members provides a way for stroke care health professionals to shift their mentality from stroke-patient-centredness to stroke-dyad-centredness to enhance their engagement with both members of the stroke dyad in goal-setting, care plan development and discharge planning. Preliminary positive outcomes on functional recovery in people with stroke, self-management capabilities and psychosocial adaptation in stroke caregivers offer opportunities for further research on dyadic interventions during the hospital-to-home transition with the aim of enhancing self-management and caregiving sustainability in the long term (Bannon et al., 2020; Pucciarelli et al., 2021). The implementation of person-centredness in stroke-dyad-centred care calls for healthcare policymakers, leaders of healthcare organisations and frontline health professionals to work in collaboration to create a culture of enabling the dyad-centred

approach in clinical practice (Byrne, Baldwin & Harvey, 2020; McCormack & McCance, 2017).

10.4 The Role of Health Professionals in Enhancing the Hospital-to-Home Transition

Post-stroke recovery and an individual's response to treatment, therapy and care is highly complex and variable (Bishop et al., 2021; Kyle et al., 2020). The present study reveals the intertwined challenges at the systemic, organisational and professional levels, including funding model constraints on service provision, imbalances between service demand and the care needs of stroke dyads, and discrepancies in priorities between stroke dyads and health professionals. These challenges may impede authentic partnerships between health professionals and stroke dyads. However, the present study also highlights the significant role that health professionals played in supporting stroke dyads transitioning through their stroke journey from an inpatient setting to the community. The interdisciplinary team approach, safety culture and team resilience enhanced the safety and continuity of care during the hospital-to-home transition for stroke dyads.

Stroke rehabilitation refers to a proactive, person-centred and goal-oriented process, with interventions designed to maximise function and reduce limitations for people with stroke in interacting with their environment (Stroke Foundation, 2022; WHO, 2021). Aligning with the *Clinical Guidelines for Stroke Management* (Stroke Foundation, 2022), rehabilitation is provided by specialised interdisciplinary health professionals throughout the stroke care continuum (Stroke Foundation, 2022). Coordinated interdisciplinary teamwork has enabled improvements in stroke care and supported stroke dyads to achieve their goals of maximising recovery and reintegration into the community (Franz et al., 2020; Kyle et al., 2020). Nevertheless, in parallel with previous studies, the present study reveals that funding, service design and availability, the trend of time-limited rehabilitation services and the lack of specialist health professionals challenged the interdisciplinary team to support stroke dyads through the hospital-to-home transition (Kyle et al., 2020; L. Chen et al., 2020). Kyle et al.
(2020) have suggested health and social care system reform to foster an environment that empowers stroke care health professionals to provide person-centred care for stroke dyads.

Working within the constraints of the above challenges, health professionals in this study demonstrated flexibility and determination to enhance the safety of care. Safety culture was evident in the findings. Patient safety culture is related to organisational culture and refers to 'a pattern of individual and organisational behaviour, based upon shared beliefs and values that continuously seeks to minimise patient harm, which may result from the process of care delivery' (ACSQHC, n.d.-a). Patient safety culture is fostered through shared perceptions of the importance of safety at all levels, constructive communication and mutual trust (ACSQHC, n.d.-a; Hessels et al., 2019). The present study supports previous studies that show that patient safety is a priority in care service delivery among interdisciplinary stroke care teams and is supported by leadership, transparency, communication, effective collaboration and mutual goals in the care team (Berry et al., 2020; Vaughn et al., 2019).

Team resilience refers to the collective capacity of all team members to respond well to change, pressure and disruption (Hartwig et al., 2020). Team resilience emerged from the present study as an important element for positively influencing health professionals' partnership and collaboration with stroke dyads in the face of constraints on service delivery (e.g., time constraints on health professionals in a busy clinical setting). In the present study, support from leaders, trust, clear communication and peer support among team members contributed to team resilience-building, team collaboration and enhanced team capability for partnering with stroke dyads in the hospital-to-home transition. With challenges in clinical practice, such as the demand for quality improvement and the rising cost of the health system (Department of Health and Aged Care, 2019; Kyle et al., 2020), resilience in health professionals (Hartmann et al., 2020; Kuntz, 2021). Kuntz (2021) advocated for promoting resilience through clear organisational communications, peer and leadership support, flexibility, role clarity and quality teamwork (Kuntz, 2021).

The present study also revealed that nursing participants seem to have limited input into their role of supporting stroke dyads during the transition. This finding parallels with previous studies that showed that nurses' contributions to patients' care experience were not fully recognised by patients and their families or identified by nurses (L. Chen et al., 2016; Godsey et al., 2020; Landis et al., 2020). Globally, nurses make up the largest healthcare workforce in the healthcare system (AIHW, 2022b; World Health Organization, 2022b). Despite their professional qualifications and unique value to care services, nurses remain undervalued and unrecognised for their contributions to the healthcare system (Landis et al., 2020; Godsey et al., 2020). Landis et al. (2020) argued that nurses are viewed by patients and other health professionals in terms of the 'visible' work they do, such as caring for basic needs and psychomotor skills for performing nursing tasks. The 'visible' work is seen as less valuable compared to that of other health professionals, such as physicians and physiotherapists. The 'invisible' work-such as patient advocacy, nursing leadership and time spent with patients and family to build rapport—is not fully recognised in patients' care experiences (Landis et al., 2020). From nurses' perspectives, Godsey et al. (2020) suggested improving the nursing professional identity; nursing education needs to focus on both technical proficiency and soft skills such as communication, critical thinking, decisionmaking and coordinating care in a multidisciplinary team. Health organisations need to develop nursing leadership development pathways and enhance nursing involvement in decision-making (Godsey et al., 2020).

'Stroke nurse liaison', 'coordinator' or 'navigator' are terms that have been used interchangeably to describe the role of advanced practice nurses to coordinate stroke care, facilitate discharge planning, provide education and promote service development (Camicia, Lutz, Summers et al., 2021; Purvis et al., 2021). This specialised stroke nurse's role is gaining international recognition and has been recommended by the National Acute Stroke Services Framework in Australia (Camicia, Lutz, Summers et al., 2021; Stroke Foundation, 2019a). However, challenges in implementing this role have been identified, including insufficient time to perform the role, a lack of role clarity and inequality in job classification,

limited funds to enable professional development, and inadequate orientation to the role (Purvis et al., 2022). To address these challenges and ensure the sustainability of this role, healthcare organisations need to conduct comprehensive reviews, establish systems to overcome barriers and allocate adequate resources (Camicia, Lutz, Summers et al., 2021; Purvis et al., 2022).

Health professionals play essential roles in post-stroke rehabilitation and hospital-tohome transition for stroke dyads. Despite the challenges they face in the reality of timelimited care services and inflexible funding models, health professionals demonstrate a resilience that is strongly influenced by patient safety culture in the workplace and strengthened through flexibility, consistent goals of the team, leadership and peer support. Team resilience enables health professionals to collaborate with stroke dyads to develop self-management capabilities and regain health, wellbeing and social integration during the hospital-to-home transition, despite challenges they may encounter in the ever-changing healthcare system. Stroke care nurses are in a unique position to lead and facilitate safe, effective and efficient care transitions for stroke dyads (Camicia, Lutz, Summers et al., 2021).

10.5 Enriching Meleis's Transitions Theory

In this study, the concepts in Meleis's transition theory (Meleis et al., 2000) that emphasise the nature of transition, transition conditions (facilitators and inhibitors) and patterns of response informed the analysis and interpretation of the potential factors affecting hospital-to-home transition care for stroke dyads (Chapter 3). Findings from this study enrich Meleis's transition theory through the perspective of stroke-specific transition care, as discussed in the following sections.

10.5.1 Nature of Transitions

10.5.1.1 Types and Patterns of Transitions

Stroke attacks the brain and instantly changes people's lives. Transitioning from hospital to home is considered the most challenging period for stroke dyads and their

families. This transition is complex and multidimensional, affecting not only people with stroke but also their immediate families and social networks. Stroke dyads' hospital-to-home transitions are interwoven into a pattern of health–illness and situational transition. These include, for example, the transition from health to life-threatening illness (stroke), survival to recovery, independence to dependence, well-controlled hospital environment to the home environment with uncertainty, employment to unemployment, and from shared family roles to caregiving and care-receiving roles.

10.5.1.2 Transitional Properties

Three critical time points emerged during the hospital-to-home transition in this study: (1) during the hospital stay, (2) arriving home and during formal rehabilitation programs and (3) after separation from formal rehabilitation programs and during community dwelling. The transition brings changes and also results in changes. Changes emerge in the transition over time (Meleis et al., 2000). In this study, before hospital discharge, some participants with stroke viewed post-stroke impairments as temporary and anticipated significant improvement after returning home. They struggled to accept the reality of having to live with these impairments even at 6 months after hospital discharge. The perception of setbacks experienced by stroke dyads reflected the fact that the transition was a nonlinear process. Therefore, a thorough assessment of care needs at different key time points is warranted.

A stroke dyad's level of awareness of transition influences their level of engagement in the transition process (Meleis et al., 2000). For example, when experiencing challenges upon arriving home, some caregivers in the study realised the inadequate preparation they had before hospital discharge. Insights into a stroke dyad's awareness of transition and the level of preparedness would empower health professionals to better engage them in the discharge planning process.

10.5.2 Transition Conditions

Transition conditions either facilitate or inhibit the transition and encompass personal conditions (e.g., meanings, preparation and knowledge, and expectations), community conditions (e.g., social support and access to community supports) and societal conditions

(e.g., stigmatisation or marginalisation; Meleis et al., 2000). These transition conditions intersect to influence the transition outcomes.

10.5.2.1 Personal Conditions

This study reveals that a stroke dyad's individual and dyadic personal conditions contribute to their transition experience. For example, both psychological distress and resilience emerged as attributes that could either enhance or challenge the transition. Sources for psychological distress and positive coping resilience differed between stroke dyad members. The study uncovered an inadequate assessment and exploration of these personal conditions before hospital discharge. Another example is caregivers' preparation for managing post-stroke complications, especially cognitive or speech impairment. Despite the information received before hospital discharge, the perceived challenges coming with the management of these impairments were beyond the capability of some caregivers at the early stage of the transition. An understanding of individual and dyadic personal conditions for stroke dyads would empower nurses and other health professionals to proactively identify care needs and mobilise the stroke dyad's inner resources, such as resilience and motivation, which contribute to a healthy transition.

10.5.2.2 Community Conditions

The social network served as a positive community condition for this study's participants, especially stroke caregivers. Emotional support and physical help from family, friends and neighbours seemed to have alleviated psychological distress and enhanced resilience for stroke dyads. Additionally, information-sharing and education on available community resources influenced stroke dyads' perceived confidence in self-management. For example, perceived support from health professionals was experienced by participants who had written the information and contact details of available stroke-specific community services. Moreover, the availability and accessibility of community rehabilitation services and peer support groups contributed to participants' transition experience. For example, people living in regional areas perceived challenges in seeking appropriate services to support their ongoing rehabilitation needs after separation from formal rehabilitation services. Participants

living in regional areas may be disadvantaged in accessing peer support groups due to the absence of groups in their region.

10.5.2.3 System Conditions

Meleis et al. (2000) described the third transition condition as societal conditions, such as social stigmatisation or marginalisation, which were not revealed in this study. Instead, participants experienced positive outcomes through the collaboration between stroke dyads and health professionals, and they also suggested improvements that could enhance stroke dyads' transition experiences and rehabilitation outcomes. All of these were rooted in the healthcare organisations and the healthcare system. For example, a focus on both dyadic members in assessing care needs, providing education and information, and engaging in decision-making and planning supported a proactive dyad-centred approach to care delivery and better targeted specific dyadic and individual needs. Additionally, a patient safety culture, supported by positive organisational leadership within the organisations, fostered clinical reflection and continuous quality improvement. Moreover, flexible funding models would potentially strengthen the integration of stakeholders on the stroke care chain to support the realisation of stroke dyads receiving the right service at the right time and in their preferred settings. Finally, as novices to the complex healthcare system, stroke dyads would benefit from a stroke nurse navigator to guide them in navigating the system, which would ease the psychological stress experienced by stroke dyads, especially the caregivers.

10.5.3 Patterns of Response

Meleis et al. (2000) presented patterns of response in process and outcome indicators and considered these indicators to be interlinked. Transition is a temporal process, and people in transition experience changes over time. Therefore, it is important to identify process indicators in the transition that either direct towards a healthy transition (referring to people feeling connected, positive intradyadic interactions, developed confidence and skills, and settling in the new role/situation) or to vulnerability and risk. Early assessment and intervention facilitate healthy outcomes (Meleis et al., 2000).

10.5.3.1 Process Indicators

In this study, several process indicators emerged. For example, both psychological distress and resilience were experienced by stroke dyads in the continuum of the transition. People with stroke and their caregivers differed in the sources and outcomes of these indicators. Psychological distress and resilience for people with stroke were related to their functional recovery. They experienced setbacks at the early stage of returning home but regained confidence and strengthened resilience with time, functional recovery and stabilisation, and support from their caregivers and health professionals. In contrast, caregivers experienced initial anxiety and distress before hospital discharge and at the early stage of returning home. The distress was balanced at the later stage of the transition, with resilience enhanced by learning to manage post-stroke complications, by support from social networks and health professionals, and by the improvement of functional recovery of their loved ones. Nurses and other health professionals would benefit from an understanding of these process indicators when interacting with stroke dyads in daily encounters. Knowledge of stroke dyads' direction of transition in the transition process enables early interventions to promote a healthy transition.

10.5.3.2 Outcome Indicators

Meleis's transition theory outlines two healthy transition outcome indicators: (1) mastery of new skills to manage the transition and (2) development of a fluid yet integrative identity, and these may be indicators for the completion of a healthy transition (Meleis et al., 2000). Meleis et al. (2000) suggested that process and outcome indicators overlap and also cautioned on the flexibility and variability in determining the completion of a transition (Meleis et al., 2000).

In this study, at the later stage of the transition (3–6 months after hospital discharge), stroke dyads were at different levels of reintegration into the community. All participants gradually resumed some pre-stroke activities. After completion of formal rehabilitation programs, some participants learned to seek stroke-specific therapies to stabilise or further enhance the regained functions. Participants also expressed gaining confidence and

described developing their own strategies for solving problems in ADLs. These are positive outcome indicators for self-management and community reintegration. Regarding developing an integrative identity, some participants in this study, especially caregivers, struggled to situate themselves in the changed relationship, which was evidenced by their perceived challenges in accepting the lost functions and adapting to the caregiving role. This may suggest that an integrative role was yet to be established. Therefore, the conclusion of hospital-to-home transition may not be determined because they were still in the process of adaptation and transformation, which is a subjective process.

10.5.4 Nursing Therapeutics

Meleis's transitions theory highlights the importance of nursing therapeutics that recognise the complexity and multidimensionality of transitions, with a focus on preventing unhealthy transitions and promoting perceived wellbeing (Meleis et al., 2000). However, the findings of this study suggest that an interprofessional approach is necessary to facilitate the transition process. Understanding the perspectives of all stakeholders involved in the transition and recognising the dynamic and reciprocal nature of the relationships between stroke dyads and other stakeholders can empower interprofessional stroke teams to develop nurse-led interdisciplinary interventions that are tailored to each individual's unique situation and transition experience (Naylor et al., 2020; Ramsay et al., 2014).

Following discharge, people with stroke in this study accessed community-based rehabilitation and care services, such as home or centre-based rehabilitation, telerehabilitation or household support services. Despite receiving support from health and social care organisations, participants expressed a desire for a central contact person to support them in navigating the complex health and social care system and addressing any queries that might arise during the transition. The participants identified nurses as suitable candidates for this role, given that care coordination is a crucial nursing competency that encompasses patient advocacy, education and the facilitation of care transitions (Camicia, Lutz, Summers et al., 2021; Nursing and Midwifery Board, 2016).

Meleis's transitions theory served as a valuable foundation for the researcher's exploration of hospital-to-home transition care for stroke dyads. However, it is important to acknowledge that Meleis's framework is generic in nature and lacks consideration for the intricate and dynamic nature of the relationship between people with stroke and their caregivers. Additionally, it overlooks the importance of involving multiple stakeholders in the transition process. To address these limitations, it is essential to integrate stroke-specific elements, such as stroke-dyad-centredness and the contemporary model of care that emphasises interprofessional collaboration. By incorporating these aspects, the framework can better capture the unique challenges faced by stroke dyads during their hospital-to-home transition. A stroke-specific hospital-to-home transition care framework would not only deepen our understanding of stroke dyads' experiences but also facilitate the development of effective strategies to enhance transition care. To address these gaps, the next chapter presents a novel stroke-specific transition care framework, inspired by the findings of this study and supported by evidence from prior research.

10.6 Summary

This chapter has discussed the complexity and multidimensionality of hospital-tohome transition care for stroke dyads, considering the perspectives of both stroke dyads and health professionals. The focus on physical functions in post-stroke recovery neglects the psychosocial needs of stroke dyads and overlooks the negative impact of the caregiving role on caregivers' mental health during the transition. Dyadic interdependence within stroke dyads can either facilitate or challenge the transition process. Moreover, a workplace culture that prioritises safety and resilience strengthens the commitment of interprofessional teams to provide safe and quality care. Stroke dyads would benefit from the guidance of a stroke nurse coordinator/navigator throughout the transition. These findings enrich Meleis's transitions theory by incorporating stroke-specific elements and highlight the need for a stroke-specific dyad-centred hospital-to-home transition care framework, which is detailed in the next chapter.

Chapter 11: Proposing a Novel Stroke-Specific Hospital-to-Home Transition Care Framework

11.1 Introduction

Chapter 10 discussed the interwoven factors that shaped the transition experiences of stroke dyads. The study reveals that caregivers tend to prioritise providing physical and psychological support to people with stroke while overlooking their own health and care needs. Such a situation compromises caregivers' psychological wellbeing and can potentially affect their self-management and caregiving capabilities. This finding highlights the importance of a dyad-centred approach to assessing care needs and planning care throughout the transition period. Furthermore, this study underscores the significance of workplace safety culture and the resilience of interprofessional teams in enhancing transition care.

To facilitate the transition, the presence of a stroke nurse coordinator/navigator is proposed to guide stroke dyads through the complexities of the healthcare system. Building upon the research evidence from this study and previous research, a novel stroke-specific hospital-to-home transition care framework (hereafter, 'the framework') is proposed. The key components of the framework are illustrated in Figure 11.1 and elaborated in the subsequent sections.

11.2 Overview of the Framework

The purpose of the framework is to engage stakeholders in planning future service development to improve stroke dyads' experience in hospital-to-home transition. The framework emphasises gaps in the transition process, opportunities to bridge the gaps and stroke-nurse-navigator-enabled interprofessional collaboration and partnership with stroke dyads during the transition. The framework consists of two interconnected components: (1) the stroke-specific hospital-to-home transition care foundation, and (2) stroke-nurse-navigator-enabled hospital-to-home transition care in the three key phases—during the

hospital stay, arriving home and during formal rehabilitation programs, and after separation from formal rehabilitation programs and during community dwelling.

11.3 The Stroke-Specific Transition Care Foundation

The foundation emphasises the collaboration and partnerships among stakeholders in transition care, including but not limited to the health and social care systems, health and social care organisations (facilities), health and social care professionals, and stroke dyads in the health and social care systems. The stroke dyads' care needs during the transition are the central focus in the foundation. Each stakeholder plays individual and collaborative roles throughout the transition process. The foundation highlights the need for stroke care stakeholders to work together by implementing strategies that support the stroke-nursenavigator-enabled hospital-to-home transition care.

11.3.1 Health and Social Care Systems

The health and social care system (hereafter, 'health system') plays a crucial role in meeting people's individual needs by providing safe, effective, accessible and appropriate treatment and services (AIHW, 2022a). Australia has a unique hybrid health system (hospital and community services) that combines public and private components, with Medicare serving as the foundation and supplemented by private health insurance. Eligible individuals in Australia can access government-subsidised health services either free of charge or at a low cost (Department of Health and Aged Care, 2019).

Figure 11.1

Proposed Stroke-Specific Hospital-to-Home Transition Care Framework



Stroke dyads use a range of rehabilitation and social care services to support the rehabilitation and community living of people with stroke after hospital discharge. These include hospital outreach services provided in homes, day hospitals or community centres; telehealth options; as well as social care services offered by community aged-care and disability care providers. Community services are typically organised based on age criteria, with government-subsidised aged-care services for people aged 65 years and over (or 50 and over for Aboriginal and Torres Strait Islander people). Disability services are primarily designed for individuals below this age threshold (My Aged Care, n.d.). However, the findings of this study align with the report of the Royal Commission into Aged Care Quality and Safety (2021), which highlighted the inequitable access to disability services for people above the age of 65, who do not meet the eligibility criteria for service such as the NDIS. The participants in this study emphasised the importance of service development based on individual needs rather than age, as it would significantly enhance transition experiences for stroke dyads.

The study's findings indicate the need for investment in psychology and mental health care resources, particularly in rural and remote areas, to enhance timely access to services and address existing gaps (Kable et al., 2019; Prior et al., 2023). The Australian National Stroke Audit (Stroke Foundation, 2020a) revealed that about 31% of rehabilitation services lack access to clinical psychologists or neuropsychologists, despite over 65% of community-dwelling people with stroke reporting unmet psychological needs within 2.5 years after stroke (Tjokrowijoto et al., 2023). Another study exploring health professionals' views on mental health services gaps in Australia advocated for system reforms and increased investment in mental health services to meet the growing demand (Petrie et al., 2021).

The expectations expressed by stroke dyads in this study call for a flexible and proactive health system to empower them to take an active role in their own care. Policy and financial innovations and reforms are necessary to enable stroke dyads to choose their preferred care facilities and providers, timings and settings for care services. Such reforms are particularly crucial for stroke caregivers, who play a vital role in supporting the

sustainability of the health system (Kokorelias et al., 2020). Quality improvement in care services resulting from health system reforms would foster greater engagement of stroke caregivers in care planning, enhance their transition experiences and strengthen sustainable caregiving for people with stroke in the community (Kokorelias et al., 2020; Lobo et al., 2021).

11.3.2 Health and Social Care Organisations (Facilities)

Health and social care organisations encompass a diverse range of public and private entities responsible for delivering health and social care services (AIHW, 2022a). Ensuring safety and quality in healthcare is a collective responsibility that relies on effective governance and management processes and the establishment of systems within these organisations (ACSQHC, n.d.-b). To provide safe and high-quality care in partnership with consumers, health and social care organisations must establish robust governance systems and implement strategies that foster a culture of safety and engage health professionals through leadership, communication and collaboration (ACSQHC, n.d.-b).

'Culture', within an organisation, refers to the shared values and beliefs among different occupational groups, which manifest in the common patterns of behaviour sustained through rituals, ceremonies and rewards of everyday organisational practices (H. T. O. Davies & Mannion, 2013). Leadership, workforce wellbeing and engagement, team coherence, and transparency are key factors that contribute to sustaining a culture of safety and high-quality care, ultimately improving the overall quality of healthcare within health organisations (Mannion & Davies, 2018; Tate et al., 2023). A study by Tate et al. (2023) emphasised the critical role of complementary and relationship-focused leadership styles and cultures in hospitals, highlighting their effect on quality care improvement and patient outcomes.

The findings of the present study reinforce the idea that health and social care organisations need to prioritise patient safety culture, emphasise organisational leadership and governance, and foster team resilience. By cultivating a culture of safety and highquality care, these organisations can promote dyad-centredness, equity, integration and

effectiveness in the provision of care services during hospital-to-home transition care. This entails an enabling environment wherein health professionals, especially stroke nurse navigators, feel empowered to collaborate, communicate effectively and contribute their expertise, ultimately resulting in improved outcomes and experiences for stroke dyads during the transition between different care settings (Adeoye et al., 2019; Lip et al., 2022).

11.3.3 Health and Social Care Professionals

Health and social care professionals involved in post-stroke care have a vital role in supporting stroke dyads throughout the hospital-to-home transition. They work closely with stroke dyads by assessing and understanding the care needs and priorities of stroke dyads; implementing and evaluating fundamentals of care and evidence-based rehabilitation interventions; and providing emotional support (Jarva et al., 2021; Theadom et al., 2018). However, the findings of this study suggest that health and social care professionals can further enhance their partnership and collaboration with stroke dyads during the transition. Incorporating this kind of knowledge into their practice would allow health professionals to better assess the care needs, goals and preferences of stroke dyads; co-develop and implement discharge plans; and proactively address their physical and psychosocial care needs. By understanding the dynamics of dyadic interdependence, professionals can cultivate resilience within stroke dyads and enable them to explore meaningful activities that lead to a purposeful life, regardless of the post-stroke disabilities experienced by the person with stroke.

Furthermore, in this study, despite physical and psychological challenges faced by the stroke dyads during the transition, they demonstrated resilience and the ability to reconcile with the changes and move forward with their changed lives. By adopting a partnership approach to care provision, health professionals can further cultivate resilience and strengthen dyadic coping in the transition period. This collaborative approach fosters a sense of empowerment, trust and shared responsibility, ultimately leading to improved

outcomes and a more positive hospital-to-home transition experience for stroke dyads (ACSQHC, n.d.-b).

Collaboration among professionals across disciplines, organisations and care settings is widely recognised as a crucial strategy to enhance person-centred care, improve patient and system outcomes and prevent silos in care delivery (McLaney et al., 2022; Rawlinson et al., 2021). The WHO defines interprofessional collaboration as the collective practice of multiple health workers from different professional backgrounds working together with patients, families, carers and communities to deliver the highest quality of care across settings (WHO, 2010). The core competencies of interprofessional collaboration encompass fostering a culture of mutual respect, trust and shared decision-making among professionals; valuing the expertise of others to assess and address healthcare needs; facilitating interprofessional communication; and strengthening teamwork to plan, deliver and evaluate safe, timely, effective and equitable healthcare services (Interprofessional Education Collaborative, 2016).

To facilitate interprofessional collaboration, health systems and organisations should establish mechanisms to optimise funding that supports collaborative practice. This can involve allocating resources specifically for interprofessional teamwork, such as dedicated time for team meetings, training programs and shared decision-making processes. Additionally, creating an enabling environment to promote a culture of caring relationships among team members is crucial. This can be achieved by fostering open and respectful communication, promoting a sense of psychological safety and encouraging collaboration and mutual support within the team. Professional development opportunities should be provided to enhance knowledge and skills related to collaborative practice, ensuring that professionals are equipped with the necessary competencies to effectively work together across disciplines. The clear delineation of roles and responsibilities is essential to facilitate interprofessional communication and coordination, minimising confusion and enhancing the efficiency and effectiveness of care delivery (Ansa et al., 2020; Rawlinson et al., 2021; H. Wei et al., 2020).

By embracing authentic partnership and collaboration, health professionals can break down the silos in care delivery and create a more integrated and dyad-centred approach to stroke transition care. This collaborative approach enhances continuity of care, reduces fragmentation, avoids duplication of services and ultimately improves the overall experience and outcomes for stroke dyads (Ansa et al., 2020; H. Wei et al., 2020). However, implementing interprofessional collaboration in daily clinical practice comes with challenges that require further research to establish insights into barriers and facilitators to interprofessional collaboration, inform the development of effective communication and coordination strategies, and promote a culture of collaboration within and across health organisations (Didier et al., 2020; Schot et al., 2020).

11.3.4 Stroke Dyads

Stroke dyads are at the core of this framework's foundation. Stroke dyads play a critical role in their own recovery and care planning. The study's findings emphasise their active involvement and desire to participate in their care, fostering a sense of ownership and empowerment throughout the hospital-to-home transition. They seek a collaborative relationship with health professionals, aiming for shared decision-making and the co-development of care plans.

In the present study, stroke dyads recognised the importance of addressing both physical and psychological care needs during the transition. While the physical rehabilitation of people with stroke is often a primary focus, the study revealed the significance of attending to the psychological wellbeing of both the person with stroke and their caregiver. Health professionals need to recognise the emotional and mental strain experienced by caregivers and provide appropriate support and resources to strengthen caregiving support systems.

Moreover, this study underscores the notion that stroke dyads' care needs evolve over the course of the transition. As dyads navigate different stages and face various challenges, it is crucial for health professionals to regularly reassess and evaluate the dyads' changing care needs. Staying attuned to these evolving needs will enable health

professionals to collaborate with stroke dyads to adjust care plans accordingly and ensure their needs are met effectively. In addition, actively engaging in ongoing communication with the dyads will inform health professionals of strategies for fostering a strong partnership with stroke dyads and promoting a dyad-centred and responsive approach to care planning.

Peer support plays an important role in supporting stroke dyads and is recommended by the *Clinical Guidelines for Stroke Management* as a means of promoting community participation and reintegration (Stroke Foundation, 2023). During the transition, health professionals should provide stroke dyads with information about peer support in the community, particularly before their separation from formal services (Stroke Foundation, 2023). Connecting with peer support allows dyads to share experiences, receive mutual support, overcome social isolation and enhance their problem-solving skills and functional capabilities (Clark et al., 2020; May et al., 2023). Peer support can be delivered individually or in group settings, offering potential benefits such as reduced psychological stress and improved self-management capabilities for the dyads (Clark et al., 2020; May et al., 2023).

Collaboration among all stroke care stakeholders is the cornerstone for successful hospital-to-home transition care for stroke dyads. This involves a stroke-dyad-centred approach, co-created goals and appropriate support from the health system. Active participation and shared decision-making between stroke dyads and health professionals are key to ensuring that care plans are tailored to the dyads' specific needs. By actively involving stroke dyads in the care planning process, their preferences and priorities can be integrated, leading to a more personalised and effective transition experience.

11.4 The Stroke-Nurse-Navigator-Enabled Hospital-to-Home Transition

This framework emphasises the need to enhance the role of the stroke nurse navigator in the hospital-to-home transition care period, as detailed in the following sections.

11.4.1 The Role of Stroke Nurse Navigator

The study's findings highlight the value of dedicated stroke nurse navigators (hereafter, 'navigators') in supporting stroke dyads throughout their journey from inpatient

settings to the community. These specialised stroke nurses, practising at an advanced level, play a crucial role in facilitating and managing care transitions, which leads to improved continuity of care, reduced service fragmentation, enhanced quality of care for stroke dyads and decreased hospital readmissions (Byrne, Hegney et al., 2020; Camicia, Lutz, Summers et al., 2021). While the role of nurse navigators was initially introduced in the United States to address gaps in oncology care (Freeman, 2006), it has also been implemented internationally, including in Australia, to benefit people with chronic conditions, including stroke, and strengthen comprehensive support (Byrne, Hegney et al., 2020; Deen et al., 2016).

In Australia, the role of stroke clinical coordinator, also known as a stroke liaison officer or stroke facilitator, has been established as an integral part of stroke care and is recommended in the National Acute Stroke Services Framework (Purvis et al., 2018; Stroke Foundation, 2019a). Stroke coordinators fulfil various responsibilities across different settings, including stroke units in acute hospitals and community settings. These responsibilities encompass facilitating the patient journey from the emergency department to discharge, discharge planning, patient and professional education, and service development (Purvis et al., 2021). Previous studies have demonstrated that the implementation of the stroke coordinator role is associated with reduced hospital length of stay and the delivery of evidence-based care in hospitals with a stroke unit (Purvis et al., 2018), as well as improved quality of care during the hospital-to-home transition (Miller et al., 2019). However, the coordinator role in these studies focused either on the acute post-stroke phase in the hospital (Purvis et al., 2018) or the post-discharge phase in the community (Miller et al., 2019). In contrast, the present study advocates for the role of a navigator to follow-up with stroke dyads throughout the entire hospital-to-home transition continuum (Camicia, Lutz, Summers et al., 2021).

In 2016, Queensland Health introduced nurse navigators as a team of advanced practice registered nurses responsible for delivering comprehensive clinical care to patients with complex health conditions, including stroke (Queensland Health, 2019). The nurse

navigators' role is guided by key principles that include establishing partnerships, coordinating consumer-centred care, facilitating system improvement and enhancing consumer outcomes (Queensland Health, 2019). The use of nurse navigators has demonstrated positive outcomes, such as reducing unplanned 30-day readmissions, improving post-discharge compliance and enhancing the quality of life for stroke dyads (Deen et al., 2016; Hudson et al., 2019; Jun-O'Connell et al., 2023). Nevertheless, in contrast to previous studies, the present study revealed the importance of stroke navigators collaborating with stroke dyads throughout the entire hospital-to-home transition and offering opportunities for stroke dyads to navigate back to the healthcare system when needed (Camicia, Lutz, Summers et al., 2021). This approach may open up a new avenue for addressing long-term care needs and better sustaining community-dwelling for stroke dyads (Scrivener et al., 2022).

To effectively support stroke dyads during the hospital-to-home transition, navigators should possess knowledge about the health systems, available resources and care services, relevant organisations, and professionals involved in post-stroke care. They should also understand dyadic interactions within stroke dyads, including their unique care needs, preferences and factors influencing their recovery and rehabilitation goals. Navigators must demonstrate proficiency in providing comprehensive support to stroke dyads, including resource and service navigation, emotional and psychological assistance, education, and coordination of services. Their interventions should aim to alleviate the stress and uncertainty associated with the transition back to home and community settings (Byrne, Hegney et al., 2020; Camicia et al., 2020).

Furthermore, navigators need to facilitate effective communication within the stroke team and across disciplines, organisations and care settings. It is crucial to rigorously evaluate the role of stroke navigators in terms of stroke dyads' overall transition experience, improving health outcomes and reducing hospital readmissions and healthcare costs (Mitchell et al., 2022; Mora et al., 2017). To ensure the presence of knowledgeable and skilled stroke navigators, governments at the federal or state levels need to fund their

position in the healthcare system. Moreover, nursing professional bodies (e.g., the Australasian Rehabilitation Nurses' Association and Acute Stroke Education Network) will need to advocate for the establishment of stroke navigators in the healthcare system. In addition, healthcare organisations need to create the position and work with nursing education providers to co-design and co-produce competent navigators using flexible, ongoing, work-integrated learning approaches or micro-credential courses based on navigators' learning needs and experiences (Desmarchelier & Cary, 2022; Karlsson et al., 2022; Purvis et al., 2022).

11.4.2 Key Transition Phases

The framework introduces a novel approach with three key transition phases derived from the findings of this study. While previous studies have examined the experiences of stroke dyads during the hospital-to-home transition period, they have not identified pivotal turning points that influence stroke dyads' care needs and progression throughout the transition, particularly the key point of arriving home (Connolly & Mahoney, 2018; Kable et al., 2019; Pereira et al., 2021). The key phases identified in this study were (1) during the hospital stay, (2) arriving home and during formal rehabilitation programs, and (3) after separation from formal rehabilitation programs.

11.4.2.1 During the Hospital Stay

The present study, along with other abundant research evidence, has demonstrated that preparation before hospital discharge and active engagement of stroke dyads in discharge planning significantly affects their ability to manage care at home, particularly for stroke caregivers who assume their role within a short timeframe (Camicia et al., 2022; L. Chen et al., 2022). The present study revealed the interdependence between stroke dyad members, highlighting the need for comprehensive care needs assessments for both the person with stroke and the caregiver. It is important to recognise that, while individuals with stroke may regain independence in walking and self-care, they may struggle with sharing household tasks with their caregivers. Consequently, caregivers often face unexpected burdens and stress after hospital discharge (L. Chen et al., 2022; Camicia et al., 2022). A

dyad-centred approach to care needs assessment becomes crucial, enabling the stroke care team to provide tailored education, share relevant information and resources, and address stroke dyads' specific needs within their unique life situations. This approach enhances engagement in discharge planning, thereby paving the way for the transition journey after leaving hospital.

Navigators play a vital role in collaboration with interdisciplinary teams at stroke units and rehabilitation wards to ensure coordinated care. They achieve this by maintaining effective and efficient communication, actively participating in dyadic care needs assessment, promoting stroke education, providing emotional support and facilitating the transition from the acute setting to rehabilitation. Prior to hospital discharge, navigators contribute to discharge planning by fostering dyadic engagement in the process. This includes facilitating knowledge and skills preparation, particularly for caregivers and collaborating with inpatient and post-discharge service providers to ensure the implementation of discharge plans that are tailored to the needs and preferences of stroke dyads.

11.4.2.2 Arriving Home and During Formal Rehabilitation Programs

The present study highlights the period immediately after arriving home as the most vulnerable phase for stroke dyads. This period refers to the first few weeks following arrival home, during which this study's participants experienced a stark contrast between the controlled hospital environment and the new post-stroke and post-hospitalisation home environment. To mitigate challenges during this phase, it is beneficial for individuals with stroke to familiarise themselves with the home environment before hospital discharge through overnight stays or day visits at home (Camicia et al., 2014; L. Chen et al., 2020). This practice enables stroke dyads, particularly those with lengthy hospital stays, to identify potential issues; facilitates clear goal-setting during inpatient rehabilitation; and addresses any concerns proactively.

Furthermore, service delivery should allow for flexibility, giving stroke dyads sufficient time to settle into their home environment before community-based rehabilitation and care

services commence. While seamless service delivery aims to bridge the gap between hospital and home, it is essential to negotiate service arrangements based on each individual's specific needs to ensure proper alignment. Mismatches between provided services and actual care needs can intensify the challenges faced by stroke dyads upon returning home. Therefore, in addition to tailored rehabilitation programs for individuals with stroke, it is imperative that there is psychosocial support for caregivers, social care support to alleviate caregiving burden and support from social networks. The stroke care team should also engage with the primary care team in the community to explore alternatives, such as peer support groups, that could help establish a support system for those with limited social networks.

Before separation from formal rehabilitation services, healthcare professionals need to build confidence for the dyads in navigating the healthcare system, developing selfmanagement capabilities and adapting to post-stroke limitations in daily life. It is crucial to engage in dyadic assessments, preparing stroke dyads for separation by facilitating goalsetting, trialling meaningful activities and sharing information about available community resources, stroke-specific rehabilitation programs, social care support and peer support groups. Additionally, informing stroke dyads about pathways to refer back to formal rehabilitation services when needed enhances ongoing support.

During this phase of the transition, navigators maintain regular contact with stroke dyads through various means, such as phone calls, telehealth technology or home visits, depending on the specific needs of the dyads. In the present study, participants expressed their expectation for stroke navigators to serve as the central point of contact, allowing them to share their concerns, discuss their thoughts about their own health and seek feedback on their future plans. Navigators also play a crucial role in assisting stroke dyads in developing self-management capabilities. This includes helping them devise strategies to enhance resilience and psychological wellbeing, acquiring knowledge about the healthcare and social care systems, identifying available support services, and establishing meaningful connections with community peer support groups or individuals. Additionally, stroke

navigators maintain collaborative relationships with community rehabilitation, primary care, and social care teams to effectively address any challenges that may arise during the transition phase.

11.4.2.3 After Separation From Formal Rehabilitation Services and During Community Dwelling

In the present study, it was observed that, 6 months after hospital discharge, the majority of participants had completed formal rehabilitation programs, with or without followup community rehabilitation services, except for one participant. The participants expressed an expectation for a gradual decrease in the intensity of rehabilitation therapies over an extended period rather than a sudden cessation at a specific point. Previous research has indicated that post-stroke functional improvement may slow down or plateau after 6 months from the onset of stroke (Grefkes & Fink, 2020; Scrivener et al., 2022). However, it is important to note that a significant number of individuals experience deterioration over time and continue to live with higher levels of disability (Scrivener et al., 2022). This is further compounded by the presence of comorbidities, such as cardiac diseases and diabetes, which can further complicate community living and adversely impact the quality of life for stroke dyads (Scrivener et al., 2022).

To continue the support for stroke dyads in the chronic phase of the stroke trajectory, the healthcare system needs to reform and ensure ongoing access to disability services for stroke dyads, irrespective of their age, to address their care needs and support their community dwelling (Scrivener et al., 2022). Navigators play a key role as intermediaries between stroke dyads and the healthcare system. They provide support to stroke dyads, facilitating connections with peer support groups, community health and social care teams, and primary care providers when the need arises. It is essential to recognise that there should be no time limit for follow-up, and navigators should be a government-subsidised, needs-driven and self-referred service available to stroke dyads. However, it is important to acknowledge that the role of navigators alone cannot fully address the complex care needs of stroke dyads without the support of health systems. Collaborative efforts between

navigators and other healthcare providers are necessary to provide comprehensive and holistic care for stroke dyads in the long term.

11.5 Summary

This chapter has discussed the proposed stroke transition care framework. The proposed framework offers a novel perspective on the hospital-to-home transition for stroke dyads, highlighting the crucial role of navigators in supporting them throughout the entire stroke trajectory. It emphasises the unique needs and necessary support at various phases of the transition, which have a significant influence on the transition experience and long-term outcomes for stroke dyads. The presentation of this framework aims to inspire meaningful discussions among stroke care stakeholders about the importance of the hospital-to-home transition and its profound impact on the ability of stroke dyads to live and thrive in the community over the long term.

The next chapter reviews the research findings, identifies study limitations, draws conclusions from the study, makes recommendations for stroke hospital-to-home transition care and discusses the implications of the findings for clinical practice and future research.

Chapter 12: Conclusion

12.1 Introduction

This two-phase qualitative study has effectively achieved its aim of exploring the care needs and expectations of stroke dyads during the hospital-to-home transition, as perceived by both stroke dyads and health professionals, in the Australian social context. The study built upon the evidence generated by a comprehensive systematic review and meta-synthesis of qualitative studies and employed Gadamer's philosophical hermeneutics to guide data collection, analysis and interpretation (Gadamer, 1979). Meleis's middle-range transition theory (Meleis et al., 2000) was also used to comprehend the complexity and multidimensionality of the transition and to aid the generation of a transition care framework. The study's findings and proposed framework offer a novel perspective on post-stroke transition care that emphasises the dyadic interdependence of caregivers and care recipients, as well as the importance of interprofessional team culture and resilience. This final chapter critically evaluates the limitations of the study; outlines implications and recommendations for practice, policy, and future research; and provides a reflective account of the researcher's experience.

12.2 Limitations

The study has several limitations that need to be acknowledged. First, the generalisability and replicability of the findings may be limited due to recruitment from a single rehabilitation centre and the exclusion of non-English speakers. Moreover, the study's hermeneutic approach is rooted in specific social and cultural contexts, which may limit the transferability of the findings to other contexts. However, the existing literature supports the study, suggesting that the findings may be transferable to similar social contexts. Furthermore, the researcher acknowledges that no participants from Aboriginal or Torres Strait Islander backgrounds or culturally and linguistically diverse backgrounds expressed

interest in participating in the study. Therefore, the findings may not represent the perspectives of these minority ethnic groups and should be interpreted accordingly.

Second, the study's sample only included participants with mild cognitive or speech impairment. Thus, the findings may not reflect the experiences of those with moderate to severe cognitive or speech impairment. Moreover, recall bias may have affected the narratives of two dyads whose first interview occurred within 2 weeks after hospital discharge. Additionally, three participants with mild cognitive impairment and their caregivers were interviewed together to cross-check the narratives, which may have suppressed free discussions from each side due to family dynamics. However, the same interviewer conducted all the interviews, which helped to achieve a level of internal consistency.

Third, the study only managed to recruit two GPs, which may limit the understanding of stroke dyads' transition experience from the perspectives of primary care teams, including community care service providers. Future research could incorporate this consideration. Furthermore, health professional participants were colleagues of the interviewer, which may have influenced the full expression of participants' thoughts. Therefore, this needs to be considered when interpreting the findings.

12.3 Summary of the Study

Stroke is a sudden brain injury that leads to long-term physical, psychological and cognitive sequelae (Grefkes & Fink, 2020; Scrivener et al., 2022). As stroke management advances, more people are surviving stroke and returning home from the hospital, where they are supported by caregivers while living with disabilities (Stroke Foundation, 2022; Zimmerman et al., 2021). The hospital-to-home transition (hereafter, 'transition') poses significant challenges for stroke dyads to navigate complex care needs and face the global trend of shortened hospital stays (Camicia, Lutz, Harvath & Joseph, 2021; L. Chen et al., 2020). Despite the demonstrated dyadic interdependence during the transition, there is a lack of research exploring the individual and dyadic care needs and expectations of stroke dyads.

Phase 1 of this research project involved a systematic review and meta-synthesis of qualitative studies with the primary aim of examining the experiences and care needs of stroke dyads during the transition. This meta-synthesis revealed various enablers and barriers in transition care. Health professionals' partnerships with stroke dyads facilitated discharge preparation and helped stroke dyads build self-management capabilities and competence in navigating health systems. Stroke dyads attributed their unmet care needs to inadequate discharge planning and post-discharge support, which adversely affected their ability to cope with post-stroke changes. Moreover, stroke dyads emphasised the need for integrated transitional care that promotes partnership and shared decision-making and empowers long-term self-management at home. These findings, encompassing a global perspective, offered a comprehensive understanding of stroke dyads' experiences, needs and expectations during transition, which paved the way for Phase 2 of this research project.

Theories and concepts applied to transition care for stroke dyads were analysed to identify a suitable theoretical framework to inform this study. The middle-range transitions theory developed by Meleis et al. (2000) was viewed as relevant to this study. Meleis's transition theory has been widely used and tested in various care settings with diverse populations. Acknowledging engagement, change, time and critical points as key properties of transition, the theory emphasises the dynamic interactions among the person, community and society, as well as the role of nursing therapeutics in facilitating healthy transition. The concepts outlined in Meleis's transition theory informed the analysis and interpretation of potential factors influencing the transition for stroke dyads in this study.

Gadamer's (1979) philosophical hermeneutics guided the qualitative study in Phase 2 of this project. Central to Gadamer's hermeneutics is a linguistically mediated dialogical approach that encourages the engagement of the researcher in the interpretation process and is referred to as a hermeneutic circle. A deeper understanding (new knowledge) of the studied subject is achieved through the cyclical fusion of horizons between the researcher's pre-knowledge (the past) and study participants' perspectives (the present). In this process, the researcher maintains an open mind and consciously identifies, provokes,

and revises the pre-knowledge (Gadamer, 1979). Gadamer's hermeneutics guided the researcher to design this study, collect and interpret data, and formulate the findings.

Phase 2 was a longitudinal qualitative study that took place in a tertiary public hospital in Adelaide, South Australia. Thirty people—people with stroke and their caregivers—had face-to-face interviews separately before hospital discharge and at 3 and/or 6 months after discharge. In addition, 31 health professionals participated in the study through face-to-face interviews or focus groups.

The study's findings highlight the complexity and multidimensionality of the transition for stroke dyads. Participants viewed that psychological care for stroke dyads could be further enhanced by health professionals' equal attention to both functional recovery and psychological care needs for stroke dyads. Additionally, stroke dyads' perceived dyadic interdependence positively impacted their adaptation to challenging situations. Caregivers played a vital role in psychological support for people with stroke, but such a role was not fully recognised nor supported in the healthcare system. Moreover, stroke dyads expected enhanced engagement and shared decision-making in discharge planning and needs-driven service provision. Furthermore, health professionals believed that their commitment to safe and quality care was enhanced by the workplace safety culture that was fostered through leadership and team resilience.

The findings of this study led to the proposal of an innovative transition care framework. The foundation of the framework is the partnership and collaboration among various stakeholders: health systems, healthcare organisations, health professionals and stroke dyads. Built on this foundation is navigator-enabled dyad-centred transition care at three key phases: (1) during the hospital stay, (2) arriving home and during formal rehabilitation programs, and (3) after separation from formal rehabilitation programs and during community dwelling. This novel transition care framework has implications for clinical practice and future research.

12.4 Implications

Findings of the longitudinal qualitative study on the transition for stroke dyads have implications for policy development, health organisations, health professional education and future research. The implications are discussed in this section.

12.4.1 Implications for Policy Development

Despite advancements in stroke management, stroke dyads in Australia face challenges in accessing equitable and timely best-practice stroke and rehabilitation services, especially in rural and regional areas (Royal Commission into Aged Care Quality and Safety, 2021; Stroke Foundation, 2019a, 2022). Policy development and resource allocation are needed to address these challenges and to support stroke dyads during the transition. The findings of this study have implications for policies and resources to enhance the health, wellbeing and social integration of stroke dyads.

12.4.1.1 Implication for Policy Development to Remove Age Threshold of Disability Services for Equal Access

Stroke dyads residing in the community have ongoing needs related to physical functioning and psychosocial wellbeing, which are crucial for maintaining their independent living (Guo et al., 2021; Scrivener et al., 2022). *Australia's Disability Strategies 2021–2031* also emphasises the importance of ensuring that individuals with disabilities have access to necessary support (National Disability Insurance Agency, 2021). However, both the findings of this study and the existing research highlight inequitable access to disability services for people with stroke aged 65 and above, who constitute 71% of the stroke population (AIHW, 2023; Royal Commission into Aged Care Quality and Safety, 2021; Scrivener et al., 2022).

Currently, discrepancies exist between the services available under the NDIS and those provided through the aged-care system, with significantly higher funding allocated to the NDIS (Royal Commission into Aged Care Quality and Safety, 2021). Research evidence emphasises the need for the government to review existing disability care and aged-care policies and eliminate the age threshold for these services. This policy change would ensure

that stroke dyads have access to services that align with their specific needs and life circumstances, regardless of their age.

12.4.1.2 Implication for Policy Development to Increase Funding in Psychological Resources for Enhanced Post-stroke Psychological Wellbeing

Research evidence highlights the significant impact of psychological wellbeing on rehabilitation outcomes for stroke dyads (Ezema et al., 2019; Perna & Harik, 2020). The importance of providing psychological support to stroke dyads is also emphasised in Australia's National Rehabilitation Stroke Services Framework (Stroke Foundation, 2022). However, the findings of this study indicate that there is a shortage of psychological care needs during the transition for stroke dyads. Therefore, adequate funding and allocation of resources are needed to establish a robust workforce capable of meeting the psychological care needs of stroke dyads, including those in rural and regional areas. Policy reforms, by investing in and expanding psychological resources, can bridge the existing gap in care and promote optimal psychological support for stroke dyads throughout their transition journey.

12.4.1.3 Implication for Policy to Invest in the Evidence-Based Nurse-Led Transition Care Services

The present study proposed a framework that holds significant implications for clinical practice. This framework presents an opportunity to establish strong connections between acute care, rehabilitation, primary care and community services to address the care needs of stroke dyads during the transition phase and to address existing gaps after formal rehabilitation programs. In the current healthcare system, the collaboration across care settings and care facilities is lacking, and stroke dyads have limited opportunities to be reassessed and referred for care services they need (Stroke Foundation, 2021).

The proposed role of navigator has the potential to bridge these gaps by serving as a central point of communication and engagement. The navigator would identify the care needs of stroke dyads, facilitate appropriate referrals to services at the right time and in the right setting, collaborate with interdisciplinary teams to solve problems, and provide

opportunities for stroke dyads to access rehabilitation services when required. Investment in research is necessary to test the feasibility, effectiveness and potential cost-effectiveness of a navigator model of care as shown in the proposed framework.

Allocating resources and funding to develop the role of navigators would greatly benefit stroke dyads as they navigate transition. It would enhance their ability and independence to cope with the challenges of the transition, improve their quality of life and support their long-term community living. Investment in evidence-based nurse-led transition care services has the potential to transform the care experiences of stroke dyads, ensuring they receive comprehensive and coordinated care throughout their transition journey.

12.4.2 Implications for Health Organisations

Health organisations play a crucial role in delivering safe, high-quality and consumercentred care (ACSQHC, 2021). The study's findings underscore the importance of a workplace safety culture and team resilience in enhancing care for stroke dyads during the transition. These implications are crucial for health organisations as they strive to improve the quality of care provided during this process.

12.4.2.1 Implication for Staff Wellbeing Focus to Enable Team Resilience Development

The study's findings highlight the importance of team resilience in fostering collaboration and partnership with stroke dyads during the transition. Health professionals often encounter numerous challenges in their daily clinical practice, including time pressure and staffing shortages. However, the study demonstrated that team resilience was nurtured through several factors, including effective leadership, open and transparent communication, trust among team members, and the presence of peer support networks. By prioritising staff wellbeing, organisations can contribute to the development of team resilience and enhance the ability of health professionals to cope with the challenges they face in daily practice (Mills et al., 2020; Wald, 2020).

Organisations can support staff wellbeing by implementing strategies such as promoting self-care and stress management, fostering a culture of support and collaboration,

and providing resources for professional development. Strong leadership is crucial for creating a positive work environment that values staff wellbeing. Leaders should actively engage with their teams, promote open communication, recognise and address concerns, and provide opportunities for feedback and reflection. (Mills et al., 2020; Wald, 2020).

12.4.2.2 Implication for Support to Develop and Sustain the Role of Stroke Nurse Navigators

The role of navigator emerged from the study's findings to enhance care transitions by supporting stroke dyads to build capacities for decision-making and self-management as they transition through the post-stroke trajectory. The nurse navigator is an established role in cancer care, primary health care and the management of chronic conditions. The role has generated positive outcomes by providing timely and culturally appropriate guidance and support and improving service integration (Byrne, Hegney et al., 2020; McMurray & Cooper, 2017). However, previous studies have identified barriers to implementing the nurse navigator's role. The challenges included role confusion, inadequate resources to support the role (e.g., insufficient time to perform the role), inconsistent methods for orientation and role implementation, and a lack of professional development opportunities (Cantril et al., 2019; Purvis et al., 2022).

Healthcare organisations should collaborate with education providers to develop tailored education modules or micro-credential courses for navigators. These programs should address their learning needs and experiences, equipping them with the necessary clinical skills and understanding of healthcare systems and available services for stroke dyads. Clear role descriptions, well-structured orientation and continuous professional development opportunities are essential. Additionally, careful evaluation of the navigator's role is crucial to ensure sufficient resources, such as time and staffing, to support their responsibilities. Workforce estimation should be aligned with individual needs and evidencebased interventions rather than a fixed staff-to-patient ratio (Stroke Foundation, 2022).

12.4.3 Implications for Health Professionals

Interdisciplinary teams, including nurses and other health professionals, play a pivotal role in the post-stroke trajectory of stroke dyads by partnering with them to set individualised goals and engage in decision-making and care planning. The findings of this study have important implications for nurses and other health professionals to enhance their collaboration with stroke dyads through dyad-centred approaches and holistic care needs assessments.

12.4.3.1 Implication for Nursing Profession to Recognise Core Competencies in Stroke Transition Care

The findings of this study shed light on the under-recognised and under-identified role of nurses in transition care for stroke dyads. To bridge this gap, it is crucial for nurses to identify and embrace their core competencies in this area. Patient education, care coordination and advocacy, counselling and emotional support, and leadership are integral nursing competencies that should be recognised and prioritised (Nursing and Midwifery Board, 2016).

Nurses need to be empowered with the knowledge and skills necessary to implement dyad-centred care and strengthen their partnership with stroke dyads. This includes a deep understanding of the physical and psychological needs of both people with stroke and their caregivers during the transition. Nursing education programs should prioritise enhancing competencies such as effective communication, conflict resolution, critical thinking and care coordination within interdisciplinary teams, enabling nurses to apply these skills in clinical practice (Godsey et al., 2020).

Healthcare organisations also play a vital role in supporting the nursing profession in stroke transition care. They should develop nursing leadership advancement pathways and provide opportunities for nurses to actively participate in health policy decision-making. By empowering nurses and recognising their expertise, organisations can create an environment that fosters professional growth and enhances the impact of nursing in stroke care (Godsey et al., 2020; Hajizadeh et al., 2021).

12.4.3.2 Implication for Health Professionals to Strengthen Partnership With Stroke Dyads

This study highlights the significance of the partnership between health professionals and stroke dyads in facilitating the transition. However, the study's findings also indicate that this partnership could be further strengthened through the adoption of dyad-centred approaches by health professionals, addressing both the physical and psychological care needs of stroke dyads equally.

Health professionals should develop insights into the significant role of psychological wellbeing in the transition for stroke dyads, particularly in relation to caregivers' psychological wellbeing. The study's findings reveal that caregivers often conceal their emotional stress to support people with stroke during the transition. Therefore, routine clinical practice should include psychological assessments for caregivers to raise awareness of their psychological care needs. By doing so, appropriate support can be provided throughout the entire post-stroke journey, with special attention given to the transition.

Stroke caregivers are vital partners in stroke care, and their involvement is invaluable for health professionals when devising tailored care plans to support post-stroke recovery. Health professionals should systematically assess caregivers to identify any risks they may face in their caregiving role and to determine the necessary support required to sustain their caregiving efforts (Camicia et al., 2022). The positive outcomes observed in this study further underscore the notion that stroke affects not only the person with stroke but also their caregivers and family members. Therefore, it is crucial for health professionals to implement dyad-centred approaches to foster authentic partnerships during the transition period. Additionally, when establishing workforces for stroke care services, health organisations should consider evidence-based practices and the specific care needs of stroke dyads, rather than solely focusing on staffing numbers, to promote dyad-centred practice (Stroke Foundation, 2022).
12.4.4 Implications for Future Research and Education

The study introduces a nurse-led framework for stroke-specific transition care, offering potential avenues for future research to investigate its implementation. Additionally, the study's findings underscore the need for enhancements in healthcare education.

12.4.4.1 Implications for Future Research

The participant sample in this study consisted solely of English-speaking Australian Caucasians. To expand the scope of knowledge in post-stroke transition care, future research should aim to include perspectives from Aboriginal or Torres Strait Islander people, as well as those from culturally and linguistically diverse populations. Incorporating their experiences and expectations would enrich our understanding of stroke transition care and contribute to the development of culturally sensitive stroke care services.

Furthermore, this study proposes the role of navigators as leaders of interdisciplinary teams to facilitate post-stroke transition care. To advance this concept, future research could focus on the development of their role in working with stroke dyads, employing a co-design approach. Additionally, pilot studies could be conducted to assess the effectiveness of nurse-led transition care interventions, incorporating health economics to evaluate their cost-effectiveness and feasibility for implementation in clinical practice. Moreover, employing a knowledge translation approach would be valuable in conducting trials of the intervention and guiding the implementation of interventions within larger stroke populations.

12.4.4.2 Implications for Healthcare Education

To enhance the continuous professional development of navigators, tailored microcredential courses at a post-graduate level would be beneficial (Desmarchelier & Cary, 2022; Karlsson et al., 2022). Additionally, for undergraduate education, it is recommended to provide equal emphasis on both technical nursing skills and interprofessional skills, including communication, critical thinking and care coordination within interdisciplinary teams. Equipping nursing graduates with these skills before entering the workforce is crucial (Godsey et al., 2020). In the context of an increasingly complex healthcare system, it is essential to integrate interprofessional education into undergraduate programs for all healthcare professionals. This approach fosters collaboration among professionals from various disciplines and breaks down silos. Ultimately, such interprofessional collaboration can significantly enhance health outcomes (van Diggele et al., 2020).

12.5 Reflection

As a nurse unit manager actively involved in the rehabilitation journey and discharge planning for stroke dyads in an inpatient setting, this study has provided the researcher with a valuable opportunity to gain profound insights into the complex and multifaceted nature of transition care for stroke dyads. Reflecting on the study's findings has prompted the researcher to examine her own clinical practice and to take proactive steps towards developing strategies for quality improvement and effecting positive changes.

The study's findings have empowered the researcher to initiate quality improvement within the inpatient rehabilitation setting. Specifically, these initiatives aim to enhance continence management and promote better information-sharing regarding post-stroke intimacy and sexuality for stroke dyads. Additionally, the findings have inspired the formation of a stroke lived-experience peer support team dedicated to improving emotional support for stroke dyads within the inpatient setting.

Moreover, the framework proposed in this study has motivated the researcher to pursue new research endeavours. These research projects will focus on the development and implementation of the navigator role within a clinical context, employing co-design and knowledge translation approaches.

Upon reflecting on the journey of this study, the researcher realised that Gadamer's philosophical hermeneutics was the most suitable methodology. This approach allowed the researcher to incorporate her pre-knowledge of transition care into the interpretation process, known as the hermeneutic circle. The researcher's pre-knowledge was acquired

269

through her extensive experience in stroke care clinical practice, self-reflection on current practices and dedication to two systematic reviews (L. Chen et al., 2020, 2021).

Furthermore, an understanding was cultivated and evolved through a fusion of horizons, achieved through dialogues between the participants and the researcher during the data collection phase, as well as dialogues between the textual data (transcripts and field notes) and the researcher during the data analysis phase. In the interpretation process, the researcher consistently connected participants' perspectives with findings from previous studies (historical awareness) to validate newly generated understandings arising from the dialogues.

Additionally, the researcher refined her pre-knowledge and developed new insights by reflecting on policies and standards that guide current clinical practices. This deeper understanding of post-stroke transition care emerged through a collaborative process between the participants and the researcher, with knowledge being co-developed through their mutual engagement.

12.6 Conclusion

The transition period for stroke dyads poses significant challenges and can be overwhelming, especially with shortened hospital stays. The study's findings shed light on both the hindrances and facilitators encountered during this transition from the perspectives of stroke dyads and health professionals.

First, enhancing the rehabilitation experience and preparation for hospital discharge requires a comprehensive approach that addresses both the physical and psychological care needs of stroke dyads. Caregivers' needs should be thoroughly assessed to ensure their wellbeing is considered alongside the needs of people with stroke.

Second, recognising the interdependence within stroke dyads can inspire health professionals to adopt a dyad-centred approach, fostering a strong partnership with stroke dyads and involving them in their own care planning.

270

Third, safety culture, team resilience and interprofessional collaboration play crucial roles in facilitating the continuity of care and enhancing the transition for stroke dyads.

Furthermore, the proposed stroke transition care framework, which incorporates the role of navigators, offers a promising avenue for improving post-stroke transition care. The study's findings have implications for policy development, health organisations and health professionals. This study has also informed future research to improve care across different settings and service providers for stroke dyads.

References

- Abreu, P., Magalhães, R., Baptista, D., Azevedo, E., Silva, M. C., & Correia, M. (2020).
 Readmissions and mortality during the first year after stroke—Data from a population-based incidence study. *Frontiers in Neurology*, *11*, Article 636.
- Aburn, G., Gott, M., & Hoare, K. (2016). What is resilience? An integrative review of the empirical literature. *Journal of Advanced Nursing*, *72*(5), 980–1000.

Accelerating Care Transformation Centre. (2023). *Improving chronic illness care*. <u>https://www.act-center.org/our-work/primary-care-transformation/chronic-illness-care</u> care/improving-chronic-illness-care

- Achilike, S., Beauchamp, J. E., Cron, S. G., Okpala, M., Payen, S. S., Baldridge, L., Okpala, N., Montiel, T. C., Varughese, T., & Love, M. (2020). Caregiver burden and associated factors among informal caregivers of stroke survivors. *Journal of Neuroscience Nursing*, *52*(6), 277–283.
- Adeoye, O., Nyström, K. V., Yavagal, D. R., Luciano, J., Nogueira, R. G., Zorowitz, R. D.,
 Khalessi, A. A., Bushnell, C., Barsan, W. G., & Panagos, P. (2019).
 Recommendations for the establishment of stroke systems of care: A 2019 update: A policy statement from the American Stroke Association. *Stroke*, *50*(7), e187–e210.
- Ahn, D.-H., Lee, Y.-J., Jeong, J.-H., Kim, Y.-R., & Park, J.-B. (2015). The effect of poststroke depression on rehabilitation outcome and the impact of caregiver type as a factor of poststroke depression. *Annals of Rehabilitation Medicine*, *39*(1), 74–80. <u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4351498/pdf/arm-39-74.pdf</u>
- Aho, K., Harmsen, P., Hatano, S., Marquardsen, J., Smirnov, V., & Strasser, T. (1980).
 Cerebrovascular disease in the community: Results of a WHO collaborative study.
 Bulletin of the World Health Organization, 58(1), 113–130.

Agency for Healthcare Research and Quality. (2014, June). Care Coordination Measures Atlas update. Chapter 2. What is care coordination? https://www.ahrg.gov/ncepcr/care/coordination/atlas/chapter2.html

Akinwuntan, A., Hu, X., Terrill, A. L., Burns, S. P., Hay, C. C., & Belagaje, S. R. (2021).
Young stroke: Resources for patients, their families, and caregivers for long-term community living. *Archives of Physical Medicine and Rehabilitation*, *102*(5), 1035–1039.

- Allen, L., Richardson, M., McIntyre, A., Janzen, S., Meyer, M., Ure, D., Willems, D., & Teasell, R. (2014). Community stroke rehabilitation teams: providing home-based stroke rehabilitation in Ontario, Canada. *Canadian Journal of Neurological Sciences*, *41*(6), 697–703.
- Almborg, A. H., Ulander, K., Thulin, A., & Berg, S. (2008). Patients' perceptions of their participation in discharge planning after acute stroke. *Journal of Clinical Nursing*, *18*(2), 199–209. <u>https://doi.org/10.1111/j.1365-2702.2008.02321.x</u>
- Almborg, A. H., Ulander, K., Thulin, A., & Berg, S. (2009). Discharge planning of stroke patients: The relatives' perceptions of participation. *Journal of Clinical Nursing*, *18*(6), 857–865. <u>https://doi.org/10.1111/j.1365-2702.2008.02600.x</u>
- Alsaigh, R., & Coyne, I. (2019). Mothers' experiences of caring for children receiving growth hormone treatment. *Journal of Pediatric Nursing*, *49*, e63–e73.

American Psychological Association. (2023). Resilience.

https://www.apa.org/topics/resilience

American Psychological Association. (2019). *The road to resilience*. https://www.apa.org/helpcenter/road-resilience

Andrew, N. E., Busingye, D., Lannin, N. A., Kilkenny, M. F., & Cadilhac, D. A. (2018). The Quality of discharge care planning in acute stroke care: Influencing factors and association with postdischarge outcomes. *Journal of Stroke and Cerebrovascular Diseases*, 27(3), 583–590. <u>https://doi.org/10.1016/j.jstrokecerebrovasdis.2017.09.043</u> Anrather, J., & Iadecola, C. (2016). Inflammation and stroke: An overview. Neurotherapeutics, 13(4), 661–670. <u>https://doi.org/10.1007/s13311-016-0483-x</u>

- Ansa, B. E., Zechariah, S., Gates, A. M., Johnson, S. W., Heboyan, V., & De Leo, G. (2020).
 Attitudes and behavior towards interprofessional collaboration among healthcare professionals in a large academic medical center. *Healthcare*, *8*(3), Article 323.
- Ashcraft, S., Wilson, S. E., Nyström, K. V., Dusenbury, W., Wira, C. R., & Burrus, T. M.
 (2021). Care of the patient with acute ischemic stroke (prehospital and acute phase of care): Update to the 2009 comprehensive nursing care scientific statement: A scientific statement from the American Heart Association. *Stroke*, *52*(5), e164–e178.
- Atteih, S., Mellon, L., Hall, P., Brewer, L., Horgan, F., Williams, D., & Hickey, A. (2015).
 Implications of stroke for caregiver outcomes: Findings from the ASPIRE-S study.
 International Journal of Stroke, *10*(6), 918–923. <u>https://doi.org/10.1111/ijs.12535</u>

Australian Bureau of Statistics. (2013). Young adults: then and now.

https://www.abs.gov.au/ausstats/abs@.nsf/lookup/4102.0main+features40april+2013

Australasian Rehabilitation Nurses' Association. (2018). *Rehabilitation nursing competency standards for registered nurses.*

https://www.arna.com.au/ARNA/ARNA/Resources/Competency_Standards.aspx

- Australian Commission on Safety and Quality in Health Care. (n.d.-a). *About patient safety culture*. <u>https://www.safetyandquality.gov.au/our-work/indicators-measurement-and-</u> <u>reporting/patient-safety-culture/about-patient-safety-culture</u>
- Australian Commission on Safety and Quality in Health Care. (n.d.-b). *National model clinical governance framework*. <u>https://www.safetyandquality.gov.au/our-work/clinical-governance/national-model-clinical-governance-framework</u>
- Australian Commission on Safety and Quality in Health Care. (2019). *Comprehensive care standard*. <u>https://www.safetyandquality.gov.au/standards/nsqhs-</u> standards/comprehensive-care-standard

Australian Commission on Safety and Quality in Health Care. (2021). National safety and quality health service standards (2nd ed.).

https://www.safetyandquality.gov.au/sites/default/files/2021-

05/national_safety_and_quality_health_service_nsqhs_standards_second_edition_-

updated may 2021.pdf

Australian Institute of Health and Welfare. (2013). Stroke and its management in Australia:

An update. Australian Government. <u>https://www.aihw.gov.au/reports/heart-stroke-</u>vascular-diseases/stroke-management-australia-update/notes

Australian Institute of Health and Welfare. (2021, 16 September). Informal carers.

https://www.aihw.gov.au/reports/australias-welfare/informal-carers

Australian Institute of Health and Welfare. (2022a, 7 July). Health system overview.

https://www.aihw.gov.au/reports/australias-health/health-system-overview

Australian Institute of Health and Welfare. (2022b, 7 July). Health workforce.

https://www.aihw.gov.au/reports/workforce/health-workforce

- Australian Institute of Health and Welfare. (2023, 30 June). *Heart, stroke and vascular disease: Australian facts*. <u>https://www.aihw.gov.au/reports/heart-stroke-vascular-diseases/hsvd-facts/contents/about</u>
- Baatiema, L., Otim, M. E., Mnatzaganian, G., de-Graft Aikins, A., Coombes, J., & Somerset,
 S. (2017). Health professionals' views on the barriers and enablers to evidencebased practice for acute stroke care: A systematic review. *Implementation Science*, *12*(1), Article 74.
- Badr, H., Bakhshaie, J., & Chhabria, K. (2019). Dyadic interventions for cancer survivors and caregivers: State of the science and new directions. *Seminars in Oncology Nursing*, 35(4), 337–341
- Bakas, T., Austin, J. K., Okonkwo, K. F., Lewis, R. R., & Chadwick, L. (2002). Needs, concerns, strategies, and advice of stroke caregivers the first 6 months after discharge. *Journal of Neuroscience Nursing*, 34(5), 242–251.

- Bakas, T., McCarthy, M. J., & Miller, E. L. (2022). Systematic review of the evidence for stroke family caregiver and dyad interventions. *Stroke*, *53*(6), 2093–2102.
- Baker, C., Worrall, L., Rose, M., Hudson, K., Ryan, B., & O'Byrne, L. (2018). A systematic review of rehabilitation interventions to prevent and treat depression in poststroke aphasia. *Disability and Rehabilitation*, 40(16), 1870–1892.
- Balasooriya-Smeekens, C., Bateman, A., Mant, J., & De Simoni, A. (2016). Barriers and facilitators to staying in work after stroke: Insight from an online forum. *BMJ Open*, 6(4), Article e009974. https://doi.org/10.1136/bmjopen-2015-009974
- Bannon, S., Lester, E. G., Gates, M. V., McCurley, J., Lin, A., Rosand, J., & Vranceanu, A.M. (2020). Recovering together: Building resiliency in dyads of stroke patients and their caregivers at risk for chronic emotional distress; a feasibility study. *Pilot and Feasibility Studies*, *6*(1), Article 75.
- Barker, I., Steventon, A., Williamson, R., & Deeny, S. R. (2018). Self-management capability in patients with long-term conditions is associated with reduced healthcare utilisation across a whole health economy: Cross-sectional analysis of electronic health records. *BMJ Quality & Safety*, 27(12), 989–999.
- Belagaje, S. R. (2017). Stroke rehabilitation. *CONTINUUM: Lifelong Learning in Neurology*, *23*(1), 238–253.
- Bernhardt, J., Urimubenshi, G., Gandhi, D. B., & Eng, J. J. (2020). Stroke rehabilitation in low-income and middle-income countries: A call to action. *The Lancet*, 396(10260), 1452–1462.
- Berry, J. C., Davis, J. T., Bartman, T., Hafer, C. C., Lieb, L. M., Khan, N., & Brilli, R. J.
 (2020). Improved safety culture and teamwork climate are associated with decreases in patient harm and hospital mortality across a hospital system. *Journal of Patient Safety*, *16*(2), 130–136.
- Bettany-Saltikov, J. (2012). *How to do a systematic literature review in nursing: A step-bystep guide*. McGraw-Hill Education.

- Bishop, M., Kayes, N., & McPherson, K. (2021). Understanding the therapeutic alliance in stroke rehabilitation. *Disability and Rehabilitation*, *43*(8), 1074–1083.
- Boehme, A. K., Kulick, E. R., Canning, M., Alvord, T., Khaksari, B., Omran, S., Willey, J. Z., &
 Elkind, M. S. (2018). Infections increase the risk of 30-day readmissions among
 stroke survivors: Analysis of the National Readmission Database. *Stroke*, *49*(12), 2999–3005.
- Booth, J., Connelly, L., Lawrence, M., Chalmers, C., Joice, S., Becker, C., & Dougall, N.
 (2015). Evidence of perceived psychosocial stress as a risk factor for stroke in adults:
 A meta-analysis. *BMC Neurology*, *15*(1), 233–245.
- Boyer, L., Baumstarck, K., Alessandrini, M., Hamidou, Z., Testart, J., Serres, M., Arquilliere,
 P., Auquier, P., Leroy, T., & Zendjidjian, X. (2017). Emotional intelligence and coping strategies as determinants of quality of life in depressed patient–caregiver dyads: An actor–partner interdependence analysis. *Comprehensive Psychiatry*, 74, 70–79.
- Bradbury-Jones, C., Sambrook, S., & Irvine, F. (2009). The phenomenological focus group: An oxymoron? *Journal of Advanced Nursing*, *65*(3), 663–671.
- Branco, J. P., Oliveira, S., Sargento-Freitas, J., Lains, J., & Pinheiro, J. (2019). Assessing functional recovery in the first 6 months after acute ischaemic stroke: A prospective, observational study. *European Journal of Physical and Rehabilitation Medicine*, *55*(1). <u>https://doi.org/10.23736/S1973-9087.18.05161-4</u>
- Brewer, L., Horgan, F., Hickey, A., & Williams, D. (2013). Stroke rehabilitation: Recent advances and future therapies. QJM: An International Journal of Medicine, 106(1), 11–25. <u>https://doi.org/10.1093/qjmed/hcs174</u>
- Bryman, A. (2016). Social research methods (5th ed.). Oxford University Press.
- Bucki, B., Spitz, E., & Baumann, M. (2019). Emotional and social repercussions of stroke on patient–family caregiver dyads: Analysis of diverging attitudes and profiles of the differing dyads. *PloS One*, *14*(4), Article e0215425.
- Bury, M. (1982). Chronic illness as biographical disruption. *Sociology of Health & Illness*, *4*(2), 167–182.

Busetto, L., Stang, C., Hoffmann, J., Amiri, H., Seker, F., Purrucker, J., Ringleb, P.A., Nagel,
S., Bendszus, M., & Wick, W. (2020). Patient-centredness in acute stroke care—A
qualitative study from the perspectives of patients, relatives and staff. *European Journal of Neurology*, *27*(8), 1638–1646.

Büttner, P., & Muller, R. (2015). Epidemiology (2nd ed.). Oxford University Press.

- Byrne, A.-L., Baldwin, A., & Harvey, C. (2020). Whose centre is it anyway? Defining personcentred care in nursing: An integrative review. *PloS One*, *15*(3), Article e0229923.
- Byrne, A.-L., Hegney, D., Harvey, C., Baldwin, A., Willis, E., Heard, D., Judd, J., Palmer, J.,
 Brown, J., & Heritage, B. (2020). Exploring the nurse navigator role: A thematic analysis. *Journal of Nursing Management*, *28*(4), 814–821.
- Cameron, J. I., Bastawrous, M., Marsella, A., Forde, S., Smale, L., Friedland, J.,
 Richardson, D., & Naglie, G. (2014). Stroke survivors', caregivers', and health care
 professionals' perspectives on the weekend pass to facilitate transition home. *Journal of Rehabilitation Medicine*, *46*(9), 858–863.
- Cameron, J. I., & Gignac, M. A. (2008). 'Timing it right': A conceptual framework for addressing the support needs of family caregivers to stroke survivors from the hospital to the home. *Patient Education and Counseling*, *70*(3), 305–314. http://www.pec-journal.com/article/S0738-3991(07)00441-7/abstract
- Cameron, J. I., Naglie, G., Silver, F. L., & Gignac, M. A. (2013). Stroke family caregivers' support needs change across the care continuum: A qualitative study using the timing it right framework. *Disability and Rehabilitation*, *35*(4), 315–324.
- Camicia, M., Black, T., Farrell, J., Waites, K., Wirt, S. & Lutz, B (with the Association of Rehabilitation Nurses Task Force). (2014). The essential role of the rehabilitation nurse in facilitating care transitions: A white paper by the Association of Rehabilitation Nursing. *Rehabilitation Nursing*, *39*, 3–15. <u>http://onlinelibrary.wiley.com/doi/10.1002/rnj.135/abstract</u>

- Camicia, M., Lutz, B. J., Harvath, T. A., & Joseph, J. G. (2021). Using the preparedness assessment for the transition home after stroke instrument to identify stroke caregiver concerns predischarge: Uncertainty, anticipation, and cues to action. *Rehabilitation Nursing Journal*, *46*(1), 33–42.
- Camicia, M., Lutz, B. J., Harvath, T., Kim, K. K., Drake, C., & Joseph, J. G. (2020).
 Development of an instrument to assess stroke caregivers' readiness for the transition home. *Rehabilitation Nursing Journal*, *45*(5), 287–298.
- Camicia, M., Lutz, B. J., Stram, D., Tucker, L.-Y., Ray, C., & Theodore, B. R. (2022).
 Improving caregiver health through systematic assessment and a tailored plan of care. Western Journal of Nursing Research, 44(3), 307–318.
- Camicia, M., Lutz, B., Summers, D., Klassman, L., & Vaughn, S. (2021). Nursing's role in successful stroke care transitions across the continuum: From acute care into the community. *Stroke*, *52*(12), e794–e805.
- Camicia, M., Lutz, B. J., & Theodore, B. R. (2023). The preparedness assessment for the transition home after stroke predicts key domains of caregiver health. *Topics in Stroke Rehabilitation*, *30*(4), 384–392.
- Camicia, M., Wang, H., DiVita, M., Mix, J., & Niewczyk, P. (2016). Length of stay at inpatient rehabilitation facility and stroke patient outcomes. *Rehabilitation Nursing*, *41*(2), 78–90.
- Cantril, C., Christensen, D., & Moore, E. (2019). Standardizing roles: Evaluating oncology nurse navigator clarity, educational preparation, and scope of work within two healthcare systems. *Clinical Journal of Oncology Nursing*, *23*(1), 52–59.
- Carey, L., Walsh, A., Adikari, A., Goodin, P., Alahakoon, D., De Silva, D., Ong., K-L., Nilsson, M., & Boyd, L. (2019). Finding the intersection of neuroplasticity, stroke recovery, and learning: Scope and contributions to stroke rehabilitation. *Neural Plasticity*, 2019, Article 5232374.

- Carpenter, C. (2017). Phenomenology and rehabilitation research. In P. Liamputtong (Ed.),
 Research methods in health: Foundations for evidence-based practice (pp. 157– 176). Oxford University Press.
- Carroll, C., Booth, A., & Lloyd-Jones, M. (2012). Should we exclude inadequately reported studies from qualitative systematic reviews? An evaluation of sensitivity analyses in two case study reviews. *Qualitative Health Research*, *22*(10), 1425–1434.
- Chen, L., Xiao, L. D., & Chamberlain, D. (2020). An integrative review: Challenges and opportunities for stroke survivors and caregivers in hospital to home transition care. *Journal of Advanced Nursing*, 76(9), 2253–2265.
- Chen, L., Xiao, L. D., & Chamberlain, D. (2022). Exploring the shared experiences of people with stroke and caregivers in preparedness to manage post-discharge care: A hermeneutic study. *Journal of Advanced Nursing*, 78(9), 2983–2999.
- Chen, L., Xiao, L. D., Chamberlain, D., & Newman, P. (2021). Enablers and barriers in hospital-to-home transitional care for stroke survivors and caregivers: A systematic review. *Journal of Clinical Nursing*, *30*(19–20), 2786–2807.
- Chen, L., Xiao, L. D., & De Bellis, A. (2016). First-time stroke survivors and caregivers' perceptions of being engaged in rehabilitation. *Journal of Advanced Nursing*, *72*(1), 73–84.
- Chen, X., Zhao, X., Xu, F., Guo, M., Yang, Y., Zhong, L., Weng, S., & Liu, X. (2022). A systematic review and meta-analysis comparing FAST and BEFAST in acute stroke patients. *Frontiers in Neurology*, *12*, Article 765069.
- Chick, N., & Meleis, A. I. (1986). Transitions: A nursing concern. In P. L. Chinn (Ed.), Nursing research methodology (pp. 237–257). Aspen.
- Chouliara, N., Fisher, R. J., Kerr, M., & Walker, M. F. (2014). Implementing evidence-based stroke early supported discharge services: A qualitative study of challenges, facilitators and impact. *Clinical Rehabilitation*, 28(4), 370–377.

- Chrissobolis, S., & Sobey, C. G. (2015). Vascular biology and atherosclerosis of cerebral vessels. In J. C. Grotta, G. W. Albers, J. P. Broderick, S. E. Kasner, E. H. Lo, A D. Mendelow, R. L. Sacco & L. K. Wong (Eds.), *Stroke pathophysiology, diagnosis, and management* (pp. 3–12). Elsevier.
- Chung, M. L., Bakas, T., Plue, L. D., & Williams, L. S. (2016). Effects of self-esteem, optimism, and perceived control on depressive symptoms in stroke survivor–spouse dyads. *The Journal of Cardiovascular Nursing*, *31*(2), E8–E16.
- Clark, E., MacCrosain, A., Ward, N. S., & Jones, F. (2020). The key features and role of peer support within group self-management interventions for stroke? A systematic review. *Disability and Rehabilitation*, 42(3), 307–316.
- Cobley, C., Fisher, R., Chouliara, N., Kerr, M., & Walker, M. (2013). A qualitative study exploring patients' and carers' experiences of early supported discharge services after stroke. *Clinical Rehabilitation*, *27*(8), 750–757.
- Condon, C., Lycan, S., Duncan, P., & Bushnell, C. (2016). Reducing readmissions after stroke with a structured nurse practitioner/registered nurse transitional stroke program. *Stroke*, *47*(6), 1599–1604.

https:/doi.org/10.1161/STROKEAHA.115.012524

- Connolly, T., & Mahoney, E. (2018). Stroke survivors' experiences transitioning from hospital to home. *Journal of Clinical Nursing*, *27*(21–22), 3979–3987. https://doi.org/10.1111/jocn.14563
- Cook, W. L., & Kenny, D. A. (2005). The actor–partner interdependence model: A model of bidirectional effects in developmental studies. *International Journal of Behavioral Development*, 29(2), 101–109.
- Coote, S., Mackey, E., Alexandrov, A. W., Cadilhac, D. A., Alexandrov, A. V., Easton, D., Zhao, H., Langenberg, F., Bivard, A., & Stephenson, M. (2022). The mobile stroke unit nurse: An international exploration of their scope of practice, education, and training. *Journal of Neuroscience Nursing*, *54*(2), 61–67.

- Coupland, A. P., Thapar, A., Qureshi, M. I., Jenkins, H., & Davies, A. H. (2017). The definition of stroke. *Journal of the Royal Society of Medicine*, *110*(1), 9–12.
- Craig, L., Hoo, Z. L., Yan, T. Z., Wardlaw, J., & Quinn, T. J. (2022). Prevalence of dementia in ischaemic or mixed stroke populations: Systematic review and meta-analysis. *Journal of Neurology, Neurosurgery & Psychiatry*, 93(2), 180–187.
- Creswell, J. W. (2013). *Qualitative inquiry and research design: Choosing among five approaches*. Sage.
- Critical Appraisal Skills Programme. (2018a). *CASP qualitative checklist*. <u>https://casp-uk.net/images/checklist/documents/CASP-Qualitative-Studies-Checklist/CASP-Qualitative-Checklist-2018_fillable_form.pdf</u>
- Critical Appraisal Skills Programme. (2018b). CASP randomised controlled trial checklist. <u>https://casp-uk.net/wp-content/uploads/2018/01/CASP-Randomised-Controlled-Trial-</u> <u>Checklist-2018.pdf</u>
- Critical Appraisal Skills Programme. (2018c). CASP systematic review checklist. <u>https://casp-uk.net/images/checklist/documents/CASP-Systematic-Review-Checklist/CASP-Systematic-Review-Checklist_2018.pdf</u>
- Crotty, M. (1996). Phenomenology and nursing research. Churchill Livingstone.
- Crotty, M. (1998). The foundations of social research: Meaning and perspective in the research process. Allen & Unwin.
- Cuff, S. (2019). Affected history: The truth of Gadamer and methods in new testament hermeneutics. *Hermeneia*, (23), 127–136.
- Da Rocha e Silva, C. E., Alves Brasil, M. A., Matos do Nascimento, E., de Bragança Pereira,
 B., & André, C. (2013). Is poststroke depression a major depression?
 Cerebrovascular Diseases, *35*(4), 385–391.
- Davidson, P. M., Halcomb, E. J., & Gholisadeh, L. (2017). Focus groups in health research.
 In P. Liamputtong (Ed.), *Research methods in health: Foundations for evidence-based practice* (pp. 84–104). Oxford University Press.

- Davies, H. T. O., & Mannion, R. (2013). Will prescriptions for cultural change improve the NHS? *BMJ*, *346*, Article f1305.
- Davies, S. (2005). Meleis's theory of nursing transitions and relatives' experiences of nursing home entry. *Journal of Advanced Nursing*, *52*(6), 658–671.
- Deen, T., Terna, T., Kim, E., Leahy, B., & Fedder, W. (2016). The impact of stroke nurse navigation on patient compliance post discharge. *Rehabilitation Nursing*, *43*(2), 65– 72
- Deloitte. (2018). 2018 Global health care outlook—The evolution of smart health care. Deloitte. <u>https://www2.deloitte.com/content/dam/Deloitte/au/Documents/life-sciences-health-care/au-lshc-global-health-care-outlook-2018-140219.pdf</u>
- Deloitte Access Economics. (2020). *The economic impact of stroke in Australia, 2020*. Deloitte.
- Denham, A. M., Wynne, O., Baker, A. L., Spratt, N. J., Loh, M., Turner, A., Margin, P., &
 Bonevski, B. (2022). The long-term unmet needs of informal carers of stroke
 survivors at home: A systematic review of qualitative and quantitative studies. *Disability and Rehabilitation*, 44(1). <u>https://doi.org/10.1080/09638288.2020.1756470</u>
- Denzin, N. K., & Lincoln, Y. S. (Eds.). (2018). *The Sage handbook of qualitative research* (5th ed.). Sage.
- Department of Health and Aged Care. (2019, 7 August). *The Australian health system*. Australian Government. <u>https://www.health.gov.au/about-us/the-australian-health-system</u>
- Desmarchelier, R., & Cary, L. J. (2022). Toward just and equitable micro-credentials: An Australian perspective. *International Journal of Educational Technology in Higher Education*, *19*(25). <u>https://doi.org/10.1186/s41239-022-00332-y</u>
- Dhamoon, M., Sciacca, R., Rundek, T., Sacco, R., & Elkind, M. (2006). Recurrent stroke and cardiac risks after first ischemic stroke: The Northern Manhattan Study. *Neurology*, *66*(5), 641–646.

- Diaz, M., Cadet, M., & Etienne, C. (2020). Resilience protects against depression in people affected by stroke. *Evidence-Based Nursing*, *24*(4), Article 135.
- Didier, A., Dzemaili, S., Perrenoud, B., Campbell, J., Gachoud, D., Serex, M., Staffoni-Donadini, L., Franco, L., Benaroyo, L., & Maya, Z.-S. (2020). Patients' perspectives on interprofessional collaboration between health care professionals during hospitalization: A qualitative systematic review. *JBI Evidence Synthesis*, *18*(6), 1208– 1270.
- Dineen-Griffin, S., Garcia-Cardenas, V., Williams, K., & Benrimoj, S. I. (2019). Helping patients help themselves: A systematic review of self-management support strategies in primary health care practice. *PloS One*, *14*(8), Article e0220116.
- Dos Santos, E. B., Rodrigues, R. A. P., Fhon, J. R. S., & Haas, V. J. (2018). Changes and predictors of psychological stress among elderly stroke survivors 6 months after hospital discharge. *Stress and Health*, *34*(3), 359–366.
- Dowling, M. (2007). From Husserl to van Manen. A review of different phenomenological approaches. *International Journal of Nursing Studies*, *44*(1), 131–142.
- Doyle, K. P., & Buckwalter, M. S. (2020). Immunological mechanisms in poststroke dementia. *Current Opinion in Neurology*, *33*(1), 30–36.
- Dutta, D., Thornton, D., & Bowen, E. (2018). Using population-based routinely collected data from the Sentinel Stroke National Audit Programme to investigate factors associated with discharge to care home after rehabilitation. *Clinical Rehabilitation*, *32*(8), 1108–1118. <u>https://doi.org/10.1177/0269215517748715</u>
- Edwards, J., & Kaimal, G. (2016). Using meta-synthesis to support application of qualitative methods findings in practice: A discussion of meta-ethnography, narrative synthesis, and critical interpretive synthesis. *The Arts in Psychotherapy*, *51*, 30–35.
- Eilertsen, G., Kirkevold, M., & Bjork, I. T. (2010). Recovering from a stroke: A longitudinal, qualitative study of older Norwegian women. *Journal of Clinical Nursing*, *19*(13–14), 2004–2013.

- Ellis-Hill, C., Robison, J., Wiles, R., McPherson, K., Hyndman, D., & Ashburn, A. (2009). Going home to get on with life: Patients and carers experiences of being discharged from hospital following a stroke. *Disability & Rehabilitation*, *31*(2), 61–72.
- Epstein, R. (2015). Cultivating resilience: Mindful practice and communities of care (SA514) [Abstract]. *Journal of Pain and Symptom Management*, *49*(2), 392.
- Erickson, F. (2018). A history of qualitative inquiry in social and educational research. In N.K. Denzin & Y. S. Lincoln (Eds.), A history of qualitative inquiry in social and educational research (pp. 36–65). Sage.
- Erikson, A., Park, M., & Tham, K. (2010). Place integration through daily activities 1 year after stroke. *OTJR: Occupation, Participation and Health*, *30*(2), 68–77.
- Ezema, C. I., Akusoba, P. C., Nweke, M. C., Uchewoke, C. U., Agono, J., & Usoro, G.
 (2019). Influence of poststroke depression on functional independence in activities of daily living. *Ethiopian Journal of Health Sciences*, *29*(1).
- Faura, J., Bustamante, A., Miró-Mur, F., & Montaner, J. (2021). Stroke-induced immunosuppression: Implications for the prevention and prediction of poststroke infections. *Journal of Neuroinflammation*, *18*(127). <u>https://doi.org/10.1186/s12974-</u> 021-02177-0
- Feigin, V. L., Brainin, M., Norrving, B., Martins, S., Sacco, R. L., Hacke, W., Fisher, M., Pandian, J., & Lindsay, P. (2022). World Stroke Organization (WSO): Global stroke fact sheet 2022. *International Journal of Stroke*, *17*(1), 18–29.
- Feigin, V. L., Norrving, B., & Mensah, G. A. (2017). Global burden of stroke. *Circulation Research*, *120*(3), 439–448. <u>https://doi.org/10.1161/CIRCRESAHA.116.308413</u>
- Feigin, V. L., Stark, B. A., Johnson, C. O., Roth, G. A., Bisignano, C., Abady, G. G.,
 Abbasifard, M., Abbasi-Kangevari, M., Abd-Allah, F., & Abedi, V. (2021). Global,
 regional, and national burden of stroke and its risk factors, 1990–2019: A systematic
 analysis for the Global Burden of Disease Study 2019. *The Lancet Neurology*,
 20(10), 795–820.

- Ferraris, G., Dang, S., Woodford, J., & Hagedoorn, M. (2022). Dyadic interdependence in non-spousal caregiving dyads' wellbeing: A systematic review. *Frontiers in Psychology*, *13*, Article 882389.
- Finch, E., Foster, M., & Fleming, J. (2021). Disrupted biographies: Making sense of minor stroke after hospital discharge. *Disability and Rehabilitation*, *43*(18), 2632–2639.
- Fleming, V., Gaidys, U., & Robb, Y. (2003). Hermeneutic research in nursing: Developing a Gadamerian-based research method. *Nursing Inquiry*, *10*(2), 113–120.
- Flemming, K., Booth, A., Hannes, K., Cargo, M., & Noyes, J. (2018). Cochrane Qualitative and Implementation Methods Group guidance series—Paper 6: Reporting guidelines for qualitative, implementation, and process evaluation evidence syntheses. *Journal* of Clinical Epidemiology, 97, 79–85.
- Frantz, J., Schopp, L., & Rhoda, A. (Eds.). (2021). Self-management in chronic illness: Principles, practice, and empowerment strategies for better health. Springer.
- Franz, S., Muser, J., Thielhorn, U., Wallesch, C. W., & Behrens, J. (2020). Inter-professional communication and interaction in the neurological rehabilitation team: A literature review. *Disability and Rehabilitation*, 42(11), 1607–1615.
- Freeman, H. P. (2006). Patient navigation: A community centered approach to reducing cancer mortality. *Journal of Cancer Education*, *21*(1 Supp.), S11–S14.
- Freytes, I. M., Sullivan, M., Schmitzberger, M., LeLaurin, J., Orozco, T., Eliazar-Macke, N., & Uphold, C. (2021). Types of stroke-related deficits and their impact on family caregiver's depressive symptoms, burden, and quality of life. *Disability and Health Journal*, *14*(2), Article 101019.
- Fryer, C. E., Luker, J. A., McDonnell, M. N., & Hillier, S. L. (2016). Self management programmes for quality of life in people with stroke. *Cochrane Database of Systematic Reviews*, (8), Article CD010442.

https://doi.org/10.1002/14651858.CD010442.pub2

Gadamer, H. (1979). *Truth and method*. Sheed and Ward.

- George, P., Wisco, D. R., Gebel, J., Uchino, K., & Newey, C. R. (2017). Nurses are as specific and are earlier in calling in-hospital stroke alerts compared to physicians. *Journal of Stroke and Cerebrovascular Diseases*, *26*(5), 917–921.
- Ghazzawi, A., Kuziemsky, C., & O'Sullivan, T. (2016). Using a complex adaptive system lens to understand family caregiving experiences navigating the stroke rehabilitation system. *BMC Health Services Research*, *16*(1), Article 538.
- Gholamzadeh, S., Aizan, H. T., Sharif, F., Hamidon, B., & Rahimah, I. (2015). Exploration the supportive needs and coping behaviors of daughter and daughter in-law caregivers of stroke survivors, Shiraz-Iran: A qualitative content analysis. *International Journal of Community-Based Nursing and Midwifery*, *3*(3), Article 205.
- Gibbs, L. A. L., Anderson, M. I., Simpson, G. K., & Jones, K. F. (2020). Spirituality and resilience among family caregivers of survivors of stroke: A scoping review. *Neurorehabilitation*, *46*(1), 41–52.
- Godsey, J. A., Houghton, D. M., & Hayes, T. (2020). Registered nurse perceptions of factors contributing to the inconsistent brand image of the nursing profession. *Nursing Outlook*, *68*(6), 808–821.
- Gorelick, P. B. (2019). The global burden of stroke: Persistent and disabling. *The Lancet Neurology*, *18*(5), 417–418.
- Govender, P., Naidoo, D., Bricknell, K., Ayob, Z., Message, H., & Njoko, S. (2019). 'No one prepared me to go home': Cerebrovascular accident survivors' experiences of community reintegration in a peri-urban context. *African Journal of Primary Health Care & Family Medicine*, *11*(1), e1–e8.
- Grant, J. S. (1996). Home care problems experienced by stroke survivors and their family caregivers. *Home Healthcare Nurse*, *14*(11), 892–902.
- Grant, J. S., Glandon, G. L., Elliott, T. R., Giger, J. N., & Weaver, M. (2006). Problems and associated feelings experienced by family caregivers of stroke survivors the second and third month postdischarge. *Topics in Stroke Rehabilitation*, *13*(3), 66–74. https://doi.org/10.1310/6UL6-5X89-B05M-36TH

- Green, T. L., McNair, N. D., Hinkle, J. L., Middleton, S., Miller, E. T., Perrin, S., Power, M., Southerland, A. M., & Summers, D. V. (2021). Care of the patient with acute ischemic stroke (posthyperacute and prehospital discharge): Update to 2009 comprehensive nursing care scientific statement: A scientific statement from the American Heart Association. *Stroke*, *52*(5), e179–e197.
- Greenwood, N., Mackenzie, A., Wilson, N., & Cloud, G. (2009). Managing uncertainty in life after stroke: A qualitative study of the experiences of established and new informal carers in the first 3 months after discharge. *International Journal of Nursing Studies*, 46(8), 1122–1133.
- Grefkes, C., & Fink, G. R. (2020). Recovery from stroke: Current concepts and future perspectives. *Neurological Research and Practice*, 2(17). <u>https://doi.org/10.1186/s42466-020-00060-6</u>
- Guo, Y., Zhang, Z., Lin, B., Mei, Y., Liu, Q., Zhang, L., Wang, W., Li, Y., & Fu, Z. (2021). The unmet needs of community-dwelling stroke survivors: A systematic review of qualitative studies. *International Journal of Environmental Research and Public Health*, *18*(4), Article 2140.
- Gustafsson, L., & Bootle, K. (2013). Client and carer experience of transition home from inpatient stroke rehabilitation. *Disability and Rehabilitation*, 35(16), 1380–1386. <u>https://doi.org/10.3109/09638288.2012.740134</u>
- Gustafsson, L., Fleming, J., Cornwell, P., Worrall, L., & Brauer, S. (2014). STRENGTH and the health care team: Changing interprofessional and client-centered practices.
 Topics in Stroke Rehabilitation, *21*(5), 413–420.
- Gyawali, P., Chow, W. Z., Hinwood, M., Kluge, M., English, C., Ong, L. K., Nilsson, M., & Walker, F. R. (2020). Opposing associations of stress and resilience with functional outcomes in stroke survivors in the chronic phase of stroke: A cross-sectional study. *Frontiers in Neurology*, *11*, Article 230.

Hajizadeh, A., Zamanzadeh, V., Kakemam, E., Bahreini, R., & Khodayari-Zarnaq, R. (2021).
Factors influencing nurses participation in the health policy-making process: A systematic review. *BMC Nursing*, *20*, Article 128.

Hankey, G. J. (2017). Stroke. *The Lancet*, 389(10069), 641–654.

Hardie, K., Hankey, G. J., Jamrozik, K., Broadhurst, R. J., & Anderson, C. (2004). Ten-year risk of first recurrent stroke and disability after first-ever stroke in the Perth Community Stroke Study. *Stroke*, *35*(3), 731–735.

http://stroke.ahajournals.org/content/strokeaha/35/3/731.full.pdf

- Harrison, M., Ryan, T., Gardiner, C., & Jones, A. (2017). Psychological and emotional needs, assessment, and support poststroke: A multi-perspective qualitative study.
 Topics in Stroke Rehabilitation, 24(2), 119–125.
- Hartmann, S., Weiss, M., Newman, A., & Hoegl, M. (2020). Resilience in the workplace: A multilevel review and synthesis. *Applied Psychology*, *69*(3), 913–959.
- Hartwig, A., Clarke, S., Johnson, S., & Willis, S. (2020). Workplace team resilience: A systematic review and conceptual development. *Organizational Psychology Review*, *10*(3–4), 169–200.
- Hersh, D., & Armstrong, E. (2021). Information, communication, advocacy, and complaint:
 How the spouse of a man with aphasia managed his discharge from hospital. *Aphasiology*, *35*(8), 1067–1083.
- Hessels, A., Paliwal, M., Weaver, S. H., Siddiqui, D., & Wurmser, T. A. (2019). Impact of patient safety culture on missed nursing care and adverse patient events. *Journal of Nursing Care Quality*, *34*(4), 287–294.
- Hodson, T., Aplin, T., & Gustafsson, L. (2016). Understanding the dimensions of home for people returning home post stroke rehabilitation. *British Journal of Occupational Therapy*, 79(7), 427–433. <u>https://doi.org/10.1177/0308022615619420</u>
- Hodson, T., Gustafsson, L., & Cornwell, P. (2019). Unveiling the complexities of mild stroke:
 An interpretative phenomenological analysis of the mild stroke experience. *Australian Occupational Therapy Journal*, 66(5), 656–664.

- Holland, J., Thomson, R., & Henderson, S. (2006). *Qualitative longitudinal research: A discussion paper*. London South Bank University.
- Hollist, M., Morgan, L., Cabatbat, R., Au, K., Kirmani, M. F., & Kirmani, B. F. (2021). Acute stroke management: Overview and recent updates. *Aging and Disease*, *12*(4), 1000–1009.
- Holm, A., & Dreyer, P. (2018). Nurse-patient communication within the context of nonsedated mechanical ventilation: A hermeneutic-phenomenological study. *Nursing in Critical Care*, 23(2), 88–94.
- Houser, J. (Ed.). (2015). *Nursing research: Reading, using, and creating evidence* (3rd ed.). Jones & Bartlett Learning.
- Huang, H. L., Shyu, Y. I. L., Chen, M. C., Chen, S. T., & Lin, L. C. (2003). A pilot study on a home-based caregiver training program for improving caregiver self-efficacy and decreasing the behavioral problems of elders with dementia in Taiwan. *International Journal of Geriatric Psychiatry*, *18*(4), 337–345.
- Hudson, A. P., Spooner, A. J., Booth, N., Penny, R. A., Gordon, L. G., Downer, T.-R., Yates,
 P., Henderson, R., Bradford, N., & Conway, A. (2019). Qualitative insights of patients and carers under the care of nurse navigators. *Collegian*, 26(1), 110–117.
- Hughes, A. K., & Cummings, C. E. (2020). Grief and loss associated with stroke recovery: A qualitative study of stroke survivors and their spousal caregivers. *Journal of Patient Experience*, *7*(6), 1219–1226.
- Hughes, A. K., Woodward, A. T., Fritz, M. C., Swierenga, S. J., Freddolino, P. P., & Reeves,
 M. J. (2020). Unmet needs of US acute stroke survivors enrolled in a transitional care intervention trial. *Journal of Stroke and Cerebrovascular Diseases*, *29*(2), Article 104462.
- Ianni, C., Nicholas, M., Magee, L., & Connor, L. (2022). Self-reported emotional health and social support but not executive function predict participation after mild stroke. *American Journal of Occupational Therapy*, 76(Supp. 1), Article 7610505048p1.

- Ingram, L. A., Butler, A. A., Brodie, M. A., Lord, S. R., & Gandevia, S. C. (2021). Quantifying upper limb motor impairment in chronic stroke: A physiological profiling approach. *Journal of Applied Physiology*, *131*(3), 949–965.
- Interprofessional Education Collaborative. (2016). Core competencies for interprofessional collaborative practice: 2016 Update. <u>https://ipec.memberclicks.net/assets/2016-</u>Update.pdf
- Jacops, R., Hopkins, C. S., & Fasolino, T. (2022). Patient–caregiver dyad: A systematic review informing a concept analysis. *International Journal of Nursing and Health Care Research*, 5(4), Article 1298. <u>https://doi.org/10.29011/2688-9501.101298</u>
- Jarva, E., Mikkonen, K., Tuomikoski, A. M., Kääriäinen, M., Meriläinen, M., Karsikas, E.,
 Meriläinen, M., Karsikas, E., Koivunen, K., Jounila-Ilola, P., & Oikarinen, A. (2021).
 Healthcare professionals' competence in stroke care pathways: A mixed-methods
 systematic review. *Journal of Clinical Nursing*, *30*(9–10), 1206–1235.
- Johnson, B. P., & Cohen, L. G. (2022). Reward and plasticity: Implications for neurorehabilitation. *Handbook of Clinical Neurology*, *184*, 331–340.
- Johnson, C. O., Nguyen, M., Roth, G. A., Nichols, E., Alam, T., Abate, D., Abd-Allah, F., Abdelalim, A., Abraha, H. N., & Abu-Rmeileh, N. M. (2019). Global, regional, and national burden of stroke, 1990–2016: A systematic analysis for the Global Burden of Disease Study 2016. *The Lancet Neurology*, *18*(5), 439–458.
- Jokinen, H., Melkas, S., Ylikoski, R., Pohjasvaara, T., Kaste, M., Erkinjuntti, T., & Hietanen,
 M. (2015). Post-stroke cognitive impairment is common even after successful clinical recovery. *European Journal of Neurology*, 22(9), 1288–1294.

https:/doi.org/10.1111/ene.12743

Jones, F., Gombert-Waldron, K., Honey, S., Cloud, G., Harris, R., Macdonald, A., McKevitt,
 C., Robert, G., & Clarke, D. (2020). Using co-production to increase activity in acute
 stroke units: The CREATE mixed-methods study. *Health Services and Delivery Research*, 8(35).

- Jun-O'Connell, A. H., Grigoriciuc, E., Gulati, A., Silver, B., Kobayashi, K. J., Moonis, M., & Henninger, N. (2023). Stroke nurse navigator utilization reduces unplanned 30-day readmission in stroke patients treated with thrombolysis. *Frontiers in Neurology*, *14*, Article 1205487.
- Kable, A., Baker, A., Pond, D., Southgate, E., Turner, A., & Levi, C. (2019). Health professionals' perspectives on the discharge process and continuity of care for stroke survivors discharged home in regional Australia: A qualitative, descriptive study. *Nursing & Health Sciences*, *21*(2), 253–261.
- Kadu, M. K., & Stolee, P. (2015). Facilitators and barriers of implementing the chronic care model in primary care: A systematic review. *BMC Family Practice*, *16*(1), Article 12.
- Karlsson, M., Hillström, L., Johnsson, A., & Pennbrant, S. (2022). Experiences of workintegrated learning in nursing education. *Journal of Further and Higher Education*, *46*(10), 1377–1390.
- Kazemi, A., Azimian, J., Mafi, M., Allen, K.-A., & Motalebi, S. A. (2021). Caregiver burden and coping strategies in caregivers of older patients with stroke. *BMC Psychology*, 9, Article 51.
- Kelley, H. H., & Thibaut, J. W. (1978). *Interpersonal relations: A theory of interdependence.* Wiley.
- Khan, K., Kunz, R., Kleijnen, J., & Antes, G. (2003). Systematic reviews to support evidencebased medicine. Royal Society of Medicine Press.
- Kilkenny, M. F., Dalli, L. L., Kim, J., Sundararajan, V., Andrew, N. E., Dewey, H. M.,
 Johnston, T., Alif, S. M., Lindley, R. I., & Jude, M. (2020). Factors associated with 90day readmission after stroke or transient ischemic attack: Linked data from the
 Australian Stroke Clinical Registry. *Stroke*, *51*(2), 571–578.
- Kim, G. M., Lim, J. Y., Kim, E. J., & Park, S. M. (2019). Resilience of patients with chronic diseases: A systematic review. *Health & Social Care in the Community*, 27(4), 797– 807.

- Koh, W. L. E., Barr, C. J., & George, S. (2014). Factors influencing poststroke rehabilitation participation after discharge from hospital. *International Journal of Therapy and Rehabilitation*, 21(6), 260–267.
- Kokorelias, K. M., Lu, F. K., Santos, J. R., Xu, Y., Leung, R., & Cameron, J. I. (2020).
 'Caregiving is a full-time job' impacting stroke caregivers' health and well-being: A qualitative meta-synthesis. *Health & Social Care in the Community*, *28*(2), 325–340.
- Kolmos, M., Christoffersen, L., & Kruuse, C. (2021). Recurrent ischemic stroke—A systematic review and meta-analysis. *Journal of Stroke and Cerebrovascular Diseases*, *30*(8), Article 105935.
- Körner, M. (2010). Interprofessional teamwork in medical rehabilitation: A comparison of multidisciplinary and interdisciplinary team approach. *Clinical Rehabilitation*, 24(8), 745–755.
- Krishnan, S., Hay, C. C., Pappadis, M. R., Deutsch, A., & Reistetter, T. A. (2019). Stroke survivors' perspectives on post-acute rehabilitation options, goals, satisfaction, and transition to home. *Journal of Neurologic Physical Therapy*, *43*(3), 160–167.
- Kunst, A. E., Amiri, M., & Janssen, F. (2011). The decline in stroke mortality. *Stroke*, *4*2(8), 2126–2130. <u>https://doi.org/10.1161/STROKEAHA.110.599712</u>
- Kuntz, J. C. (2021). Resilience in times of global pandemic: Steering recovery and thriving trajectories. *Applied Psychology*, *70*(1), 188–215.
- Kuriakose, D., & Xiao, Z. (2020). Pathophysiology and treatment of stroke: Present status and future perspectives. *International Journal of Molecular Sciences*, *21*(20), Article 7609.
- Kyle, J., Skleparis, D., Mair, F. S., & Gallacher, K. I. (2020). What helps and hinders the provision of healthcare that minimises treatment burden and maximises patient capacity? A qualitative study of stroke health professional perspectives. *BMJ Open*, *10*(3), Article e034113.

- Lachal, J., Revah-Levy, A., Orri, M., & Moro, M. R. (2017). Metasynthesis: An original method to synthesize qualitative literature in psychiatry. *Frontiers in Psychiatry*, *8*, Article 269.
- Lamontagne, M.-E., Richards, C., Azzaria, L., Rosa-Goulet, M., Clément, L., & Pelletier, F. (2019). Perspective of patients and caregivers about stroke rehabilitation: The Quebec experience. *Topics in Stroke Rehabilitation*, *26*(1), 39–48.
- Landis, T. T., Severtsen, B. M., Shaw, M. R., & Holliday, C. E. (2020). Professional identity and hospital-based registered nurses: A phenomenological study. *Nursing Forum*, *55*(3), 389–394
- Langhorne, P., Baylan, S., & Early Supported Discharge Trialists. (2017). Early supported discharge services for people with acute stroke (Review). *Cochrane Database of Systematic Reviews*, (7), Article CD000443.

https://doi.org/10.1002/14651858.CD000443.pub4

- Langhorne, P., Bernhardt, J., & Kwakkel, G. (2011). Stroke rehabilitation. *The Lancet*, 377(9778), 1693–1702. <u>https://doi.org/10.1016/S0140-6736(11)60325-5</u>
- Laver, K., Walker, M., & Ward, N. (2022). Telerehabilitation for stroke is here to stay. But at what cost? *Neurorehabilitation and Neural Repair*, *36*(6), 331–334.

Le Danseur, M. (2020). Stroke rehabilitation. Critical Care Nursing Clinics, 32(1), 97–108.

- Lee, S. A., Park, E.-C., Shin, J., Ju, Y. J., Choi, Y., & Lee, H.-Y. (2019). Patient and hospital factors associated with 30-day unplanned readmission in patients with stroke. *Journal of Investigative Medicine*, *67*(1), 52–58.
- Lewin, S., Bohren, M., Rashidian, A., Munthe-Kaas, H., Glenton, C., Colvin, C. J., Garside, R., Noyes, J., Booth, A., & Tunçalp, Ö. (2018). Applying GRADE-CERQual to qualitative evidence synthesis findings—Paper 2: How to make an overall CERQual assessment of confidence and create a summary of qualitative findings table. *Implementation Science*, *13*(1), 11–23.
- Liamputtong, P. (Ed.). (2017). *Research methods in health: Foundations for evidence-based practice*. Oxford University Press.

- Liamputtong, P., & Serry, T. (2017). Making sense of qualitative data. In P. Liamputtong (Ed.), *Research methods in health: Foundations for evidence-based practice* (pp. 421–436). Oxford University Press.
- Lin, B., Zhang, Z., Mei, Y., Wang, C., Xu, H., Liu, L., & Wang, W. (2021). Cumulative risk of stroke recurrence over the last 10 years: A systematic review and meta-analysis. *Neurological Sciences*, 42, 61–71.
- Lin, S., Xiao, L. D., Chamberlain, D., Ullah, S., Wang, Y., Shen, Y., Chen, Z., & Wu, M. (2022). Nurse-led health coaching programme to improve hospital-to-home transitional care for stroke survivors: A randomised controlled trial. *Patient Education and Counseling*, *105*(4), 917–925.

Lincoln, Y. S., & Guba, E. G. (1985). Naturalistic inquiry. Sage.

- Lincoln, Y. S., Lynham, S. A., & Guba, E. G. (2018). Paradigmatic controversies, contradictions, and emerging confluences, revisited. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage handbook of qualitative research* (5th ed, pp. 108–150). Sage.
- Lindblom, S., Ytterberg, C., Elf, M., & Flink, M. (2020). Perceptive dialogue for linking stakeholders and units during care transitions—A qualitative study of people with stroke, significant others and healthcare professionals in Sweden. *International Journal of Integrated Care*, *20*(1), Article 11.
- Linsay, M., Norrving, B., Furie, K., Donnan, G., Langhorne, P., & Davis, S. (2017). *Global stroke guidelines and action plan: A road map for quality stroke care*. World Stroke Organization. <u>http://www.world-</u>

stroke.org/images/GSGAAP/Global_Stroke_Guidelines_and_Action_Plan_All_in_one .pdf

Lip, G. Y., Lane, D. A., Lenarczyk, R., Boriani, G., Doehner, W., Benjamin, L. A., Fisher, M., Lowe, D., Sacco, R. L., & Schnabel, R. (2022). Integrated care for optimizing the management of stroke and associated heart disease: A position paper of the European Society of Cardiology Council on Stroke. *European Heart Journal*, 43(26), 2442–2460.

- Liu, K. F., Lin, H. R., Lee, T.-Y., & Lin, K. C. (2022). Time-varying risk factors associated with the progress of functional recovery and psychological distress in first-ever stroke patients. *Journal of Neuroscience Nursing*, *54*(2), 80–85.
- Liu, Z., Zhou, X., Zhang, W., & Zhou, L. (2019). Factors associated with quality of life early after ischemic stroke: The role of resilience. *Topics in Stroke Rehabilitation*, *26*(5), 335–341.
- Liu, Z., Zhou, X., Zhang, W., & Zhou, L. (2021). Resilience is an independent correlate of the course of quality of life in patients with first-ever ischemic stroke. *International Psychogeriatrics*, 33(6), 567–575.
- Lobo, E. H., Abdelrazek, M., Grundy, J., Kensing, F., Livingston, P. M., Rasmussen, L. J.,
 Islam, S. M. S., & Frølich, A. (2021). Caregiver engagement in stroke care:
 Opportunities and challenges in Australia and Denmark. *Frontiers in Public Health*, 9,
 Article 758808.
- Loft, M. I., Martinsen, B., Esbensen, B. A., Mathiesen, L. L., Iversen, H. K., & Poulsen, I.
 (2019). Call for human contact and support: An interview study exploring patients' experiences with inpatient stroke rehabilitation and their perception of nurses' and nurse assistants' roles and functions. *Disability and Rehabilitation*, *41*(4), 396–404.
- Long, J., Briggs, M., Long, A., & Astin, F. (2016). Starting where I am: A grounded theory exploration of mindfulness as a facilitator of transition in living with a long-term condition. *Journal of Advanced Nursing*, *72*(10), 2445–2456.
- Lord, A. S., Lewis, A., Czeisler, B., Ishida, K., Torres, J., Kamel, H., Woo, D., Elkind, M. S., & Boden-Albala, B. (2016). Majority of 30-day readmissions after intracerebral hemorrhage are related to infections. *Stroke*, *47*(7), 1768–1771.
 https://doi.org/10.1161/STROKEAHA.116.013229
- Lou, S., Carstensen, K., Moldrup, M., Shahla, S., Zakharia, E., & Nielsen, C. P. (2017). Early supported discharge following mild stroke: A qualitative study of patients' and their partners' experiences of rehabilitation at home. *Scandinavian Journal of Caring Sciences*, *31*(2), 302–311. <u>https://doi.org/10.1111/scs.12347</u>

- Love, B., Vetere, A., & Davis, P. (2020). Should interpretative phenomenological analysis (IPA) be used with focus groups? Navigating the bumpy road of 'iterative loops', idiographic journeys, and 'phenomenological bridges'. *International Journal of Qualitative Methods*, *19*, Article 1609406920921600.
- Lutz, B. J., Young, M. E., Creasy, K. R., Martz, C., Eisenbrandt, L., Brunny, J. N., & Cook, C. (2017). Improving stroke caregiver readiness for transition from inpatient rehabilitation to home. *Gerontologist*, *57*(5), 880–889. <u>https://doi.org/10.1093/geront/gnw135</u>
- MacIntosh, B. J., Edwards, J. D., Kang, M., Cogo-Moreira, H., Chen, J. L., Mochizuki, G.,
 Herrmann, N., & Swardfager, W. (2017). Poststroke fatigue and depressive
 symptoms are differentially related to mobility and cognitive performance. *Frontiers in Aging Neuroscience*, *9*, Article 343. <u>https://doi.org/10.3389/fnagi.2017.00343</u>
- Magnus, T., & Liesz, A. (2023). Unveiling the immunopathology of stroke: A comprehensive view on brain-immune interaction. *Seminars in Immunopathology*, *45*(3), 279–280.
- Mannion, R., & Davies, H. (2018). Understanding organisational culture for healthcare quality improvement. *BMJ*, *363*, Article k4907.
- Martz, E., & Livneh, H. (2016). Psychosocial adaptation to disability within the context of positive psychology: Findings from the literature. *Journal of Occupational Rehabilitation*, *26*(1), 4–12.
- May, C., Bieber, K., Chow, D., Mortenson, W. B., & Schmidt, J. (2023). Experiences of adults with stroke attending a peer-led peer-support group. *Brain Impairment*, Advance online publication.
- McCarthy, M. J., Sanchez, A., Garcia, Y. E., Lyons, K. S., & Bakas, T. (2021). Feasibility of the hand in hand relationship intervention for stroke survivor–caregiver dyads: A randomized trial. *Research on Social Work Practice*, *31*(1), 75–89.
- McCormack, B., & McCance, T. (2017). *Person-centred nursing and health care: Theory and practice*. Wiley.

- McGreal, R., & Olcott, D., Jr. (2022). A strategic reset: Micro-credentials for higher education leaders. *Smart Learning Environments*, *9*(1), Article 9.
- McLaney, E., Morassaei, S., Hughes, L., Davies, R., Campbell, M., & Di Prospero, L. (2022).
 A framework for interprofessional team collaboration in a hospital setting: Advancing team competencies and behaviours. *Healthcare Management Forum*, *35*(2), 112–117.
- McMurray, A., & Cooper, H. (2017). The nurse navigator: An evolving model of care. *Collegian*, 24(2), 205–212.
- Medeiros, G. C., Roy, D., Kontos, N., & Beach, S. R. (2020). Poststroke depression: A 2020 updated review. *General Hospital Psychiatry*, *66*, 70–80.
- Meleis, A. (2010). *Transitions theory: Middle-range and situation-specific theories in nursing research and practice.* Springer.
- Meleis, A. I. (1975). Role insufficiency and role supplementation: A conceptual framework. *Nursing Research*, *24*(4), 264–271.
- Meleis, A. I., Sawyer, L. M., Im, E.-O., Messias, D. K. H., & Schumacher, K. (2000). Experiencing transitions: An emerging middle-range theory. *Advances in Nursing Science*, 23(1), 12–28.
- Meyers, E. E., McCurley, J., Lester, E., Jacobo, M., Rosand, J., & Vranceanu, A.-M. (2020). Building resiliency in dyads of patients admitted to the neuroscience intensive care unit and their family caregivers: Lessons learned from William and Laura. *Cognitive and Behavioral Practice*, *27*(3), 321–335.
- Mi, Y. (2016). Development of a transition process scale for high-risk infant's caregiver. *Iranian Journal of Public Health*, *45*(2), 158–169.
- Miller, K. K., Lin, S. H., & Neville, M. (2019). From hospital to home to participation: A position paper on transition planning poststroke. *Archives of Physical Medicine and Rehabilitation*, *100*(6), 1162–1175.

- Mills, J., Ramachenderan, J., Chapman, M., Greenland, R., & Agar, M. (2020). Prioritising workforce wellbeing and resilience: What COVID-19 is reminding us about self-care and staff support. *Palliative Medicine*, *34*(9), 1137–1139.
- Minshall, C., Castle, D. J., Thompson, D. R., Pascoe, M., Cameron, J., McCabe, M., Apputhurai, P., Knowles, S. R., Jenkins, Z., & Ski, C. F. (2020). A psychosocial intervention for stroke survivors and carers: 12–month outcomes of a randomized controlled trial. *Topics in Stroke Rehabilitation*, 27(8), 563-576.
- Minshall, C., Ski, C. F., Apputhurai, P., Thompson, D. R., Castle, D. J., Jenkins, Z., & Knowles, S. R. (2021). Exploring the impact of illness perceptions, self-efficacy, coping strategies, and psychological distress on quality of life in a poststroke cohort. *Journal of Clinical Psychology in Medical Settings*, *28*, 174–180.
- Mishel, M. H. (1990). Reconceptualization of the uncertainty in illness theory. *Image: The Journal of Nursing Scholarship*, 22(4), 256–262.
- Mitchell, E., Reynolds, S. S., Mower-Wade, D., Raser-Schramm, J., & Granger, B. B. (2022). Implementation of an advanced practice registered nurse–led clinic to improve followup care for post–ischemic stroke patients. *Journal of Neuroscience Nursing*, *54*(5), 193–198.
- Mohan, K. M., Crichton, S. L., Grieve, A. P., Rudd, A. G., Wolfe, C. D. A., & Heuschmann, P. U. (2009). Frequency and predictors for the risk of stroke recurrence up to 10 years after stroke: The South London Stroke Register. *Journal of Neurology, Neurosurgery* & *Psychiatry*, *80*(9), 1012–1018.
- Moher, D., Liberati, A., Tetzlaff, J., & Altman, D. G. (2009). Preferred reporting items for systematic reviews and meta-analyses: The PRISMA statement. *BMJ*, 339, Article b2535.
- Monroe, S. M., & Slavich, G. (2016). Psychological stressors: Overview. In G. Fink (Ed.), Stress: Concepts, cognition, emotion, and behavior (pp. 109–115). Academic Press.
- Moon, K., & Blackman, D. (2014). A guide to understanding social science research for natural scientists. *Conservation Biology*, *28*(5), 1167–1177.

- Mora, K., Dorrejo, X. M., Carreon, K. M., & Butt, S. (2017). Nurse practitioner-led transitional care interventions: An integrative review. *Journal of the American Association of Nurse Practitioners*, *29*(12), 773–790.
- Morse, J. (2018). Reframing rigor in qualitative inquiry. In N. K. Denzin & Y. S. Lincoln (Eds.), *The Sage handbook of qualitative research* (pp. 796–817). Sage.
- Mosqueda-Díaz, A., Vílchez-Barboza, V., Valenzuela-Suazo, S., & Sanhueza-Alvarado, O. (2014). Critical theory and its contribution to the nursing discipline. *Investigación y Educación en Enfermería*, *32*(2), 356–363.
- Mou, H., Wong, M. S., & Chien, W. T. (2021). Effectiveness of dyadic psychoeducational intervention for stroke survivors and family caregivers on functional and psychosocial health: A systematic review and meta-analysis. *International Journal of Nursing Studies*, *120*, Article 103969. <u>https://doi.org/10.1016/j.ijnurstu.2021.103969</u>
- Mountain, A., Patrice Lindsay, M., Teasell, R., Salbach, N. M., de Jong, A., Foley, N.,
 Bhogal, S., Bains, N., Bowes, R., & Cheung, D. (2020). Canadian stroke best
 practice recommendations: Rehabilitation, recovery, and community participation
 following stroke. Part two: Transitions and community participation following stroke. *International Journal of Stroke*, *15*(7), 789–806.
- Munce, S. E., Graham, I. D., Salbach, N. M., Jaglal, S. B., Richards, C. L., Eng, J. J., Desrosiers, J., MacKay-Lyons, M., Wood-Dauphinee, S., & Korner-Bitensky, N. (2017). Perspectives of health care professionals on the facilitators and barriers to the implementation of a stroke rehabilitation guidelines cluster randomized controlled trial. *BMC Health Services Research*, *17*, Article 440.
- My Aged Care. (n.d.). Support for people living with disability. Australian Government. https://www.myagedcare.gov.au/support-people-living-with-disability
- My Aged Care. (2020, September). Your guide to Home Care Package services. Australian Government. <u>https://www.myagedcare.gov.au/sites/default/files/2022-12/your-guide-to-home-care-package-services_0_0.pdf</u>

Nasreddine, Z. S., Phillips, N. A., Bédirian, V., Charbonneau, S., Whitehead, V., Collin, I.,
Cummings, J. L., & Chertkow, H. (2005). The Montreal Cognitive Assessment,
MoCA: A brief screening tool for mild cognitive impairment. *Journal of the American Geriatrics Society*, *53*(4), 695–699.

National Disability Insurance Agency. (2021). *Australia's disability strategy 2021–2031*. <u>https://www.ndis.gov.au/understanding/australias-disability-strategy-2021-2031</u>

National Disability Insurance Agency. (2022, 13 December). Understanding the NDIS. <u>https://www.ndis.gov.au/understanding/what-ndis</u>

National Health and Medical Research Council. (2018). *National statement on ethical conduct in human research 2007 (Updated 2018)*. Australian Government. <u>https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018</u>

National Institute of Neurological Disorders and Stroke. (n.d.). *Brain basics: Know your brain*. National Institutes of Health. <u>https://www.ninds.nih.gov/health-information/public-</u> education/brain-basics/brain-basics-know-your-brain

Naylor, M. D., Hirschman, K. B., & McCauley, K. (2020). Meeting the transitional care needs of older adults with COVID-19. *Journal of Aging & Social Policy*, *3*2(4–5), 387–395.

Nelson, M. L., Hanna, E., Hall, S., & Calvert, M. (2016). What makes stroke rehabilitation patients complex? Clinician perspectives and the role of discharge pressure. *Journal of Comorbidity*, *6*(2), 35–41.

Nie, J., & Yang, X. (2017). Modulation of synaptic plasticity by exercise training as a basis for ischemic stroke rehabilitation. *Cellular and Molecular Neurobiology*, *37*, 5–16.

Nielsen, C., Agerskov, H., Bistrup, C., & Clemensen, J. (2020). Evaluation of a telehealth solution developed to improve follow-up after kidney Transplantation. *Journal of Clinical Nursing*, 29(7–8), 1053–1063.

- Nordin, A., Sunnerhagen, K. S., & Axelsson, A. B. (2015). Patients' expectations of coming home with very early supported discharge and home rehabilitation after stroke—An interview study. *BMC Neurology*, *15*, Article 235. <u>https://doi.org/10.1186/s12883-015-0492-0</u>
- Nowell, L. S., Norris, J. M., White, D. E., & Moules, N. J. (2017). Thematic analysis: Striving to meet the trustworthiness criteria. *International Journal of Qualitative Methods*, *16*(1), Article 1609406917733847.
- Nursing and Midwifery Board. (2016). *Registered nurse standards for practice*. Australian Health Practitioner Regulation Agency.

https://www.nursingmidwiferyboard.gov.au/Codes-Guidelines-

Statements/Professional-standards/registered-nurse-standards-for-practice.aspx

- O'Brien, C. L., Moore, G., Rolley, J. X., Ski, C. F., Thompson, D. R., Lautenschlager, N. T., Gonzales, G., Hsueh, Y., & Castle, D. (2014). Exploring health care providers' perceptions of the needs of stroke carers: Informing development of an optimal health program. *Topics in Stroke Rehabilitation*, *21*(5), 421–431.
- Olazarán, J., Trincado, R., & Bermejo-Pareja, F. (2013). Cumulative effect of depression on dementia risk. *International Journal of Alzheimer's Disease*, *2013*, Article 457175.
- Olofsson, A., Andersson, S. O., & Carlberg, B. (2005). 'If only I manage to get home I'll get better'—Interviews with stroke patients after emergency stay in hospital on their experiences and needs. *Clinical Rehabilitation*, *19*(4), 433–440.
- Olson, D. M., & Juengst, S. B. (2019). The hospital to home transition following acute stroke. *Nursing Clinics*, *54*(3), 385–397.
- Osborne, C. L., & Neville, M. (2019). Understanding the experience of early supported discharge from the perspective of patients with stroke and their carers and health care providers: A qualitative review. *Nursing Clinics*, *54*(3), 367–384.
- Ouellette, D., Timple, C., Kaplan, S., Rosenberg, S., & Rosario, E. (2015). Predicting discharge destination with admission outcome scores in stroke patients. *Neurorehabilitation*, 37(2), 173–179. <u>https://doi.org/10.3233/NRE-151250</u>

Oxford Learners' Dictionaries. (n.d.). Philosophy. Oxford University Press.

https://www.oxfordlearnersdictionaries.com/definition/american_english/philosophy#: ~:text=noun-,noun,philosophy%20a%20degree%20in%20philosophy

- Pagram, H., Bivard, A., Lincz, L. F., & Levi, C. (2017). Immunity and stroke, the hurdles of stroke research translation. *International Journal of Stroke*, *12*(2), 123–131. <u>https://doi.org/10.1177/1747493016676622</u>
- Palmer, S., Glass, T. A., Palmer, J. B., Loo, S., & Wegener, S. T. (2004). Crisis Intervention with individuals and their families following stroke: A model for psychosocial service during inpatient rehabilitation. *Rehabilitation Psychology*, 49(4), 338–343.
- Pearson, M., Hunt, H., Cooper, C., Shepperd, S., Pawson, R., & Anderson, R. (2015).
 Providing effective and preferred care closer to home: A realist review of intermediate care. *Health & Social Care in the Community*, 23(6), 577–593.
 https://doi.org/10.1111/hsc.12183
- Pereira, C. M., Greenwood, N., & Jones, F. (2021). From recovery to regaining control of life—The perspectives of people with stroke, their carers and health professionals. *Disability and Rehabilitation*, *43*(20), 2897–2908.
- Perna, R., & Harik, L. (2020). The role of rehabilitation psychology in stroke care described through case examples. *Neurorehabilitation*, *46*(2), 195–204.
- Perry, L., & Middleton, S. (2011). An investigation of family carers' needs following stroke survivors' discharge from acute hospital care in Australia. *Disability and Rehabilitation*, 33(19–20), 1890–1900.
- Petrie, K., Baldwin, P., Crawford, J., & Harvey, S. B. (2021). The voice of mental health practice in Australia: A mixed-method cross-sectional study of gaps and areas of need. *Australian and New Zealand Journal of Public Health*, *45*(4), 318–324.
- Phillips, J. L., Rondon, A. J., Vannello, C., Fillingham, Y. A., Austin, M. S., & Courtney, P. M. (2019). A nurse navigator program is effective in reducing episode-of-care costs following primary hip and knee arthroplasty. *The Journal of Arthroplasty*, *34*(8), 1557–1562.
- Piccenna, L., Lannin, N. A., Gruen, R., Pattuwage, L., & Bragge, P. (2016). The experience of discharge for patients with an acquired brain injury from the inpatient to the community setting: A qualitative review. *Brain Injury*, *30*(3), 241–251. <u>https://doi.org/10.3109/02699052.2015.1113569</u>
- Pindus, D. M., Mullis, R., Lim, L., Wellwood, I., Rundell, A. V., Abd Aziz, N. A., & Mant, J. (2018). Stroke survivors' and informal caregivers' experiences of primary care and community healthcare services—A systematic review and meta-ethnography. *PloS One*, *13*(2), Article e0192533. <u>https://doi.org/10.1371/journal.pone.0192533</u>
- Plant, S. E., Tyson, S. F., Kirk, S., & Parsons, J. (2016). What are the barriers and facilitators to goal-setting during rehabilitation for stroke and other acquired brain injuries? A systematic review and meta-synthesis. *Clinical Rehabilitation*, *30*(9), 921–930. <u>https://doi.org/10.1177/0269215516655856</u>
- Pluye, P., Robert, E., Cargo, M., Bartlett, G., O'Cathain, A., Griffiths, F., Boardman, F., Gagnon, M.P., & Rousseau, M.C. (2011). Proposal: A mixed methods appraisal tool for systematic mixed studies reviews. Retrieved on September 3,2023 from <u>http://mixedmethodsappraisaltoolpublic.pbworks.com</u> Archived by WebCite® at <u>http://www.webcitation.org/5tTRTc9yJ</u>
- Pluye, P., & Hong, Q. N. (2014). Combining the power of stories and the power of numbers:
 Mixed methods research and mixed studies reviews. *Annual Review of Public Health*, 35, 29–45.
- Polit, D. F., & Beck, C. T. (2012). *Nursing research: Generating and assessing evidence for nursing practice* (9th ed.). Wolters Kluwer Health.
- Polit, D. F., & Beck, C. T. (2017). *Nursing research: Generating and assessing evidence for nursing practice* (10th ed.). Wolters Kluwer Health.
- Pont, W., Groeneveld, I., Arwert, H., Meesters, J., Mishre, R. R., Vliet Vlieland, T., & Goossens, P. (2020). Caregiver burden after stroke: Changes over time? *Disability and Rehabilitation*, *42*(3), 360–367.

- Pringle, J., Drummond, J. S., & McLafferty, E. (2013). Revisioning, reconnecting and revisiting: The psychosocial transition of returning home from hospital following a stroke. *Disability and Rehabilitation*, 35(23), 1991–1999.
- Prior, S. J., Mather, C. A., & Campbell, S. J. (2023). Redesigning rural acute stroke care: A person-centered approach. *International Journal of Environmental Research and Public Health*, 20(2), Article 1581.
- Pucciarelli, G., Lommi, M., Magwood, G. S., Simeone, S., Colaceci, S., Vellone, E., & Alvaro,
 R. (2021). Effectiveness of dyadic interventions to improve stroke patient–caregiver
 dyads' outcomes after discharge: A systematic review and meta-analysis study.
 European Journal of Cardiovascular Nursing, 20(1), 14–33.

https://doi.org/10.1177/1474515120926069

- Pugh, J. D., McCoy, K., Needham, M., Jiang, L., Giles, M., McKinnon, E., & Heine, K.
 (2022). Evaluation of an Australian neurological nurse-led model of postdischarge care. *Health & Social Care in the Community*, *30*(4), e962–e973.
- Purvis, T., Kilkenny, M. F., Middleton, S., & Cadilhac, D. A. (2018). Influence of stroke coordinators on delivery of acute stroke care and hospital outcomes: An observational study. *International Journal of Stroke*, *13*(6), 585–591.
- Purvis, T., Middleton, S., Alexandrov, A. W., Kilkenny, M. F., Coote, S., Kuhle, S., & Cadilhac, D. A. (2021). Understanding coordinator roles in acute stroke care: A national survey. *Journal of Stroke and Cerebrovascular Diseases*, *30*(12), Article 106111.
- Purvis, T., Middleton, S., Alexandrov, A. W., Kilkenny, M. F., Coote, S., Kuhle, S., & Cadilhac, D. A. (2022). Exploring barriers to stroke coordinator roles in Australia: A national survey. *Collegian*, *29*(4), 423–429.
- Queensland Health. (2019, 27 September). *Nurse navigators: Information sheet.* Queensland Government.

https://www.health.qld.gov.au/__data/assets/pdf_file/0032/666437/nn-info-sheet.pdf

- Quinn, K., Murray, C., & Malone, C. (2014). Spousal experiences of coping with and adapting to caregiving for a partner who has a stroke: A meta-synthesis of qualitative research. *Disability and Rehabilitation*, *36*(3), 185–198.
- Quintas, R., Cerniauskaite, M., Ajovalasit, D., Sattin, D., Boncoraglio, G., Parati, E. A., & Leonardi, M. (2012). Describing functioning, disability, and health with the international classification of functioning, disability, and health brief core set for stroke. *American Journal of Physical Medicine & Rehabilitation*, *91*(13), S14–S21.
- Qureshi, A., Swain, N., Aldabe, D., & Hale, L. (2022). Exploring challenges affecting resilience in carers of stroke survivors: A qualitative descriptive study. *Disability and Rehabilitation*, *44* (24), 1-9.
- Ramazanu, S., Chiang, V. C. L., & Valimaki, M. (2021). The experiences and evaluation of a complex intervention for couples coping with stroke. *Journal of Neuroscience Nursing*, *53*(1), 18–23.
- Ramsay, P., Huby, G., Thompson, A., & Walsh, T. (2014). Intensive care survivors' experiences of ward-based care: Meleis' theory of nursing transitions and role development among critical care outreach services. *Journal of Clinical Nursing*, 23(5–6), 605–615.
- Rawlinson, C., Carron, T., Cohidon, C., Arditi, C., Hong, Q. N., Pluye, P., Peytremann-Bridevaux, I., & Gilles, I. (2021). An overview of reviews on interprofessional collaboration in primary care: Barriers and facilitators. *International Journal of Integrated Care*, *21*(2), Article 32.
- Reeves, M. J., Fritz, M. C., Woodward, A. T., Hughes, A. K., Coursaris, C. K., Swierenga, S. J., Nasiri, M., & Freddolino, P. P. (2019). Michigan stroke transitions trial: A clinical trial to improve stroke transitions. *Circulation: Cardiovascular Quality and Outcomes*, *12*(7), Article e005493.

- Reeves, M. J., Hughes, A. K., Woodward, A. T., Freddolino, P. P., Coursaris, C. K.,
 Swierenga, S. J., Schwamm, L. H., & Fritz, M. C. (2017). Improving transitions in acute stroke patients discharged to home: The Michigan Stroke Transitions Trial (MISTT) protocol. *BMC Neurology*, *17*, Article 115.
- Rhudy, L. M., Wells-Pittman, J., & Flemming, K. D. (2020). Psychosocial sequelae of stroke in working age adults: A pilot study. *The Journal of Neuroscience Nursing*, 52(4), 192–199.
- Rittman, M., Boylstein, C., Hinojosa, R., Hinojosa, M. S., & Haun, J. (2007). Transition experiences of stroke survivors following discharge home. *Topics in Stroke Rehabilitation*, *14*(2), 21–31.
- Rittman, M., Faircloth, C., Boylstein, C., Gubrium, J. F., Williams, C., Van Puymbroeck, M., &
 Ellis, C. (2004). The experience of time in the transition from hospital to home
 following stroke. *Journal of Rehabilitation Research & Development*, *41*(3A), 259–268.
- Robertson, L. G., Anderson, T. L., Hall, M. E. L., & Kim, C. L. (2019). Mothers and mental labor: A phenomenological focus group study of family-related thinking work. *Psychology of Women Quarterly*, *43*(2), 184–200.
- Robinson, R. G., & Jorge, R. E. (2016). Poststroke depression: A review. *American Journal* of *Psychiatry*, *173*(3), 221–231.
- Rohde, D., Gaynor, E., Large, M., Conway, O., Bennett, K., Williams, D. J., Callaly, E.,
 Dolan, E., & Hickey, A. (2019). Stroke survivor cognitive decline and psychological wellbeing of family caregivers five years poststroke: A cross-sectional analysis. *Topics in Stroke Rehabilitation*, *26*(3), 180–186.
- Rosewilliam, S., Sintler, C., Pandyan, A. D., Skelton, J., & Roskell, C. A. (2016). Is the practice of goal-setting for patients in acute stroke care patient-centred and what factors influence this? A qualitative study. *Clinical Rehabilitation*, *30*(5), 508–519. https://doi.org/10.1177/0269215515584167

Rost, N. S., Brodtmann, A., Pase, M. P., van Veluw, S. J., Biffi, A., Duering, M., Hinman, J.
D., & Dichgans, M. (2022). Poststroke cognitive impairment and dementia. *Circulation Research*, *130*(8), 1252–1271.

Royal Commission into Aged Care Quality and Safety. (2021, 26 February). *Final report: Care, dignity and respect*. Commonwealth of Australia.

https://agedcare.royalcommission.gov.au/publications/final-report

- Rusbult, C. E., & Van Lange, P. A. (2008). Why we need interdependence theory. *Social and Personality Psychology Compass*, *2*(5), 2049–2070.
- Sadler, E., Daniel, K., Wolfe, C. D., & McKevitt, C. (2014). Navigating stroke care: The experiences of younger stroke survivors. *Disability & Rehabilitation*, *36*(22), 1911–1917.
- Sadler, E., Wolfe, C. D., Jones, F., & McKevitt, C. (2017). Exploring stroke survivors' and physiotherapists' views of self-management after stroke: A qualitative study in the UK. *BMJ Open*, *7*(3), Article e011631.
- SA Health. (2022, 2 April). *Transition care program*. Government of South Australia <u>https://www.sahealth.sa.gov.au/wps/wcm/connect/public+content/sa+health+internet/</u> <u>services/health+services+for/seniors+health+services/older+people+services/support</u> ing+older+peoples+independence/transition+care+program
- SA Health. (2023, 21 June). *Telerehabilitation*. Government of South Australia. <u>https://www.sahealth.sa.gov.au/wps/wcm/connect/public+content/sa+health+internet/</u> <u>services/hospitals/rehabilitation+services/telerehabilitation/telerehabilitation</u>
- Sandelowski, M., & Barroso, J. (2006). *Handbook for synthesizing qualitative research*. Springer.
- Sandelowski, M., Leeman, J., Knafl, K., & Crandell, J.L. (2013). Text-in-context: a method for extracting findings in mixed-methods mixed research synthesis studies. *Journal of Advanced Nursing*, 69(6),1428–1437.

- Satink, T., Josephsson, S., Zajec, J., Cup, E. H., de Swart, B. J., & Nijhuis-van der Sanden,
 M. W. G. (2016). Negotiating role management through everyday activities:
 Narratives in action of two stroke survivors and their spouses. *Disability & Rehabilitation*, *38*(24), 2354–2364.
- Schot, E., Tummers, L., & Noordegraaf, M. (2020). Working on working together. A systematic review on how healthcare professionals contribute to interprofessional collaboration. *Journal of Interprofessional Care*, *34*(3), 332–342.
- Schumacher, K. L., & Meleis, A. I. (1994). Transitions: A central concept in nursing. *Image: The Journal of Nursing Scholarship*, *26*(2), 119–127.
- Schwarz, B., Claros-Salinas, D., & Streibelt, M. (2018). Meta-synthesis of qualitative research on facilitators and barriers of return to work after stroke. *Journal of Occupational Rehabilitation*, *28*(1), 28–44. <u>https://doi.org/10.1007/s10926-017-9713-2</u>
- Scrivener, K., Dean, C., Cameron, I. D., & Ada, L. (2022). Stroke in Australia: Long term survivors have fallen into a black hole. *The Medical Journal of Australia*, *217*(6), 290– 291.
- Sennfält, S., & Ullberg, T. (2020). Informal caregivers in stroke: Life impact, support, and psychological well-being—A Swedish Stroke Register (Riksstroke) study. *International Journal of Stroke*, *15*(2), 197–205.
- Serry, T., & Liamputtong, P. (2017). The in-depth interviewing method in health. In P. Liamputtong (Ed.), Research methods in health: Foundations for evidence-based practice (pp. 67–83). Oxford University Press.
- Sexton, E., McLoughlin, A., Williams, D. J., Merriman, N. A., Donnelly, N., Rohde, D.,
 Hickey, A., Wren, M., & Bennett, K. (2019). Systematic review and meta-analysis of
 the prevalence of cognitive impairment no dementia in the first year poststroke. *European Stroke Journal*, *4*(2), 160–171.
- Shyu, Y. I. L., Chen, M. C., Chen, S. T., Wang, H. P., & Shao, J. H. (2008). A family caregiver-oriented discharge planning program for older stroke patients and their family caregivers. *Journal of Clinical Nursing*, *17*(18), 2497–2508.

- Simeone, S., Savini, S., Cohen, M. Z., Alvaro, R., & Vellone, E. (2015). The experience of stroke survivors three months after being discharged home: A phenomenological investigation. *European Journal of Cardiovascular Nursing*, *14*(2), 162–169.
- Sörös, P., Harnadek, M., Blake, T., Hachinski, V., & Chan, R. (2015). Executive dysfunction in patients with transient ischemic attack and minor stroke. *Journal of the Neurological Sciences*, *354*(1–2), 17–20. <u>https://doi.org/10.1016/j.jns.2015.04.022</u>
- Stein, L. K., Kornspun, A., Erdman, J., & Dhamoon, M. S. (2020). Readmissions for depression and suicide attempt following stroke and myocardial infarction. *Cerebrovascular Diseases Extra*, *10*(2), 94–104.
- Stinear, C. M., Lang, C. E., Zeiler, S., & Byblow, W. D. (2020). Advances and challenges in stroke rehabilitation. *The Lancet Neurology*, *19*(4), 348–360.
- Stokman-Meiland, D. C., Groeneveld, I. F., Arwert, H. J., van der Pas, S. L., Meesters, J. J., Rambaran Mishre, R. D., Vliet Vlieland, T. P., & Goossens, P. H. (2022). The course of depressive symptoms in the first 12 months poststroke and its association with unmet needs. *Disability and Rehabilitation*, 44(3), 428–435.
- Stroke Foundation. (2018, 30 November). Stroke facts. InformMe.

https://informme.org.au/learning-modules/stroke-facts

Stroke Foundation. (2019a). Acute services report 2019. InformMe.

https://informme.org.au/media/nwsfwre1/acute_strokeservicesreport_2019.pdf

- Stroke Foundation. (2019b). *Stroke foundation audits—A 20 year retrospective*. InformMe.https://informme.org.au/en/stroke-data/About-stroke-audits
- Stroke Foundation. (2020a). *National stroke audit: Rehabilitation services report 2020*. InformMe.

https://informme.org.au/media/drtlcbvp/rehab_strokeservicesreport_2020.pdf

Stroke Foundation. (2020b, 2 June). *Stroke facts and secondary prevention*. InformMe. <u>https://informme.org.au/learning-modules/take?id=29609444-1260-47df-b036-</u> <u>d9dd60afd27e</u> Stroke Foundation. (2021). *National stroke audit: Acute services report 2021*. InformMe. https://informme.org.au/media/0bzk42xf/2021_acute-services-national-report.pdf

- Stroke Foundation. (2022). *National rehabilitation stroke services framework*. InformMe. <u>https://informme.org.au/media/sqfevyko/stroke-rehabilitation-framework-2022-</u> <u>update-final.pdf</u>
- Stroke Foundation. (2023). *Clinical guidelines for stroke management*. InformMe. https://informme.org.au/guidelines/living-clinical-guidelines-for-stroke-management
- Su, F., & Xu, W. (2020). Enhancing brain plasticity to promote stroke recovery. *Frontiers in Neurology*, *11*, Article 554089.
- Taku, K. (2014). Relationships among perceived psychological growth, resilience and burnout in physicians. *Personality and Individual Differences*, *59*, 120–123.
- Tan, E., Gao, L., Collier, J. M., Ellery, F., Dewey, H. M., Bernhardt, J., & Moodie, M. (2022).
 The economic and health burden of stroke among younger adults in Australia from a societal perspective. *BMC Public Health*, 22(1), 218–228.
- Tate, K., Penconek, T., Dias, B. M., Cummings, G. G., & Bernardes, A. (2023). Authentic leadership, organizational culture and the effects of hospital quality management practices on quality of care and patient satisfaction. *Journal of Advanced Nursing*.
- Tawakol, A., Ishai, A., Takx, R. A., Figueroa, A. L., Ali, A., Kaiser, Y., Truong, Q. A., Solomon, C., Calcagno, C., & Mani, V. (2017). Relation between resting amygdalar activity and cardiovascular events: A longitudinal and cohort study. *The Lancet*, *389*(10071), 834–845.
- Taylor, A., Monsanto, X., Kilgour, H., Smith, C., & Hale, L. (2019). Attitudes of healthcare professionals towards self-management for patients with stroke. *New Zealand Journal of Physiotherapy*, *47*(1), 25–36.
- Taylor, B., & Francis, K. (2013). *Qualitative research in the health sciences: Methodologies, methods and processes.* Routledge.

Tempest, S., Harries, P., Kilbride, C., & De Souza, L. (2013). Enhanced clarity and holism:
 The outcome of implementing the ICF with an acute stroke multidisciplinary team in
 England. *Disability and Rehabilitation*, *35*(22), 1921–1925.
 https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3812702/pdf/DRE-35-1921.pdf

Theadom, A., Rutherford, S., Kent, B., McPherson, K., & Group, A. I. (2018). The process of adjustment over time following stroke: A longitudinal qualitative study. *Neuropsychological Rehabilitation*, *29*(9), 1464–1474.

- Thomas, J., & Harden, A. (2008). Methods for the thematic synthesis of qualitative research in systematic reviews. *BMC Medical Research Methodology*, *8*, Article 45.
- Timothy, E. K., Graham, F. P., & Levack, W. M. (2016). Transitions in the embodied experience after stroke: Grounded theory study. *Physical Therapy*, *96*(10), 1565– 1575.
- Tjokrowijoto, P., Stolwyk, R. J., Ung, D., Kilkenny, M. F., Kim, J., Dalli, L. L., Cadilhac, D. A.,
 & Andrew, N. E. (2023). Factors associated with mental health service access among Australian community-dwelling survivors of stroke. *Disability and Rehabilitation*, *45*(3), 504–511.
- Tobin, M. K., Bonds, J. A., Minshall, R. D., Pelligrino, D. A., Testai, F. D., & Lazarov, O.
 (2014). Neurogenesis and inflammation after ischemic stroke: What is known and where we go from here. *Journal of Cerebral Blood Flow & Metabolism*, *34*(10), 1573–1584.
- Tong, A., Sainsbury, P., & Craig, J. (2007). Consolidated criteria for reporting qualitative research (COREQ): A 32-item checklist for interviews and focus groups. *International Journal for Quality in Health Care*, *19*(6), 349–357. <u>https://doi.org/10.1093/intqhc/mzm042</u>
- Tong, A., Flemming, K., McInnes, E., Oliver, S., & Craig, J. (2012). Enhancing transparency in reporting the synthesis of qualitative research: ENTREQ. *BMC Medical Research Methodology*, *12*(1), 181–189.

- Toye, C., Parsons, R., Slatyer, S., Aoun, S. M., Moorin, R., Osseiran-Moisson, R., & Hill, K.
 D. (2016). Outcomes for family carers of a nurse-delivered hospital discharge intervention for older people (the Further Enabling Care at Home Program): Single blind randomised controlled trial. *International Journal of Nursing Studies*, *64*, 32-41.
- Treacy, M., & Stayt, C. L. (2019). To identify the factors that influence the recognizing and responding to adult patient deterioration in acute hospitals. *Journal of Advanced Nursing*, *75*(12), 3272–3285.
- Uchino, B. N., Trettevik, R., Kent de Grey, R. G., Cronan, S., Hogan, J., & Baucom, B. R. (2018). Social support, social integration, and inflammatory cytokines: A metaanalysis. *Health Psychology*, *37*(5), 462–471.
- Ulin, K., Olsson, L.-E., Wolf, A., & Ekman, I. (2016). Person-centred care–An approach that improves the discharge process. *European Journal of Cardiovascular Nursing*, *15*(3), e19–e26. <u>https://doi.org/10.1177/1474515115569945</u>
- Uniform Data System for Medical Rehabilitation. (2012). *The FIM Instrument: Its background, structure and usefulness*. University of Buffalo.
- United Nations Department of Economic and Social Affairs. (2020). *World economic situation and prospects*. United nations.

https://www.un.org/development/desa/dpad/publication/world-economic-situationand-prospects-2020/

- van den Kommer, T. N., Comijs, H. C., Aartsen, M. J., Huisman, M., Deeg, D. J., & Beekman, A. T. (2013). Depression and cognition: How do they interrelate in old age? *The American Journal of Geriatric Psychiatry*, *21*(4), 398–410.
- Van Diggele, C., Roberts, C., Burgess, A., & Mellis, C. (2020). Interprofessional education: Tips for design and implementation. *BMC Medical Education*, *20*(2), 455–460.
- Vaughn, V. M., Saint, S., Krein, S. L., Forman, J. H., Meddings, J., Ameling, J., Winter, S.,
 Townsend, W., & Chopra, V. (2019). Characteristics of healthcare organisations
 struggling to improve quality: Results from a systematic review of qualitative studies.
 BMJ Quality & Safety, 28(1), 74–84.

- Vlachos, G., Hege, I.-H., Wyller, T. B., Brækhus, A., Mangset, M., Hamre, C., & Brynjar, F.
 (2021). Cognitive and emotional symptoms in patients with first-ever mild stroke: The syndrome of hidden impairments. *Journal of Rehabilitation Medicine*, *53*(1).
- Von Koch, L. (2019). Individualised care and rehabilitation. In R. Suhonen, M. Stolt & E. Papastavrou (Eds.), *Individualized care* (pp. 151–162). Springer. <u>https://doi.org/10.1007/978-3-319-89899-5_14</u>
- Wagner, E. H., Ludman, E. J., Bowles, E. J. A., Penfold, R., Reid, R. J., Rutter, C. M., Chubak, J., & McCorkle, R. (2014). Nurse navigators in early cancer care: A randomized, controlled trial. *Journal of Clinical Oncology*, 32(1), 12–18.
- Wainwright, C., McClure, J., & McDowall, J. (2017). Misconceptions about stroke: Causal attributions for stroke-related symptoms reflect the age of the survivor. *Brain Impairment*, 18(3), 299–309. <u>https://doi.org/10.1017/BrImp.2017.13</u>
- Wald, H. S. (2020). Optimizing resilience and wellbeing for healthcare professions trainees and healthcare professionals during public health crises–Practical tips for an 'integrative resilience' approach. *Medical Teacher*, *42*(7), 744–755.
- Wang, Y., Yang, F., Shi, H., Yang, C., & Hu, H. (2017). What type of transitional care effectively reduced mortality and improved ADL of stroke patients? A meta-analysis. *International Journal of Environmental Research and Public Health*, *14*(5), Article 510.
- Ward, M. E., De Brún, A., Beirne, D., Conway, C., Cunningham, U., English, A., Fitzsimons, J., Furlong, E., Kane, Y., & Kelly, A. (2018). Using co-design to develop a collective leadership intervention for healthcare teams to improve safety culture. *International Journal of Environmental Research and Public Health*, *15*(6), Article 1182.
- Waring, J., Bishop, S., & Marshall, F. (2016). A qualitative study of professional and carer perceptions of the threats to safe hospital discharge for stroke and hip fracture patients in the English National Health Service. *BMC Health Services Research*, *16*, Article 297.

- Waring, J., Bishop, S., Marshall, F., Tyler, N., & Vickers, R. (2019). An ethnographic study comparing approaches to inter-professional knowledge sharing and learning in discharge planning and care transitions. *Journal of Health Organization and Management*, 33(6), 677–694.
- Webb, B. (2002). Using focus groups as a research method: A personal experience. *Journal of Nursing Management*, *10*(1), 27–35.
- Webb, C., & Kevern, J. (2001). Focus groups as a research method: A critique of some aspects of their use in nursing research. *Journal of Advanced Nursing*, 33(6), 798–805.
- Webb, A. J., Pendlebury, S. T., Li, L., Simoni, M., Lovett, N., Mehta, Z., & Rothwell, P. M. (2014). Validation of the Montreal cognitive assessment versus mini-mental state examination against hypertension and hypertensive arteriopathy after transient ischemic attack or minor stroke. *Stroke*, *45*(11), 3337–3342.
- Wei, H., Corbett, R. W., Ray, J., & Wei, T. L. (2020). A culture of caring: The essence of healthcare interprofessional collaboration. *Journal of Interprofessional Care*, *34*(3), 324–331.
- Wei, N., Yong, W., Li, X., Zhou, Y., Deng, M., Zhu, H., & Jin, H. (2015). Poststroke depression and lesion location: A systematic review. *Journal of Neurology*, 262(1), 81–90.
- Welten, J., Cox, V., Kruithof, W., Visser-Meily, J., Post, M., van Heugten, C., & Schepers, V. (2023). Intra- and interpersonal effects of coping style and self-efficacy on anxiety, depression and life satisfaction in patient–partner couples after stroke. *Neuropsychological Rehabilitation*, 33(5), 849–870.
- White, C. L., Brady, T. L., Saucedo, L. L., Motz, D., Sharp, J., & Birnbaum, L. A. (2015).
 Towards a better understanding of readmissions after stroke: Partnering with stroke survivors and caregivers. *Journal of Clinical Nursing*, *24*(7/8), 1091–1100.
 https://doi.org/10.1111/jocn.12739

- White, C. L., Korner-Bitensky, N., Rodrigue, N., Rosmus, C., Sourial, R., Lambert, S., & Wood-Dauphinee, S. (2007). Barriers and facilitators to caring for individuals with stroke in the community: The family's experience. *Canadian Journal of Neuroscience Nursing*, *29*(2), 5–12.
- Williams, C. L., Rittman, M. R., Boylstein, C., Faircloth, C., & Haijing, Q. (2005). Qualitative and quantitative measurement of depression in veterans recovering from stroke.
 Journal of Rehabilitation Research and Development, *42*(3), 277–290.
 https://doi.org/10.1682/JRRD.2004.02.0017
- Williams, S. (2000). Chronic illness as biographical disruption or biographical disruption as chronic illness? Reflections on a core concept. *Sociology of Health & Illness*, 22(1), 40–67.
- Winkler, D. (2022, 10 August). NDIS participants are left waiting for too long in hospital beds due to bureaucratic delays. The Conversation. <u>https://theconversation.com/ndis-</u> <u>participants-are-left-waiting-for-too-long-in-hospital-beds-due-to-bureaucratic-delays-</u> <u>188439</u>
- Wong, F. K., & Yeung, S. M. (2015). Effects of a 4-week transitional care program for discharged stroke survivors in Hong Kong: a randomised controlled trial. *Health & social care in the community*, 23(6), 619–631.
- Wood, J. P., Connelly, D. M., & Maly, M. R. (2010). 'Getting back to real living': A qualitative study of the process of community reintegration after stroke. *Clinical Rehabilitation*, 24(11), 1045–1056.
- World Health Organization. (1948). *Constitution of the World Health Organization*. <u>https://apps.who.int/gb/bd/PDF/bd47/EN/constitution-en.pdf?ua=1</u>

World Health Organization. (2007). Everybody's business: Strengthening health systems to improve health outcomes who's framework for action. <u>https://apps.who.int/iris/bitstream/handle/10665/43918/9789241596077_eng.pdf?seq</u> <u>uence=1&isAllowed=y</u> World Health Organization. (2010). *Framework for action on interprofessional education* & *collaborative practice*. <u>https://www.who.int/publications/i/item/framework-for-action-on-interprofessional-education-collaborative-practice</u>

World Health Organization. (2013). How to use the ICF—A practical manual for using the international classification of functioning, disability and health.

https://www.who.int/publications/m/item/how-to-use-the-icf---a-practical-manual-forusing-the-international-classification-of-functioning-disability-and-health

World Health Organization. (2015). WHO global strategy on people-centred and integrated health services. <u>https://www.who.int/servicedeliverysafety/areas/people-centred-</u> care/global-strategy/en/

World Health Organization. (2016). Integrated care models: An overview. <u>https://www.researchgate.net/publication/315493946_Integrated_care_models_an_o</u> <u>verview_Copenhagen_WHO_Regional_Office_for_Europe_2016</u>

World Health Organization. (2017a). *Depression and other common mental disorders: Global health estimates*. <u>https://apps.who.int/iris/handle/10665/254610</u>

World Health Organization. (2017b). *Strengthening resilience: A priority shared by Health* 2020 and the Sustainable Development Goals.

http://www.euro.who.int/en/countries/monaco/publications/strengthening-resilience-a-

priority-shared-by-health-2020-and-the-sustainable-development-goals-2017

World Health Organization. (2021). Health promotion glossary of terms.

https://www.who.int/publications/i/item/9789240038349

World Health Organization. (2023a). Stroke, Cerebrovascular accident.

https://www.emro.who.int/health-topics/stroke-cerebrovascular-accident/index.html

World Health Organization. (2023b). Rehabilitation.

https://www.who.int/news-room/fact-sheets/detail/rehabilitation

World Health Organization. (2022a, 2 December). Global report on health equity for persons

with disabilities: Executive summary.

https://www.who.int/publications/i/item/9789240063624

World Health Organization. (2022b, 18 March). Nursing and midwifery.

https://www.who.int/news-room/fact-sheets/detail/nursing-and-midwifery

- World Stroke Organization. (n.d.). *Learn about stroke*. <u>https://www.world-stroke.org/world-</u> <u>stroke-day-campaign/why-stroke-matters/learn-about-stroke</u>
- Wottrich, A. W., Astrom, K., & Lofgren, M. (2012). On parallel tracks: Newly home from hospital—People with stroke describe their expectations. *Disability & Rehabilitation*, 34(14), 1218–1224.
- Wu, C.-Y., Skidmore, E. R., & Rodakowski, J. (2019). Relationship consensus and caregiver burden in adults with cognitive impairments 6 months following stroke. *PM&R*, *11*(6), 597–603.
- Yeung, E. H., Szeto, A., Richardson, D., Lai, S. H., Lim, E., & Cameron, J. I. (2015). The experiences and needs of Chinese-Canadian stroke survivors and family caregivers as they re-integrate into the community. *Health & Social Care in the Community*, 23(5), 523–531.
- Yeung, S. M., Wong, F. K. Y., & Mok, E. (2011). Holistic concerns of Chinese stroke survivors during hospitalization and in transition to home. *Journal of Advanced Nursing*, 67(11), 2394–2405. <u>http://onlinelibrary.wiley.com/doi/10.1111/j.1365-</u> 2648.2011.05673.x/abstract
- Young, M. E., Lutz, B. J., Creasy, K. R., Cox, K. J., & Martz, C. (2014). A comprehensive assessment of family caregivers of stroke survivors during inpatient rehabilitation. *Disability and Rehabilitation*, 36(22), 1892–1902.

https://doi.org/10.3109/09638288.2014.881565

- Zapater-Fajarí, M., Crespo-Sanmiguel, I., Pulopulos, M. M., Hidalgo, V., & Salvador, A. (2021). Resilience and psychobiological response to stress in older people: The mediating role of coping strategies. *Frontiers in Aging Neuroscience*, *13*, Article 632141.
- Zhang, W., Liu, Z., Zhou, X., & Zhou, L. (2020). Resilience among stroke survivors: A cohort study of the first 6 months. *Journal of Advanced Nursing*, *76*(2), 504–513.

- Zhang, X.-Y., Sha, K.-H., Ma, X.-Y., Li, X.-M., & Zhang, M.-H. (2023). Dyadic psycho-social interventions for stroke survivors and family caregivers: A systematic review and meta-analysis of randomized controlled trials. *Journal of Advanced Nursing*. Advance online publication.
- Zhong, W., Geng, N., Wang, P., Li, Z., & Cao, L. (2016). Prevalence, causes and risk factors of hospital readmissions after acute stroke and transient ischemic attack: A systematic review and meta-analysis. *Neurological Sciences*, *37*(8), 1195–1202. <u>https://doi.org/10.1007/s10072-016-2570-5</u>
- Zimmerman, W. D., Grenier, R. E., Palka, S. V., Monacci, K. J., Lantzy, A. K., Leutbecker, J. A., Geng, X.& Denny, M. C. (2021). Transitions of care coordination intervention identifies barriers to discharge in hospitalized stroke patients. *Frontiers in Neurology*, *12*, Article 573294.

Appendices

Appendix 1: Published Integrative Review Protocol

NIHR National Institute for Health Research PROSPERO International prospective register of systematic reviews

Challenges and opportunities for stroke survivors and carers in hospital to home transition care: a systematic integrated review protocol Langduo Chen, Lily Xiao, Diane Chamberlain

Citation

Langduo Chen, Lily Xiao, Diane Chamberlain. Challenges and opportunities for stroke survivors and carers in hospital to home transition care: a systematic integrated review protocol. PROSPERO 2019 CRD42019136068 Available from: https://www.crd.york.ac.uk/prospero/display_record.php?ID=CRD42019136068

Review question

1. What are the opportunities to improve hospital to home transition care?

2. What are the challenges faced by stroke survivors and carers in hospital to home transition care?

Searches

The following electronic databases will be searched for eligible literature: CINAHL (Cumulative Index to Nursing and Allied Health Literature), MEDLINE, Web of Science, ProQuest, Scopus, and ScienceDirect. A manual search of reference lists of eligible studies and reviews will also be conducted.

Search dates: 18 June 2018 to 31 October 2018.

Restrictions:

- Publication in English
- · No publication period restrictions

The above sources will be re-run prior to the final analysis to identify further studies for inclusion.

Types of study to be included Inclusion:

Original qualitative, quantitative, mixed method studies, systematic reviews.

Exclusion:

Pilot studies, scoping reviews, narrative reviews, reports, conference proceedings, discussions or commentaries.

Condition or domain being studied

Discharge from hospital to home marks a beginning for stroke survivors and their family carers to take charge of the care and cope with challenges arisen from the transition period (Ellis-Hill, et al., 2009, Nordin, Sunnerhagen, & Axelsson, 2015). In the context of shortened hospital stay and early discharge in most healthcare systems in the world due to cost-saving pressure, stroke survivors and carers are expected to deal with more complex healthcare issues at home, e.g. prevention of stroke associated complications, managing health conditions underlying stroke, while making efforts to gain physical and psychosocial recovery (Burgan 2015, Langhome et al. 2017). Although studies on the poststroke hospital to home transition are abundant, there is a gap in the literature to synthesize these studies to gain a greater understanding of issues of concern in this increasingly important field.

This review examines the period from before discharge up to 6 months after hospital discharge as the majority of studies on the hospital to home transition reported a 3 to 6 months follow-up.

PROSPERO



International prospective register of systematic reviews

Reference:

Langhome, P., Baylan, S., & Early Supported Discharge Trialists. (2017). Early supported discharge services for people with acute stroke (Review). Cochrane Database of Systematic Reviews(7).

Participants/population Inclusion:

inclusion.

Adult stroke survivors aged 18 years and above.

 Carers: Carers are defined as unpaid carers including spouse or partner, family members, friends, or significant others who provide physical and emotional support to stroke survivors.

- Healthcare professionals: multidisciplinary stroke care team members and community service providers.

Exclusion:

Stroke survivors younger than 18 years old.

 Multiple patient populations including brain tumour, traumatic brain injury, other neurological condition and/or other

diagnoses where stroke specific data cannot be separated.

Intervention(s), exposure(s) Inclusion:

Studies that explored the experience, care needs, expectations of stroke survivors, their carers and healthcare professionals during the hospital to home transition or a transition facilitating intervention up to 6 months after discharge.

Exclusion:

- Studies that examined
- · Poststroke long term experiences and needs (over 6 months after hospital discharge).
- · Discharges to facilities other than home.

 Studies conducted across multiple periods of stroke trajectory where the hospital to home care transition period cannot be separated.

Comparator(s)/control Not applicable

Context

Studies in hospital on patient wards, outpatient clinics, in community.

Main outcome(s)

The primary outcomes are (1) professional practices, services, care approaches, care models, self-care, patient/carer educations and resources that enabled the hospital to home transition and resulted in positive outcomes of post stroke recovery and rehabilitation: (2) factors that impeded the hospital to home transition and contributed to poor care outcomes.

* Measures of effect

This review examines the period from before discharge up to 6 months after hospital discharge as the majority of studies on the hospital to home transition reported a 3 to 6 months follow-up.

Page: 2/6



PROSPERO International prospective register of systematic reviews

Additional outcome(s) None. * Measures of effect

Not applicable.

Data extraction (selection and coding) Study Selection:

The screening process will be carried out by three reviewers.

Stage 1

Using EndNote, the initial citations will be shared amongst reviewers. Reviewer 1 will screen each article based on the title and abstracts against the inclusion and exclusion criteria. The rest of the reviewers will screen the title and abstracts of half of the citations. The screening results will be compared, and any disagreements will be resolved through discussions in regular team meetings. Citations that are not selected by two reviewers will be excluded for full-text review.

Stage 2

Reviewer 1 will source full text and save PDF versions into EndNote database from the title and abstracts screening process. Each full text of all included articles will be reviewed by two reviewers independently. Any discrepancies will be discussed and resolved in team meetings. All decisions will be recorded in EndNote.

Data Extraction:

 Data that will be extracted from study documents include information about the study, study design, methodology, participant demographics, main findings, limitations, and implications.

· Interview excerpts will be extracted from qualitative studies for analysis.

 The text-in-context method developed by Sandelowski and associates (2013) will be applied to extract data from the quantitative, mixed-method and systematic review studies. Data extraction form will be used to transform quantitative data into context-preserved restatements of findings (Sandelowski, Leeman, Knafl, & Crandell, 2013).

Reviewer 1 will extract data. Reviewer 2 and 3 will review half of the extracted data.

 Any discrepancies will be discussed and resolved in team meetings. A third party will be involved to solve disagreement if required.

· Original study authors will be contacted for missing data or clarification of data details.

. The software NVivo 11 will be used to record and manage data extraction.

Page: 3/6

NIHR National Institute for Health Research

PROSPERO International prospective register of systematic reviews

Risk of bias (quality) assessment

Critical Appraisal Skills Programme (CASP) tools (CASP 2018) will be used to evaluate the qualitative studies, randomised controlled trials and systematic reviews. Quality of the cross-sectional and descriptive quantitative studies will be assessed with a set of questions developed by Büttner and Muller (2015) for appraising epidemiological studies. The mixed method study will be appraised by using the 'Mixed Methods Appraisal Tool (MMAT) -Version 2011' developed by Pluye and colleagues (2011). All studies that meet the inclusion criteria will be included disregarding the study quality as methodologically weak studies may provide novel understandings that may not be produced in methodologically strong studies (Carroll et al. 2012).

References:

Büttner, P., & Muller, R. (2015). Epidemiology (2nd ed.). England, UK: Oxford University Press.

Carroll, C, Booth, A, and Lloyd-Jones, M 2012, 'Should we exclude inadequately reported studies from

qualitative systematic reviews? An evaluation of sensitivity analyses in two case study reviews', Qualitative

Health Research, vol. 22, no. 10, pp. 1425-34.

Pluye, P., Robert, E., Cargo, M., Bartlett, G., O'cathain, A., Griffiths, F., . . . Rousseau, M. (2011). Proposal: A mixed methods appraisal tool for systematic mixed studies reviews. Retrieved from http://mixedmethodsap praisaltoolpublic.pbworks.com/w/file/fetch/84371689/MMAT%202011%20criteria%20and%20tutorial%20201 1-06-29updated2014.08.21.pdf

Strategy for data synthesis

Synthesis of data will be conducted based on the integrated design described by Sandelowski, Voils, and Barroso (2006). Qualitative and quantitative data are viewed as being able to be transformed into each other and generate findings to serve the same study purpose. The aim of data analysis is to translate findings so they can be combined to answer the same research questions (Sandelowski, et al., 2006). Underpinned by the principles of this synthesis design and based on the review questions, data will be highlighted in different colours and categorised into challenges and opportunities. Under these two categories, data for the first 6 months post-discharge will be compared constantly within the same category and across categories to identify relevant findings.

Reference:

Sandelowski, M., Voils, C. I., & Barroso, J. (2006). Defining and designing mixed research synthesis studies. Research in the schools: a nationally refereed journal sponsored by the Mid-South Educational Research Association and the University of Alabama, 13(1), 29-40.

Analysis of subgroups or subsets Not planned.

Contact details for further information Langduo Chen chen0365@flinders.edu.au

Organisational affiliation of the review Southern Adelaide Local Health Network www.flinders.sa.gov.au

Review team members and their organisational affiliations Ms Langduo Chen. Southern Adelaide Local Health Network, Flinders Medical Centre Professor Lily Xiao. Flinders University of South Australia Assistant/Associate Professor Diane Chamberlain. Flinders University of South Australia

Page: 4/6



PROSPERO International prospective register of systematic reviews

Type and method of review Systematic review

Anticipated or actual start date 10 July 2018

Anticipated completion date 30 June 2019

Funding sources/sponsors None

Conflicts of interest

Language English

Country Australia

Stage of review Review Ongoing

Subject index terms status Subject indexing assigned by CRD

Subject index terms Caregivers; Hospitals; Humans; Stroke; Survivors; Transitional Care

Date of registration in PROSPERO 23 July 2019

Date of publication of this version 23 July 2019

Details of any existing review of the same topic by the same authors

Stage of review at time of this submission

Stage	Started	Completed
Preliminary searches	Yes	No
Piloting of the study selection process	Yes	No
Formal screening of search results against eligibility criteria	Yes	No
Data extraction	Yes	No
Risk of bias (quality) assessment	Yes	No
Data analysis	Yes	No
Versions		

23 July 2019

PROSPERO

This information has been provided by the named contact for this review. CRD has accepted this information in good faith and registered the review in PROSPERO. The registrant confirms that the information supplied for this submission

Page: 5/6



PROSPERO International prospective register of systematic reviews

is accurate and complete. CRD bears no responsibility or liability for the content of this registration record, any associated files or external websites.

Page: 6/6

Appendix 2: Medline Search Strategy

#▲	Searches	Results
1	(stroke or cerebrovascular accident* or cerebrovascular apoplexy or brain vascular accident*).tw.	209027
2	(survivor or patient).tw.	1953801
3	1 and 2	34658
4	"continuity of patient care"/ or transitional care/	17784
5	limit 4 to english language	16393
6	patient discharge/	25383
7	limit 6 to english language	23007
8	community.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]	485494
9	service.mp. [mp=title, abstract, original title, name of substance word, subject heading word, floating sub-heading word, keyword heading word, protocol supplementary concept word, rare disease supplementary concept word, unique identifier, synonyms]	319769
10	8 and 9	40807
11	4 or 6	41103
12	10 or 11	80631
13	3 and 12	644

Appendix 3: Summary of Selected Studies

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
1	Almborg et al., 2008 <i>Sweden</i>	To describe stroke patients' perceptions of their participation in the discharge planning process and identify correlates of perceived participation	Consecutive sampling In a hospital in Southern Sweden 188 stroke patients	Quantitative Questionnaire via interview 2–3 weeks after discharge Exploratory factor analysis Prospective, descriptive cross- sectional study	 Most satisfied with information related to illness, medication & rehabilitation. Poorly involved in medical treatment decision-making, goal-setting and evaluation of needs. Younger patients perceived higher participation than older patients—may be older patients wanted HCP to take initiatives. Patients with shorter hospital stay were more satisfied. Patients with high-level dependency perceived poorer participation than those with low-level dependency. 	 The exclusion of patients with aphasia or moderate to severe cognitive impairment Single unit 	Understanding of stroke patients' participation and factors influencing the participation helps clinicians to facilitate a more person-centred discharge planning.
2	Almborg et al., 2009 <i>Sweden</i>	To describe relatives' perception of participation in discharge planning of patients with stroke To identify the association between patient characteristics and relatives' perceptions Participation	Consecutive sampling In a hospital in Southern Sweden 152 relatives of stroke patients	Quantitative Questionnaire via interview 2–3 weeks after discharge Multiple regression models Prospective, descriptive cross- sectional study	 Most relatives did not receive sufficient information on stroke, medication, rehabilitation, care and community support. 80% of them perceived no participation particularly in goal-setting and identification of care needs. Longer hospital stay, higher educated and female patients and relatives were associated with higher participation. 	Single unit	Understanding of relatives' participation and factors influencing the participation helps clinicians to facilitate a more person-centred discharge planning.
3	Bakas et al., 2002 USA	To determine the self-reported needs and concerns of family caregivers of stroke survivors soon after discharge to the home setting	Convenience sampling An outpatient neurology clinic in an inner-city hospital in a large Midwestern city of the USA. 14 family caregivers of stroke survivors	Qualitative Telephone interviews within 6 months after discharge Using a pre- determined code list to identify categories of interview data	 Information: need to be better educated with stroke signs and symptoms, and management of complications. To meet this need, a network of support groups sessions facilitated by professionals was needed prior to discharge. Emotions and behaviours: need help in providing emotional support and managing behavioural problems manifested by the stroke survivor. Physical care: 	 All participants are female caregivers; male caregivers may have different needs and concerns Stroke severity was not assessed—may affect the findings 	Understanding the needs and concerns of stroke family caregivers in transition from hospital to home helps professionals better identify caregiver needs so interventions could be in place before discharge. The findings could serve as a guide to

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
					 concerns about medication management, motivating survivor, proper diet, fall prevention. Instrumental care: managing finances, providing transportation, finding respite care. Personal response to caregiving: managing own emotions, adapting to changed social life, own health challenges; those were ignored by professionals. 		develop tailored educational activities before discharge to enable a smooth transition home.
4	Cameron et.al., 2014 <i>Canada</i>	To explore stroke survivors', caregivers', and professionals' perceptions of Weekend Pass (WP, weekend stay at home) offered during inpatient rehabilitation and its role in facilitating the transition	Purposive Sampling First interview 16 patients & 15 caregivers Second interview 11 patients & 11 caregivers 20 HCP: three focus groups & one interview A rehabilitation facility in a large urban centre in Canada	Qualitative Descriptive First interview during the week following the first WP. Second interview 4 weeks post-discharge. Three focus groups or an in-depth interview with HCP.	 Preparing for patients to be safe at home: Professionals discussed & assessed patient readiness including home assessment patients/carers desired to be better prepared with expectations for WP and training of family Professionals' insufficient time to train and discuss. Gaining insight through WP: provides insight into ability of care needs. informs future inpatient therapy sessions. Professionals' caseload and work hours limited ability to interact with caregivers who work. Need to develop a formal procedure when returning from WP. 	 Exclusion of non- English speakers Single site Absence of physician's views 	WP has therapeutic value—guides inpatient rehabilitation after WP. Highlighted emotional support needed in the process. WP helps stroke survivors and family manage most challenges in transition care.
5	Chouliara et al., 2014 <i>UK</i>	To explore professionals' perspectives of ESD service	Purposive sampling 35 professionals Two ESD services in the East Midlands of UK	Qualitative Cross-sectional Semi-structured interviews	 Facilitators: specific and flexible eligibility criteria the systems needed to be reflective of individuality of each case flexibility of the length of the intervention the support of rehabilitation assistants to sustain the service collaborative links with other services by staff rotations across services, participation in meetings and common training events. Challenges: lack of clarity regarding the referral decision-making process 	 Nature of qualitative study—not generalisable The findings only reflected service provision in urban and semi-urban setting 	The findings highlighted the contribution of ESD service to early transition home for stroke patients. Further exploration was warranted to evaluate strategies aiming to improve interactions between professionals in different settings to ensure a seamless transition.

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
6	Cobley et al., 2013 UK	To investigate patients' and carers' experiences of ESD services	Purposive sampling 27 stroke patients and 15 carers Community stroke services in Nottinghamshire, UK	Qualitative Semi-structured interviews	 fragmented stroke care pathway—(1) the process of securing social care input was delayed; (2) lack of specialist community services that could respond to the needs of more disabled patients. duplication of assessments across services. Perceived impact of Early Supported Discharge: reduction of the length of hospital stay without compromising the intensity of rehabilitation input contributed (bridging) to the improvement of collaboration between the acute and community stroke services. ESD was intensive, stroke-specific and patient-centred intervention emotional issues could be addressed timely smoother transition to home. Limited support in dealing with carer strain. Lack of education and training of carers. Inadequate provision and delivery of stroke-related information. Disjointed transition between ESD and ongoing rehabilitation services. Patients and carers were satisfied with home-based rehabilitation. 	 Participants' reports open to biases related to self- evaluation Majority of carers were female Participants' willingness to provide feedback may lead to exaggerated and inaccurate descriptions Design did not allow 	The study provided unique insights into patients' and carers' experiences of ESD services. It highlighted areas for service development and provision.
7		To develop the		Qualitativa		over time	The findings provide
1	al., 2010	understanding of what constitutes a 'good' or 'poor' experience in relation to the transition from	20 stroke survivors and 13 carers within 1 month of being discharged from hospital following a stroke	Semi-structured interviews	 Stroke survivors recovery changes over time. Provision of services need to be based on the stroke survivors' and health providers' shared models of recovery. Communication of discharge plans was critical in hospital-to-home transition. 	 Only mobility was explored in-depth. Lack of in-depth analysis of other stroke-related changes Exclusion of people with severe 	some understanding of stroke survivors' and carers' needs when discharged from hospital to home. The researchers' recommendation may

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
		hospital to home following a stroke	Participants were recruited from an acute stroke ward in the district general hospital		 Understanding of patients' model of recovery revealed expectations and training needs of carers. Many patients were discharged with unanswered questions about stroke. 	communication or cognitive deficit • Single-site study	inform the discharge planning process.
8	Greenwood et al., 2009 <i>UK</i>	To investigate the experiences of informal carers of stroke survivors in the first 3 months after discharge	Purposive sample 31 stroke caregivers 3 months after discharge	Qualitative ethnography In-depth interviews at three time points— prior to discharge and 1 month and 3 months post-discharge	 New carers were struggling to provide home care in early stage of home care after discharge Established carer were more confident in coping with home care Unreliable services Coping strategies: establishing routines focusing on the present reduced uncertainty relying on established coping skills identifying positive aspects of caregiving different views on future care for stroke survivors. 	 Only followed up to 3 months after discharge Survivors' present may have impacted on the openness of caregivers during interviews The research setting was an ethnically diverse area in London 	Caregivers' experiences changed over time during the transition. Findings have implications for assessment and support provision. Differences between new caregivers and established caregivers help clinicians to develop more individualised interventions.
9	Gustafsson & Bootle, 2013 <i>Australia</i>	To explore and contrast the transition to home experience of stroke survivors and their carers during discharge and first month at home	Purposive sampling 5 first-time stroke patients 5 carers Recruited from a major metropolitan hospital in Queensland	Descriptive qualitative design Semi-structured interviews	 Inability to participate in meaningful activities during inpatient rehabilitation may have prevented adaptation and recovery. Weekend leave, overnight leave and family meetings were identified as positive strategies for hospital-to-home transition care. Lack of information on a contact person when having questions after discharge (unclear, check and change) Carers needs more support due to change of role. Uncertainty about ongoing care after completion of rehabilitation program. 	 Small sample size Recruitment from one single site Study designs did not allow investigation of changes over time 	Useful insights gained from this study are important to inform further research to explore alternate approaches for rehabilitation service delivery and transition to home.
10	Grant et al., 2006 USA	To identify problems and associated feelings experienced by family caregivers of stroke survivors during the second and	Purposive sampling 22 family caregivers assigned to the intervention group of an RCT: 1 week post-discharge, 3- hour home visit with training on the use of a positive	Descriptive quantitative Data form and biweekly telephone contacts	 Week 6: concerns about safety—falls issues with managing personal care issues with managing emotional, cognitive and behavioural changes felt unsure, scared, helpless, frustrated. Week 8: 	 Small sample size Recruitment of participants from an intervention group of an RCT Stroke survivors with moderate level of disability 	The findings help development of interventions targeting specific issues at varied time frames and involving both consumers and professionals.

No.	Author and vear	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
		third month after discharge from a rehabilitation facility to home	problem orientation and problem-solving skills		 same issues as Week 6; 50% felt to be able to manage better. Difficulties in managing symptoms. Week 10: same issues as Week 8, significantly more positive and able to manage issues as survivors become more independent more evident of issues with finances and tenuous family and social relationships. Week 12: same issues as Week 10; continued to be positive towards management caregivers were able to establish realistic goals for self to participate in social activities and for the care of the stroke survivors acknowledged that cognitive and behavioural issues may be long-term. 	Lack of data analysis details	
11	Hall et al., 2012 <i>Australia</i>	To explore the transition experiences of individuals with non-traumatic brain injury using mixed-methods approach	Using maximum of variation sampling <i>N</i> = 6 Acquired Brain Injury (Subarachnoid Haemorrhage & stroke)	Qualitative: in-depth interview 6 months after discharge Quantitative: self- reported questionnaires Assessment prior to hospital discharge and 6 months after discharge	 Qualitative findings: changes in role performance and relationships; tapering off social relationships inadequate support due to distance, time and pre-injury family relationships participants were satisfied with hospital extended post-discharge services lack of support groups, long waiting between reviews in the community difficulties in navigating healthcare system—unaware of available services, confused about process, unsure how to use information people in the community lacked understanding of brain injury Quantitative findings at 6 months: psychosocial integration remained the same for half of participants functional ability also remained the same for half of participants emotional function remained the same for most participants improved perceived health status for most participants. 	 Sample representation: 4 SAH due to aneurysm Single site Single time point for interviews 	Findings provide evidence for clinical practice and service development.

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
12	Langhorne et al., 2017 UK	To establish if ESD services can result in a better patient recovery and if they are as acceptable and affordable as usual services	17 RCTs from countries—Canada, Australia, India, Thailand, Norway, Portugal, Ireland, Denmark, UK and Sweden. 2422 participants (moderate degree of disability) ESD team—multi-D coordination— including at least physiotherapist, OT, doctor and speech therapist; nurses not always included No ESD team—no multi-D coordination	Systematic review	 Reduced the length of hospital stay by 6 days in the ESD group compared to the usual care group Reduced death or institutional care in the ESD group compared to the usual care group Small improvement of ADLs in the intervention group compared to the usual care group More satisfaction with services in the ESD group compared to the usual care group More satisfaction with services in the ESD group compared to the usual care group Greatest reduction of death and dependency was identified in the coordinated ESD team with case manager in subgroup analyses Cost of ESD was a reduction to a moderate increase 	 Insufficient information for 3 studies Missing data for small amount of patients Systematic review authors being authors of individual trials 	Effectiveness of ESD services was delivered by multi-D team (including Nurses). The identification of multi-D-team-involved services warrants further trial targeting stroke population with moderate level of disability.
13	Lou et al., 2017 <i>Denmark</i>	To investigate how mild stroke patients and their partners experience and manage everyday life in the context of ESD	Purposive sample <i>N</i> = 40 (22 stroke survivors, 18 partners) Central Denmark region	Qualitative 3–6 weeks after stroke onset	 Theme 1—Home as a healing place: minority of participants felt inadequate information on their condition coming home was a relief and as own space compared to structured day in hospital ESD services supported a safe transition. Theme 2—Flow of everyday life: physical and cognitive impairment impact on daily activity for some collaboration between stroke survivors and their partners maintains flow of daily life patients with chronic illnesses adjusted better due to experience with healthcare system uncertainty—e.g. recognition of symptoms of another stroke. Theme 3—Professional safety net: home a better arena for rehabilitation satisfied with quality of ESD—safe and highly accessible ESD did not focus on goal of returning to work. 	 Recruitment via ESD team—unsure consistency of criteria and total recruitment Interviewing survivor and caregiver as a dyad together— potential under- report of issues Relatively optimistic responses to outcomes shortly after discharge Responses to longer time (i.e. after completion of ESD remain unknown. Homogeneous sample 	ESD services are a valuable solution for mild stroke survivors who return home after discharge from hospital. ESD could be trialled in a broader region.

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
14	Nordin et al., 2015 <i>Sweden</i>	To describe patients' expectations of coming home very early after stroke with support and rehabilitation at home	10 participants from the intervention group in an RCT project	Descriptive qualitative Interviews 0–5 days before discharge (after rehab goal-setting was agreed)	 Mixed feelings about coming home—fear, insecurity, important milestone of recovery. Participants had confidence of support to gain independence from the team. Expectation that the visits from the team would reduce participants' insecurity. Unrealistically high expectation of functional status. 	 Small sample size Single site Shorter interviews with three participants, two interviews before setting rehab goals—may affect the description of experience 	 Implications for clinical practice: exploring patients' expectations prior to discharge may help professionals and stroke survivors/carers to set agreed and realistic goals for recovery and rehabilitation. identify issues that might compromise the transition; Building on the trust and partnership to achieve the goals.
15	O'Brien et al., 2014 <i>Australia</i>	Explore HCP perceptions of stroke carer roles and support needs; Examine carer needs across the stroke care trajectory to assist with the development of an OHP to support carers	Purposive sampling 23 professionals who represented a range of disciplines An Australian metropolitan public health service 4 clinical settings— acute care, rehabilitation, transitional care and community rehabilitation	Focus group—dual- moderated focus groups approach	 Support for carers was crucial, especially during hospital-to-home transition. The role of HCP and carer changes over time. Professionals' role in supporting carer may be challenged by workload. Carers devote to the role rather than asking for help. Challenge for carer to access support. Timing and volume of information provision was a challenge to HCP. Carers experience loss and financial disadvantage. 	 Single site Only included perception from professionals 	Informing development of OHP. Training professionals with OHP may improve consistency in the provision of carer support across stroke pathways.
16	Perry & Middleton, 2011 <i>Australia</i>	To advance understanding of stroke caregiving in Australia	Convenience sampling Acute stroke patients from two hospitals in two Australian states— Sydney (New South Wales) and Brisbane (Queensland)	Quantitative Descriptive with survey instruments through interviews at 1 and 3 months follow-up after discharge	 Inadequate preparation for discharge. Lack of information provision. Lack of involvement in discharge arrangements. Services were least satisfactory where they were most needed. 	 Small sample size Assessment at different time post- discharge—i.e. assessment at one site was later than the other Lack of validity of assessment tools 	The findings of this study provide insights into current issues with informal stroke caregiving in Australia. It may help formulate consumer-guided questions for future research.

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
			32 patient–carer dyads				
17	Pringle et al., 2008 UK	To understand the experiences of stroke patients and their carers during the early days following discharge from hospital	28 articles 943 participants References examined for the first month after discharge home	Systematic review 26 qualitative studies 1 quantitative 1 mixed-method	 Little evidence available on early discharge experience, especially in the UK. Younger stroke patients' and carers' needs are neglected. However, young and the old shared many similar experiences. First month at home was most distressful time. Few studies made reference to the time of discharge home. Most studies excluded aphasic patients. 	 Heterogeneity of original studies 	The issues stroke patients and carers described during their early days at home will inform the preparation for discharge. More focused follow-up services to meet individual needs in the community are needed.
18	Pringle et al., 2013 <i>UK</i>	To investigate and improve understanding of the experiences of patients and their carers during the first month at home following discharge from hospital	12 patient/carer dyads Setting—local hospital ward. Other information unclear	Qualitative Interpretive phenomenology analysis Double hermeneutic process In-depth interviews and self-reported diaries	 Revisioning: 1 month post-discharge a dynamic time altered concept of image and identity loss and grief lack of information—needs unpredictable while in hospital. Reconnecting: family and social support regaining control and exit sick role. Revisiting: reflection on previous life recounting stroke event. 	 Small sample size Localised nature of study 	Understanding of patient/carer experience will benefit advance planning and preparatory training from health professionals to assist with psychosocial transitions. Early identification of dysfunction may lead to fewer negative outcomes.
19	Rittman et al., 2007 <i>USA</i>	To describe three domains of psychosocial experiences of stroke survivors during the first months following discharge for acute stroke	125 stroke survivors and caregivers	Qualitative Semi-structured interviews and field observations during home visits at 1 month post-discharge	 The experiences are characterised by a process of adapting to changes in sense of self, connections with other and community participation. All participants, regardless of their functional assessment scores, could benefit from interventions that prepare them to manage multiple psychosocial changes during the transition. 	Limited sample— veterans	Given that the transition period is a time of adjustment, interventions are needed to support stroke survivors and their caregivers. The result also highlighted importance to assess and match the level of support with the needs of stroke survivors.
20	Shyu et al., 2008 <i>Taiwan</i>	To examine the effects of a discharge planning program that targets preparation of	158 older stroke patients and their caregivers from four homogeneous neurological wards of a 3,000-bed	Quantitative Randomised experimental design A nurse-directed, family-oriented,	 Experimental group showed significantly better discharge preparation, greater satisfaction with meeting discharge needs. No significant difference was found between caregiver groups in perceived balance of competing needs. 	 Attrition of 26.8% Lack of baseline data for some assessment The risk of contamination between control and 	This caregiver-oriented discharge planning program, with its emphasis on individualised health education and home visits following

No.	Author and vear	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
		family caregivers, in Taiwan	medical centre in northern Taiwan	discharge-intervention program Three assessments— within 3 days of admission, before discharge, 1 month after discharge		 experimental groups because of randomisation within the same hospital Lack of blinded assessment may affect quality of assessment by professionals. 	discharge may improve caregivers' preparation and the satisfaction of their needs during the discharge transition. It warrants a larger scale of research to further examine the effectiveness of the program.
21	Simeone et al., 2015 <i>Italy</i>	To describe the experience of stroke survivors 3 months after being discharged home from rehabilitation hospitals	Purposive sampling From a longitudinal study 15: Rome = 13; Naples = 2 Stroke patients	Qualitative phenomenology	 Deeply change lives Vivid memory of acute phase of stroke Slowed lives Relief after recovering from stroke Being a burden to family members 	 Mostly male patients Mostly independent patients 	Understanding of stroke patients early discharge experience helps HCP reflect on how to deliver person- centred care.
22	Wong & Yeung, 2015 <i>Hong Kong</i>	To test the effectiveness of a nurse-led transitional care program	Purposive sampling From three regional hospitals within the same cluster in Hong Kong 54 each in controlled and intervention group	Quantitative RCT Follow-up at 4 and 8 weeks	 The intervention group had significantly higher quality of life and satisfaction in interpersonal skills and health advice. The intervention group had significantly less depressive symptoms and emergency department visit. The ADL performance was better for the intervention group. 	 Single region Missing values replaced by group means Only followed up for 8 weeks 	The effectiveness of this nurse-led transition care intervention warrants further research and clinical trial to see the effect for the transition period up to 6 months. The findings confirmed that nurses are in the appropriate position to lead and coordinate transition care programs.
23	Wood et al., 2010 <i>Canada</i>	To examine the process of community reintegration after stroke for their perspective	Purposive sampling N = 10 (four women, six men) 2 rehabilitation units	Qualitative Grounded theory Interviews Observational field notes Total of 46 interviews at pre-discharge and 2 weeks, 3 months, 6 months and 1 year post-discharge	 Community reintegration was a process of transitioning through a series of goals. Early stage of discharge home was overwhelming from appointments, family and HCP visits. Decreased confidence due to altered physical ability and dependence on others—e.g. driving, ADLs. 3 months onwards after discharge, participants adjusted expectations and participated more meaningful activities in the community. 	 Sample was from a sub cohort of a larger trial Participants had high functioning Exclusion of aphasic and cognitive impaired 	Understanding of the process of community reintegration helps clinicians to address the importance of creating balance between capacity and expectations by setting realistic and attainable goals. Peer support groups may also benefit stroke

No.	Author and year	Aim/objective	Sample and setting	Methodology and methods	Main findings	Limitations and rigour/validity	Significance to the review
							patients' social reintegration.
24	Wottrich et al., 2012 <i>Sweden</i>	To investigate how stroke survivors describe their expectations when they were discharged from rehabilitation ward and during the first 3 months at home	Purposive sampling 5 stroke survivors From two rehabilitation centres in Stockholm, Sweden	Qualitative Semi-structured interview	 Prior to discharge: home practice before discharge was beneficial for transition. physical training plan was important to maintain hope for recovery. 2-4 weeks after discharge adjustment and waiting for recovery dissatisfied with reduced physical training physical and psychological deterioration. 3-4 months after discharge: acceptance disappointment about physical training concern about new stroke lower expectations to self and others strong desire for independence. 	 Small sample size Exclusion of people with aphasia and dementia Setting is limited to two rehabilitation centres in the same region; hence, limits generalisability 	Understanding of stroke survivors' experience during rehabilitation process may contribute to improved quality of care via more realistic goal-setting, discussion of prognosis prior to discharge and home practice before returning home.
25	Young et al., 2014 USA	To understand the needs of spousal caregivers of stroke survivors during the transition from rehabilitation to home	Purposive sampling 14 spousal caregivers of first- time stroke survivors 2 inpatient rehabilitation facilities in the USA	Qualitative Grounded theory	 Caregiver unprepared, unclear caregiving expectations. Beneficial to conduct a caregiver assessment at an early stage of rehabilitation. A health professional to assist, advocate tailored to needs of caregivers may smooth transition. Making transitional plan to address post-discharge care needs. Home-based family counselling service may help develop coping skills in transition. 	 Options given to spouses to be interviewed together or separately. 	The result demonstrates research and practice gaps in improving post- rehabilitation outcomes. A comprehensive assessment of caregivers may provide a solid base for a discharge plan that reflects assessment findings. Further research should focus on the development of a caregiver assessment protocol.

Note. HCP = healthcare professional/provider; USA = United States of America; WP = weekend pass; UK = United Kingdom; ESD = early supported discharge.; RCT = randomised controlled trial; OT = occupational therapist; OHP = optimal health program; ADL = activity of daily living.

Appendix 4: Quality Appraisal of Reviewed Studies

Study	Question*										
	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Q9	Q10	
Bakas et al., 2002	Y	Y	Y	Y	Y	?	Y	Y	Y	Y	
Cameron et al., 2014	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y	
Chouliara et al., 2014	Y	Y	Y	Y	Υ	Y	Y	Y	Y	Y	
Cobley et al., 2013	Y	Y	Y	Y	Υ	Y	Y	Y	Y	Y	
Ellis-Hill et al., 2009	Y	Y	Y	Y	Υ	Y	Y	Y	Y	Y	
Greenwood et al., 2009	Y	Y	Y	Y	Υ	Y	Y	Y	Y	Y	
Gustafsson & Bootle, 2013	Y	Y	Υ	Υ	Υ	Y	Y	Υ	Y	Y	
Lou et al., 2017	Y	Y	Y	Y	Υ	Y	Y	Y	Y	Y	
Nordin et al., 2015	Y	Y	Υ	Υ	Υ	Y	Y	Υ	Y	Y	
O'Brian et al., 2014	Y	Y	Υ	Υ	Υ	?	Υ	Υ	Y	Y	
Pringle et al., 2013	Y	Y	Υ	Υ	Υ	Y	Υ	Υ	Y	Y	
Rittman et al., 2007	Y	Y	Υ	Υ	Υ	?	Υ	Υ	Y	Y	
Simeone et al., 2015	Y	Y	Υ	Υ	Υ	?	Y	Υ	Y	Y	
Wood et al., 2010	Y	Y	Υ	Υ	Υ	Y	Υ	Υ	Y	Y	
Wottrich et al., 2012	Y	Y	Y	Y	Υ	Y	Y	Υ	Y	Y	
Young et al., 2014	Y	Y	Y	Y	Υ	Y	Y	Y	Y	Y	

Appendix 4a: Evaluation of Qualitative Studies Included for Review

Note. Y = yes; N = no; ? = cannot tell. Adapted from: Critical Appraisal Skills Programme. (2018a). *CASP qualitative checklist*. <u>https://casp-uk.net/wp-content/uploads/2018/01/CASP-Qualitative-Checklist.pdf</u>

*Q1: Was there a clear statement of the aims of the research?

*Q2: Is a qualitative methodology appropriate?

*Q3: Was the research design appropriate to address the aims of the research?

*Q4: Was the recruitment strategy appropriate to the aims of the research?

*Q5: Was the data collected in a way that addressed the research issue?

*Q6: Has the relationship between the researcher and participants been adequately considered?

*Q7: Have ethical issues been taken into consideration?

*Q8: Was the data analysis sufficiently rigorous?

*Q9: Is there a clear statement of findings?

*Q10: How valuable is the research?

Appendix 4b: Evaluation of Randomised Controlled Trials Included for Review

Study	Question*												
	Q1 Q2 Q3 Q4 Q5 Q6 Q7 Q8 Q9 Q									Q10	Q11		
Shyu et al., 2008	Y	Y	Y	?	Y	Y	Y	Y	Y	?	Y		
Wong & Yeung, 2015	Y	Y	Y	Y	Y	Y	Y	?	?	Y	Y		

Note. Y = yes; N = no; ? = cannot tell. Adapted from: Critical Appraisal Skills Programme. (2018b). *CASP randomised controlled trial checklist*. <u>https://casp-uk.net/wp-content/uploads/2018/03/CASP-Randomised-Controlled-Trial-Checklist-Download.pdf</u>

*Q1 Did the trial address a clearly focused issue?

*Q2 Was the assignment of patients to treatments randomised?

*Q3 Were all of the patients who entered the trial properly accounted for at its conclusion?

*Q4 Were patients, health workers and study personnel 'blind' to treatment?

*Q5 Were the groups similar at the start of the trial?

*Q6 Aside from the experimental intervention, were the groups treated equally?

*Q7 How large was the treatment effect? Is the primary outcome clearly specified?

*Q8 How precise was the estimate of the treatment effect?

*Q9 Can the results be applied to the local population or in your context?

*Q10 Were all clinically important outcomes considered?

*Q11 Are the benefits worth the harms and costs?

Appendix 4c: Evaluation of Systematic Reviews Included for Review

Study	Question*											
	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Q9	Q10		
Langhorne et al., 2017	Y	Y	Y	Y	Y	Y	Y	Y	Y	Y		
Pringle et al., 2008	Y	Υ	Υ	Υ	?	?	?	Y	Ν	?		

Note. Y = yes; N = no; ? = cannot tell. Adapted from: Critical Appraisal Skills Programme. (2018c). *CASP systematic review checklist*. <u>https://casp-uk.net/wp-content/uploads/2018/03/CASP-Systematic-Review-Checklist-Download.pdf</u>

*Q1 Did the review address a clearly focused question?

*Q2 Did the authors look for the right type of papers?

*Q3 Do you think all the important, relevant studies were included?

*Q4 Did the review's authors do enough to assess the quality of the included studies?

*Q5 If the results of the review have been combined, was it reasonable to do so?

*Q6 What are the overall results of the review?

*Q7 How precise are the results?

*Q8 Can the results be applied to the local population?

*Q9 Were all important outcomes considered?

*Q10 Are the benefits worth the harms and costs?

Study	Question*											
	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Q9			
Almborg et al., 2008	Y	Y	Y	Y	Y	Y	Y	Y	?			
Almborg et al., 2009	Y	Υ	Y	Y	Y	Y	Y	Υ	?			
Grant et al., 2006	Y	Υ	Y	Υ	Ν	Ν	?	Ν	?			
Perry & Middleton, 2011	Y	Y	Y	Y	Y	Y	Y	Y	Y			

Appendix 4d: Evaluation of Quantitative Studies Included for Review

Note. Y = yes; N = no; P = partly; ? = unclear. Adapted from a set of questions developed by Büttner and Muller (2015) to assess epidemiological studies.

*Q1 Was the research question clearly stated?

*Q2 Was the design clearly reported?

*Q3 Was the design appropriate for the purpose of the study?

*Q4 Was the sample described in detail?

*Q5 Did the sample represent the target population?

*Q6 Was the outcome assessed with a validated tool?

*Q7 Was confounding bias controlled in the study?

*Q8 Were the main results presented in a statistically appropriate way?

*Q9 Does the study add to the question of causality between study factor and outcome?

Appendix 4e: Evaluation of the Mixed Method Study Included for Review

Study	Question*						Quali	tative		Quantitative			
	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Q9	Q10	Q11	Q12	Q13
Hall et al., 2012	Y	Y	Y	Y	?	Y	Y	Y	?	Y	Y	Y	Y

Note. Y = yes; N = no; ? = cannot tell. Adapted from: Pluye, P., Robert, E., Cargo, M., Bartlett, G., O'Cathain, A., Griffiths, F., Boardman, F., Gagnon, M.P., & Rousseau, M.C. (2011). Proposal: A mixed methods appraisal tool for systematic mixed studies reviews. Retrieved on September 3,2023 from http://mixedmethodsappraisaltoolpublic.pbworks.com Archived by WebCite® at http://www.webcitation.org/5tTRTc9yJ

*Q1 Is there a clear mixed methods question (or objective)?

*Q2 Do the collected data allow address the research question (objective)?

*Q3 Is the mixed methods research design relevant to address the qualitative and quantitative research questions (or objectives)?

*Q4 Is the integration of qualitative and quantitative data (or results) relevant to address the research question (objective)?

*Q5 Is appropriate consideration given to the limitations associated with this integration, e.g., the divergence of qualitative and quantitative data (or results) in a triangulation design?

*Q6 Are the sources of qualitative data (archives, documents, informants, observations) relevant to address the research question (objective)?

*Q7 Is the process for analysing qualitative data relevant to address the research question (objective)?

*Q8 Is appropriate consideration given to how findings relate to the context, e.g., the setting in which the data were collected?

*Q9 Is appropriate consideration given to how findings relate to researchers' influence, e.g., through their interactions with participants?

*Q10 Is the sampling strategy relevant to address the quantitative research question (quantitative aspect of the mixed methods question)?

*Q11 Is the sample representative of the population understudy?

*Q12 Are measurements appropriate (clear origin, or validity known, or standard instrument)?
*Q13 Is there an acceptable response rate (60% or above)?

Appendix 5: Exemplar—Data Extraction by Rule of Thumb for Anchoring Text to Context in Restatements of Findings (Sandelowski et al., 2013, p. 1432)

Shyu, Chen, Chen, Wang & Shao (2008). A family caregiver-oriented discharge planning program for older stroke patients and their family caregivers (nursing intervention)

Sample

- 65–95 years stroke patients from Northern Taiwan who were discharged to home and screened as high-risk patients requiring discharge planning services.
- 18–85-year-old family caregivers who were primarily responsible for providing direct care to or supervising the care of the patient.

Source of information

Caregivers

Time

Perceived balance between competing needs measured one month after discharge indicated no significant difference between the two groups. The competing needs were moderately well-balanced, with the experimental group slightly better than the control group.

Findings

Program: 1) needs assessment; 2) individualised health information 4–5 times at bedside based on needs assessment; 3) assess competing needs. Consult with caregivers before discharge on handling competing needs (together with bedside health information); 4) consult by telephone within one week; visit at home within one week and at one month after discharge to support including demonstration of skills and problem-solve.

Caregiver preparation evaluation before discharge

- Nurse evaluated: both groups before program moderately prepared, after program significantly improved. Experimental group slightly better than the control.
- Self-evaluated before discharge: experimental group significantly improved than the control.

Caregiver satisfaction with discharge needs being met

- Experimental group: moderately satisfied before discharge, significantly improved one month after discharge.
- Control group: mildly satisfied before discharge, significantly improved one month after discharge.
- Experimental group had a significantly higher satisfaction score than the control group on month after discharge.

Perceived balance between competing needs

Assessment only at one month after discharge. No significant difference between groups.

Appendix 6: Themes With Direct Participant Quotations

	Theme 1. Health and social care systems factors				
	Subtheme 1. Enhanced transition care via well-coordinated services				
Quotations	'[The specialist service played a critical role in] bridging a big part of the gap in community-based rehabilitation [and it is] what sets the service aside from other community services.'	Chouliara et al., 2014, p. 374			
	'In this process I've felt completely safe. Like, they didn't just send me home and hope for the best; I knew someone would check up on me, that I was not alone.'	Lou et al., 2017, p. 306			
	'When people get home, they realise what they can do and what they can't do, their limitations, and they realise, okay, this is what I need to work on.'	Cameron et al., 2014, p. 861			
	Subtheme 2. Compromised transition due to silos in service provision				
Quotations	'I felt like I was a little old Eskimo woman that they put on this ice block, chopped it off and sailed it out into the middle of the ocean.'	Young et al., 2014, p. 1895			
	'Yes. I mean, with hindsight, it would obviously have been better for me to stay in C [place] because I would have had access to getting physio every day, but I realise that they need the beds. I 'm not saying they were trying to get rid of me, they weren't.'	Ellis-Hill et al., 2009, p. 67			
	'Cos I said, you know, "it's been over a week now" and I feel it's really bad that he's had nothing, professionally, only what I've been doing at home, nothing professionally, and I think it should be ongoing because he's very positive about the future and, um, you know, this is not helping.'	Ellis-Hill et al., 2009, p. 68			
	'all of a sudden it's like, "Oh, we've referred you to the hospital again to get the physio", which has took, like three months. So I've had intense physio for six weeks and then, for three months, I've had nothing.'	Cobley et al., 2013, p. 754			
	'There is dependence and reliance, which is difficult to navigate, mainly because people need long-term support and we are the end of the line as community rehabilitation. Especially discharge conversation. "Yeah but where are we going? We still need support with this." There is still much more improvement to make, which is right but there no one else.'	O'Brien et al., 2014, p. 425			
	Theme 2. Health professionals and stroke survivors/caregivers engagement factors	·			
	Subtheme 1. Tailored transition built on partnership with consumers				
Quotations	'it's never just one of us that is addressing the problem, if the whole team works together brainstorm together with the family member and the patient—we can come up with hopefully the best ideas, the best solutions possible.'	Cameron et al., 2014, p. 860			

	'It is less about a body in a bed that needs a bit of fixing; to me, it feels more of a holistic service; just being in people's houses, seeing what problems they actually have and adapting the service around that.'	Chouliara et al., 2014, p. 375
	'[they] learned that both the caregiver and survivor were important in the caregiving process and ways to take care of both that there were people who cared about them and made them realise that they, as well as the stroke survivor, were important, which made them feel good.'	Grant et al., 2006, p. 72
	'I think a lot of what we do in terms of carers is subtle supportive counselling just trying to have an understanding to know what it might be like to walk in their shoes.'	O'Brien et al., 2014, p. 427
	Subtheme 2. The lack of individualised support for clients	·
Quotations	'I would like to give more time to families but feel constrained by my workload for the shift and sometimes you see husbands or wives, or carers and you know in your heart they are finding things difficult and this is very hard and I feel it is part of my work.'	O'Brien et al., 2014, p. 425
	'If you've got any questions you could ask them [the Early Supported Discharge team after discharge], whereas when you're in a hospital, I feel that I can't take up the people's time because they haven't really got time.'	Cobley et al., 2013, p. 753
	'Er the uncertainty. Even now I'm not sure whether one should expect another stroke the biggest problem is not knowing what the future holds No they haven't [talked about that]. I haven't asked. Perhaps that's the reason they haven't mentioned it. Perhaps they feel that I'm quite lucid and comfortable with it, but I do feel a little bit uncertain.'	Ellis-Hill et al., 2009, p. 68
	'I don't think they told me anything. I was just left out in the cold I didn't have a clue what was going on.'	Cobley et al.,
	'His depression I just don't know what to do. I cannot cope because I don't know what to do to stop it. When he is continuously crying, it can be really wearing because you are helpless.'	2013, p. 754
	'They just don't know—that is what is quite scary, it is the unknown. You just don't know what it is going to be like in a year's time because everybody is just so different.'	Greenwood et al., 2009, p. 1125
	'I wasn't physically shown the best way to support him it was all trial and error.'	Cobley et al., 2013, p. 754
	'because [they are just] saying I can come home and do 90 percent of what I used to be able to, but they haven't actually assessed that I can do it.'	Gustafsson & Bootle, 2013, p. 1383

	Theme 3. Stroke survivors/caregivers self-management capability factors		
	Subtheme 1. Resilience as a key indicator of a smooth transition		
Quotations	'There is one lady that she has a stroke before Mum and she hasn't gained any physical movement at all … I think with Mum—well, I think she has improved a lot anyway.'	Greenwood et al., 2009, p. 1128	
	'I was in hospital with people that were really bad, that couldn't walk properly, couldn't dress I could have been so much worse. And each day when I see a slight improvement, and I do every day, I'm so thankful.'	Pringle et al., 2013, p. 1995	
	'One thing that—whatever happens, I always say that it is best. Better. God never wants to harm you, it is your destiny. And there is something always positive—good.'	Greenwood et al., 2009, p. 1129	
	'I've always been Catholic. I've always believed in God. So that's, I'm sure there's a reason for all this.'	Rittman et al., 2007, p. 25	
	'Yeah, that is one thing, giving them hope, that's one thing we are really good at as a team we never, never extinguish hope. It may look bleak, but we never extinguish that hope.'	O'Brien et al., 2014, p. 427	
	'My hope is to recover to my previous level, to be able to walk as well as before and to use the arm, to get as close as possible to where I was before, preferably all the way there. I'm not sure if it's going to be possible, it's just training.'	Wottrich et al., 2012, p. 1220	
	'She's (her daughter) got bad behavioural problems … (laughs) It sounds very throw away but it's just … Yes … it's a stroke … it's nothing really.'	Greenwood et al., 2009, p. 1126	
	'I didn't have to figure out anything because we had taken care of her mother for 17 years with a stroke. And so, I more or less knew what we had to do.'	Young et al., 2014, p. 1897	
	'It was a bit difficult at first with everything new, but it gets better and better, one enjoys being at home more now than one did when one left hospital.'	Wottrich et al., 2012, p. 1222	
	'You try to provide education and strategies for communication at home and provide examples to support someone to be independent in the community and at home.'	O'Brien et al., 2014, p. 426	
	'We have learnt from last time. Umm P doesn't have a lot of trousers with buttons now. It is more jogging bottoms. He doesn't have a lot of shirts with buttons because he had difficulty fastening buttons. To give him his independence, we have altered all that so that he can still dress himself and he doesn't rely on me.'	Greenwood et al., 2009, p. 1128	
	'When it comes to social events and going out I have to thank my friend because she drives and the same thing with my son it's encouraging that these people are around me.'	Wood et al., 2010, p. 1051	

Subtheme 2. Sense of loss and burdensome on others		
Quotations	'I have lost this person, who is this new person who I have to take care of? I have lost my husband and I don't know who it is.'	O'Brien et al., 2014, p. 427
	'I feel I've got a different man home than he was before he had the stroke, because before he was a very well educated man, and although we tended to make decisions together, I always took his lead and now he's not—he's not even interested in anything.'	Pringle et al., 2013, p. 1995
	'I've always been independent//so it is a completely new situation and I can't know how it is going to be//it might be great, or it can go badly.'	Nordin et al., 2015, p. 5
	'Even people that have minimal physical impairments can be really anxious because their whole life has changed.'	Chouliara et al., 2014, p. 375
	'I think the support around the emotional journey, [for] both client and families, is a big gap because if we had access to psychology or counselling, we could start educating about stroke offering what they need as things change such as adjusting to their emotions.'	O'Brien et al., 2014, p. 245
	'it took me a long time to buy into the rehab therapist being pleased with small little gains and I tried to look at it from their point [of view]. They know nothing about him. They do not see him until the first day he is here. So that is the history they have of him, so every little thing that he does that gets better they see as great progress. It's very difficult [when they didn't know the way he was before the stroke.'	Young et al., 2014, p. 1898
	'the 24/7 care gave her a 'whole different perspective' about how difficult caregiving was.'	Young et al., 2014, p. 1897
	'The financial disadvantage of the client is the biggest hurdle if you have a family that can buy a lifting machine and a special bed and employ a physio, so they can get a better outcome, they have the potential to have a better outcome than those who are lower socioeconomic.'	O'Brien et al., 2014, p. 426
	'Here we are getting her better and then, boom, we'll run out of money, and our social system encourages you to confine them into a nursing home and that will kill her.'	Bakas et al., 2002, p. 248
	' pressured to take the patient home despite feeling unprepared.'	Young et al., 2014, p. 1895
	'Carers have their issues before the person had their stroke and it pushes them to the limit of what their capacity is. They might not have been managing with what they had before, and this comes and just tops it off.'	O'Brien et al., 2014, p. 427
	'it's really, really dependent on the family and I don't want to say their level of involvement, but their readiness to be involved in this stage a lot of families are really overwhelmed it is not that they don't want to help, it's just that they can't kind of get their heads around helping.'	Cameron et al., 2014, p. 860

'I've hurt my back I will do this until the day that we finally both crash to the floor or I break a hip, or he breaks a rib. Then we will both end up in a nursing home together there are things worse than death, and this is it. This is the living dead. You're looking at us right here, he and I.'	Young et al., 2014, p. 1896
'I don't wanna be a burden to anybody else Now I can't seem to get out and do what I wanna do it's depressing All of a sudden find out that you can't do much of anything, you gotta depend on people.'	Rittman et al., 2007, p. 27
'Do you see the condition in which I put my son? I forced my son to bounce among work, home, his family, and me. My son seems like a yo-yo attached to the rope running back and forth, poor man!'	Simeone et al., 2015, p. 166

Appendix 7: PRISMA (Preferred Reporting of Items for Systematic Reviews and Meta-Analysis) Statement

Section/topic	Section/topic # Checklist item		Reported in manuscript page #
TITLE	·		
Title	1	Identify the report as a systematic review, meta-analysis, or both.	1
		ABSTRACT	
Structured summary	Structured summary 2 Provide a structured summary including, as applicable: background; objectives; data sources; study eligibility criteria, participants, and interventions; study appraisal and synthesis methods; results; limitations; conclusions and implications of key findings; systematic review registration number.		1–3
		INTRODUCTION	
Rationale	3	Describe the rationale for the review in the context of what is already known.	3–6
Objectives	4	Provide an explicit statement of questions being addressed with reference to participants, interventions, comparisons, outcomes, and study design (PICOS).	7
		METHODS	
Protocol and registration5Indicate if a review protocol exists, if and where it can be accessed (e.g., Web address), and, if available, provide registration information, including registration number.		7	
Eligibility criteria 6 Specify study characteristics (e.g., PICOS, length of follow-up) and report characteristics (e.g., years considered, language, publication status) used as criteria for eligibility, giving rationale. 7		7–8, Supp file S2	
Information sources7Describe all information sources (e.g., databases with dates of coverage, contact with study authors to identify additional studies) in the search and date last searched.7–8		7–8	
Search 8 Present the full electronic search strategy for at least one database, including any limits used, such that it could be repeated.		Supp file S1	
Study selection 9 State the process for selecting studies (i.e., screening, eligibility, included in a systematic review, and, if applicable, included in the meta-analysis).		8, Figure 1	
Data collection process	Data collection10Describe the method of data extraction from reports (e.g., piloted forms, independently, in duplicate) and any processes for obtaining and confirming data from investigators.		9
Data items	Data items 11 List and define all variables for which data were sought (e.g., PICOS, funding sources) and any assumptions and simplifications made.		9, Supp file S5
Risk of bias in individual studies	12	Describe methods used for assessing risk of bias of individual studies (including specification of whether this was done at the study or outcome level), and how this information is to be used in any data synthesis.	8, Supp file S4

Section/topic	Section/topic # Checklist item		Reported in manuscript page #
Summary measures	Summary measures 13 State the principal summary measures (e.g., risk ratio, difference in means).		Table 1 & Supp file S3
Synthesis of results	14	Describe the methods of handling data and combining results of studies, if done, including measures of consistency (e.g., β) for each meta-analysis.	9, Supp file S6
Risk of bias across studies	15	Specify any assessment of risk of bias that may affect the cumulative evidence (e.g., publication bias, selective reporting within studies).	8, Supp file S4
Additional analyses	16	Describe methods of additional analyses (e.g., sensitivity or subgroup analyses, meta-regression), if done, indicating which were pre-specified.	NA
	•	RESULTS	
Study selection	17	Give numbers of studies screened, assessed for eligibility, and included in the review, with reasons for exclusions at each stage, ideally with a flow diagram.	Figure 1
Study characteristics	18	For each study, present characteristics for which data were extracted (e.g., study size, PICOS, follow-up period) and provide the citations.	Table 1, Supp file S3
Risk of bias within19Present data on risk of bias of each study and, if available, any outcome level assessment (see item 12).studies		8, Supp file S4	
Results of individual studies20For all outcomes considered (benefits or harms), present, for each study: (a) simple summary data for each intervention group (b) effect estimates and confidence intervals, ideally with a forest plot.		NA	
Synthesis of results 21 Present results of each meta-analysis done, including confidence intervals and measures of consistency.		9–18	
Risk of bias across studies	Risk of bias across 22 Present results of any assessment of risk of bias across studies (see Item 15). studies 22 Present results of any assessment of risk of bias across studies (see Item 15).		8, 21, Supp file S4
Additional analysis	23	Give results of additional analyses, if done (e.g., sensitivity or subgroup analyses, meta-regression [see Item 16]).	NA
		DISCUSSION	
Summary of evidence	Summary of evidence 24 Summarise the main findings, including the strength of evidence for each main outcome; consider their relevance to key groups (e.g., healthcare providers, users, and policymakers).		18–22, Figure 2
Limitations	25	Discuss limitations at study and outcome level (e.g., risk of bias) and at review level (e.g., incomplete retrieval of identified research, reporting bias).	21
Conclusions	26	Provide a general interpretation of the results in the context of other evidence, and implications for future research.	22
		FUNDING	

Section/topic	#	Checklist item	Reported in manuscript page #
Funding	27	Describe sources of funding for the systematic review and other support (e.g., supply of data); role of funders for the systematic review.	Title Page

No.	Checklist items	Guide and description	Page no. in manuscript
1	Aim	State the research question the synthesis addresses.	1, 6
2	Synthesis methodology	Identify the synthesis methodology or theoretical framework that underpins the synthesis and describe the rationale for the choice of methodology.	6, 7
3	Approach to searching	Indicate whether the search was pre-planned (comprehensive search strategies to seek all available studies) or iterative (to seek all available concepts until they theoretical saturation is achieved).	7
4	Inclusion criteria	Specify the inclusion/exclusion criteria (e.g. in terms of population, language, year limits, type of publication, study type).	7,8 Table 1
5	Data sources	Describe the information sources used (e.g. electronic databases (MEDLINE, EMBASE, CINAHL, psycINFO, Econlit), grey literature databases (digital thesis, policy reports), relevant organisational websites, experts, information specialists, generic web searches (Google Scholar) hand searching, reference lists) and when the searches conducted; provide the rationale for using the data sources.	1, 7
6	Electronic Search strategy	Describe the literature search (e.g. provide electronic search strategies with population terms, clinical or health topic terms, experiential or social phenomena related terms, filters for qualitative research, and search limits).	7
7	Study screening methods	Describe the process of study screening and sifting (e.g. title, abstract and full text review, number of independent reviewers who screened studies).	7, 8
8	Study characteristics	Present the characteristics of the included studies (e.g. year of publication, country, population, number of participants, data collection, methodology, analysis, research questions).	10, Table 2,
9	Study selection results	Identify the number of studies screened and provide reasons for study exclusion (e.g. for comprehensive searching, provide numbers of studies screened and reasons for exclusion indicated in a figure/flowchart; for iterative searching describe reasons for study exclusion and inclusion based on modifications to the research question and/or contribution to theory development).	8, Figure 1
10	Rationale for appraisal	Describe the rationale and approach used to appraise the included studies or selected findings (e.g. assessment of conduct (validity and robustness), assessment of reporting (transparency), assessment of content and utility of the findings).	8, Table 3
11	Appraisal items	State the tools, frameworks and criteria used to appraise the studies or selected findings (e.g. Existing tools: CASP, QARI, COREQ; reviewer developed tools; describe the domains assessed: research team, study design, data analysis and interpretations, reporting).	8, Table 3

Appendix 8: ENTREQ (Enhancing Transparency in Reporting the Synthesis of Qualitative Research) Statement

No.	Checklist items	Guide and description	Page no. in manuscript
12	Appraisal process	Indicate whether the appraisal was conducted independently by more than one reviewer and if consensus was required.	8
13	Appraisal results	Present results of the quality assessment and indicate which articles, if any, were weighted/excluded based on the assessment and give the rationale.	8, Table 3
14	Data extraction	Indicate which sections of the primary studies were analysed and how were the data extracted from the primary studies? (e.g. all text under the headings 'results /conclusions' were extracted electronically and entered into a computer software).	8, 9
15	Software	State the computer software used, if any.	EndNote
16	Number of reviewers	Identify who was involved in coding and analysis.	7–9
17	Coding	Describe the process for coding data (e.g. line-by-line coding to search for concepts).	9
18	Study comparison	Describe how were comparisons made within and across studies (e.g. subsequent studies were coded into pre-existing concepts, and new concepts were created when deemed necessary).	9, Appendix 1
19	Derivation of themes	Explain whether the process of deriving the themes or constructs was inductive or deductive.	9
20	Quotations	Provide quotations from the primary studies to illustrate themes/constructs, and identify whether the quotations were participant quotations of the author's interpretation.	11–22
21	Synthesis output	Present rich, compelling and useful results that go beyond a summary of the primary studies (e.g. new interpretation, models of evidence, conceptual models, analytical framework, development of a new theory or construct).	11–22, Table 4

Appendix 9: Published Systematic Review Protocol

PROSPERO

International prospective register of systematic reviews



Stroke survivors' and informal caregivers' experience in hospital to home transition: a metasynthesis protocol Langduo Chen, Lily (Dongxia) Xiao, Diane Chamberlain, Peter Newman

Citation

Langduo Chen, Lily (Dongxia) Xiao, Diane Chamberlain, Peter Newman. Stroke survivors' and informal caregivers' experience in hospital to home transition: a meta-synthesis protocol. PROSPERO 2018 CRD42018091345 Available from: http://www.crd.york.ac.uk/PROSPERO/display_record.php?ID=CRD42018091345

Review question

 What are barriers and facilitators during hospital to home transition for stroke survivors and their caregivers?

 What are the views and expectations/preferences of stroke survivors and informal caregivers on healthcare system/professionals in hospital to home transition?

Searches

The following electronic databases will be searched for eligible literature: CINAHL Plus, MEDLINE, Scopus, Web of Science and ProQuest and ProQuest Dissertations and Theses.

Keywords and search terms using PICO framework:

The Population, Intervention, Comparison and Outcomes (PICO) and Population, Exposure and Outcome (PEO) framework will be used to develop search terms and keywords (Khan et al. 2003).

Population: Stroke Cerebral vascular accident Patients Survivors Caregivers Family carers

Exposure: Patient discharge "continuity of patient care"/ or transitional care/ Aftercare Stroke rehabilitation Hospital to home Care coordination

Outcome: Experience experiences perception* perceptive perspective*

Each term will be entered into database with truncation where appropriate. All individual searches will be combined using the "OR" Boolean into a single group. Each group will then be combined using the "AND" with Qualitative filters indicating qualitative data collection. The final citation will be saved in Endnote and screened for duplications. Studies published in English or with an abstract published in English will be included.

Reference: Khan, K.S., Kunz, R., Kleijnen, J. and Antes, G. (2003) Systematic reviews to support evidencebased medicine: how to review and apply findings of healthcare research. London: Royal Society of Medicine Press.

NHS National Institute for Health Research

Types of study to be included

Inclusion:

Qualitative original studies published in English (including qualitative data from mixed methods studies) focusing on the experiences of the hospital to home transition period, self-care or self-management approaches and expectations of improved care services that enable hospital to home transition care.

Exclusion:

Studies using quantitative methods or mixed methods where the qualitative data cannot be separated. Qualitative studies involving statistical data as reporting results. Conference proceedings, discussions, reviews or commentaries.

Condition or domain being studied

Hospital to home transition (the transition) is the most stressful period for stroke survivors and their informal caregivers. The widely reported challenges and difficulties for stroke survivors include self-care deficits, complications associated with stroke related functional declines and risk factors, isolation, depression, cognitive and personality changes (Rittman et al. 2007, Langhorne, Bernhardt and Kwakkel 2011). The widely reported challenges for caregivers are lack of information, resources, knowledge, skills and confidence to manage care at home and to coordinate required treatment, rehabilitation and various care services. They experience high levels of caregiving burden and burnout (Greenwood and McKenzie 2010, Grant et al. 2014). It is evident that personal coping skills, service providers' factors and the health care system factors contribute to these undesirable situations that stroke survivors and their caregivers encounter during hospital to home transition (Wissle et al. 2013, Chen et al. 2016).

This review examines the Preparation and Implementation stage of post stroke recovery trajectory defined by the 'Time It Right' framework (Cameron and Gignac 2008). Data up to 6 months after hospital discharge will be analysed as majority of studies on hospital to home transition reported a 3 to 6 month follow-up.

References:

Cameron, JI and Gignac, MA 2008, "Timing It Right": A conceptual framework for addressing the support needs of family caregivers to stroke survivors from the hospital to the home', Patient education and counselling, vol. 70, no. 3, pp. 305-14.

Chen, L, Xiao, LD and De Bellis, A 2016, 'First?time stroke survivors and caregivers' perceptions of being engaged in rehabilitation', Journal of Advanced Nursing, vol. 72, no. 1, pp. 73-84.

Grant, J, Hunt, C and Steadman, L 2014, 'Common caregivers issues and nursing interventions after a stroke', Stroke, vol. 45, pp. e151-e3.

Participants/population

Inclusion:

· Adult stroke survivors and informal caregivers aged 18 years and above.

 Informal caregivers are defined as unpaid caregivers including spouse or partner, family members, friends, or significant others who provide physical and emotional support to stroke survivors.

Exclusion:

· Studies that focus on stroke survivors less than 18 years old.

Page: 2/6



 Studies on multiple patient populations such as traumatic brain injuries, brain tumour, dementia, hip fracture where data relevant to stroke cannot be separated.

 Studies of multiple populations e.g. health care professionals, service provider, where perspectives of stroke survivors and caregivers cannot be separated.

Intervention(s), exposure(s)

Inclusion:

Studies that explore the impact and perceptions of stroke, the experience of stroke survivors and family caregivers, perspectives or needs in hospital to home transition up to 6 months after hospital discharge.

Exclusion:

Studies examining post stroke long term experiences and needs (over 6 months after hospital discharge). Studies on discharges to facilities other than home. Studies conducted across multiple period of stroke trajectory where the transition period cannot be separated.

Comparator(s)/control

Not applicable

Context

Studies in hospital on patient wards, outpatient clinics, in community.

Main outcome(s)

The primary outcomes are qualitatively derived experiences and needs of stroke survivors and informal caregivers in hospital to home transition.

Timing and effect measures

This review examines the Preparation and Implementation stage of post stroke recovery trajectory defined by the 'Time It Right' framework (Cameron and Gignac 2008). Data up to 6 months after hospital discharge will be analysed as majority of studies on hospital to home transition reported a 3 to 6 month follow-up.

Reference:

Cameron, JI and Gignac, MA 2008, "Timing It Right": A conceptual framework for addressing the support needs of family caregivers to stroke survivors from the hospital to the home', Patient education and counselling, vol. 70, no. 3, pp. 305-14.

Additional outcome(s)

The secondary outcomes are the views and expectations/preferences of stroke survivors and informal caregivers on healthcare system/professionals in hospital to home transition.

Timing and effect measures

Same as the Primary outcome(s).

Data extraction (selection and coding)

The screening process for the review will be carried out by three reviewers in 2 stages. The citations will be distributed in the following manner:

- · Reviewer 1 (primary reviewer) will review each citation.
- Each articles is independently reviewed by two reviewers.

Page: 3/6

NHS National Institute for Health Research

Stage 1: Screening based on title and abstract

Using EndNote, the initial citations will be shared amongst reviewers. Reviewer 1 will screen each article based on title and abstract against the inclusion and exclusion criteria. The rest of reviewers will screen title and abstract of half of the citations based on the inclusion and exclusion criteria. The screening results will be discussed and citations not selected by two reviewers will be excluded for full text review and maintained for future reference. The included citations will be reviewed in stage 2.

Stage 2 Full text review

Reviewer 1 will source full text and save PDF versions into EndNote database from the title and abstract screening process. Each reviewer will independently review full text of all included articles. Any discrepancies will be discussed and resolved until consensus is achieved.

Data will be extracted from the studies selected for inclusion following the data extraction methods developed by Bettany-Saltikov (2012).

Reference:

Bettany-Saltikov, J 2012, How to do a systematic literature review in nursing: a step-by-step guide, McGraw-Hill Education (UK).

Risk of bias (quality) assessment

The rigour of the qualitative studies will be appraised using the CERQual approach (Confidence in the Evidence from Reviews of Qualitative Research) (Lewin et al. 2015). Studies will be reviewed by three independent review authors. The CERQual approach consists of 4 components:

 Methodological limitations refers to any problems in the design or conduct of the included primary studies. The Critical Appraisal Skills Programme checklist (CASP) (2017) will be used as a tool to evaluate the methodological strengths and weaknesses of the primary studies.

Relevance is considered as the extent to which the evidence of the primary studies is aligned with the context of the review question. This includes study population, phenomenon of interest, setting and time frame.

Coherence assesses the confidence in the study findings. Whether or not these findings are well grounded in the data and provide explanation to support the pattern found in these data.

4. Adequacy can be defined as richness and quantity of data supporting a finding.

The quality assessment in this review serves to offer further insights into the contribution of the included studies at the later stage of the review (Carroll and Booth 2015). All studies that meet the inclusion criteria will be included disregarding the study quality as methodologically weak studies may provide novel understandings that may not be produced in methodologically strong studies (Carroll et al. 2012).

Reference:

Carroll, C, Booth, A and Lloyd-Jones, M 2012, 'Should we exclude inadequately reported studies from qualitative systematic reviews? An evaluation of sensitivity analyses in two case study reviews', Qualitative Health Research, vol. 22, no. 10, pp. 1425-34.

Carroll, C and Booth, A 2015, 'Quality assessment of qualitative evidence for systematic review and synthesis: Is it meaningful, and if so, how should it be performed?', Research synthesis methods, vol. 6, no.

Page: 4/6

NHS National Institute for Health Research

2, pp. 149-54.

Lewin, S., Glenton, C., Munthe-Kaas, H., Carlsen, B., Colvin, C.J., Gülmezoglu, M., Noyes, J., Booth, A., Garside, R. and Rashidian, A. (2015) Using Qualitative Evidence in Decision Making for Health and Social Interventions: An Approach to Assess Confidence in Findings from Qualitative Evidence Syntheses (GRADE-CERQual). PLOS Medicine, 12(10), e1001895.

Strategy for data synthesis

The qualitative studies will be analysed using Thomas and Hare's approach to metasynthesis (Thomas and Hare, 2008). This involves 3 steps:

Step 1: Free line-by-line coding of findings of primary studies to enable "Translation" of concepts from one study to another.

Step 2: Developing descriptive themes by grouping similarities and differences between the codes and creating new codes.

Step 3: Generating analytical themes based on review questions.

Reference:

Thomas, J and Harden, A 2008, 'Methods for the thematic synthesis of qualitative research in systematic reviews', BMC medical research methodology, vol. 8, no. 1, p. 45.

Analysis of subgroups or subsets Not planned

Contact details for further information Langduo Chen chen0365@flinders.edu.au

Organisational affiliation of the review Flinders University of South Australia

https://www.flinders.edu.au/

Review team members and their organisational affiliations Ms Langduo Chen. Flinders University of South Australia Assistant/Associate Professor Lily (Dongxia) Xiao. Flinders University of South Australia Assistant/Associate Professor Diane Chamberlain. Flinders University of South Australia Mr Peter Newman. Flinders University of South Australia

Anticipated or actual start date 01 December 2017

Anticipated completion date 31 December 2018

Funding sources/sponsors None

Conflicts of interest

Language English

Page: 5/6

NHS National Institute for Health Research

Country Australia

Stage of review Review_Ongoing

Subject index terms status Subject indexing assigned by CRD

Subject index terms Caregivers; Hospitals; Humans; Stroke; Survivors

Date of registration in PROSPERO 21 March 2018

Date of publication of this version 21 March 2018

Details of any existing review of the same topic by the same authors Not applicable

Stage of review at time of this submission

Stage	Started	Completed
Preliminary searches	Yes	Yes
Piloting of the study selection process	Yes	Yes
Formal screening of search results against eligibility criteria	Yes	Yes
Data extraction	No	No
Risk of bias (quality) assessment	No	No
Data analysis	No	No

Versions 21 March 2018

PROSPERO

This information has been provided by the named contact for this review. CRD has accepted this information in good faith and registered the review in PROSPERO. CRD bears no responsibility or liability for the content of this registration record, any associated files or external websites.

Page: 6/6

Appendix 10: Search Strategies

Medline

Ovid MEDLINE(R) Epub Ahead of Print, In-Process & Other Non-Indexed Citations, Ovid

MEDLINE(R) Daily, Ovid MEDLINE and Versions(R)

#	Search
1	exp Stroke/
2	(stroke or cerebrovascular accident* or cerebrovascular apoplexy or brain vascular
	accident*).tw.
3	1 or 2
4	"continuity of patient care"/ or transitional care/
5	Stroke Rehabilitation/
6	aftercare/
7	patient discharge/
8	Patient Education as Topic/
9	(transition* or aftercare or aftercare or discharge plan* or (patient* adj2 discharg*) or
	(care adj2 coordinat*) or continuity of care or (discharg* adj3 folllow-up*) or
	(patient* adj2 educat*) or rehabilitation).tw.
10	or/4-9
11	Independent Living/
12	(home or homes or (independent* adj2 living) or communit*).tw.
13	11 or 12
14	interviews as topic/ or focus groups/ or narration/ or qualitative research/
15	((semi-structured or semistructured or unstructured or informal or "in-depth" or
	indepth or "face-to-face" or structured or guide? or group*) adj3 (discussion* or
	questionnaire*)).tw.
16	(Interview* or focus group*).tw.

Search					
((context* or semantic or content) adj2 analys*).tw.					
(narrat* or qualitative* or ethnograph* or phenomenolog* or hermeneutic*).tw.					
((personal* or patient* or inpatient* or participant* or lived) adj2 (experience or					
experiences or perception* or perceptive or perspective*)).tw.					
or/14-19					
3 and 10 and 13 and 20					

CINAHL Database

S1	(MH "Stroke+")
S2	TI (stroke or cerebrovascular accident* or cerebrovascular apoplexy or brain
	vascular accident*)
S3	S1 OR S2
S4	(MH "Continuity of Patient Care") OR (MH "Transitional Care")
S5	(MH "Rehabilitation")
S6	(MH "After Care")
S7	(MH "Patient Discharge")
S8	(MH "Patient Education")
S9	(MH "Discharge Planning") OR (MH "Patient Discharge Education")
S10	TI ((transition* or aftercare or aftercare or "discharge plan*" or (patient* N2
	discharg*) or (care N2 coordinat*) or "continuity of care" or (discharg* N3 folllow-
	up*) or (patient* N2 educat*) or rehabilitation)) OR AB ((transition* or aftercare
	or aftercare or "discharge plan*" or (patient* N2 discharg*) or (care N2
	coordinat*) or "continuity of care" or (discharg* N3 folllow-up*) or (patient* N2
	educat*) or rehabilitation))
S11	S4 OR S5 OR S6 OR S7 OR S8 OR S9 OR S10

S12 (MH "Community Living+")

S13 TI ((home or homes or (independent* N2 living) or communit*)) OR AB ((home or homes or (independent* N2 living) or communit*))

S14 S12 OR S13

- S15 (MH "Interviews+") OR (MH "Focus Groups") OR (MH "Narratives") OR (MH "Qualitative Studies+")
- S16 TI ((("semi-structured" or semistructured or unstructured or informal or "in-depth" or indepth or "face-to-face" or structured or guide? or group*) N3 (discussion* or questionnaire*))) OR AB ((("semi-structured" or semistructured or unstructured or informal or "in-depth" or indepth or "face-to-face" or structured or guide? or group*) N3 (discussion* or questionnaire*)))
- S17 TI ((Interview* or focus group*)) OR AB ((Interview* or focus group*))
- S18 TI (. ((context* or semantic or content) N2 analys*)) OR AB (. ((context* or semantic or content) N2 analys*))
- S19 TI ((narrat* or qualitative* or ethnograph* or phenomenolog* or hermeneutic*)) OR AB ((narrat* or qualitative* or ethnograph* or phenomenolog* or hermeneutic*))
- S20 TI (((personal* or patient* or inpatient* or participant* or lived) N2 (experience or experiences or perception* or perceptive or perspective*))) OR AB (((personal* or patient* or inpatient* or participant* or lived) N2 (experience or experiences or perception* or perceptive or perspective*)))
- S21 S15 OR S16 OR S17 OR S18 OR S19 OR S20
- S22 S3 AND S11 AND S14 AND S21

PsycINFO

#	Search
1	Cerebrovascular Accidents/ or exp Cerebral Ischemia/ or Cerebral Hemorrhage/
2	(stroke or cerebrovascular accident* or cerebrovascular apoplexy or brain vascular
	accident*).tw.

#	Search
3	1 or 2
4	Continuum of Care/
5	Rehabilitation/
6	aftercare/
7	hospital discharge/ or discharge planning/
8	Client Education/
9	(transition* or aftercare or aftercare or discharge plan* or (patient* adj2 discharg*)
	(care adj2 coordinat*) or continuity of care or (discharg* adj3 folllow-up*) or
	(patient* adj2 educat*) or rehabilitation).tw.
10	or/4-9
11	Self-Care Skills/
12	(home or homes or (independent* adj2 living) or communit*).tw.
13	11 or 12
14	interviews/ or interviewing/ or narratives/ or qualitative research/
15	((semi-structured or semistructured or unstructured or informal or "in-depth" or
	indepth or "face-to-face" or structured or guide? or group*) adj3 (discussion* or
	questionnaire*)).tw.
16	(Interview* or focus group*).tw.
17	((context* or semantic or content) adj2 analys*).tw.
18	(narrat* or qualitative* or ethnograph* or phenomenolog* or hermeneutic*).tw.
19	((personal* or patient* or inpatient* or participant* or lived) adj2 (experience or
	experiences or perception* or perceptive or perspective*)).tw.
20	or/14-19
21	3 and 10 and 13 and 20

Scopus

(TITLE-ABS-KEY (({stroke} OR "cerebrovascular accident*" OR "cerebrovascular apoplexy" OR "brain vascular accident*"))) AND (TITLE-ABS-KEY ((transition* OR aftercare OR aftercare OR "discharge plan*" OR (patient* W/2 discharg*) OR (care W/2 coordinat*) OR "continuity of care" OR (discharg* W/3 folllow-up*) OR (patient* W/2 educat*) OR rehabilitation))) AND (TITLE-ABS-KEY ((home OR homes OR (independent* W/2 living) OR communit*))) AND (((TITLE-ABS-KEY ((interview* OR focus AND group*)) OR TITLE-ABS-KEY (((context* OR semantic OR content) W/2 analys*)) OR TITLE-ABS-KEY (((narrat* OR qualitative* OR ethnograph* OR phenomenolog* OR hermeneutic*)) OR TITLE-ABS-KEY (((personal* OR patient* OR inpatient* OR participant* OR lived) W/2 (experience OR experiences OR perception* OR perceptive OR perspective*))))) OR (TITLE-ABS-KEY ((("semi-structured" OR semistructured OR unstructured OR informal OR "indepth" OR indepth OR "face-to-face" OR structured OR guide? OR group*) W/3 (discussion* OR questionnaire*))))

Web of Science

TS=(("stroke" OR "cerebrovascular accident*" OR "cerebrovascular apoplexy" OR "brain vascular accident*")) AND TS=((transition* OR aftercare OR aftercare OR "discharge plan*" OR (patient* NEAR/2 discharg*) OR (care NEAR/2 coordinat*) OR "continuity of care" OR (discharg* NEAR/3 folllow-up*) OR (patient* NEAR/2 educat*) OR rehabilitation)) AND TS=((home OR homes OR (independent* NEAR/2 living) OR communit*)) AND TS=((("semistructured" OR semistructured OR unstructured OR informal OR "in-depth" OR indepth OR "face-to-face" OR structured OR guide? OR group*) NEAR/3 (discussion* OR questionnaire*)) OR (Interview* OR focus group*) OR ((context* OR semantic OR content) NEAR/2 analys*) OR (narrat* OR qualitative* OR ethnograph* OR phenomenolog* OR hermeneutic*) OR ((personal* OR patient* OR inpatient* OR participant* OR lived) NEAR/2 (experience OR experiences OR perception* OR perceptive OR perspective*)))

ProQuest

all("stroke" OR "cerebrovascular accident*" OR "cerebrovascular apoplexy" OR "brain vascular accident*") AND all(transition* OR aftercare OR aftercare OR "discharge plan*" OR (patient* NEAR/2 discharg*) OR (care NEAR/2 coordinat*) OR "continuity of care" OR (discharg* NEAR/3 folllow-up*) OR (patient* NEAR/2 educat*) OR rehabilitation) AND all(home OR homes OR (independent* NEAR/2 living) OR communit*) AND all((("semistructured" OR semistructured OR unstructured OR informal OR "in-depth" OR indepth OR "face-to-face" OR structured OR guide? OR group*) NEAR/3 (discussion* OR questionnaire*)) OR Interview* OR focus group* OR ((context* OR semantic OR content) NEAR/2 analys*) OR narrat* OR qualitative* OR ethnograph* OR phenomenolog* OR hermeneutic* OR ((personal* OR patient* OR inpatient* OR participant* OR lived) NEAR/2 (experience OR experiences OR perception* OR perceptive OR perspective*)))

Appendix 11: Summary of Reviewed Studies

Author and year	Aim/objective	Sample and setting	Time post- discharge	Methodology and methods	Summary of findings
1. Bakas et al., 2002 <i>USA</i>	To determine the self-reported needs and concerns of family caregivers of stroke survivors soon after discharge to home.	N = 14 stroke family carers discharged from hospital wards. An outpatient neurology clinic	Within 6 months	 Telephone interviews Using a pre-determined code list to identify categories 	 Preparation required before discharge on management of stroke complications. Caregivers' emotions and needs ignored by healthcare professionals.
2. Cameron et al., 2013 Canada	Using Time It Right framework to determine stroke caregivers' needs, how these needs change over time and the support received.	N = 38 (24 caregivers, 14 health professionals) Inpatient rehab centre & community programs	1 month to 1 year	 Face-to-face or phone interviews Qualitative framework analysis 	 More information needs at preparation and implementation phase. Needs change over time. Caregivers need to be involved in inpatient rehab phase. Support from health professionals needed after discharge.
3. Cobley et al., 2013 UK	To investigate patients' and carers' experiences of ESD services.	<i>N</i> = 42 (27 stroke patients and 15 carers) Two stroke units	<i>M</i> = 69.5 ± 40.5 days.	Face-to-face semi- structured interviewsThematic analysis	 Lack of education and support for carers. Disjointed transition between ESD and ongoing rehabilitation services. Home-based rehabilitation satisfactory.
4. Connolly & Mahoney, 2018 USA	To investigate the experiences of ischaemic stroke survivors during the first 4 weeks after discharge home directly from the hospital.	N = 31 Two general medical and surgical neurological wards	Within 4 weeks	 Exploratory descriptive design Semi-structured phone interviews Qualitative analysis 	 A journey to new self hindered by lack of preparation, delayed follow-up services. A cohesive plan developed with survivors facilitated adjustment. Personal views of situation decided adjustment.
5. Eilersten et al., 2010 <i>Norway</i>	To examine older women's experience and the characteristics of their recovery process during the first 2 years post-stroke.	<i>N</i> = 6 Two stroke units	0–24 months	 Gadamer's phenomenological hermeneutics In-depth interviews 	 Transition home = taking on more responsibilities as more demanding than anticipated. Difficulty functioning in home environment. Loss of ability to perform pre-morbid activities = loss of value and meaning.
6. Erikson et al., 2010 <i>Sweden</i>	To investigate the meaning of acting with others in different places during the first year of rehabilitation after stroke.	N = 9 A rehabilitation department	1, 3, 6, 12 months	Grounded theoryInterviews	 Unable to gain understanding from friends during social interaction. Sensed to be a burden to family and caregiver. Establishment of sense of belonging through engaging with others.
7. Finch et al., 2021 <i>Australia</i>	To explore how people with minor stroke make sense of the stroke in the first weeks following hospital discharge to home.	N = 17 A hospital	<i>M</i> = 17 days	 Qualitative descriptive approach Semi-structured interviews Content analysis 	 Post-stroke symptoms continued but not knowing how to follow up as hospital clinicians deemed no need for further treatment. Minor stroke impacted on social participation and personal relationships.
8. Ghazzawi et al., 2016 <i>Canada</i>	To explore family carers' perceptions of navigating the stroke rehab system in transition from rehab facility to home.	N = 14 An outpatient rehabilitation facility	4–12 weeks	 Qualitative exploratory design Interviews Inductive content analysis 	 Carers forced to become system navigator due to non-linear and dynamic nature of stroke rehab system. Challenges included own health, insufficient information, unavailability of services and financial constraint

Author and year	Aim/objective	Sample and setting	Time post- discharge	Methodology and methods	Summary of findings
9. Gholamzadeh et al., 2015 <i>Iran</i>	To explore the supportive needs, concerns and coping behaviours of daughter and daughter-in-law caregivers of stroke survivors.	<i>N</i> = 17 Two large university hospitals	1 month	 Semi-structured interviews & focus groups Content & thematic analysis 	 Inadequate preparation before discharge on carer skills and management of stroke complications. Early discharge complicated with lack of follow-up and unavailability of community support services. Carers coped by spiritual emotion-focused strategies.
10. Grant, 1996 <i>USA</i>	To explore home care problems experienced by stroke survivors and their family caregivers after discharge home.	N = 20 (10 dyads) Rehabilitation ward Outpatient clinic	Within 4 months of stroke onset	 Interviews Constant comparative method 	 Loss of survivors' familiar identity. Struggling with personal care activities, mainly depending on carers. Difficulty obtaining support, especially emotional, financial support and community services.
11. Greenwood et al., 2009 <i>UK</i>	To investigate the experiences of informal carers of stroke survivors in the first 3 months after discharge.	<i>N</i> = 31 Rehabilitation Units	3 months	 Ethnography Interviews before discharge, 1 and 3 months post-discharge Thematic analysis 	 New carers struggled with care at early stage while established carers more confident. Community services delivery unreliable. Coping strategies included establishing routines, focusing on the present, identifying positives.
12. Gustafsson & Bootle, 2013 <i>Australia</i>	To explore and contrast the transition to home experience of stroke survivors and their carers during discharge and first month at home.	N = 10 (5 dyads) An inpatient rehab unit	1 month	 Descriptive qualitative design Semi-structured interviews Thematic analysis 	 Weekend & overnight leave and family meetings as positive strategies for the transition care. Lack of information on a contact person after discharge. Carers needs more support due to changes. Uncertainty about ongoing care after rehab program.
13. Gustafsson et al., 2014 <i>Australia</i>	To investigate experiences and expectations of stroke survivors in hospital-to-home transition and after participating in inpatient outreach program.	<i>N</i> = 7 An inpatient rehab unit	4–6 weeks	 Descriptive qualitative design Interviews Thematic analysis 	 Therapy at home before discharge (outreach program)—very positive in preparation for discharge. Inadequate communication on rationale for therapy. Community support and services helped finding alternative way of living for some participants.
14. Lou et al., 2017 <i>Denmark</i>	To investigate how mild stroke patients' and their partners' experience and manage life in a context of ESD.	<i>N</i> = 40 (22 survivors, 18 partners) Stroke units	3–6 weeks after stroke onset	Semi-structured interviewsThematic analysis	 ESD services supported a safe transition—safe and highly accessible. ESD did not focus on goals of returning home. Patients with chronic illnesses adjusted better due to experience with healthcare system.
15. Lutz et al., 2017 <i>USA</i>	To develop a theoretical framework for improving stroke caregiver readiness that is grounded in the experiences of stroke family caregivers.	N = 40 Two inpatient rehabilitation facilities	1–6 months	 Grounded theory Interviews Dimensional analysis and constant comparative method 	 Consider contextual factors in getting carers ready. Risk assessment including survivors' functional status, carers' capacity and commitment. Before discharge identify gaps and problem-solve. Planning for self-care to enhance sustainability.
16. Olofsson et al., 2005 <i>Sweden</i>	To understand stroke patients' experiences and needs in their journey from falling ill to discharge and after discharge.	N = 9 (5 survivors, 4 carers) Stroke centre	Within 4 months of hospital admission	Face-to-face interviewsContent analysis	 Inconsistent involvement in discharge planning and information provision. Uncertainty on the function and organisation of care. Wish for support from healthcare services.

Author and year	Aim/objective	Sample and setting	Time post- discharge	Methodology and methods	Summary of findings
17. Pereira et al., 2021 <i>Portugal</i>	To understand how stroke survivors and carers adapt over time, and how health professionals support transition to home.	 N = 18 (8 survivors, 10 carers) (Health professionals' views were excluded from this review) Hospital and rehabilitation unit 	1 & 6 months	Prospective qualitative approachInterviews	 Family-centred team approach to manage challenges in rehabilitation and adaptation process. Interest from health professionals helped adaptation. Differences in expectations and collaboration between survivors/family and health professionals.
18. Pringle et al., 2013 <i>UK</i>	To investigate the experiences of patients and their carers during the first month after discharge from hospital.	N = 24 (12 dyads) Stroke unit	1 month	 Interpretive phenomenology In-depth interviews and self-reported diaries Thematic analysis 	 One month post-discharge a dynamic time. Loss and grief over altered concept of identity. Lack of information—needs unpredictable while in hospital. Family and social support key to reconnect with life.
19. Rittman et al., 2004 <i>USA</i>	To describe dimensions of the experience of time during hospital- to-home transition.	N = 96 Inpatient ward	1 month	 Grounded theory Semi-structured interviews and field observation Constant comparative analysis 	 Challenges during transition included change of role, disruption of daily routines. Griefs on loss of old self and pre-morbid roles. The transition is a multidimensional, not only adjusting to impairments.
20. Rittman et al., 2007 <i>USA</i>	To describe three domains of psychosocial experiences of stroke survivors during the first months following discharge for acute stroke.	N = 125 Inpatient ward	1 month	Semi-structured interviews and field observations	 Psychological experience is a process of adapting to changes in self and community participation. Participants could benefit from interventions that prepare them to manage psychosocial changes. Depression is related to low community participation.
21. Sadler et al., 2014 <i>UK</i>	To investigate younger stroke survivors' experiences in navigating stroke care, and in adjusting to life after stroke.	N = 31 A population-based stroke register and a tertiary rehabilitation unit	6 weeks and 28 months after stroke onset	Semi-structured interviewsThematic analysis	 Stroke has significant impact on finance. Mismatch between patients' expectation and community service provider's focus and capacity. Inadequate involvement in care before discharge. Lack of vocational support to return to work.
22. Timothy et al., 2016 <i>New Zealand</i>	To explore embodiment and stroke through understanding of participants' bodily experience and how this related to their lived following discharge from hospital.	N = 7 Stroke rehabilitation ward	1 month	Grounded theorySemi-structured interviews	 Loss of freedom linked to loss of pre-morbid body performance. Confusion resulted from mismatch of embodied experience and self-identity. Positive attitude help anchor to a more personally acceptable embodied experience.
23. C. L. Williams et al., 2005 USA	To explore the emotional experience of veterans following hospitalisation for stroke.	N = 51 Medical centres	1 month	 Structured interview schedule, field notes and observation 	 <u>Non-depressed participants:</u> Facing adversity with strength; Finding meaning in a setback; Staying present; Hope for the future. <u>Depressed participants:</u> Taking in multiple losses; Battling a new reality; Preparing for the worst.

Author and year	Aim/objective	Sample and setting	Time post- discharge	Methodology and methods	Summary of findings
24. White et al., 2007 <i>Canada</i>	To explore the perceptions of stroke carers on barriers and facilitators to undertaking the caregiving role, particularly as related to healthcare system.	N = 14 Acute hospitals	50% < 6 months 50% > 1 year	 Interviews and focus group Content analysis 	 <u>Barriers</u>: limited involvement in discharge planning process; insufficient skills preparation; lack of community support; negative lifestyle changes. <u>Facilitators</u>: coordinated care; integration of changes to daily life; mastery of caregiving skills; accessible community resources.
25. White et al., 2015 <i>Canada</i>	To describe the perspectives of older stroke survivors and caregivers related to a readmission after stroke.	N = 20 Acute hospitals	2 weeks	 Descriptive qualitative design Interviews with medical records review 	 Inconsistency in preparation before discharge. Inadequate information provision e.g. medications. Lack of post-discharge follow-up. Inconsistency in community care planning. Financial and social support constraints.
26. Wottrich et al., 2012 <i>Sweden</i>	To investigate stroke survivors' expectations after being discharged from rehabilitation ward and during the first 3 months at home.	<i>N</i> = 5 Two rehabilitation centres	3–4 months	Semi-structured interviewContent analysis	 Home practice before discharge beneficial. <u>2-4 weeks</u>: Adjustment and waiting for recovery; physical and psychological deterioration. <u>3-4 months</u>: Acceptance with lower expectations to self and others; strong desire for independence.
27. Yeung et al., 2011 <i>Hong Kong</i>	To explore the phenomenon of concerns as experienced by Chinese stroke survivors during hospitalisation and in transition to home.	N = 15 Two stroke wards	1 week	 Phenomenological design Semi-structured interviews Thematic analysis 	 Focus of holistic concerns different hospital vs. home. Feeling of a burden to family. Social disengagement due to post-stroke changes. Needs physical as well as social, psychological and spiritual support. Inadequate provision of information.
28. Yeung et al., 2015 <i>Canada</i>	To investigate experiences of Chinese stroke survivors and family caregivers as they return to community living.	<i>N</i> = 18 A rehabilitation centre	Average 6 months	 Interviews Framework analysis (Time It Right framework) 	 Information and training needs highest during rehabilitation and returning to community phase esp. social support and long-term plan. Limited educational resources in Chinese language. Inadequate pre-discharge education.
29. Young et al., 2014 <i>USA</i>	To understand the needs of spousal caregivers of stroke survivors during the transition from rehabilitation to home.	N = 14 Two inpatient rehabilitation facilities	3–6 months	 Grounded theory Semi-structured interviews Dimensional constant comparative analysis 	 Carers unprepared with clear caregiving expectations. Beneficial to conduct a caregiver assessment at early stage of rehabilitation. A health professional to assist, advocate tailored to needs of caregivers may smooth transition. Home-based family counselling service may help develop coping skills in transition.

Note. USA = United States of America; ESD = early supported discharge.

Appendix 12: Developing Descriptive Themes Example



Appendix 13: CERQual Evidence Profile

Summary of review findings	Studies contributing to the review finding	Methodological limitations	Coherence	Adequacy	Relevance	CERQual assessment of confidence in the evidence	Explanation of CERQual assessment
Factors enabling transitional care							
Partnership approach to individualised transitional care Stroke survivors and caregivers considered engaging in own care as an essential part of the discharge planning. They valued health professionals' compassionate care and family- oriented approach, which enhanced their experience during the transition. Being able to practice in home environment prior to discharge and caregivers' building of care skills empowered them with self- management capabilities.	2–4, 12–15, 17, 20, 21, 24, 25	No or very minor concerns (Ten studies with no or very minor concerns and two study with minor concern on unclear reflexivity (3,20))	No or very minor concerns	No or very minor concerns	No or very minor concerns (Nine studies with direct relevance; three studies with indirect relevance (2,13,25))	High confidence	Twelve studies with no or very minor concerns about methodological limitations, coherence, adequacy and relevance.
 Being prepared to navigate health and social care services Stroke survivors and caregivers perceived that having a contact person after discharge that can offer help, continuity of rehabilitation and community support services generated sense of safety and facilitated post-discharge rehabilitation. 	3, 4, 13, 15, 25	No or very minor concerns (Four studies with no or very minor concerns and one study with minor concern on unclear reflexivity (3))	No or very minor concerns	Moderate concerns (Only five studies contributed to this review finding with relatively thin data)	No or very minor concerns	Moderate confidence	Five studies with no or very minor concerns about methodological limitations, coherence and relevance, moderate concerns on adequacy.

Summary of review findings	Studies contributing to the review finding	Methodological limitations	Coherence	Adequacy	Relevance	CERQual assessment of confidence in the evidence	Explanation of CERQual assessment
Developing self-management capabilities Positive thoughts, positive self- appraisal of functional improvement and self-efficacy, and family and social support contributed to the development of self-management capabilities for stroke survivors and caregivers.	4, 11, 12, 14–17, 20– 22, 26, 27, 29	No or very minor concerns (Ten studies with no or very minor concerns and three studies with minor concern on unclear reflexivity (16, 20, 21))	No or very minor concerns	No or very minor concerns	No or very minor concerns (Twelve studies with direct relevance, one study with indirect relevance (16))	High confidence	Thirteen studies with no or very minor concerns about methodological limitations, coherence, adequacy and relevance.
Barriers to effective transitional ca	re	•	•	•	•	•	
Gaps in discharge planning Stroke survivors and caregivers identified gaps in the discharge planning process including discrepancies in goals of care between survivors and health professionals, emotional needs negligence, disengaging survivors and caregivers in discharge preparation. These gaps hindered a satisfactory transition.	1, 3, 4, 9, 16, 17, 21, 24, 25, 27	Minor concerns (Six studies have no or very minor concerns of methodological limitations. Four studies have minor concerns on reflexivity (1,9,16,21))	No or very minor concerns	No or very minor concerns	No or very minor concerns about relevance (Seven studies with direct relevance. Three studies with indirect relevance (3,16,27))	High confidence	Ten studies with no or minor concerns about coherence, adequacy and relevance. Five studies have no or very minor concerns of methodological limitations. Four studies have minor concerns on methodological limitations.

Summary of review findings	Studies contributing to the review finding	Methodological limitations	Coherence	Adequacy	Relevance	CERQual assessment of confidence in the evidence	Explanation of CERQual assessment
Factors affecting self-care at home Stroke survivors' self-care capability was inhibited by discontinuation of support in community, lack of post-discharge follow-up, and significant gaps between inpatient and community rehabilitation services. Caregivers found themselves in an unexpected challenging situation due to inadequate preparation before discharge. Financial constraints impacted on their ability to achieve rehabilitation goals.	1–3, 5, 8–10, 15, 19, 21, 24, 25, 28, 29	Minor concerns (Eight studies have no or very minor concerns of methodological limitations. Six studies have minor concerns on reflexivity (1,8–10, 19, 21))	No or very minor concerns	No or very minor concerns	No or very minor concerns about relevance (Ten studies with direct relevance. Four studies with indirect relevance (3,5,19,25))	High confidence	Fourteen studies have no or very minor concerns about coherence, adequacy and relevance. Eight studies have no or very minor concerns on methodological limitations. Six studies have minor concerns on methodological limitations.
Inability to cope with challenges Stroke survivors' inability to accept the post-stroke impairments prevented them from developing coping strategies, which resulted in social disengagement. Caregivers struggled to cope with increased responsibilities and providing care while managing own health conditions. Lack of social interaction and emotional support added burden to the already challenging caregiving situation.	1, 6, 7, 9, 10, 13, 14, 15,18, 20, 22–24, 26, 27, 29	Minor concerns (Nine studies have no or very minor concerns. Seven studies have minor concerns on reflexivity (1,6,9,10,16,20, 23))	No or very minor concerns	No or very minor concerns	No or very minor concerns	High confidence	Sixteen studies with no or very minor concerns on coherence, adequacy and relevance. Nine studies have no or very minor concerns on methodological limitations. Seven studies have minor concerns on

Summary of review findings	Studies contributing to the review finding	Methodological limitations	Coherence	Adequacy	Relevance	CERQual assessment of confidence in the evidence	Explanation of CERQual assessment
							methodological limitations.
Expectation of integrated transitional care Stroke survivors and caregivers desired a person-centred integrated transitional care encompassing compassionate and holistic care approach from health professionals and adequate information provision before discharge, especially about their condition and the healthcare system. They also expected a central point of contract after discharge to help and problem-solve. Timely post- discharge follow-up and community support services would benefit achievement of rehabilitation goals.	2–4, 9, 10, 15, 24, 25	No or very minor concerns (Six studies have no or very minor concerns. Two studies have minor concerns on reflexivity (9,10))	No or very minor concerns	No or very minor concerns	No or very minor concerns	High confidence	Six studies have no or very minor concerns on methodological limitations, coherence, relevance and adequacy. Two studies have minor concerns on methodological limitations.

Appendix 14: Authorisation Letter From the Southern Adelaide Local Health Network Ethics Committee

From: Health:SALHN Office for Research <Health.SALHNOfficeforResearch@sa.gov.au>
Sent: Thursday, 2 May 2019 8:58:45 AM
To: Langduo Chen
Cc: Lily Xiao; Diane Chamberlain; Saxty, Karen (Health)
Subject: RE: SAC HREC review - 39.19 - Exploring the care needs of stroke survivors and informal carers in hospital to home transition.

Dear Langduo,

RE: Exploring the care needs of stroke survivors and informal carers in hospital to home transition. (39.19)

Thank you for your email below. This study has now been approved by the SAC HREC.

This approval is subject to governance authorisation. You should not proceed with any aspect of this study until governance (the Site Specific Assessment) has been authorised. If you have any questions regarding the governance authorisation process you should contact Karen Saxty at this email address.

The formal approval letter will be provided to you once the governance has been authorised. The SAC HREC wishes you every success with your research.

Kind Regards,

Dominic How

Ethics Officer, Office for Research

Southern Adelaide Local Health Network Flinders Medical Centre, Ward 6C, Room 6A219 Tel: (08) 8204 6285 Email: <u>Dominic.How@sa.gov.au</u> / <u>Health.SALHNOfficeforResearch@sa.gov.au</u> Website: http://www.sahealth.sa.gov.au/SALHNresearch To: Lily Xiao <lily.xiao@flinders.edu.au>; Prendergast, Justin (Health) <Justin.Prendergast@sa.gov.au> Subject: 39.19 authorisation letter

Good morning

The SSA form for the application **39.19** has been authorised and the project may now commence.

Thank you

Dani Eley Administration Officer Executive Assistant | Paula Davies, Director Research Operations

Southern Adelaide Local Health Network Flinders Medical Centre, Ward 6C, Room 6A219 Phone: 8204 6453 | Email: <u>Danielle.eley@sa.gov.au</u> / <u>health.SALHNOfficeforResearch@sa.gov.au</u> Website: <u>http://www.sahealth.sa.gov.au/SALHNresearch</u>

Monday to Friday 800 - 1600

The <u>Office for Research</u> is keen to work with researchers and we have organised education initiatives each week at FMC (Level 6, Room 6A219). We run a <u>drop in advisory clinic session</u> Wednesday 9am – 11am but we are happy to discuss any concern you may have relating to new or ongoing research at any stage via <u>Health.SALHNOfficeforResearch@sa.gov.au</u> or 8204 6453. We also can provide tailored team training sessions by appointment.

This email may contain confidential information, which may be legally privileged. Only the intended recipient(s) may access, use, distribute or copy this email. If this email is received in error, please inform the sender by return email and delete the original. If there are doubts about the validity of this message, please contact the sender by telephone. It is the recipient's responsibility to check the email and any attached files for viruses.
Appendix 15: Participant Information and Consent Form

TitleExploring the care needs of stroke survivors and
informal carers in hospital to home transitionShort TitleStroke hospital to home transitionCoordinating Principal
Investigator/ Principal
InvestigatorProfessor Lily XiaoAssociate Investigator(s)Langduo Chen, A/Professor Diane ChamberlainLocationFlinders Medical Centre

Participant Information Sheet/Consent Form

Part 1 What does my participation involve?

1. Introduction

You are invited to participate in a research project to investigate the care needs and the approaches to meeting the care needs of stroke survivors and their informal carers during the first 6 months after discharge from hospital to home. Stroke survivors in this study are defined as stroke patients who are 18 years and older and are discharged to home with ongoing rehabilitation and community support services after hospital stay. Participants in the study are required to use English in interviews and diary entry to share their experience. This Participant Information Sheet/Consent Form tells you about the study. It explains the activities involved. Knowing what is involved will help you decide if you want to participate in the research project.

Please read this information carefully. Ask questions about the project that you do not understand or want to know more about. Before deciding whether to take part, you might want to talk about it with a relative and friend.

Participation in this research is voluntary. If you do not wish to take part, you do not have to.

If you decide to take part in the research, you will be asked to sign the consent section.

By signing it you are telling us that you:

- Understand what you have read;
- Consent to take part in the research;
- Consent to the use of your personal and health information as described.

You will be given a copy of this Participant Information and Consent Form to keep.

2. What is the purpose of this research?

The aim of this research is to explore the care needs and the approaches to meeting the care needs for stroke survivors and their informal carers during hospital to home transition. The first 6 months after discharge from hospital is the most stressful period for stroke survivors and their informal carers. They usually face enormous challenges to adapt to their home care environments for continuing recovery after discharge. They also encounter great difficulties in identifying suitable rehabilitation and care services. Findings of this research will inform the improvement of transition care services for stroke survivors and their informal carers.

The results of this research will be used by the researcher Langduo Chen to obtain a *PhD in Nursing* degree.

3. What does participation in this research involve?

You are invited to participate in 2-3 times of a face to face, 30-45 minutes interview with the researcher at those time periods: prior to discharge, 3 months and 6 months after returning home from hospital. The interview will occur at a venue of your choice. Hospital to home transition is a continuum process and care needs of stroke survivors and carers change over time. Interviews at different time points help the researchers to identify these changes. Each interview will be audio recorded for data analysis. You are also required to take a structured diary at home and record key activities/challenges they face during the follow up period. The consent form will be signed before you participate in this research. The interview will be guided by the overarching question 'what was it like to be discharged from hospital after stroke'. The interviews provide you with an opportunity to tell your experience, care needs and expectations of health and social care system during hospital to home transition. You have the rights to refuse or withdraw from the research at any stage. You also have the right to refuse to answer any questions.

The interview will be audio recorded and later transcribed word by word by the researcher. You will be given the opportunity to edit your response before data analysis if you wish to do so. All records containing personal information will remain confidential and no information which could lead to your identification will be released, except as required by law. Data will be stored at the university's computer and protected by a password to ensure security. On the completion of the research, you will receive a written summary of research findings.

If you decide to take part in the research project, you will first be contacted by a researcher to make an appointment with you before discharge. The Researcher will conduct a cognitive function assessment for you to determine if you are eligible to take part. The assessment will take about 10 minutes to complete. Based on the assessment outcome, the researcher will discuss with you if you are eligible to take part in the project.

This research project has been designed to make sure the researchers interpret the results in a fair and appropriate way and avoids researchers or participants jumping to conclusions.

There are no costs associated with participating in this research project, nor will you be paid. However, you will be reimbursed for any reasonable travel and parking expenses associated with the research project visit. Participant who completes at least 2 times of interviews and diary entries will receive a \$50 complementary Woolworth gift card.

4. Other relevant information about the research project

This research is part of a PhD study for the PhD candidate, Ms Langduo Chen at Flinders University, South Australia. The research project will be conducted at Flinders Medical Centre. Langduo will undertake interviews with stroke survivors and their informal carers.

5. Do I have to take part in this research project?

Participation in this research is voluntary. You do not have to if you do not wish to take part. If you decide to take part and later change your mind, you are free to withdraw from the research at any stage.

Your decision whether to take part or not to take part, or to take part and then withdraw, will not affect your routine care, your relationship with professional staff or your relationship with Flinders Medical Centre.

There are a few ways to express your interest.

- When approached for expression of intention, you can verbally inform the researcher your intention to participate. A Research Assistant will make an appointment with you for further discussion.
- You can fill in and return the response slip in a prepaid envelope and post to the researcher.
- You can text message or ring the researcher (Ms Langduo Chen, phone number: 08 8404 2062) to express your interest in participation.
- You can email the researcher to express your interest in participation (Ms Langduo Chen, email: langduo.chen@flinders.edu.au)

On receipt of your response, the researcher will contact you to confirm your participation. More detailed information related to the research will be explained to you by the Research Assistant if you would like to know. If you do decide to take part, you will be given this Participant Information and Consent Form to sign and you will be given a copy to keep.

6. What are the possible benefits of taking part?

We cannot guarantee or promise that you will receive any benefits from this research. You will be reimbursed for any reasonable travel and parking expenses associated with the research project visit. Participant who completes at least 2 times of interviews and diary data entries will receive a \$50 complementary Woolworth gift card.

Your participation will be a valuable addition to our research and findings could lead to greater understanding of the care needs and expectations during hospital to home transition after stroke. The findings will assist health professionals and health/social care service providers to improve transition care services. The findings will also inform staff education and training associated with transition care for stroke survivors.

7. What are the possible risks and disadvantages of taking part?

There are no foreseen risks associated with the research. You may feel that some of the questions we ask are stressful or upsetting. If you do not wish to answer a question, you may skip it and go to the next question, or you may stop immediately. If you become upset or distressed as a result of your participation in the research project, upon your request and permission, the research team will be able to arrange for counselling or other appropriate support. Any counselling or support will be provided by qualified staff who are not members of the research team. This counselling will be provided free of charge.

8. What if I withdraw from this research project?

If you do consent to participate, you may withdraw at any time. If you decide to withdraw from the project, please notify Langduo Chen before you withdraw. Langduo will inform you if there are any special requirements linked to withdrawing. If you do withdraw, you will be asked to complete and sign a 'Withdrawal of Consent' form; this will be provided to you by the research team.

If you decide to leave the research project, the researchers will not collect additional personal information from you, although personal information already collected will be retained to ensure that the results of the research project can be measured properly and to comply with law. You should be aware that data collected up to the time you withdraw will form part of the research project results. If you do not want your data to be included, you must tell the researchers when you withdraw from the research project.

9. Could this research project be stopped unexpectedly?

This research project may be stopped unexpectedly for a variety of reasons. These may include reasons such as the PhD student researcher stops the PhD study. You will be advised if this occurs.

10. What happens when the research project ends?

After the research project is completed, a summary of the global findings will be sent to you by email or posting to you with your permission.

Part 2 How is the research project being conducted?

11. What will happen to information about me?

By signing the consent form you agree to the study team to accessing health records in this research project. By signing the consent form you also consent to the research team collecting and using personal information about you for the research project. Any information obtained in connection with this research project that can identify you will remain confidential and you will not be individually identifiable in the resulting report and thesis. Any information provided by you will be collected in a de-identifiable form. Therefore, the data will not be linked to the individual or organisation. Data collected through this research will be stored in a locked cabinet in the researcher's office and electronic data will be kept on Flinders University computer server, which is password protected. All study-related data will only be accessible to the researchers. All data will be retained for fifteen years, and then it will be destroyed. Your information will only be used for the purpose of this research project and it will only be disclosed with your permission, except as required by law. The personal information that the research team collects and uses is your age, gender, education level, diagnosis, functional impairment scores, admission and discharge date, community rehabilitation services you receive.

Information about your diagnosis, type of stroke and functional impairment may be obtained from your health records held at this hospital for the purpose of this research. By signing the consent form you agree to the research team accessing health records if they are relevant to your participation in this research project.

It is anticipated that the results of this research will be published and/or presented in a variety of forums. In any publication and/or presentation, information will be provided in such a way that you cannot be identified, except with your permission. All collected data will be de-identified. The transcripts of interviews and focus groups will be coded so participants' responses are de-identified. Data will be analysed collectively. Pseudonyms will be used in the PhD thesis and all relevant publications.

In accordance with relevant Australian privacy and other relevant laws, you have the right to request access to the information about you that is collected and stored by the research team. You also have the right to request that any information with which you disagree be

corrected. Please inform the research team member named at the end of this document if you would like to access your information.

Any information obtained for the purpose of this research project that can identify you will be treated as confidential and securely stored. It will be disclosed only with your permission, or as required by law.

12. Complaints and compensation

There is no foreseen adverse event and outcomes associated with your participation in this project. If you suffer any distress or psychological injury as a result of this research project, you should contract the research team as soon as possible. You will be assisted with arranging appropriate treatment and support. The complaints contact is

Position	Asst. Director, Office for Research
Telephone	8204 6453
Email	Health.SALHNofficeforresearch@sa.gov.au

13. Who is organising and funding the research

As part of the PhD study, this research project is being conducted by the PhD candidature Langduo Chen and the PhD student supervisors.

14. Who has reviewed the research project?

All research in Australia involving humans is reviewed by an independent group of people called the Human Research Ethics Committee (HREC). The ethical aspects of this research have been approved by the Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC).

This research will be carried out according to the National Statement on Ethical Conduct in Human Research (2007 updated 2018). This statement has been developed to protect the interests of people who agree to participate in human research studies.

15. Further information and who to contact

If you have any complaints about any aspect of the project, the way it is being conducted or any questions about being a research participant in general, then you may contact:

Reviewing HREC approving this research and HREC Executive Officer details

Reviewing HREC name	Southern Adelaide Clinical
Position	Executive Officer
Telephone	8204 6453
Email	Health.SALHNofficeforresearch@sa.gov.au

Local HREC Office contact (Single Site -Research Governance Officer)

Position	Research Governance Officer
Telephone	8204 6453
Email	Health.SALHNofficeforresearch@sa.gov.au

Thank you for taking the time to read this information sheet and we hope that you will accept our invitation to be involved.

For further information or questions, please contact:

Name: Langduo Chen

Phone: 08 8404 2062

Email: Langduo.chen@flinders.edu.au

Consent Form

Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Stroke hospital to home transition
Coordinating Principal Investigator/ Principal Investigator	A/Professor Lily Xiao
Associate Investigator(s)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Declaration by Participant

I (being over the age of 18 years) have read the Participant Information Sheet or someone has read it to me in a language that I understand.

I understand the purposes, procedures and risks of the research described in the project.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the project without affecting my future care.

I freely agree to audio recording of my information and participation.

I give permission for health professionals and hospitals to release information to the principle investigator and associate investigators concerning my disease for the purposes of this research. I understand that such information will remain confidential.

I understand that I will be given a signed copy of this document to keep.

I am aware that I should retain a copy of the Information Sheet and Consent Form for future reference.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I understand that:

- I may not directly benefit from taking part in this research.
- I am free to withdraw from the research at any time and am free to decline to answer particular questions.

- While the information gained in this study will be published as explained, I will not be identified, and individual information will remain confidential.
- Whether I participate or not, or withdraw after participating, it will have no effect on any treatment or service that is being provided to me.
- I may ask that the recording/observation be stopped at any time, and that I may withdraw at any time from the session or the research without disadvantage.

Name of Participant (please print)	
Signature	Date
Name of Participant's Carer (please print)	
Signature	Date
Declaration by Researcher†	
I have given a verbal explanation of the research, the participant has understood that explanation.	its procedures and risks and I believe that
Name of Researcher (please print)	
Signature	Date
† An appropriately qualified member of the resear and information concerning, the research project.	ch team must provide the explanation of,

Note: All parties signing the consent section must date their own signature.

Response Slip

I..... (name) would like to participate in the research.

or Email.....

Alternatively, you can text message or ring the researcher (Ms Langduo Chen, phone number: 08 8404 2062) or email the researcher to express your interest in participating (Ms Langduo Chen, email: langduo.chen@flinders.edu.au).

Form for Withdrawal of Participation

Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Stroke hospital to home transition
Coordinating Principal Investigator/ Principal Investigator	A/Professor Lily Xiao
Associate Investigator(s)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Declaration by Participant

I wish to withdraw from participation in the above resea	arch project and understand that such
vithdrawal will not affect my routine care, or my relationships with the researchers or	
Flinders Medical Centre.	
Name of Participant (please print)	
Signature	Date

In the event that the participant's decision to withdraw is communicated verbally, the Senior Researcher must provide a description of the circumstances below.

Declaration by Researcher[†]

I have given a verbal explanation of the research, its procedures and risks and I believe that the participant has understood that explanation.

Name of Researcher (please print)

Signature _____

Date _____

† An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Note: All parties signing the consent section must date their own signature.

Participant Information Sheet/Consent Form

(Stroke carers)

Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Stroke hospital to home transition
Coordinating Principal Investigator/ Principal Investigator	Professor Lily Xiao
Associate Investigator(s)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Part 1 What does my participation involve?

1. Introduction

You are invited to participate in a research project to investigate the care needs and the approaches to meeting the care needs of stroke survivors and their informal carers during the first 6 months after discharge from hospital to home. Informal carers of stroke survivors are persons who are 18 years and older, principal carers who have daily contact with stroke survivors. Participants in the study are required to use English in interviews and diary entry to share their experience. This Participant Information Sheet/Consent Form tells you about the study. It explains the activities involved. Knowing what is involved will help you decide if you want to participate in the research project.

Please read this information carefully. Ask questions about the project that you do not understand or want to know more about. Before deciding whether to take part, you might want to talk about it with a relative and friend.

Participation in this research is voluntary. If you do not wish to take part, you do not have to.

If you decide to take part in the research, you will be asked to sign the consent section.

By signing it you are telling us that you:

- Understand what you have read;
- Consent to take part in the research;
- Consent to the use of your personal and health information as described.

You will be given a copy of this Participant Information and Consent Form to keep.

2. What is the purpose of this research?

The aim of this research is to explore the care needs and the approaches to meeting the care need for stroke survivors and their informal carers during hospital to home transition. The first 6 months after discharge from hospital is the most stressful period for stroke survivors and their informal carers. They usually face enormous challenges to adapt to their home care environments for continuing recovery after discharge. They also encounter great difficulties in identifying suitable rehabilitation and care services. Findings of this research will inform the improvement of transition care services for stroke survivors and their informal carers.

The results of this research will be used by the researcher Langduo Chen to obtain a *PhD in Nursing* degree.

3. What does participation in this research involve?

You are invited to participate in 2-3 times of a face to face, 30-45 minutes interview with the researcher at those time periods: prior to discharge, 3 months, 6 months after returning home from hospital. The interview will occur at a venue of your choice. Hospital to home transition is a continuum process and care needs of stroke survivors and carers change over time. Interviews at different time points help the researchers to identify these changes. Each interview will be audio recorded for data analysis. The participant is also required to take a structured diary at home and record key activities/challenges they face during the follow up period. The consent form will be signed before you participate in this research. The interview will be guided by the overarching question 'what was it like to be discharged from hospital after stroke'. The interviews provide you with an opportunity to tell your experience, care needs and expectations of health and social care system during hospital to home transition. You have the rights to refuse or withdraw from the research at any stage. You also have the right to refuse to answer any questions.

The interview will be audio recorded and later transcribed word by word by the researcher. You will be given the opportunity to edit your response before data analysis if you wish to do so. All records containing personal information will remain confidential and no information which could lead to your identification will be released, except as required by law. Data will be stored at the university's computer and protected by a password to ensure security. On the completion of the research, you will receive a written summary of research findings.

If you decide to take part in the research project, you will firstly be contacted by a researcher to make an appointment with you before the discharge of your relative. The research will ask about your age, relevant information about relation with your relative, discharge date and community rehabilitation services that your relative will receive after discharge. Answering these questions will take approximately 5-10 minutes. Based on the assessment outcome, the researcher will discuss with you if you are eligible to take part in the project.

This research project has been designed to make sure the researchers interpret the results in a fair and appropriate way and avoids study researchers or participants jumping to conclusions.

There are no costs associated with participating in this research project, nor will you be paid. However, you will be reimbursed for any reasonable travel and parking associated with the research project visit. Participant who completes at least 2 times of interviews and diary entry will receive a \$50 complementary Woolworth gift card.

4. Other relevant information about the research project

This research is part of a PhD study for the PhD student, Ms. Langduo Chen at Flinders University, South Australia. The research project will be conducted at Flinders Medical Centre. Langduo will undertake interviews with stroke survivors and their informal carers.

5. Do I have to take part in this research project?

Participation in this research is voluntary. You do not have to if you do not wish to take part. If you decide to take part and later change your mind, you are free to withdraw from the research at any stage.

Your decision whether to take part or not to take part, or to take part and then withdraw, will not affect your routine care, your relationship with professional staff or your relationship with Flinders Medical Centre.

There are a few ways to express your interest.

- When approached for expression of intention, you can verbally inform the researcher your intention to participate. A Research Assistant will make an appointment with you for further discussion.
- You can fill in and return the response slip in a prepaid envelope and post to the researcher.
- You can text message or ring the researcher (Ms Langduo Chen, phone number: 08 8404 2062) to express your interest in participation.
- You can email the researcher to express your interest in participation (Ms Langduo Chen, email: langduo.chen@flinders.edu.au)

On receipt of your response, the researcher will contact you to confirm your participation. More detailed information related to the research will be explained to you by the researcher if you would like to know. If you do decide to take part, you will be given this Participant Information and Consent Form to sign and you will be given a copy to keep.

6. What are the possible benefits of taking part?

We cannot guarantee or promise that you will receive any benefits from this research. You will be reimbursed for any reasonable travel and parking associated with the research project visit. Participant who completes at least 2 times of interviews and diary entry will receive a \$50 complementary Woolworth gift card.

Your participation will be a valuable addition to our research and findings could lead to greater understanding of the care needs and expectations during hospital to home transition after stroke. The findings will assist health professionals and health/social care service providers to improve transition care services. The findings will also inform staff education and training associated with transition care for stroke survivors.

7. What are the possible risks and disadvantages of taking part?

There are no foreseen risks associated with the research. You may feel that some of the questions we ask are stressful or upsetting. If you do not wish to answer a question, you may skip it and go to the next question, or you may stop immediately. If you become upset or distressed as a result of your participation in the research project, upon your request and permission, the research team will be able to arrange for counselling or other appropriate support. Any counselling or support will be provided by qualified staff who are not members of the research team. This counselling will be provided free of charge.

8. What if I withdraw from this research project?

If you do consent to participate, you may withdraw at any time. If you decide to withdraw from the project, please notify Langduo Chen before you withdraw. Langduo will inform you if there are any special requirements linked to withdrawing. If you do withdraw, you will be asked to complete and sign a 'Withdrawal of Consent' form; this will be provided to you by the research team.

If you decide to leave the research project, the researchers will not collect additional personal information from you, although personal information already collected will be retained to ensure that the results of the research project can be measured properly and to comply with law. You should be aware that data collected up to the time you withdraw will form part of the research project results. If you do not want your data to be included, you must tell the researchers when you withdraw from the research project.

9. Could this research project be stopped unexpectedly?

This research project may be stopped unexpectedly for a variety of reasons. These may include reasons such as the PhD student researcher stops the PhD study. You will be advised if this occurs.

10. What happens when the research project ends?

After the research project is completed, a summary of the global findings will be sent to you by email or posting to you with your permission.

Part 2 How is the research project being conducted?

11. What will happen to information about me?

By signing the consent form you consent to the research team collecting and using personal information about you for the research project. Any information obtained in connection with this research project that can identify you will remain confidential and you will not be individually identifiable in the resulting report and thesis. Any information provided by you will be collected in a de-identifiable form. Therefore, the data will not be linked to the individual or organisation. Data collected through this research will be stored in a locked cabinet in the researcher's office and electronic data will be kept on Flinders University computer server, which is password protected. All study-related data will only be accessible to the researchers. All data will be retained for fifteen years, and then it will be destroyed. Your information will only be used for the purpose of this research project and it will only be disclosed with your permission, except as required by law. The personal information that the research team collects and uses is your age, gender, education level, diagnosis, functional impairment, discharge date, community rehabilitation services you receive.

Information about your diagnosis, type of stroke and functional impairment may be obtained from your health records held at this hospital for the purpose of this research. By signing the consent form you agree to the research team accessing health records if they are relevant to your participation in this research project.

It is anticipated that the results of this research will be published and/or presented in a variety of forums. In any publication and/or presentation, information will be provided in such a way that you cannot be identified, except with your permission. All collected data will be de-identified. The transcripts of interviews and focus groups will be coded so participants' responses are de-identified. Data will be analysed collectively. Pseudonyms will be used in the PhD thesis and all relevant publications.

In accordance with relevant Australian privacy and other relevant laws, you have the right to request access to the information about you that is collected and stored by the research team. You also have the right to request that any information with which you disagree be

391

corrected. Please inform the research team member named at the end of this document if you would like to access your information.

Any information obtained for the purpose of this research project that can identify you will be treated as confidential and securely stored. It will be disclosed only with your permission, or as required by law.

12. Complaints and compensation

There is no foreseen adverse event and outcomes associated with your participation in this project. If you suffer any distress or psychological injury as a result of this research project, you should contract the research team as soon as possible. You will be assisted with arranging appropriate treatment and support. The complaints contact is

Position	Asst. Director, Office for Research
Telephone	8204 6453
Email	Health.SALHNOfficeforResearch@sa.gov.au.

13. Who is organising and funding the research

As part of the PhD study, this research project is being conducted by Langduo Chen and the PhD student supervisors.

14. Who has reviewed the research project?

All research in Australia involving humans is reviewed by an independent group of people called the Human Research Ethics Committee (HREC). The ethical aspects of this research have been approved by the Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC).

This research will be carried out according to the National Statement on Ethical Conduct in Human Research (2007 updated 2018). This statement has been developed to protect the interests of people who agree to participate in human research studies.

15. Further information and who to contact

If you have any complaints about any aspect of the project, the way it is being conducted or any questions about being a research participant in general, then you may contact:

Reviewing HREC approving this research and HREC Executive Officer details

Reviewing HREC name	Southern Adelaide Clinical
Position	Executive Officer
Telephone	8204 6453

Email	Health.SALHNofficeforresearch@sa.gov.au
-------	---

Local HREC Office contact (Single Site -Research Governance Officer)

Position	Research Governance Officer
Telephone	8204 6453
Email	Health.SALHNofficeforresearch@sa.gov.au

Thank you for taking the time to read this information sheet and we hope that you will accept our invitation to be involved.

For further information or questions, please contact:

Name: Langduo Chen

Phone: 08 8404 2062

Email: Langduo.chen@flinders.edu.au

Consent Form

Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Stroke hospital to home transition
Coordinating Principal Investigator/ Principal Investigator	A/Professor Lily Xiao
Associate Investigator(s)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Declaration by Participant

I (being over the age of 18 years) have read the Participant Information Sheet or someone has read it to me in a language that I understand.

I understand the purposes, procedures and risks of the research described in the project.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the project without affecting my future care.

I freely agree to audio recording of my information and participation.

I give permission for health professionals and hospitals to release information to the principle investigator and associate investigators concerning my disease for the purposes of this research. I understand that such information will remain confidential.

I understand that I will be given a signed copy of this document to keep.

I am aware that I should retain a copy of the Information Sheet and Consent Form for future reference.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I understand that:

- I may not directly benefit from taking part in this research.
- I am free to withdraw from the research at any time and am free to decline to answer particular questions.

- While the information gained in this study will be published as explained, I will not be identified, and individual information will remain confidential.
- Whether I participate or not, or withdraw after participating, it will have no effect on any treatment or service that is being provided to me.
- I may ask that the recording/observation be stopped at any time, and that I may withdraw at any time from the session or the research without disadvantage.

Name of Participant (please print)		
Signature	Date	

Declaration by Researcher†

I have given a verbal explanation of the research, its procedures and risks and I believe that the participant has understood that explanation.

Name of Researcher (please print)

Signature	Date

† An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Note: All parties signing the consent section must date their own signature.

Response Slip

I..... (name) would like to participate in the research.

or Email.....

Alternatively, you can text message or ring the researcher (Ms Langduo Chen, phone number: 08 8404 2062) or email the researcher to express your interest in participating (Ms Langduo Chen, email: langduo.chen@flinders.edu.au).

Form for Withdrawal of Participation

Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Stroke hospital to home transition
Coordinating Principal Investigator/ Principal Investigator	A/Professor Lily Xiao
Associate Investigator(s)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Declaration by Participant

I wish to withdraw from participation in the above research project and understand that such
withdrawal will not affect my routine care, or my relationships with the researchers or
Flinders Medical Centre.

Name of Participant (please print)	
· · · · · · · ·	

Signature	Date	
0		

In the event that the participant's decision to withdraw is communicated verbally, the Senior Researcher must provide a description of the circumstances below.

Declaration by Researcher†

I have given a verbal explanation of the research, its procedures and risks and I believe that the participant has understood that explanation.

Name of Researcher (please print)_____

Signature _____ Date _____

† An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Note: All parties signing the consent section must date their own signature.

Participant Information Sheet/Consent Form

(Н	ealth professionals)
Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Hospital to home transition for stroke survivors
Coordinating Principal Investigator/ Principal Investigator	Professor Lily Xiao
Associate Investigator(s) (if required by institution)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Part 1 What does my participation involve?

1. Introduction

You are invited to participate in a research project to investigate the care needs and the approaches to meeting the care need of stroke survivors and their informal carers during the first 6 months after discharge from hospital to home. Health professionals in this study are defined as staff members of stroke/rehabilitation care team (not students) and General Practitioners (GPs) who are older than 18 years and are involved in the care of stroke patients. This Participant Information Sheet/Consent Form tells you about the study. It explains the activities involved. Knowing what is involved will help you decide if you want to participate in the research project.

Please read this information carefully. Ask questions about the project that you do not understand or want to know more about. Before deciding whether to take part, you might want to talk about it with a relative, friend or your colleague.

Participation in this research is voluntary. If you do not wish to take part, you do not have to.

If you decide to take part in the research, you will be asked to sign the consent section.

By signing it you are telling us that you:

- Understand what you have read;
- Consent to take part in the research;
- Consent to the use of your personal information as described.

You will be given a copy of this Participant Information and Consent Form to keep.

2. What is the purpose of this research?

The aim of this research is to explore the care needs and the approaches to meeting the care need for stroke survivors and their informal carers during hospital to home transition. The first 6 months after discharge from hospital is the most stressful period for stroke survivors and their informal carers. They usually face enormous challenges to adapt to their home care environments for continuing recovery after discharge. They also encounter great difficulties in identifying suitable rehabilitation and care services. Increased knowledge on their care needs will inform changes to transition care for these populations.

Findings of this research will inform the development of a transition care model and pathway to provide better patient centred care for stroke survivors and their informal carers.

The results of this research will be used by the researcher Langduo Chen to obtain a *PhD in Nursing* degree.

3. What does participation in this research involve?

You are invited to participate in one of the 3-5 focus groups lasting about 60-90 minutes. If you are unable to attend focus groups, you will be invited to a one on one, face to face interview lasting about 45 to 60 minutes. General practitioners will participate in a face to face, 45-60 minutes interview with the researcher. The focus groups/interview will occur in a meeting room at participants' workplace. Each focus group/interview will be audio recorded for data analysis. You will be asked to sign a consent form before you participate in this research. The focus group/interview will be guided by the overarching question 'what are the approaches to identifying and meeting transition care needs of the stroke survivors and their informal carers'. The focus group/interview provides you with an opportunity to share your thoughts, ideas and experience in the care of stroke survivors and their carers during hospital to home transition. You have the rights to refuse or withdraw from the research at any stage. You also have the right to refuse to answer any questions.

The interview will be audio recorded and later transcribed word by word by the researcher. You will be given the opportunity to read your response before data analysis if you wish to do so. All records containing personal information will remain confidential and no information which could lead to your identification will be released, except as required by law. Data will be stored at the university's computer and protected by a password to ensure security. On the completion of the research, you will receive a written summary of research findings.

If you decide to take part in the research project, you will first be contacted by the researcher, Langduo Chen, asking about your age, occupation and experience in stroke care. This will determine if you are eligible to take part. Answering these questions will take approximately 5-10 minutes. If the screening questions shows that you meet the requirements, then you will be able to participate in the research project.

This research project has been designed to make sure the researchers interpret the results in a fair and appropriate way and avoids study researchers or participants jumping to conclusions.

There are no costs associated with participating in this research project, nor will you be paid. However, you will be reimbursed for any reasonable travel and parking associated with the research project visit.

4. Other relevant information about the research project

This research is part of a PhD study for the PhD candidate, Langduo Chen at Flinders University, Australia. The research project will be conducted at Flinders Medical Centre. Langduo will undertake interviews with stroke survivors and their informal carers as well as the General Practitioners. She will also facilitate focus groups for stroke health professionals.

5. Do I have to take part in this research project?

Participation in this research is voluntary. You do not have to if you do not wish to take part. If you decide to take part and later change your mind, you are free to withdraw from the research at any stage.

Your decision whether to take part or not to take part, or to take part and then withdraw, will not affect your employment status.

There are a few ways to express your interest.

- You can fill in and return the response slip in a prepaid envelope and post to the researcher.
- You can text message or ring the researcher (Ms Langduo Chen, phone number: 08 8404 2062) to express your interest in participation.
- You can email the researcher to express your interest in participation (Ms Langduo Chen, email: langduo.chen@flinders.edu.au)

On receipt of your response, the researcher will contact you to confirm your participation. More detailed information related to the research will be explained to you by the researcher if you would like to know. If you do decide to take part, you will be given this Participant Information and Consent Form to sign and you will be given a copy to keep.

6. What are the possible benefits of taking part?

We cannot guarantee or promise that you will receive any benefits from this research. However, your participation will be a valuable addition to our research and findings could lead to greater understanding of the care needs and how to meet the care needs during hospital to home transition after stroke. The evidence will assist the development of a transition care model and pathway that will be beneficial for stroke survivors and their carers. The result will also inform staff development and training in stroke care.

7. What are the possible risks and disadvantages of taking part?

There are no foreseen risks associated with the research. You may feel that some of the questions we ask are stressful or upsetting. If you do not wish to answer a question, you may skip it and go to the next question, or you may stop immediately. Whilst all care will be taken to maintain privacy and confidentiality, you may experience uncomfortable if one of the group members were to repeat things said in a confidential group meeting. If you become upset or distressed as a result of your participation in the research project, the research team will be able to arrange for counselling or other appropriate support. Any counselling or support will be provided by qualified staff who are not members of the research team. This counselling will be provided free of charge.

8. What if I withdraw from this research project?

If you do consent to participate, you may withdraw at any time. If you decide to withdraw from the project, please notify Langduo Chen before you withdraw. Langduo will inform you if there are any special requirements linked to withdrawing.

If you decide to leave the research project, the researchers will not collect additional personal information from you, although personal information already collected will be retained to ensure that the results of the research project can be measured properly and to comply with law. You should be aware that data collected up to the time you withdraw will form part of the research project results. If you do not want your data to be included, you must tell the researchers when you withdraw from the research project.

9. Could this research project be stopped unexpectedly?

This research project may be stopped unexpectedly for a variety of reasons. These may include reasons such as the PhD student researcher stops the PhD study. You will be advised if this occurs.

10. What happens when the research project ends?

After the research project is completed, a summary of the global findings will be sent to you by email or posting to you with your permission.

Part 2 How is the research project being conducted?

11. What will happen to information about me?

By signing the consent form you consent to the research team collecting and using personal information about you for the research project. Any information obtained in connection with

this research project that can identify you will remain confidential and you will not be individually identifiable in the resulting report and thesis. Any information provided by you will be collected in a de-identifiable form. Therefore, the data will not be linked to the individual or organisation. Data collected through this research will be stored in a locked cabinet in the researcher's office and electronic data will be kept on Flinders University computer server, which is password protected. All study-related data will only be accessible to the researchers. All data will be retained for fifteen years, and then it will be destroyed. Your information will only be used for the purpose of this research project and it will only be disclosed with your permission, except as required by law. The personal information that the research team collects and uses is your age, gender, education level, occupation, length of time in the position.

It is anticipated that the results of this research will be published and/or presented in a variety of forums. In any publication and/or presentation, information will be provided in such a way that you cannot be identified, except with your permission. All collected data will be de-identified. The transcripts of interviews and focus groups will be coded so participants' responses are de-identified. Data will be analysed collectively. Pseudonyms will be used in the PhD thesis and all relevant publications.

In accordance with relevant Australian privacy and other relevant laws, you have the right to request access to the information about you that is collected and stored by the research team. You also have the right to request that any information with which you disagree be corrected. Please inform the research team member named at the end of this document if you would like to access your information.

Any information obtained for the purpose of this research project that can identify you will be treated as confidential and securely stored. It will be disclosed only with your permission, or as required by law.

12. Complaints and compensation

There is no foreseen adverse event and outcomes associated with your participation in this project. If you suffer any distress or psychological injury as a result of this research project, you should contract the research team as soon as possible. You will be assisted with arranging appropriate treatment and support. The complaints contact is

Position	Asst. Director, Office for Research
Telephone	8204 6453
Email	Health.SALHNofficeforresearch@sa.gov.au

13. Who is organising and funding the research

As part of the PhD study, this research project is being conducted by Langduo Chen and the PhD student supervisors.

14. Who has reviewed the research project?

All research in Australia involving humans is reviewed by an independent group of people called the Human Research Ethics Committee (HREC). The ethical aspects of this research have been approved by the Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC).

This research will be carried out according to the National Statement on Ethical Conduct in Human Research (2007 updated 2018). This statement has been developed to protect the interests of people who agree to participate in human research studies.

15. Further information and who to contact

If you have any complaints about any aspect of the project, the way it is being conducted or any questions about being a research participant in general, then you may contact:

Reviewing HREC name	Southern Adelaide Clinical
Position	Executive Officer
Telephone	8204 6453
Email	Health.SALHNofficeforresearch@sa.gov.au

Reviewing HREC approving this research and HREC Executive Officer details

Local HREC Office contact (Single Site -Research Governance Officer)

Position	Research Governance Officer
Telephone	8204 6453
Email	Health.SALHNofficeforresearch@sa.gov.au

Thank you for taking the time to read this information sheet and we hope that you will accept our invitation to be involved.

For further information or questions, please contact:

Name: Langduo Chen

Phone: 08 8404 2062

Email: Langduo.chen@flinders.edu.au

Consent Form

Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Stroke hospital to home transition
Coordinating Principal Investigator/ Principal Investigator	A/Professor Lily Xiao
Associate Investigator(s)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Declaration by Participant

I (being over the age of 18 years) have read the Participant Information Sheet or someone has read it to me in a language that I understand.

I understand the purposes, procedures and risks of the research described in the project.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I freely agree to participate in this research project as described and understand that I am free to withdraw at any time during the project without affecting my work.

I freely agree to audio recording of my information and participation.

I understand that I will be given a signed copy of this document to keep.

I am aware that I should retain a copy of the Information Sheet and Consent Form for future reference.

I have had an opportunity to ask questions and I am satisfied with the answers I have received.

I understand that:

- I may not directly benefit from taking part in this research.
- I am free to withdraw from the research at any time and am free to decline to answer particular questions.
- While the information gained in this study will be published as explained, I will not be identified, and individual information will remain confidential.
- Whether I participate or not, or withdraw after participating, it will have no effect on my work.

• I may ask that the recording/observation be stopped at any time, and that I may withdraw at any time from the session or the research without disadvantage.

Name of Participant (please print)	
Signature	Date
Declaration by Researcher [†]	
I have given a verbal explanation of the research, it the participant has understood that explanation.	s procedures and risks and I believe that
Name of Researcher (please print)	
Signature	Date

† An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Note: All parties signing the consent section must date their own signature.

Response Slip

I..... (name) would like to participate in the research.

or Email.....

Alternatively, you can text message or ring the researcher (Ms Langduo Chen, phone number: 08 8404 2062) or email the researcher to express your interest in participating (Ms Langduo Chen, email: langduo.chen@flinders.edu.au).

Form for Withdrawal of Participation

Title	Exploring the care needs of stroke survivors and informal carers in hospital to home transition
Short Title	Stroke hospital to home transition
Coordinating Principal Investigator/ Principal Investigator	A/Professor Lily Xiao
Associate Investigator(s)	Langduo Chen, A/Professor Diane Chamberlain
Location	Flinders Medical Centre

Declaration by Participant

I wish to withdraw from participation in the above research project and understand that such withdrawal will not affect my routine care, or my relationships with the researchers or Flinders Medical Centre.

Name of Participant (please print)_____

Signature _____ Date _____

In the event that the participant's decision to withdraw is communicated verbally, the Senior Researcher must provide a description of the circumstances below.

Declaration by Researcher[†]

I have given a verbal explanation of the research, its procedures and risks and I believe that the participant has understood that explanation.

Name of Researcher (please print)_____

Signature _____

† An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Date

Note: All parties signing the consent section must date their own signature.

Appendix 16: Recruitment Poster

Recruitment poster for stroke survivor and carer participants

Participants wanted for study on Stroke

We are undertaking a study to explore care needs of persons with stroke and their family carers during hospital to home transition in South Australia.

You and your family carer are invited to the study if you/your relative:

- Are aged ≥18 years
- Have diagnosed with stroke
- Discharge home

Activities involved in participation:

• Prior to consent, potential stroke survivor participant will need

to undergo a 10- minute cognitive assessment. A decision on inclusion or exclusion of participation will be based on the assessment outcome.

- Take part in 2-3 times of a 30-45 minutes interview within 6 months of discharge
- Use a diary designed by the researchers to record your experiences during the 6month transition care period.

If you are interested in participating, please fill in and return the response slip to a survey drop box in your ward/clinics. The researcher will contact you shortly to explain more details of the study and sign a consent form with you.

The project has been approved by the *Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC).*

RESPONSE SLIP





Recruitment poster for health professional participants

Participants wanted for study on Stroke

We are undertaking a research study to explore care needs of persons with stroke and their unpaid carers during hospital to home transition in South Australia.

You are invited to the study if you:

• Have worked with people with stroke and their informal carers

Activities involved in participation:

• 60 to 90 minutes focus group ONCE only



If you are interested in participating, please fill in and return the response slip to a survey drop box in your work place. The researcher will contact you shortly to explain more details of the study and sign a consent form with you.

The project has been approved by the

Southern Adelaide Clinical Human Research Ethics Committee (SAC HREC).

	F	RESPONSE SLIP		

Name	Name	Name	Name
-	-	-	-
Contact Number	Contact Number	Contact Number	Contact Number



Appendix 17: Montreal Cognitive Assessment

Appendix 18: Functional Independence Measure

HOBPITAL BURANAZE LUMAS FUNCTIONAL INDEPENDENCE MEASURE (FIM) ASSESSMENT OOB GENDER NARD: Interface Notestand Notestand SOCTOR START FINISH TELEPHONE Episode Dates //// //// Notestand AROC Impairment Code Interface Interface (Bee over page for listing of veils codes) Interface Interface Refor to RF FIM Performance Levels ADMIS SION DISCHARGE Assessment Date						Please u	se I.D. label or bloc	k print	
FUNCTIONAL INDEPENDENCE MEASURE (FIM) ASSESSMENT DOB Dehoden VARD. ADRESS POSTCODE VARD. TELEPHONE TELEPHONE START FINISH International assistance with helper AROC Impairment Code		+	IOSPITAL	SURNAME				LIMRN	
(FIM) A SSESSMENT ADDRESS POSTCODE ARDD: TELEPHONE TELEPHONE OCTOR: TOTAL FINISH FIM Performance Levels POSTCODE	FL	UNCTIONAL	SURF	GIVEN NAME	ES			DOB	GENDER
OCTOR: TELEMIONE TELEMIONE FIN START FIN SH I - Total contact assistance with helper 2 - Maximal contact assistance with helper 2 - Comptete independence with helper 3 - Compteter Body 3 - Compteter Body 4	(FIM) ASSESSMENT			ADDRESS	ADDRESS				POSTCODE
START FINISH Episode Dates	OCTOR:							TELEPHONE	
Episode Dates /_// /_// 1 - Total contact assistance with helper AROC Impairment Code			STA	RT	FINISH	I	FIM Perf	ormance L	evels
FIM A SSE SSMENT Refer to the FILV Performance Level for available scores ADMISSION DISCHARGE Assessment Date	Episode Dates _/_/ AROC Impairment Code					_/	 Total contact assistance with helper Maximal contact assistance with helper Moderate contact assistance with helper Minimal contact assistance with helper Supervision or setup with helper Modified independence with helper Complete independence 		
Refer to the FIU Performance Level for available scores ADMIS SION DISCHARGE Assessment Date /_/				FIMAS	SESSME	NT			
Assessment Date	Refer to the Fill	/ Performance Level for a	wallable sco	res		AD	MISSION	DISC	CHARGE
Keting Grooming Bathing Dressing Upper Body Dressing Upper Body Dressing Lower Body Toileting Dressing Lower Body Bladder Management Bowel Management Bowel Management Transfer bed/ohair/wheelohair Transfer bed/ohair/wheelohair Transfer toilet Transfer bed/ohair/wheelohair Transfer bed/ohair/wheelohair Transfer bed/ohair/wheelohair Transfer bed/ohair/wheelohair Transfer bed/ohair/wheelohair Stairs Motor Subtotal Score (max. score 20) Comprehension Expression Social interaction Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 36) sseessor Details Image [intera pint] grave Unterprint	Assessmen	nt Date					//	/	
Grooming Bathing Bathing Dressing Upper Body Dressing Lower Body Dressing Lower Body Toileting Bladder Management Bowel Management Bladder Management Transfer bed/chair/wheelohair Transfer bed/chair/wheelohair Transfer bed/chair/wheelohair Transfer toilet Transfer bed/chair/wheelohair Transfer bed/chair/wheelohair Torsser bath/shower Decomotion Stairs Stairs Motor Subtotal Score (max score 97) Comprehension Expression Social interaction Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max score 36) TOTAL FIM SCORE (max score 120) Insersiblese prit Drespine [phone prit] graduer Drespine [phone prit]		Eating							
Bathing Dressing Upper Body Dressing Lower Body Dressing Lower Body Toileting Image: Status Bladder Management Image: Status Bowel Management Image: Status Transfer bed/chair/wheelchair Image: Status Transfer toilet Image: Status Transfer toilet Image: Status Transfer bath/shower Image: Status Locomotion Image: Status Status Image: Status Motor Subtotal Score (max: score 9/) Image: Status Cognition Social interaction Problem solving Image: Status Memory Image: Status Cognition Subtotal Score (max: score 58) Image: Status TOTAL FIM SCORE (max: score 58) Image: Status Image: Status Image: Status		Grooming							
Motor subscale Dressing Upper Body Dressing Lower Body Toileting Bladder Management Bladder Management Transfer bed/ohair/wheelchair Transfer bed/ohair/wheelchair Transfer bed/ohair/wheelchair Transfer bed/ohair/wheelchair Transfer bath/shower Locomotion Stairs Motor Subtotal Score (max. score 31) Cognition subscale Comprehension Expression Social interaction Problem solving Memory Cognition Subtotal Score (max. score 35) TOTAL FIM SCORE (max. score 35) TOTAL FIM SCORE (max. score 36) Sesessor Details		Bathing	R-+-						
Motor subscale Motor subscale Motor Subscale Motor Subscale Motor Stairs Motor Stairs Motor Stairs Motor Subtotal Score (max.score 21) Comprehension Expression Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max.score 28) TOTAL FIM SCORE (max.score 28) Sessessor Details Anama gives Details Deta		Dressing Uppe	per Body						
Motor subscale Bladder Management Bowel Management Imagement Transfer bed/ohair/wheelchair Transfer toilet Transfer bath/shower Imagement Locomotion Imagement Stairs Motor Subtotal Score (max. score 01) Cognition Social interaction Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 36) Imagement TOTAL FIM SCORE (max. score 120) Seegnates (linear prot)		Dressing Lowe	wer Body						
Motor subscale Bladder Management Bowel Management Transfer bed/chair/wheelchair Transfer bed/chair/wheelchair Transfer toilet Transfer bath/shower Locomotion Stairs Motor Subtotal Score (max. score 21) Comprehension Expression Social interaction Expression Social interaction Problem solving Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 36) ISSessor Details dirare (place prif) Desgnates (place prif)		Toileting							
Subscale Bowel Management Transfer bed/chair/wheelchair Transfer bed/chair/wheelchair Transfer toilet Transfer bath/shower Locomotion Stairs Motor Subtotal Score (max. score 91) Comprehension Expression Social interaction Problem solving Memory Cognition Subtotal Score (max. score 35) TOTAL FIM SCORE (max. score 35) TOTAL FIM SCORE (max. score 120)	Motor	Bladder Manag	jement						
Transfer bed/chair/wheelchair Transfer toilet Transfer bath/shower Locomotion Stairs Motor Subtotal Score (max. score 91) Comprehension Expression Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 35) TOTAL FIM SCORE (max. score 35) Itarme (places pnrt) Use	subscale	Bowel Manage	ment						
Transfer toilet Transfer toilet Transfer bath/shower Locomotion Stairs Motor Subtotal Score (max. score Pt) Comprehension Expression Expression Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 26) TOTAL FIM SCORE (max. score 26) TOTAL FIM SCORE (max. score 26)		Transfer bed/cl	hair/whee	elchair					
Image: Cognition Subtotal Score (max. score P1) Cognition subscale Social interaction Problem solving Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 126)		Transfer toilet							
Locomotion Stairs Stairs Motor Subtotal Score (max. score 21) Comprehension Expression Expression Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 35) TOTAL FIM SCORE (max. score 120) Image: score 120)		Transfer bath/s	hower						
Stairs Motor Subtotal Score (max. score P1) Comprehension Expression Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 128)		Locomotion							
Motor Subtotal Score (max. score P1) Comprehension Expression Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 120)		Stairs							
Cognition subscale Comprehension Problem solving Image: Cognition Solution S		Motor Su	ıbtotal S	core (max. :	score 91)				
Cognition Social interaction Social interaction Problem solving Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 126) Assessor Details Imme (please print) Desprintion (please print) Desprintion (please print) Desprintion (please print)		Comprehensio	n						
Cognition subscale Social interaction Problem solving Problem solving Memory Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 120) Insessor Details Inserse (places pmt) geture Desgreton (places pmt) Use		Expression							
subscale Problem solving Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 126) sssessor Details Imme (please part) Desgnature Desgnature Detail	Cognition Social interaction								
Memory Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 128) Issessor Details Insre (please print) Usegration (please print) Data	subscale	Problem solvin	9						
Cognition Subtotal Score (max. score 36) TOTAL FIM SCORE (max. score 120) Assessor Details all mms (please print) Desgnation (please print) Units		Memory							
TOTAL FIM SCORE (max. score 126) Assessor Details If mare (please part) Despreton (please part) Deta		Cognition Su	ıbtotal S	core (max. s	score 36j				
Inseres (please print) Uee preton (please print) pretore Uee		TOTAL F	IM SCO	RE (max. sc	ore 126)				
di name (please print) Despition (please print) gradure Date	ssessor De	etails							
gnèure Dés	ali name (please prin	e)			Designatio	an (please print	0		
· · · · · · · · · · · · · · · · · · ·	grature				Uate				
AUSTRALIA SIAN REHABILITATION OUTCOMES CENTRE IMPAIRMENT CODES VERSION 2

STROKE

Haemorrhagic

- 1.11 Left body involvement (right brain) 1.12 Right body involvement (left brain
- Bilateral involvement 1.13
- 1.14 No paresis
- 1.19 Other stroke

Ischaemic

1.21	Left body involvement (right
	brain)
1.22	Right body involvement (left
	brain)
1.23	Bilateral involvement

- 1.24 No paresis
- Other stroke 1.29

BRAIN DYSFUNCTION

Non-traumatic

- 2.11 Sub-arachnoid haemorrhage 2.12 Anoxic brain damage 2.13 Other non-traumatic brain
- dysfunction

Traumatic 2.21

Open injury Closed injury 2 22

NEUROLOGICAL CONDITIONS

- Multiple Sclerosis 3.1
- 3.2 Parkinsonism
- 3.3 Polyneuropathy
- 3.4 Guillain-Barre
- 3.5 Cerebral Palsy
- Neuromuscular disorders 3.8 3.9
- Other neurological disorders

SPINAL CORD DYSFUNCTION Non traumatic

- Paraplegia, incomplete 4.111 Paraplegia, complete 4 1 1 2
- 4.1211 Quadriplegia, incomplete
- C1-4
- 4.1212 Quadriplegia, incomplete C5-8
- 4.1221 Quadriplegia, complete C1-4
- 4.1222 Quadriplegia, complete C5-8
- 4.13 Other non-traumatic spinal cord dysfunction

Traumatic

- 4 2 1 1 Paraplegia, incomplete Paraplegia, complete 4.212 4.2211 Quadriplegia, incomplete C1-4 4.2212 Quadriplegia, incomplete C5-8 4.2221 Quadriplegia, complete C1-4 4.2222 Quadriplegia, complete C5-8
- 4.23 Other traumatic spinal cord
 - dysfunction

AMPUTATION OF LIMB

Not resulting from trauma 5.11

- Single upper above elbow 5.12 Single upper below elbow 5.13 Single lower above knee
- (includes through knee) 5.14 Single lower below knee
- 5.15 Double lower above keen (includes through knee)
- Double lower above/below 5.16

knee

- 5.17 Double lower below knee
- 5.18 Partial foot (single or double)
- 5.19 Other amputation not from trauma

Resulting from trauma

- 5.21 Single upper above elbow 5.22 Single upper below elbow 5.23Single lower above knee (includes through knee) 5.24 Single lower below knee 5.25 Double lower above knee (includes through knee)
- 5.26 Double lower above/below knee
- 5 27 Double lower below knee Partial foot (single or 528
- double) 5 29 Other amputation from
- trauma

ARTHRITIS

- Rheumatoid arthritis 61
- 62 Osteoarthritis
- 69 Other arthritis

PAIN SYNDROMES

- 7.1 Neck pain
- 7.2 Back pain
- Extremity pain Headache (includes 7.3
- 7.4
- migraine) 75
- Multi-site pain 7.9
 - Other pain (includes abdominal/chest wall)

ORTHOAPEDIC CONDITIONS

- Fractures (includes dislocation) 8.111 Fracture of hip, unilateral
- (includes #NOF) 8.112 Fracture of hip, bilateral
- (includes #NOF)
- 8 12 Fracture of shaft of femur
- 8.13 Fracture of pelvis
- 8.141 Fracture of knee
- 8.142 Fracture of lower leg, ankle, foot 8.15 Fracture of upper limb
- 8.16 Fracture of spine
- 8.17 Fracture of multiple sites
- 8.19 Other orthopaedic fracture

412

Post orthopaedic surgery 8.211 Unilateral hip replacement

- 8.212 Bilateral hip replacement
- 8.221 Unilateral knee replacement
- 8.222 Bilateral knee replacement
- 8.231 Knee and hip replacement
 - same side
- 8.232 Knee and hip replacement different sides
- 8.24 Shoulder replacement or repair
- 8.25 Post spinal surgery
- 8.26 Other orthopaedic surgery

SOFT TISSUE INJURY

- 8.3 Soft tissue injury
- CARDIAC
- 9.1 Following recent onset of
- new cardiac impairment 9.2 Chronic cardiac insufficiency
- 9.3 Heart and heart/lung transplant

PULMONARY

- Chronic obstructive 10.1
- pulmonary disease

CONGENITAL DEFORMITIES

Lymphoedema

a specific group

Brain + multiple

DEVELOPMENT DISABILITIES

surgery

medical illness

fracture/amputation

Spinal cord + multi

fracture/amputation

Other multiple trauma

Developmental disabilities

(excludes cerebral palsy)

Re-conditioning following

Re-conditional following

RE-CONDITIONING/RESTORATIVE

Cancer rehabilitation

MAJOR MULTIPLE TRAUMA

Conversion disorder

Other congenital deformity

Other disabling impairments

that cannot be classified into

Brain + spinal cord injury

Spina bifida

- 10.2 Lung transplant
- Other pulmonary 10.9

Burns

OTHER DISABLING

IMPAIRMENTS

BURNS

11

12.1

12.9

13.1

13.3

13.9

14.1

14.2

14.3

14.9

15.1

16.1

16.2

16.3

Appendix 19: Simplified Participant Information and Consent Form





Information Sheet

Care needs of stroke survivors and carers in Title:

hospital to home transition

Main researcher: Ms. Langduo Chen



You are invited to take part in research.

39.19 PICF Survivors simplified v5 28.04.19

Page 1 of 13

What is this project about?

Stroke changes people's life.



It affects you and your family.

Returning home from hospital can be a very hard time.



We want to know your experience.



39.19 PICF Survivors simplified v5 28.04.19

Page 2 of 13

What do I need to do?

You will need to have a cognition assessment.

You will need to sign a consent form.

You will be asked for your personal information.

- - Age
 - Occupation
 - Health information

You will need to attend 2-3 times of interviews and each

interview will take about 30 to 45 minutes.

- Before discharge
- 3 months after discharge
- 6 months after discharge

Where will I need to go?

39.19 PICF Survivors simplified v5 28.04.19







Page 3 of 13

The interview takes place at a location of your choice.

What will happen to the information I give?

• You will be audio recorded.



We will need to read your medical notes.



- We will keep the information you give us and recordings at The Flinders University of South Australia.
- All information will be kept safely.



39.19 PICF Survivors simplified v5 28.04.19

Page 4 of 13



- All information will be confidential.
- To protect your privacy, all information about you will be labelled with a code number.

What will happen to the information I give?

The results may appear in journal publications and conference presentations.



- Your identity in publications will be kept private.
- We will send all participants a summary of the results.



How long will you keep my information for after the

project is completed?

• All your information from the study will be stored securely for

15 years.

39.19 PICF Survivors simplified v5 28.04.19

Page 5 of 13

Are there any risks?

• There is **no danger** in doing this research.

Are there any benefits?

- We are doing this study to learn how to improve health services for people with stroke.
- It might not help you right now.

What if I decide I do not want to be involved?

- You do not have to participate. It is your choice.
- You can start participating and then change your mind.
- You can ask questions at any time.
- Saying no will not affect your relationship with staff at Flinders Medical Centre.
- You can stop at any time.



39.19 PICF Survivors simplified v5 28.04.19

Page 6 of 13

To stop doing the research contact:



Ms. Langduo Chen

Phone: 08 8404 2062



Email: Langduo.chen@flinders.edu.au

Who has approved this research?

• This study has been **approved** by the Southern Adelaide Clinical Human Research **Ethics Committees**.

What if I have a complaint?

• If you would like to speak to an officer of the Committee **not**

involved in the study, you may contact:

The Ethics Officer at the Ethics Committee



8204 6453



Health.SALHNofficeforresearch@sa.gov.au

What if I have more questions?

39.19 PICF Survivors simplified v5 28.04.19

Page 7 of 13

Please contact:





Email: Langduo.chen@flinders.edu.au

How can I find out the results of the study

• We can mail our findings out to you.



• Do you want us to email our findings to you?

(Please circle)



Yes







Page 8 of 13

39.19 PICF Survivors simplified v5 28.04.19





Consent Form

Title: Care needs of stroke survivors and carers in

hospital to home transition

Main researcher:

Ms. Langduo Chen







Phone: 08 8404 2062.

Email:

Langduo.chen@flinders.edu.au

39.19 PICF Survivors simplified v5 28.04.19

Page 9 of 13



I will answer questions about my experience and thoughts.



Yes

I will allow researchers to look at my medical notes.



I will take part in 2-3 interviews about my experience.

I will **allow** the interview to be **audio**-

recorded and typed up.



Yes



I understand that people will be told about the study but

my name will never be used.



Page 11 of 13

I will allow the information I supply to b	be used
anonymously in journal publications an	d conference
presentations.	Yes
I would like to see the results of the stu is completed.	dy when it
Name of Participant (please print)	Yes
Signature D	ate
Name of Participant's Carer (please print	nt)
Signature D	ate
Declaration by Researcher ⁺	
I have given a verbal explanation of the	research, its procedures

and risks and I believe that the participant has understood that 39.19 PICF Survivors simplified v5 28.04.19 Page 12 of 13 explanation.

Name of Researcher (please print)

Signature _____

Date ______

⁺ An appropriately qualified member of the research team must provide the explanation of, and information concerning, the research project.

Note: All parties signing the consent section must date their own signature.

39.19 PICF Survivors simplified v5 28.04.19

Page 13 of 13

Appendix 20: Interview Guide

Appendix 20a: Interview Guide Before Hospital Discharge (the First Interview)

Question	Stroke patient	Stroke caregiver
Question 1	How have you been since the stroke?	How have you been since the stroke of your relative?
	• How have you been dealing with the fact that you had a stroke?	How have you been coping with the fact that your relative had a
	How do you think your family cope with what happened?	stroke?
		• How do you think your relative is dealing with what happened?
Question 2	Now you are going home soon, please tell me about your	Now your relative is going home soon, please tell me about your
	rehabilitation experience.	thoughts on the caregiving role.
	 How prepared do you think you are for returning home? 	• How prepared do you think you are for supporting your relative at
		home?
Question 3	What are your expectations after returning home?	What are your expectations after your relative returns home?
	• Expectations on your rehabilitation/recovery.	 Expectations on the relative's rehabilitation/recovery.
	• Expectations on community health and social care services.	 Expectations on the support services.

Appendix 20b: Interview Guide After Hospital Discharge (the Second and Third

Interviews)

Interview guide for stroke survivors (2nd interview)

The following questions will serve as a guide in the interview. Questions may be asked in a different order or some omitted or added, depending on the participants' responses.

- 1. Could you please tell me your experience in the preparation for discharge?
 - Level of involvement?
 - Information/training received?
 - Practice sufficient for home?
- 2. What were your expectations in discharge preparation? Any suggestions?
- 3. Now you have come home, how do you feel being at home after stroke? Can you please tell me a typical day? How do you manage daily activities?
 - Meaning of home
 - Differences comparing to life before
 - Strategies to manage daily activities
- 4. What health and social care services do you receive after discharge from hospital?
- 5. How have these care services helped you to recover at home?
- 6. What changes regarding your recovery and health have you experienced since your discharge from hospital?
 - Examples of the progress of recovery that you are satisfied with
 - Examples of the progress of recovery that you are not satisfied with
 - Thoughts and ideas about health and social care that may help you to achieve your goal for recovery
- 7. Is there anything else that you would like to talk about and add?
- 8. Is there anything that you have talked about that you would like me not to include in my research report?

Interview guide for stroke survivors (3rd interview)

- 1. How have you been since our last meeting?
 - Health condition
 - Hospital visit
 - Issues and concerns
- 2. Now you have come home for XX weeks. Tell me about a typical day? How do you manage daily activities?
- 3. How has your life changed since the last meeting?
 - Relationship with your family/carer/friends
 - How the changes impacted on the recovery
- 4. What do you think about the community support/rehabilitation you received?
 - Helped or hindered you to adapt to the new life after stroke
 - Suggestions
- 5. What changes in health and social care do you want to see in order to help you to achieve the heist level of recovery and adapt to the new life after stroke?
 - Process of discharge from hospital
 - Health professionals
 - Health and social care system
- 6. Is there anything else that you would like to talk about and add?
- 7. Is there anything that you have talked about that you would like me not to include in my research report?

Interview guide for caregivers (2nd Interview)

- 1. Please tell me about your experience in the preparation for discharge.
 - Level of involvement
 - Training/skills building
 - Practice sufficient?
 - What helped/didn't?
- 2. Now your relative has come home, would you please tell me how you feel being at home? Describe a typical day? How do you manage daily activities?
 - Differences comparing to life before
 - Strategies to manage daily activities
 - Relaxing and have time for yourself
- 3. What commitment you have made in order to support your relative to go through rehabilitation after discharge from hospital? How have these commitments impacted on your life?
 - Physical e.g. manual handling
 - Mental e.g. emotional support to persist
 - Financial e.g. giving up work or reducing working hours
 - Time e.g. how many hours per day or per week
- 4. Please tell me about the support and services your relative received after discharge?
 - Source of information
 - Knowledge of all available services
 - Helped transition home?
- 5. Is there anything else that you would like to talk about and add?
- 6. Is there anything that you have talked about that you would like me not to include in my research report?

Interview guide for caregivers (3rd Interview)

- 1. How have you been since our last meeting?
 - Health condition
 - Hospital visit
 - Issues and concerns
- 2. Now your relative have come home for XX weeks, could you please tell me how you feel being at home? Describe a typical day? How do you manage daily activities?
- 3. How has your life changed since the last meeting?
 - Relationship with your family/carer
 - How the changes impacted on the recovery
- 4. How have your relationship with your relative changed? How have these changes impacted on your life?
- 5. What do you think about the support/rehabilitation your relative received after discharge?
 - Impact on your life style/employment
 - Helped/hindered transition home
 - Care needs met?
- 6. What changes do you want to see in order to make the process of discharging home easier?
 - Discharge planning process
 - Health professionals
 - Health care system
- 7. Is there anything else that you would like to talk about and add?
- 8. Is there anything that you have talked about that you would like me not to include in my research report?

Appendix 20c: Interview Guide for General Practitioners

General question: full name, years of experience, experience with stroke population.

The following questions will serve as a guide in the interview. Questions may be asked in a different order or some omitted or added, depending on the participants' responses.

- 1. Could you please talk about your involvement in the care of stroke survivors?
- 2. What are your thoughts on factors that facilitate the continuity of care for stroke survivors after discharge from hospital?
- 3. What are your thoughts on factors that impede the continuity of care for stroke survivors after discharge from hospital?
- 4. What are your expectations of the discharging hospital? What recommendations do you have?
- 5. What community rehabilitation services and supports do you know that are available for stroke survivors and their informal carers? From which sources you have obtained the information?
- What community rehabilitation services and supports do you usually refer the stroke survivors to? How well do these services support them? Please provide some examples.
- 7. What challenges do you face when caring for stroke survivors and working with their informal carers? Any recommendations on how to better manage these challenges?
- 8. What changes do you want to see regarding the process of discharging from hospital?
- 9. What changes do you want to see for the health care and social care system to support stroke survivors and their carers during hospital to home transition?
- 10 Is there anything else that you would like to talk about and add?
- 11 Is there anything that you have talked about that you would like me not to include in my research report?

Appendix 20d: Focus Group and Interview Guide

Focus Group and interview Guide

Generic Questions

- 1. What are your involvements in the care of stroke survivors?
- 2. What strategies do you usually use to identify care needs for stroke survivors during hospital to home transition? Are those strategies workable? Examples?
- 3. What strategies do you usually use to identify the need to prepare carers to care for stroke survivors prior to discharge? Are those strategies workable? Examples?
- 4. What do you do to ensure that the transition care and rehabilitation needs for stroke survivors are met? Examples?
- 5. What impact do the early discharge and shortened hospital stay have on the hospital to home transition for stroke survivors and their family?
- 6. What are the factors contributing to the implementation of effective discharge planning? Examples especially for patients with cognitive impairment?
- 7. What are system factors that enable or impede the continuity of health care and social care services in hospital to home transition?
 - Examples
 - What could be done better?

Discipline-specific questions

Social worker

For young stroke survivors (<65 years, <45 years Aboriginal population), what factors facilitate or impede a smooth transition? Examples?

Physiotherapist/Occupational therapist

What are the differences in discharge preparation approaches between young and old stroke survivors?

Speech pathologist

For survivors with speech and swallowing impairment, what factors facilitate or impede a smooth transition? Examples?

Medical and Nursing

What are the challenges in secondary prevention? Examples?

For survivors with continence issues, what are strategies to ensure continuity of care at home?

Appendix 21: Structured Diary

Structured Diary Entry Guide for Person with Stroke

The following tables will serve as a guide for your diary entry. There is no restriction on writing or recording more or less; depending on your choice.

Name: (Option).....Code Number (for researchers to enter):....Date.....

Your experience	Your experiences, thoughts and comments on the Significant Events
In coping with physical	
difficulties	
In coping with other health	
problems	
With your family carers	
In the use of care services	
In engaging support group	
Any difficulties you	
encounter and how you	
cope with the difficulties	

Structured Diary Entry Guide for Caregiver of Person with Stroke

The following tables will serve as a guide for your diary entry. There is no restriction on writing or recording more or less; depending on your choice.

Name: (Option)..... Code Number (for researchers to enter):..... Date.....

Your experience	Your experiences, thoughts and comments on the Significant Events
In coping with post stroke difficulties	
In coping with other health	
problems	
With your own health	
problems	
In coping with caregiving	
In daily activities	
In the use of care services	
In engaging support group	
Any difficulties you	
encounter and how you	
cope with the difficulties	

Торіс	ltem	Guide question/descriptions	Reported in
			manuscript on
		Domain 1: Research team and reflexivity	page no.
Personal characteristics			
Interviewer/facilitator	1	Which author/s conducted the interview or focus group?	10
Credentials	2	What were the researcher's credentials? E.g. PhD, MD	9
Occupation	3	What was their occupation at the time of the study?	10
Gender	4	Was the researcher male or female?	NA
Experience and training	5	What experience or training did the researcher have?	9
Relationship with participants			•
Relationship established	6	Was a relationship established prior to study commencement?	10
Participant knowledge of the	7	What did the participants know about the Researcher? e.g. personal goals, reasons	10—stated in
interviewer		for doing the research	participant
			information
			sheet
Interviewer characteristics	8	What characteristics were reported about the interviewer/facilitator? e.g. Bias,	10
		assumptions, reasons and interests in the research topic	
		Domain 2: Study design	
Theoretical framework			1
Methodological orientation	9	What methodological orientation was stated to underpin the study? e.g. grounded	7–13
and theory		theory, discourse analysis, ethnography, phenomenology, content analysis	
Participant selection			1
Sampling	10	How were participants selected? e.g. purposive, convenience, consecutive,	9, Table 1
		snowball	
Method of approach	11	How were participants Approached? e.g. face to face, telephone, mail, email	9–10
Sample size	12	How many participants were in the study?	10
Non-participation	13	How many people refused to participate or dropped out? Reasons?	10
Setting			
Setting of data collection	14	Where was the data collected? e.g. home, clinic, workplace	9, 10
Presence of non-participants	15	Was anyone else present besides the participants and researchers?	10
Description of sample	16	What are the important characteristics of The sample? e.g. demographic data, date	Table 4

Appendix 22: COREQ (COnsolidated Criteria for REporting Qualitative Research) Checklist

Торіс	ltem	Guide question/descriptions	Reported in
			manuscript on page no.
Data collection	-		
Interview guide	17	Were questions, prompts, guides provided by the authors? Was it pilot tested?	10, Table 2
Repeat interviews	18	Were repeat interviews carried out? If yes, how many?	NA
Audio/visual recording	19	Did the research use audio or visual recording to collect the data?	10–11
Field notes	20	Were field notes made during and/or after the interview or focus group?	8, 11
Duration	21	What was the duration of the interviews or focus group?	11
Data saturation	22	Was data saturation discussed?	NA
Transcripts returned	23	Were transcripts returned to participants for comment and/or correction?	NA, statements
			cross-checked
			during interview
		Domain 3: Analysis and findings	
Data analysis	•	1	
Number of data coders	24	How many data coders coded the data?	12
Description of coding tree	25	Did authors provide a description of the coding tree?	Table 3
Derivation of themes	26	Were themes identified in advance or derived from the data?	11–12
Software	27	What software, if applicable, was used to manage the data?	11
Participant checking	28	Did participants provide feedback on the findings?	NA
Reporting			
Quotations presented	29	Were participant quotations presented to illustrate the themes/findings?	14–26
		Was each quotation Identified? e.g. participant number	
Data and finding consistent	30	Was there consistency between the data presented and the findings?	14–26
			SuppInfo Table
			2
Clarity of major themes	31	Were major themes clearly presented in the findings?	14–26
Clarity of minor themes	32	Is there a description of diverse cases or discussion of minor themes?	14–26

Adapted from: Tong, A., Sainsbury, P., Craig, J. (2007). Consolidated criteria for reporting qualitative research (COREQ): A 32-item checklist for interviews and focus groups. *International Journal for Quality in Health Care*, *19*(6), 349–357. <u>https://doi.org/10.1093/intqhc/mzm042</u>

Appendix 23: Themes With Quotations

Theme	Subtheme	Direct quotation
Psychological 'I a stress to	'sychological tress 'I am looking forward to going home but feeling anxious' 'Not knowing what the future holds'	The, the only thing I have to be just mindful of is my, probably my physical strength. I don't have as much physical strength as I used to. (SS03)
		[I feel] a bit depressed, worried about having another one [stroke]. And not going back to my previous life. Worried about my independence yeah Just as I said just a bit fearful of the future, I guess. Who knows what it will bring. (SS09)
		It is frustrating, I'm annoyed about it and I wrote this morning well a page in the notebook it probably took me half an hour to write and I made some mistakes, which I corrected in it. (SS11)
		I am looking forward to going home because this a fairly lengthy time to be in hospital but at the same time, I'm also feeling anxious about my level of functioning in the home situation to see what I can do, what I can't do, what impact it's going to have on me and how is it going to perhaps affect me emotionally as well? Because all those things that under normal circumstances you take for granted, suddenly it's a bit like having electricity on in the house where you just switch it on, take it as a given thing that it's there. As soon as the power gets switched off, you think no electricity. (SS12)
		A big focus for me was the walking part because that was quite devastating to find that you'd stand up and you couldn't put any weight on the left hand side. And when I started using the walker, it was kind of very stilted it became I suppose a focal point for me to actually keep doing that. (SS12)
		Well, I think mainly I mean if his needs are ongoing as far as not being able to stay at home by himself, that will create a whole new set of issues, and at the moment I am willing to support his needs 24/7 simply because he is mobile. So the main concern I think might be down the track as far as getting some sort of assistance for me to be able to leave the home at times. (SC01)
		My blood pressures up. Look one of—one of the—one of the I guess the fall—some of the fall out of—of I do [worry] because he's not always easy to tell him to slow down, and we'll stress at each other over those things. And just not knowing what the future holds, how I can care for him and help him better. (SC06)

Theme	Subtheme	Direct quotation
		[I feel] really stressed, depressed, just, just generally a lot of sadness because of not knowing what the future holds for him. Just coping, coping not as good as I used to be. (SC03)
		I do [worry] because he's not always easy to tell to slow down, and we'll stress at each other over those things. And just not knowing what the future holds, how I can care for him and help him better [I feel] really stressed, depressed, just generally a lot of sadness because of not knowing what the future holds for him. Just coping, coping not as good as I used to be. (SC03)
		No, I just know that really there can't be expectations because we don't know what it's going to—my commitments only to put the work in, to put the therapies in place, that's what I requested from all of them to give me as much as we can with therapy because he's so young and because obviously the boys expect him to be playing soccer with them and doing stuff and so but you know we'll just keep strengthening him and see what we can glean out of it, but they're very excited, they want me to pick them up because Dad's coming home. (SC05)
		And the 20-year-old was XX's [husband's] apprentice for 5 years, so XX [husband] worked for himself so as the main breadwinner of the family and we found him unconscious the morning my son went to write his final exam, so it's been quite emotional, quite difficult for the family to work through because it was quite traumatic. (SC05)
		Well it's been 12 weeks of coming in and out of hospital with four children, so juggling family/work and home and hospital, it's been tricky. So it's been hard and emotionally it's been hard especially because the children are quite—well, two of the children are quite young. (SC05)
		Well because XX [the husband] was the main breadwinner we've got a mortgage, young family, kids in school, I needed to stabilise my work But I've cut back one day so I'm 3 days a week, but I need to build back up to doing just so that we can pay off our [bills]. (SC05)
		The initial shock was terrible- Because we nearly lost him, it was really bad. Unreal—it seemed unreal, but gradually, I know from the moment he had the stroke, my life will change, so will his. (SC08)
		When I knew that Mum had had—she had two mini strokes and yeah it was quite upsetting and then yeah to see her in the hospital as well was sad yeah. (SC13)
		I mean to see her like this after being so well for 91 years and not having any sicknesses or anything and then to see her have to go through something like this, it has been pretty stressful

Theme	Subtheme	Direct quotation
		and you do have your ups and downs, but I'm so glad now that she seems to be on the road. (SC14)
		It's been difficult Just running everywhere, making sure that everything's okay when he comes home. I'm usually very organised things have sort gone a bit haywire. I lose my keys, forget my phone I don't know what to expect. I believe someone spoke to XX [the husband]. But he's not reliable. Here we are we're going home. I'd like to know who I'm expecting to come into my home, how often. (SC18)
		I worry about him at home. I've been the one that's found him on the floor. And I worry about that [another stroke] and falling again. (SC18)
Resilience	'I've come from zero to hero'	I—I can accept that with the major stroke I have, there might be some limitations If you could say, I've come from zero to hero a fortnight ago I couldn't sit up in bed. I was in XX [acute ward], And physios came and tried to sit my up in bed, and I couldn't do it. So, I've gone from that to walking, with a frame. I can walk some distance independently You know, I mean I was—I quite surprised myself, yesterday I actually put sneakers on, and I was able to tie the shoelaces. And to me, I know some adults that still can't—they still can't tie their shoelaces. (SS03)
		I'm reasonably confident. I just know that I have to be careful and go slowly, be sensible. All those things I'm not terribly good at but I'll try my brother will fix the shopping and I will do the cooking. And we'll go back to how we were before. (SS09)
		After really accepting that I had a stroke and, and it seemed to be a slow improvement, most days going forward. But I, I appreciate I understand, at least I believe I'm not going to get back to where I was, and things will be all slower I can live with my calculation ability and I think I can live with my speech it's not the best but it could be far worse And I see around here a lot of people are far worse and I'm certainly happy with my physical side of things in many ways. (SS11)
		I wouldn't say that it's nice to sort of be face with things like that but I think you start to adjust your way of thinking a little bit to—and you've got to—to try to use those things. (SS12)
		I saw the awful disabilities that had happened to them and I saw this man and he was just so badly stricken with his stroke And it made me realise how lucky I was that I had just had a minor stroke, was still able to think and feel (SS13)

Theme	Subtheme	Direct quotation
		Well I'm just so looking forward to going home. And sleeping in my own bed and doing stuff for myself when I want to do it and I'm, I'm probably going to make mistakes but I've got things organised, I've got a cleaner organised, my driver's license, which is my lifeline, is being organised. And that's really great because I live on top of a ridge, in a bushfire prone area and I'm right on the top. So either way it's very steep going down. In the hills at Bridgewater I don't have footpaths, streetlights or any of those sophisticated things, and driving and being able to get my license back is an absolutely wonderful, it's a God send. (SS15)
		I think my objective up there [acute ward] was to walk out of here one day cured or fit and well and able to walk. See the significance on able to walk, I guess I'm gauging my wellbeing and mobility through being able to walk and out of here being symbolic with, hey I've done well, I've recovered from potentially a very dangerous incident or event and I'm on my way again. (SS17)
	'I look at this as a change in the road for life'	I have done a lot of mindfulness practice in the past and that has been hugely beneficial. And I would really recommend that there is support for people in that area to learn—to be able to manage their anxiety or anything that might come up. (SC01)
		My job is working with people that live at home with disabilities. So, it's more setting up rosters and doing care plans and working with the NDIS. So, I have set up support for XX [the husband] with the NDIS just for when I'm working. (SC05)
		I face things front-on, so this is what it is, look at the positives. I've got him and he's lucky and I'm lucky, the family's lucky. So we don't look at this as a disability, we look at this as a change in the road for life. That's all it is. All my life and we've had a lot of ups and downs Business, financial, but you've got to look at the positives. I just accept it and work from there and try and make my husband's life through speech therapy and us working at home as best as he can get it, and that's the goal. (SC08)
		I am as prepared—I've been prepared the whole time he's been in care. Well now, I can't tell you how prepared. I'm a strong person, I think I can manage, in fact, I'm positive, I know I can manage, but I'm sure there would be times where I would feel a little bit angry because my life has changed. (SC08)
		I do have the facility and I've already had it, when there's a drama, I gone dead calm and deal with it. My personality, yes. (SC11)

Theme	Subtheme	Direct quotation
		Yeah, that's alright; I mean we've talked about dying, we're very straight forward about the future, we've talked about death and we say death and dying. We've got things in place with the family and with our executor as to what we want and we both realise he's 82, I'm nearly 80 that our life is coming towards the end. We're not going to live forever, no one does. (SC11)
		[when we feel stressed] we talk a lot, spend time with friends because my son and close friends are still in Melbourne, I'm on the phone a lot, I read, I go walking and I sleep; I suppose they're the things that I've been trying to do. (SC14)
		But we are both in our 70s. We're quite realistic about the—the reality of mortality. But there's no reason why she can't have a good quality of life, and enjoy life for the foreseeable future. And that's what my expectations are. (SC16)
Certainty and uncertainty about	'They boost you up': certainty about post-	But I'd like to be able to strive to get back to a normal life … I'm hopeful that given a week at home, I can push a vacuum around. We've got a Dyson vacuum; very lightweight. (SS03)
post-discharge	ge discharge rehabilitation	I don't know exactly what the services are at this stage. (SS11)
		I believe that I'll be getting some in-home rehab and that's one of the things. The other things are sort of not specifically sure in the home situation. (SS12)
		Well to start with the physios showed me how to put, how to walk what it was like, even though I had to hang onto things all the time and then they worked on my balance and they worked on my walking and coordination and getting those neural pathways back. And explaining what was happening all the time, and so they've taught me a lot of the things I need to know. And it's been terrific and they've given me a lot of confidence too, that I could do it Because they said, 'Oh you're wonderful you did it all really well'. And so apart from the actual showing you how to do things, they boost you up and so do the nurses. (SS15)
		But I think probably week, 10 days I'll be managing to walk, my works nearly finished for this year. So I don't know if I'll get to my studio because there's also big stairs there. There are stairs at home, well I won't be able to do those yet because they're extremely steep, but I think eventually I'll be able to do all those things. (SS15)
		Well just their expertise, it can be no, no more complicated than someone holding me down in the gym, while I'm trying to develop a balance, improve balance. So there's the expertise, they know where I'm comfortable to work within that risk parameter but if they weren't as professional I may be nervous and not as willing to undertake that next exercise. My walking or

Theme	Subtheme	Direct quotation
		my mobility has improved from a walking frame to a mobile walking frame and now it's a walking stick and I even suggest I'm not far off just walking naturally. And that's all been with their guidance, so I think that answers your question. (SS17)
		I'm confident, I'm not intimidated or nervous by it, rehab will go on but in a different way from the home and visits from here. So I'll still get that support, but I'm just thrilled to be going home, particularly Christmas season. And I was informed by the doctor here that the best thing I could do had, them having observed my progress, was to go home into the work environment—sorry the home environment where improvement should be more pronounced. So of course, I would do it. (SS17)
	'We don't know what's going to happen': uncertainty about post- discharge care	I also feel confident—I am being told all the time that I can access support here, ask any questions. That will be important for anything that comes up. (SC01)
		and I asked him to do a family meeting, so we had a family meeting to discuss with the children what Michael's needs would be at home and how he would be different. And that they understand that he can't come back in the same capacity and that they need to be looking out for him, so that was a very beneficial meeting. (SC05)
		No, I just know that really there can't be expectations because we don't know what's going to [happen]—my commitment is only to put the work in, to put the therapies in place, that's what I requested from all of them [service providers] to give me as much as we can with therapy but you know we'll just keep strengthening him and see what we can glean out of it. (SC05)
		The past few weeks is it not knowing what the future holds of course. But also not knowing on the support side what's available, what's going to be available. It—it really was a matter of trying to get information from My Aged Care. (SC06)
		Okay well we had a lot of phone calls from the hospital just to say that Mum would be coming home on this date and that we were expected to go up there as a family but that didn't happen. Then I don't know it just never came about so Mum was just discharged and when I rang she'd already left. Well it was a little bit disappointing because you know we had all the phone calls about we were going to do—go to the hospital and all get together and have a chat about Mum and it just never happened and then she was discharged so that side of it didn't happen. (SC13)
		Totally and where we've thought that we haven't had an answer quickly enough, we've rung; so there was a time where somebody didn't get back to us and so we followed it up straight away.

Theme	Subtheme	Direct quotation
		So no, totally involved and as I said, we've pushed a little bit where we've felt that we haven't had an answer quickly enough. So yeah and I don't think that's going to change, I think we'll be totally involved because at the end of the 12 weeks there has to be something as well, so we'll keep pushing. (SC14)
		As prepared as I can be because it's a bit unknown. I think we've now got a reasonable amount of stuff in the house and we've got RDNS coming today, so once I get to know who those people are going to be and how we're going to then work together in the house I think I'll be a lot calmer; so today there's a little bit of trepidation, but then I'm a person that kind of just keeps going with the flow, so we get the next group of people and we'll learn how to deal with the next group of people and work out how it's going to be at home. I think it's going to be great; I've always said I thought Mum would heal sooner at home than in this environment and I think the couple of days we've had her home on leave has demonstrated that. So I'm really—I'm as prepared as I can be for what's going to come, but I really don't know what's going to come. I think there are going to be days where we're at each other's throats and I think there'll be days that will just be bliss, I think it's just going to be again, a rollercoaster for some time to come. (SC14)
		Well I don't know too much about the details. I understand we're going to receive recommendations about grips for the shower, etc. And the sooner that happens the better, because until I know what's required, I can't do anything about it. It has been explained to XX [wife] that there are various forms of assistance available. And she has much more detail—knowledge with that than I do. I'm going to be spending as much time with her as I can. (SC16)
		I came in on Friday to talk to the social worker, she wasn't here. So then I asked the desk, I said look I don't know what's happening with rehab next week. I don't know what to expect. And she said oh I'll get somebody here at 3:30 to talk to you. I come back and I didn't speak to anybody. I believe someone came and spoke to Ian. But he's not reliable, yeah he's not reliable to tell me exactly what is being said, yeah. So I think there's been a bit of a problem there yeah. And here we are we're going home. And I'm hoping to talk to somebody today. Well there's nobody—I thought there would be another family meeting, I thought. That's why I wanted to talk to the social worker on Friday, but she's not here. So therefore I suppose it's because of Christmas, and people are not around, so yeah. (SC18)
		About the rehab. Well I've got no idea. It's just the OTs that have said oh yes you'll have this, and you'll have that. But no—I believe I'm going to get a pack with some notes in. I'd like to

Theme	Subtheme	Direct quotation
		know who I'm expecting to come into my home, how often. And now because of Christmas it's not going to happen Wednesday, and Thursday on public holidays. Or the week after it's not going to happen on the public holiday. I did ask do we get an extension of the time for them to come. No, you don't. So it's just bad luck that you've got the public holidays in there. But I'd just like to know what is set up. I don't want to feel—and I worry about his new medication, I want to talk to the pharmacist and get the list. (SC18)

Note. SS = person with stroke; SC = stroke caregiver; NDIS = National Disability Insurance Scheme; RDNS = Royal District Nursing Service; OT = occupational therapi